

Modular affine Hecke category and regular centralizer Roman Bezrukavnikov, Simon Riche

▶ To cite this version:

Roman Bezrukavnikov, Simon Riche. Modular affine Hecke category and regular centralizer. 2022. hal-03691331

HAL Id: hal-03691331 https://hal.science/hal-03691331

Preprint submitted on 9 Jun 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

MODULAR AFFINE HECKE CATEGORY AND REGULAR CENTRALIZER

ROMAN BEZRUKAVNIKOV AND SIMON RICHE

ABSTRACT. In this paper we provide a "combinatorial" description of the category of tilting perverse sheaves on the affine flag variety of a reductive algebraic group, and its free-monodromic variant, with coefficients in a field of positive characteristic. This provides a replacement for the familiar "Soergel theory" for characteristic-0 coefficients. As an application we deduce character formulas for indecomposable tilting perverse sheaves on the affine flag variety in terms of ℓ -Kazhdan–Lusztig polynomials.

Contents

1. I	troduction	1				
2. (Coherent sheaves on the Steinberg variety					
3. 8	Some Hecke categories					
4. (Constructible sheaves on affine flag varieties					
5. (. Construction of the monodromic regular quotient					
6. I	ee-monodromic perverse sheaves	58				
7. I	7. Free-monodromic central sheaves and Wakimoto sheaves					
8. I	8. Perverse sheaves on G/U					
9.]	uncation of perverse sheaves	94				
10.	10. Perverse sheaves on $\widetilde{\mathrm{Fl}}_G$					
11.	A Soergel-type description of tilting perverse sheaves 1	.30				
12.	12. The case of Fl_G					
13.	13. Applications					
Appe	dix A. Verdier quotients, Serre quotients, and t-structures 1	52				
Appe	dix B. Flat modules in categories 1	55				
Appe	dix C. Infinitesimal flatness 1	57				
Appe	dix D. Complements on the perverse t-structure on the completed					
	category 1	.58				
Refei	nces 1	61				

1. INTRODUCTION

1.1. **Presentation.** The present paper is the second step in our project (initiated in [BRR], joint with L. Rider) of constructing "modular tamely ramified local Langlands equivalences" adapting to positive-characteristic coefficients the constructions of the first author in [B2]. These equivalences will relate some categories of Iwahori-constructible or Iwahori-equivariant perverse sheaves on the affine flag variety Fl_G of a connected reductive algebraic group G (or a natural torsor \widetilde{Fl}_G over Fl_G),

with coefficients in an algebraically closed field k of characteristic ℓ (or a "freemonodromic completion"), to some categories of equivariant coherent sheaves on the Steinberg variety of the reductive group G_k^{\vee} over k which is Langlands dual to G (or some variants). Our strategy involves a description of both sides of this equivalence in "Soergel theoretic terms;" the present paper finishes the first half of this construction, by providing such a description on the constructible side of the picture. The second half will be completed in a future publication, based of the localization theory for representations of reductive Lie algebras over fields of positive characteristic developed by the first author with Mirković and Rumynin, see [BMR, BM]. (This part of the work is closely related to our work in [BR2].)

The main result of the paper is therefore an equivalence of categories relating an appropriate category of perverse sheaves on \widetilde{Fl}_G to a category of "Soergel bimodules." This equivalence is fundamental for our project explained above, but it is also of independent interest, given the importance of categories of perverse sheaves on affine flag varieties in recent work on representation theory of reductive algebraic groups over fields of positive characteristic (see in particular [RW1, RW2]).

As a direct application of our constructions, we prove that characters of indecomposable tilting perverse sheaves on Fl_G can be computed using the combinatorics of the ℓ -canonical basis of the affine Hecke algebra. The analogous result for characteristic-0 coefficients (involving the "ordinary" Kazhdan–Lusztig basis) is well known, see e.g. [Yu], but such a result in the positive-characteristic setting was known only for ordinary ("finite dimensional") flag varieties of reductive groups, see [AR1]. A similar claim for "mixed" perverse sheaves was proved in [AMRW], but these "mixed perverse sheaves" are a priori unrelated to the ordinary perverse sheaves we consider here; another consequence of our constructions is precisely a construction of a "degrading functor" relating these two categories.

1.2. Soergel bimodules. We need to explain what we mean by "Soergel bimodules" in our present context. Ordinarily, Soergel bimodules are attached to a Coxeter system (W', S) and a suitable representation V of W', and defined as a certain category of graded $\mathscr{O}(V)$ -bimodules, see [So]. The case we are interested in here is when (W', S) is the affine Weyl group¹ attached to G with its standard Coxeter generators, and V is the dual of the Lie algebra of a fixed maximal torus T_{\Bbbk}^{\vee} in G_{\Bbbk}^{\vee} , with the action factoring through the canonical action of the (finite) Weyl group $W_{\rm f}$ of $(G_{\Bbbk}^{\vee}, T_{\Bbbk}^{\vee})$. This representation does not satisfy the technical conditions imposed in [So], so that a different definition of Soergel bimodules is required.

Various alternative definitions of categories of Soergel bimodules have been proposed in the recent years by Fiebig [Fi] (involving sheaves on moment graphs), Elias–Williamson [EW] (in terms of a monoidal category defined by generators and relations involving planar diagrams), Juteau–Mautner–Williamson [JMW] (involving parity complexes) or Abe [Ab1] (involving bimodules with extra structures). Some of these definitions apply in our context, and give rise to essentially equivalent categories; in fact some will be used in the body of the paper. But they all have one drawback, which is that they do not "see" an essential ingredient that is specific to the case of affine Weyl groups (rather than a general Coxeter system), namely the relation with the geometric Satake equivalence [MV] and the Langlands

¹The main player in this paper will rather be the *extended* affine Weyl group, for which we reserve the notation W.

dual group G_{\Bbbk}^{\vee} . For that reason we will consider a different (although equivalent) realization of the category of Soergel bimodules, in terms of certain representations of the universal centralizer associated with G_{\Bbbk}^{\vee} . Earlier instances of relations between Soergel bimodules for affine Weyl groups and representations of universal centralizers can be found in [Dod, MR, BR2].

1.3. Categories of perverse sheaves. Now we can start explaining the definition of the categories we aim at describing. Let \mathbb{F} be an algebraically closed field of characteristic p > 0, and let G be a connected reductive algebraic group over \mathbb{F} , with a choice of (negative) Borel subgroup B and maximal torus $T \subset B$. Let z be an indeterminate, and let LG and L^+G be the associated loop group and arc group, i.e. the group ind-scheme, resp. group scheme, over \Bbbk representing the functor

$$R \mapsto G(R((z))), \text{ resp. } R \mapsto G(R[[z]]).$$

Let also $I \subset L^+G$ be the Iwahori subgroup obtained as the preimage of B under the canonical morphism $L^+G \to G$, and let I_u be its pro-unipotent radical, namely the preimage of the unipotent radical U of B. The affine flag variety Fl_G is defined as the fppf quotient LG/I; it is known to be represented by an ind-projective indscheme. For technical reasons we will also consider the fppf quotient $\widetilde{Fl}_G := LG/I_u$; this quotient is represented by an ind-scheme of ind-finite type, and the natural morphism $\widetilde{Fl}_G \to Fl_G$ is a (Zariski locally trivial) T-torsor.

Let k be an algebraic closure of \mathbb{F}_{ℓ} , where ℓ is a prime number different from p. We will denote by $\mathsf{D}_{I,I}$ and $\mathsf{D}_{I_u,I}$ the I-equivariant and I_u -equivariant derived categories of constructible étale k-sheaves on Fl_G , respectively. We will also denote by $\mathsf{P}_{I,I}$ and $\mathsf{P}_{I_u,I}$ the hearts of the perverse t-structures on these categories. The category $\mathsf{D}_{I,I}$ admits a natural convolution operation \star_I , which endows it with a monoidal structure.

Let Gr_G be the affine Grassmannian of G, defined as the fppf quotient LG/L^+G (which is, again, represented by an ind-projective ind-scheme over \mathbb{F}). Denote by D_{L^+G,L^+G} the L^+G -equivariant derived category of constructible étale k-sheaves on Gr_G , and by P_{L^+G,L^+G} the subcategory of perverse sheaves. This category is the main ingredient in the geometric Satake equivalence. Namely, convolution defines a monoidal product \star_{L^+G} on D_{L^+G,L^+G} , which turns out to be exact with respect to the perverse t-structure, hence to induce a monoidal product on P_{L^+G,L^+G} . The geometric Satake equivalence, proved in this setting in [MV], provides an equivalence of monoidal categories

(1.1)
$$(\mathsf{P}_{L^+G,L^+G},\star_{L^+G}) \cong (\mathsf{Rep}(G^{\vee}_{\Bbbk}),\otimes)$$

where G_{\Bbbk}^{\vee} is the connected reductive algebraic group over \Bbbk which is Langlands dual to G and $\operatorname{Rep}(G_{\Bbbk}^{\vee})$ is its category of finite-dimensional algebraic representations.

A way to "upgrade" L^+G -equivariant perverse sheaves on Gr_G to I-equivariant perverse sheaves on Fl_G has been proposed by Gaitsgory [Ga1]. More specifically, this construction provides a canonical t-exact monoidal functor

$$\mathsf{Z}: (\mathsf{D}_{L^+G,L^+G},\star_{L^+G}) \to (\mathsf{D}_{\mathrm{I},\mathrm{I}},\star_{\mathrm{I}}),$$

which admits various favorable properties and structures (in particular, a "monodromy" automorphism) that will not be recalled in detail here.

Our goal is therefore to provide "combinatorial" descriptions of the categories $\mathsf{P}_{I,I}$ and $\mathsf{P}_{I_u,I}$, in a way compatible with the functor Z and the geometric Satake equivalence.

$\mathbf{A}_n \ (n \ge 1)$	$\mathbf{B}_n \ (n \ge 2)$	$\mathbf{C}_n \ (n \ge 3)$	$\mathbf{D}_n \ (n \ge 4)$	\mathbf{E}_{6}	\mathbf{E}_7	\mathbf{E}_8	\mathbf{F}_4	\mathbf{G}_2
1	n	2	2	3	19	31	3	3

FIGURE 1.1. Bounds on ℓ

1.4. The equivariant regular quotient. A first step in this direction has been obtained (jointly with L. Rider) in [BRR]. Contrary to the case of $\mathsf{D}_{L+G,L+G}$, the monoidal structure on $\mathsf{D}_{I,I}$ is unfortunately not compatible with the perverse t-structure, in the sense that \star_{I} is neither left nor right exact. In order to take advantage of this structure while staying in the world of abelian categories, we consider a category that we call the "regular quotient" $\mathsf{P}_{I,I}^0$ of $\mathsf{P}_{I,I}$. Recall that the simple objects in $\mathsf{P}_{I,I}$ are in a canonical bijection with the elements in the extended affine Weyl group W associated with G. Then $\mathsf{P}_{I,I}^0$ is defined as the Serre quotient of $\mathsf{P}_{I,I}$ by the Serre subcategory generated by the simple objects whose label has positive length (equivalently, whose support is not a point). The simple objects in $\mathsf{P}_{I,I}^0$ are uninteresting (they are labeled by length-0 elements in W), but the extensions between these objects are more subtle. The convolution product \star_{I}^0 on $\mathsf{P}_{I,I}^0$, which is exact on both sides.

The main result of [BRR] is the following claim. Here we denote by \mathscr{Z}^0 the composition

$$\mathsf{Rep}(G^{\vee}_{\Bbbk}) \xrightarrow{(1.1)} \mathsf{P}_{L^+G, L^+G} \xrightarrow{\mathsf{Z}} \mathsf{P}_{\mathrm{I},\mathrm{I}} \to \mathsf{P}^0_{\mathrm{I},\mathrm{I}}$$

where the third functor is the natural quotient functor. (This composition is easily seen to be monoidal.)

Theorem 1.1. Assume that the following conditions are satisfied:

- (1) the quotient of $X^*(T)$ by the root lattice of (G,T) is free;
- (2) the quotient of $X_*(T)$ by the coroot lattice of (G,T) has no ℓ -torsion;
- (3) for any indecomposable factor in the root system of (G,T), ℓ is strictly bigger than the corresponding value in Figure 1.1.

Let $\mathbf{u} \in G_{\Bbbk}^{\vee}$ be a regular unipotent element. There exists an equivalence of monoidal categories

$$\Phi_{\mathrm{I},\mathrm{I}}:(\mathsf{P}^0_{\mathrm{I},\mathrm{I}},\star^0_{\mathrm{I}})\xrightarrow{\sim}(\mathsf{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})),\otimes)$$

such that $\Phi_{\mathrm{I},\mathrm{I}} \circ \mathscr{Z}^0$ identifies with the restriction functor $\operatorname{\mathsf{Rep}}(G^{\vee}_{\Bbbk}) \to \operatorname{\mathsf{Rep}}(Z_{G^{\vee}_{\Bbbk}}(\mathsf{u})).$

- Remark 1.2. (1) The first two assumptions in Theorem 1.1 are rather standard. The third assumption is an artefact of the method of proof of one of the crucial claims in [BRR]; we expect that this assumption can be weakened at least to the requirement that ℓ is good for G.
 - (2) For V in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$, the functor $\Phi_{I,I}$ sends the monodromy automorphism of $\mathscr{Z}^{0}(V)$ to the action of \mathfrak{u} on V.

The proof of Theorem 1.1 is based on a general result regarding central functors whose domain is a category of representations of an affine group scheme over a field; see [BRR, Lemma 3.1] for a precise statement. (This statement first appeared in [B1].) This statement immediately provides an equivalence between a certain full subcategory of $\mathsf{P}^0_{I,I}$ and the category of representations of a closed subgroup scheme of G^{\vee}_{\Bbbk} ; what remains to be checked in order to obtain the theorem is that this full subcategory is the whole of $\mathsf{P}^0_{I,I}$, and that the subgroup is $Z_{G_k^{\vee}}(\mathsf{u})$ for a certain regular unipotent element u . Note that the freedom in the choice of the element u is closely related to the fact that there does not exist any "canonical" fiber functor on the category $\mathsf{P}^0_{I,I}$; in fact, a choice of such a fiber functor is essentially equivalent to a choice of a regular unipotent element u .

1.5. The monodromic regular quotient. Theorem 1.1 has a clean formulation, but it has the drawback that the category $D_{I,I}$ cannot be recovered from the regular quotient $P_{I,I}^0$. In order to solve this issue one might be tempted to play the same game with the category $D_{I_{u,I}}$ instead of $D_{I,I}$. (In fact, standard arguments guarantee that if $P_{I_u,I}^0$ is defined by the same procedure as $P_{I,I}^0$, then the quotient functor $P_{I_u,I} \rightarrow P_{I_u,I}^0$ is fully faithful on tilting perverse sheaves, and the category $D_{I_u,I}$ identifies with the bounded homotopy category of the category of tilting perverse sheaves. In this sense, $D_{I_u,I}$ can be recovered from $P_{I_u,I}^0$.)

However, in this process one looses the monoidal product, which was crucial for the construction of the functor $\Phi_{I,I}$. To remedy this, one needs to work instead with the category D_{I_u,I_u} which is the full triangulated subcategory of the derived category of constructible complexes of k-sheaves on \widetilde{Fl}_G generated by complexes obtained by pullback from I_u -equivariant complexes on Fl_G . This category admits a natural perverse t-structure, whose heart will be denoted P_{I_u,I_u} . Once again the simple objects in P_{I_u,I_u} are in bijection with W, and we can define the monodromic regular quotient $P^0_{I_u,I_u}$ as the quotient by the Serre subcategory generated by simple objects whose label has positive length. There exists a natural convolution product \star_{I_u} on D_{I_u,I_u} , which is not exact, but which induces in a natural way a monoidal product $\star^0_{I_u}$ on $P^0_{I_u,I_u}$. (This product is only right exact on both sides.) The pullback functor $D_{I,I} \rightarrow D_{I_u,I_u}$ induces an exact monoidal functor

(1.2)
$$(\mathsf{P}^0_{\mathrm{I},\mathrm{I}},\star^0_{\mathrm{I}}) \to (\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},\star^0_{\mathrm{I}_{\mathrm{u}}}).$$

Compared with $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$, the category $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ has two new "directions of deformation," corresponding to the replacement of I by I_{u} on both sides. On the geometric side, the corresponding process is the replacement of $Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})$ by a group scheme constructed out of the "universal centralizer" over G_{\Bbbk}^{\vee} . This universal centralizer is an affine group scheme over G_{\Bbbk}^{\vee} whose fiber over a closed point $g \in G_{\Bbbk}^{\vee}$ is the scheme-theoretic centralizer $Z_{G_{\Bbbk}^{\vee}}(g)$. Its restriction to the regular locus in G_{\Bbbk}^{\vee} is smooth. We consider a "Steinberg section" $\Sigma \subset G_{\Bbbk}^{\vee}$, a certain section of the adjoint quotient $G_{\Bbbk}^{\vee} \to G_{\Bbbk}^{\vee}/G_{\Bbbk}^{\vee} \cong T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$ which consists of regular elements. (Here, T_{\Bbbk}^{\vee} is the canonical maximal torus in G_{\Bbbk}^{\vee} , and W_{f} is the Weyl group of $(G_{\Bbbk}^{\vee}, T_{\Bbbk}^{\vee})$, which identifies with the Weyl group of (G, T).) The restriction \mathbb{J}_{Σ} of the universal centralizer to Σ is therefore a smooth affine group scheme over the affine scheme Σ . We consider $\mathbf{D} := T_{\Bbbk}^{\vee} \times T_{\Bbbk}^{\vee}/W_{\mathrm{f}} T_{\Bbbk}^{\vee}$, seen as a scheme over Σ via the identification $\Sigma \cong T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$, and set

$$\mathbb{J}_{\mathbf{D}} := \mathbf{D} \times_{\Sigma} \mathbb{J}_{\Sigma},$$

a smooth affine group scheme over **D**. We consider the category

$$\mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$$

of representations of $\mathbb{J}_{\mathbf{D}}$ whose underlying $\mathscr{O}_{\mathbf{D}}$ -module is coherent and set-theoretically supported on $\{(e, e)\} \subset \mathbf{D}$. The tensor product of $\mathscr{O}(T_{\Bbbk}^{\vee})$ -bimodules endows this category with a monoidal product \circledast . The point in Σ corresponding to the image of the unit $e \in T_{\Bbbk}^{\vee}$ in $T_{\Bbbk}^{\vee}/W_{\rm f}$ is a regular unipotent element u, and by construction the fiber of $\mathbb{J}_{\mathbf{D}}$ over (e, e) is $Z_{G_{\Bbbk}^{\vee}}(u)$. We therefore have a natural exact monoidal functor

(1.3)
$$\operatorname{Rep}(Z_{G_{\mathbb{H}}^{\vee}}(\mathsf{u})) \to \operatorname{Rep}_{0}(\mathbb{J}_{\mathbf{D}})$$

induced by pushforward along the embedding $\{(e, e)\} \hookrightarrow \mathbf{D}$.

The first main result of the present paper is the following claim.

Theorem 1.3. Assume that the conditions in Theorem 1.1 are satisfied. There exists an equivalence of monoidal categories

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}:\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\xrightarrow{\sim}\mathsf{Rep}_{0}(\mathbb{J}_{\mathbf{D}})$$

such that the following diagram commutes:

$$\begin{array}{c|c} \mathsf{P}^{0}_{\mathrm{I},\mathrm{I}} & \xrightarrow{\Phi_{\mathrm{I},\mathrm{I}}} & \mathsf{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})) \\ (1.2) & & & & \\ \mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} & \xrightarrow{\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}} & \mathsf{Rep}_{0}(\mathbb{J}_{\mathbf{D}}). \end{array}$$

The general result that was the basis for the proof of Theorem 1.1 has no version for group schemes over a base. Our proof of Theorem 1.3 therefore follows a different, and more specific, approach. We make the functor appearing in Theorem 1.1 more explicit, and then provide an explicit "deformation" of this functor by deforming each of its constituents. Precisely defining these deformations requires recalling and generalizing a number of known tools in this domain (in particular, from [B2]), and proving that the functor constructed in this way has the expected properties turns out to be long and technical; these tasks occupy a large part of the paper.

The functor Φ_{I_u,I_u} also admits some compatibility properties with an appropriate version of the central functor Z. These properties are however more difficult to spell out, and will not be discussed in this introduction.

Remark 1.4. The action of G on the base point in $\widetilde{\mathrm{Fl}}_G$ induces a closed embedding $G/U \hookrightarrow \widetilde{\mathrm{Fl}}_G$. As in the definition of $\mathrm{D}_{\mathrm{I}_u,\mathrm{I}_u}$, we define the category $\mathrm{D}_{U,U}$ as the triangulated subcategory of the derived category of U-equivariant constructible \Bbbk -sheaves on G/U generated by objects obtained by pullback from U-equivariant complexes on G/B. This category admits a natural perverse t-structure, and the simple objects in its heart $\mathsf{P}_{U,U}$ are in bijection with W_{f} . We can then define $\mathsf{P}_{U,U}^0$ as the Serre quotient of $\mathsf{P}_{U,U}$ by the Serre subcategory generated by simple objects whose label has positive length. A variation on the constructions of [BR1] (explained in Section 8) provides an equivalence of monoidal categories between $\mathsf{P}_{U,U}^0$ and the category $\mathsf{Coh}_0(\mathbf{D})$ of coherent sheaves on \mathbf{D} which are set-theoretically supported on $\{(e, e)\}$. The equivalence $\Phi_{\mathrm{I}_u,\mathrm{I}_u}$ is also compatible with this equivalence, using the functor $\mathsf{P}_{U,U}^0 \to \mathsf{P}_{\mathrm{I}_u,\mathrm{I}_u}^0$ induced by pushforward along the closed embedding $G/U \hookrightarrow \widetilde{\mathrm{Fl}}_G$ and the functor $\mathsf{Coh}_0(\mathbf{D}) \to \mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$ sending a coherent sheaf to itself with the trivial structure as a representation.

1.6. Description of tilting perverse sheaves. As explained in §1.5, one motivation for considering the category $P^0_{I_u,I_u}$ is that one can reconstruct the category D_{I_u,I_u} out of it. This can however not be done directly as in the case of $D_{I_u,I}$,

essentially because there is no appropriate notion of tilting perverse sheaves in this category. For this one needs to use a "completed" version $\mathsf{D}_{I_u,I_u}^{\wedge}$ of D_{I_u,I_u} , constructed following a procedure developed by Yun in an appendix to [BY]. The category $\mathsf{D}_{I_u,I_u}^{\wedge}$ is triangulated, it contains D_{I_u,I_u} as a full triangulated subcategory, and it is endowed with a perverse t-structure whose heart is denoted $\mathsf{P}_{I_u,I_u}^{\wedge}$. In $\mathsf{P}_{I_u,I_u}^{\wedge}$ we have an appropriate notion of tilting objects; the full subcategory of tilting perverse sheaves will be denoted $\mathsf{T}_{I_u,I_u}^{\wedge}$. We have a canonical equivalence of categories

(1.4)
$$K^{\mathrm{b}}\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \xrightarrow{\sim} \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},$$

so that $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ can be reconstructed from $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

The monoidal product \star_{I_u} admits a natural "extension" to a (triangulated) monoidal product $\hat{\star}$ on $D^{\wedge}_{I_u,I_u}$, which stabilizes the subcategory $T^{\wedge}_{I_u,I_u}$. With these structures, (1.4) becomes an equivalence of monoidal categories.

On the coherent side, we consider the scheme \mathbf{D}^{\wedge} obtained as the spectrum of the completion of $\mathscr{O}(\mathbf{D})$ with respect to the ideal of the point (e, e). We also set

$$\mathbb{J}^{\wedge}_{\mathbf{D}} := \mathbf{D}^{\wedge} \times_{\mathbf{D}} \mathbb{J}_{\mathbf{D}}.$$

The category $\operatorname{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ of representations of the group scheme $\mathbb{J}^{\wedge}_{\mathbf{D}}$ on coherent $\mathscr{O}_{\mathbf{D}^{\wedge}}$ modules admits once again a canonical monoidal product \circledast . It also contains natural objects \mathscr{B}^{\wedge}_{s} attached to simple reflections $s \in S$, and natural objects $\mathscr{M}^{\wedge}_{\omega}$ attached to length-0 elements ω in W. We define the category $\operatorname{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ of "Soergel representations" of $\mathbb{J}^{\wedge}_{\mathbf{D}}$ as the full subcategory generated under \circledast , direct sums and direct summands by the objects \mathscr{B}^{\wedge}_{s} and $\mathscr{M}^{\wedge}_{\omega}$.

The second main result of the paper is the following claim.

Theorem 1.5. Assume that the conditions in Theorem 1.1 are satisfied. There exists an equivalence of monoidal categories

$$(\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},\widehat{\star}) \xrightarrow{\sim} (\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}}),\circledast).$$

From the description in Theorem 1.5 one obtains (at least in theory) a description of the category $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$ and its subcategory $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, and also of the category $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. Making this description explicit, in terms of coherent sheaves on the Steinberg variety of G_{\Bbbk}^{\vee} , will be the subject of the third part of this project.

The proof of Theorem 1.5 proceeds by describing each side of the equivalence in terms of the categories appearing in Theorem 1.3. (This description is similar to the description of finitely generated modules over the completion A^{\wedge} of a noetherian ring A with respect to an ideal I in terms of sequences of modules over the quotients A/I^m for $m \ge 0$.) On the coherent side, this uses restriction to the various infinitesimal neighborhoods of the preimage of the base point in $T_{\Bbbk}^{\vee}/W_{\rm f}$. On the perverse side, this uses some "truncation" functors which "kill" powers of the ideal of this base point acting by monodromy.

1.7. **Applications.** We finish the paper with some direct applications of Theorem 1.5. The first of these applications is concerned with the study of tilting perverse sheaves in $P_{I_u,I}$. The general theory of highest weight categories shows that isomorphism classes of indecomposable tilting perverse sheaves are in a canonical bijection with W; the indecomposable object attached to $w \in W$ will be denoted

 \mathscr{T}_w . The main combinatorial information one might hope to compute on these objects is encoded in the multiplicities $(\mathscr{T}_w : \Delta_y^{\mathrm{I}})$ of a given standard perverse sheaf Δ_y^{I} in a standard filtration of \mathscr{T}_w .

The following claim is obtained as a consequence of Theorem 1.5.

Corollary 1.6. Assume that the conditions in Theorem 1.1 are satisfied, and assume moreover that either $G = GL_n$ or ℓ is very good for G. For any $y, w \in W$ we have

$$(\mathscr{T}_w:\Delta^1_y) = {}^{\ell}h_{y,w}(1),$$

where the polynomials $({}^{\ell}h_{y,w} : y, w \in W)$ are the ℓ -Kazhdan-Lusztig polynomials attached to W (and its standard realization) as in [JW].

Similar descriptions of tilting multiplicities in terms of Kazhdan–Lusztig-type polynomials were previously obtained in the characteristic-0 setting (for the flag variety of any Kac–Moody group, see [Yu, BY]), and for positive-characteristic coefficients on "finite" flag varieties G/B, see [AR1]. Such formulas are also proved for "mixed" perverse sheaves (on the flag variety of any Kac–Moody group) in [AMRW]; from such information one can however not directly deduce information on the "ordinary" perverse sheaves we consider here.

In order to prove Corollary 1.6 we relate the category $\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ to the "ordinary" Hecke category attached to W as in [EW, JW, RW1]. This involves a comparison of $\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ with an "additive" version defined in terms of the universal centralizer over the Lie algebra of G^{\vee}_{\Bbbk} , and then the use of the recent "bimodule theoretic" description of the Hecke category by Abe [Ab1]. (This second step has already been used in our previous paper [BR2].)

These methods also provide an alternative way of constructing the "Koszul duality" of [AMRW] in the affine case, following the method used in the "finite" case in [AR2]; as a byproduct, we also obtain a construction of a "degrading" functor relating mixed perverse sheaves to ordinary perverse sheaves (again, in the affine case); see §13.4 for details.

Finally, in §13.5 we explain that our results also imply a weak form of the Finkelberg–Mirković conjecture [FM] relating the principal block of a semisimple algebraic group with Iwahori-constructible perverse sheaves on the affine Grassmannian of the dual group, which provides yet another proof of Lusztig's conjecture on characters of simple modules in large characteristics. (A stronger form of the Finkelberg–Mirković conjecture will be obtained in the third part of this project.)

1.8. **Contents.** In Section 2 we prove or recall a number of facts regarding the geometry of the (multiplicative) Steinberg variety. In particular we explain the construction of the Steinberg section and recall some facts about the universal centralizer group scheme. In Section 3 we explain various incarnations of the "Hecke category" attached to an affine Weyl group.

Sections 4–7 are devoted to reminders on essentially known constructions regarding categories of constructible sheaves on affine flag varieties. More specifically, in Section 4 we introduce these categories, recall Gaitsgory's construction relating the Satake category to Iwahori-equivariant perverse sheaves on the affine flag variety, and recall the construction of the main player of [BRR] (the equivariant regular quotient) and the main result of that paper (an equivalence of categories relating this equivariant regular quotient to representations of the centralizer of a regular unipotent element in the dual group). In Section 5 we introduce one of the main

players of the present paper, the "monodromic regular quotient" category. In Section 6 we recall Yun's construction of the "free-monodromic derived category" and its perverse t-structure. Finally, in Section 7 we discuss a free-monodromic variant of Gaitsgory's construction, following a similar construction in [B2].

Sections 8–11 contain our main constructions, and the proofs of our main results. First, in Section 8 we reinterpret the results of [BR1] from a slightly different perspective, thereby providing a "reconstruction" of the base scheme **D** from perverse sheaves on G/U. In Section 9 we prove some technical results regarding a "truncation" operation on perverse sheaves that will be required later. In Section 10 we construct our functor Φ_{I_u,I_u} and prove a slightly more precise version of Theorem 1.3 (see Theorem 10.1). Then in Section 11 we explain how to deduce a slightly more precise version of Theorem 1.5 (see Theorem 11.2).

In Section 12 we explain variants of Theorems 1.3 and 1.5 which describe the more familiar category $P_{I_u,I}$. Finally, in Section 13 we prove the applications of our results described in §1.7. (In particular, we prove Corollary 1.6 in Corollary 13.7.)

The paper finishes with four appendices, each discussing a technical construction which is required in the course of our proofs.

1.9. Notation and conventions. Unless otherwise specified, a "module" will mean a left module, and a "comodule" will mean a right comodule. If A is a ring, we will denote by Mod(A), resp. $Mod_r(A)$, the category of A-modules, resp. of right A-modules. If A is noetherian, resp. right noetherian, the subcategory of finitely generated modules will be denoted $Mod^{fg}(A)$, resp. $Mod_r^{fg}(A)$.

Given an affine group scheme H over a noetherian scheme X, we will denote by $\operatorname{\mathsf{Rep}}^{\infty}(H)$ the category of representations of H, and by $\operatorname{\mathsf{Rep}}(H)$ the full subcategory of representations whose underlying \mathscr{O}_X -module is coherent. In most cases we will encounter below X will be affine; in this case we will identify $\operatorname{\mathsf{Rep}}^{\infty}(H)$ with the category of comodules over the $\mathscr{O}(X)$ -Hopf algebra $\mathscr{O}(H)$, and $\operatorname{\mathsf{Rep}}(H)$ with the subcategory of comodules which are finitely generated as $\mathscr{O}(X)$ -modules.

In this paper we will make extensive use of the theory of pro-objects and indobjects in categories, for which we refer to [KS, Chap. 6]. (For us, all pro-objects will be parametrized by $\mathbb{Z}_{\geq 0}$ with the standard order. On the other hand, indobjects will be parametrized by arbitrary filtrant categories.) In particular, we will repeatedly use the property that any functor "extends" in a canonical way to a functor between categories of pro-objects or ind-objects; see [KS, Proposition 6.1.9]. We will also use the fact that the category of ind-objects in an abelian category is itself abelian; see [KS, Theorem 8.6.5(i)].

1.10. Acknowledgements. We thank A. Bouthier for useful discussions on Steinberg's sections, and S. Cotner for enlightening exchanges on centralizers of regular elements and sending a preliminary version of [Co]. Finally we thank P. Achar for stimulating discussions.

This paper is the second step of a project we initially started as a joint work with L. Rider. We thank her for her early contributions to this program.

R.B. was supported by NSF Grant No. DMS-1601953. This project has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (S.R., grant agreement No. 101002592).

2. Coherent sheaves on the Steinberg variety

In this section we collect a number of results on the geometry of various varieties associated with a connected reductive algebraic group, and coherent sheaves on such varieties, that will be required in later sections.

2.1. Notation. We fix an algebraically closed field \Bbbk of characteristic ℓ , and a connected reductive algebraic group \mathbf{G} over \Bbbk whose derived subgroup $\mathscr{D}\mathbf{G}$ is simply connected. We choose a Borel subgroup $\mathbf{B} \subset \mathbf{G}$ and a maximal torus $\mathbf{T} \subset \mathbf{B}$, and denote by \mathbf{U} the unipotent radical of \mathbf{B} . The respective Lie algebras of \mathbf{G} , \mathbf{B} , \mathbf{T} will be denoted \mathbf{g} , \mathbf{b} , \mathbf{t} . The Borel subgroup of \mathbf{G} opposite to \mathbf{B} (with respect to \mathbf{T}) will be denoted \mathbf{B}^+ , and its unipotent radical will be denoted \mathbf{U}^+ ..

We will denote by \mathbf{W}_{f} the Weyl group of (\mathbf{G}, \mathbf{T}) , and by \mathbf{U} the unipotent radical of \mathbf{B} . We will also denote by $\mathfrak{R} \subset X^*(\mathbf{T})$ the root system of (\mathbf{G}, \mathbf{T}) , and by $\mathfrak{R}^{\vee} \subset X_*(\mathbf{T})$ the corresponding coroots; for any root $\beta \in \mathfrak{R}$, we will denote by β^{\vee} the associated coroot. The choice of \mathbf{B} determines a system $\mathfrak{R}_+ \subset \mathfrak{R}$ of positive roots, chosen so that the \mathbf{T} -weights on Lie (\mathbf{U}) are the negative roots. The associated basis of \mathfrak{R} will be denoted $\mathfrak{R}_{\mathrm{s}}$. We will denote by $X^*_+(\mathbf{T}) \subset X^*(\mathbf{T})$ the submonoid of weights which are dominant with respect to \mathfrak{R}_+ , and by \preceq the order on $X^*(\mathbf{T})$ such that $\lambda \preceq \mu$ iff $\mu - \lambda$ is a sum of positive roots. The choice of \mathfrak{R}_+ also determines a system $\mathbf{S}_{\mathrm{f}} \subset \mathbf{W}_{\mathrm{f}}$ of Coxeter generators; the longest element with respect to this structure will be denoted w_{\circ} .

The following classical result of Steinberg (after earlier work of Pittie, see [S2]) will be crucial in later sections.

Theorem 2.1. The $\mathcal{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})$ -module $\mathcal{O}(\mathbf{T})$ is free of rank $\#\mathbf{W}_{\mathrm{f}}$.

Remark 2.2. In [S2] the author assumes that the group under consideration is semisimple (and simply connected). However the proof applies in above setup, as checked in detail in [Go2, §10.1.1].

For any connected reductive algebraic group \mathbf{H} over \Bbbk , we will denote by $\mathbf{H}_{\text{reg}} \subset \mathbf{H}$ the open subscheme of regular elements, i.e. the unique open subscheme whose \Bbbk -points are the elements $g \in \mathbf{H}$ whose centralizer $Z_{\mathbf{H}}(g)$ has dimension the rank of \mathbf{G} (i.e. the minimal possible dimension). If \mathbf{H}' is another connected reductive group and $\varphi : \mathbf{H}' \to \mathbf{H}$ is a finite central isogeny, then for any closed point $h \in \mathbf{H}'$ we have dim $Z_{\mathbf{H}'}(h) = \dim Z_{\mathbf{H}}(\varphi(h))$ (see e.g. [Ku, Proposition 2.3]); as a consequence we have

(2.1)
$$\mathbf{H}'_{\mathrm{reg}} = \varphi^{-1}(\mathbf{H}_{\mathrm{reg}}).$$

2.2. The adjoint quotient and Steinberg's section. Consider the adjoint quotient G/G. It is a classical fact that the embedding $T \subset G$ induces an isomorphism

(2.2)
$$\mathbf{T}/\mathbf{W}_{\mathrm{f}} \xrightarrow{\sim} \mathbf{G}/\mathbf{G}_{\mathrm{f}}$$

see e.g. [Lee] for a proof of this theorem over any commutative ring (and for any reductive group admitting a maximal torus). We will denote by $\chi : \mathbf{G} \to \mathbf{T}/\mathbf{W}_{\mathrm{f}}$ the composition of the adjoint quotient morphism with the identification (2.2).

The algebra $\mathscr{O}(\mathbf{G}/\mathbf{G})$ admits a basis (as a k-vector space) parametrized by $X^*_+(\mathbf{T}) \subset X^*(\mathbf{T})$ and defined as follows. For any M in the category $\operatorname{Rep}(\mathbf{G})$ of finite-dimensional algebraic \mathbf{G} -modules, we denote by $\operatorname{ch}(M) : \mathbf{G} \to \mathbb{A}^1_{\mathbb{k}}$ the function sending $g \in \mathbf{G}$ to the trace of its action on M. For $\lambda \in X^*_+(\mathbf{T})$ we denote

by $\mathsf{L}(\lambda)$ the simple **G**-module with highest weight λ , i.e. the socle of the induced module $\operatorname{Ind}_{\mathbf{B}}^{\mathbf{G}}(\lambda)$. Then $(\operatorname{ch}(\mathsf{L}(\lambda)) : \lambda \in X_{+}^{*}(\mathbf{T}))$ is a k-basis of $\mathscr{O}(\mathbf{G}/\mathbf{G})$. More generally, for any family $(M_{\lambda} : \lambda \in X_{+}^{*}(\mathbf{T}))$ of **G**-modules such that

(2.3)
$$[M_{\lambda} : \mathsf{L}(\lambda)] = 1 \quad \text{and} \quad [M_{\lambda} : \mathsf{L}(\mu)] \neq 0 \Rightarrow \mu \preceq \lambda,$$

the family $(ch(M_{\lambda}) : \lambda \in X^*(\mathbf{T}))$ is a k-basis of $\mathscr{O}(\mathbf{G}/\mathbf{G})$.

Under our assumption that $\mathscr{D}\mathbf{G}$ is simply connected, the adjoint quotient can be described more explicitly, as follows. First, recall that (without any assumption) the quotient $\mathbf{G}/\mathscr{D}\mathbf{G}$ of \mathbf{G} by its normal subgroup $\mathscr{D}\mathbf{G}$ is a torus, whose lattice of characters is determined by the fact that the pullback under the composition

$$(2.4) \mathbf{T} \hookrightarrow \mathbf{G} \to \mathbf{G} / \mathscr{D}\mathbf{G}$$

provides an identification

(2.5)
$$X^*(\mathbf{G}/\mathscr{D}\mathbf{G}) \xrightarrow{\sim} \{\lambda \in X^*(\mathbf{T}) \mid \forall \alpha \in \mathfrak{R}_{\mathrm{s}}, \langle \lambda, \alpha^{\vee} \rangle = 0\}$$

If λ belongs to the right-hand side, then the **G**-module $L(\lambda)$ is one-dimensional, with the **G**-action given by the associated character of **G**.

On the other hand, the intersection $\mathbf{T} \cap \mathscr{D}\mathbf{G}$ is a maximal torus of $\mathscr{D}\mathbf{G}$, and we have a surjective restriction morphism $X^*(\mathbf{T}) \to X^*(\mathbf{T} \cap \mathscr{D}\mathbf{G})$. For each $\alpha \in \mathfrak{R}$, the coroot α^{\vee} takes values in $\mathbf{T} \cap \mathscr{D}\mathbf{G}$; the map $\langle -, \alpha^{\vee} \rangle$ therefore factors through a map $X^*(\mathbf{T} \cap \mathscr{D}\mathbf{G}) \to \mathbb{Z}$, which identifies with the similar map for the root $\alpha_{|\mathbf{T} \cap \mathscr{D}\mathbf{G}}$ of $\mathscr{D}\mathbf{G}$. For each $\alpha \in \mathfrak{R}_s$ we have the fundamental weight $\varpi_{\alpha} \in X^*(\mathbf{T} \cap \mathscr{D}\mathbf{G})$, which is characterized by the property that $\langle \varpi_{\alpha}, \beta^{\vee} \rangle = \delta_{\alpha,\beta}$ for $\beta \in \mathfrak{R}_s$. Let us fix, for each $\alpha \in \mathfrak{R}_s$, a lift $\omega_{\alpha} \in X^*(\mathbf{T})$ of ϖ_{α} . Then, using the identification (2.5) we obtain an isomorphism of \mathbb{Z} -modules

(2.6)
$$\mathbb{Z}^{\mathfrak{R}_{s}} \times X^{*}(\mathbf{G}/\mathscr{D}\mathbf{G}) \xrightarrow{\sim} X^{*}(\mathbf{T})$$

given by

$$((m_{\alpha}: \alpha \in \mathfrak{R}_{\mathrm{s}}), \lambda) \mapsto \lambda + \sum_{\alpha \in \mathfrak{R}_{\mathrm{s}}} m_{\alpha} \omega_{\alpha},$$

which in turn provides an identification

(2.7)
$$\mathbf{T} \cong (\mathbb{G}_{\mathrm{m}})^{\mathfrak{R}_{\mathrm{s}}} \times \mathbf{G}/\mathscr{D}\mathbf{G},$$

such that $(\mathbb{G}_{\mathbf{m}})^{\mathfrak{R}_{\mathbf{s}}} \times \{e\}$ corresponds to $\mathbf{T} \cap \mathscr{D}\mathbf{G}$ and the projection $\mathbf{T} \to \mathbf{G}/\mathscr{D}\mathbf{G}$ coincides with (2.4).

For $\alpha \in \mathfrak{R}_{s}$, we set $\chi_{\alpha} := ch(\mathsf{L}(\omega_{\alpha})) : \mathbf{G} \to \mathbb{A}^{1}_{\Bbbk}$.

Lemma 2.3. The morphisms $(\chi_{\alpha} : \alpha \in \mathfrak{R}_{s})$ and the projection $\mathbf{G} \to \mathbf{G}/\mathscr{D}\mathbf{G}$ induce an isomorphism

$$\mathbf{G}/\mathbf{G} \xrightarrow{\sim} \mathbb{A}^{\mathfrak{R}_{\mathrm{s}}}_{\Bbbk} \times (\mathbf{G}/\mathscr{D}\mathbf{G}).$$

Proof. Since $\mathbf{G}/\mathscr{D}\mathbf{G}$ is a torus we have

$$\mathscr{O}(\mathbf{G}/\mathscr{D}\mathbf{G}) = \bigoplus_{\nu \in X^*(\mathbf{G}/\mathscr{D}\mathbf{G})} \Bbbk \nu,$$

and our morphism is induced by the morphism

$$\mathscr{O}(\mathbf{G}/\mathscr{D}\mathbf{G}) \otimes_{\Bbbk} \Bbbk[X_{\alpha} : \alpha \in \mathfrak{R}_{\mathrm{s}}] \to \mathscr{O}(\mathbf{G}/\mathbf{G})$$

sending $\nu \otimes \prod_{\alpha} X_{\alpha}^{m_{\alpha}}$ to

(2.8)
$$\operatorname{ch}(\mathsf{L}(\nu)\otimes\bigotimes_{\alpha}\mathsf{L}(\omega_{\alpha})^{\otimes m_{\alpha}}).$$

Now (2.6) restricts to a bijection

 $(\mathbb{Z}_{\geq 0})^{\mathfrak{R}_{\mathrm{s}}} \times X^*(\mathbf{G}/\mathscr{D}\mathbf{G}) \xrightarrow{\sim} X^*_+(\mathbf{T}).$

The family of characters (2.8) can therefore be considered as parametrized by $X^*_+(\mathbf{T})$. As such, the corresponding family of **G**-modules satisfies the conditions spelled out in (2.3), and these characters therefore form a basis of $\mathscr{O}(\mathbf{G}/\mathbf{G})$. Our algebra morphism sends a k-basis to a k-basis, hence is an isomorphism.

We now explain how to construct a "Steinberg section" for χ , i.e. a closed subscheme $\Sigma \subset \mathbf{G}$ contained in \mathbf{G}_{reg} such that the composition $\Sigma \hookrightarrow \mathbf{G} \xrightarrow{\chi} \mathbf{T}/\mathbf{W}_{\text{f}}$ is an isomorphism. (This construction is due to Steinberg [S1] in the case \mathbf{G} is semisimple; the extension to reductive groups is due to De Concini–Maffei [DCM].) Let us fix a numbering $(\alpha_1, \dots, \alpha_r)$ of \mathfrak{R}_s . For $i \in \{1, \dots, r\}$, we will denote by \mathbf{U}_{α_i} and $\mathbf{U}_{-\alpha_i}$ the root subgroups of \mathbf{G} associated with α_i and $-\alpha_i$ respectively. We will also chose a lift $n_i \in N_{\mathbf{G}}(\mathbf{T})$ of the simple reflection $s_i \in \mathbf{W}_{\mathbf{f}}$ associated with α_i which belongs to $\mathscr{D}\mathbf{G}$. Let us denote by $\mathbf{A} \subset \mathbf{T}$ the subtorus given by the image of $\{e\} \times \mathbf{G}/\mathscr{D}\mathbf{G}$ under the identification (2.7). We then set

$$\Sigma := \mathbf{A} \cdot \mathbf{U}_{\alpha_1} n_1 \cdot (\cdots) \cdot \mathbf{U}_{\alpha_r} n_r$$

Standard properties of the Bruhat decomposition (see e.g. [Hu, §4.15] for details) show that the map

$$(u_1, \cdots, u_r) \mapsto u_1 n_1 u_2 n_2 \cdots u_r n_r \cdot (n_1 \cdots n_r)^{-1}$$

induces a closed embedding $\mathbf{U}_{\alpha_1} \times \cdots \times \mathbf{U}_{\alpha_r} \hookrightarrow \mathbf{U}^+$; this shows that Σ is a closed subscheme of \mathbf{G} , isomorphic to $\mathbf{A} \times \mathbb{A}^r_{\Bbbk}$. The other properties of Σ announced above are proved in the following proposition.

Proposition 2.4. (1) We have $\Sigma \subset \mathbf{G}_{reg}$. (2) The composition

$$\Sigma \hookrightarrow \mathbf{G} \xrightarrow{\chi} \mathbf{T} / \mathbf{W}_{\mathrm{f}}$$

is an isomorphism.

Proof. (1) It is clear that we have

$$\Sigma = \mathbf{U}_{\alpha_1} n_1 \cdot (\cdots) \cdot \mathbf{U}_{\alpha_r} n_r \cdot ((n_1 \cdots n_r)^{-1} \mathbf{A} n_1 \cdots n_r) \subset \mathbf{U}_{\alpha_1} n_1 \cdot (\cdots) \cdot \mathbf{U}_{\alpha_r} n_r \cdot \mathbf{T},$$

so that it is enough to prove that the right-hand side is contained in \mathbf{G}_{reg} . Now if \mathbf{Z} is the neutral component of the reduced center $Z(\mathbf{G})_{\text{red}}$, as explained in [Ja, §1.18] multiplication induces a finite central isogeny

$$\varphi : \mathbf{Z} \times \mathscr{D}\mathbf{G} \to \mathbf{G}.$$

By (2.1) we have

$$\varphi^{-1}(\mathbf{G}_{\mathrm{reg}}) = (\mathbf{Z} \times \mathscr{D}\mathbf{G})_{\mathrm{reg}} = \mathbf{Z} \times (\mathscr{D}\mathbf{G})_{\mathrm{reg}}.$$

Now by Steinberg's results for semisimple groups (see [S1] and [Hu, §4.20]) we have

$$\mathbf{U}_{\alpha_1}n_1 \cdot (\cdots) \cdot \mathbf{U}_{\alpha_r}n_r \cdot (\mathbf{T} \cap \mathscr{D}\mathbf{G}) \subset (\mathscr{D}\mathbf{G})_{\mathrm{reg}}.$$

(Steinberg's results involve elements in $\mathbf{U}_{\alpha_1}n_1 \cdot (\cdots) \cdot \mathbf{U}_{\alpha_r}n_r$, but the choice of the elements n_i can be arbitrary; for the various choices of these elements, the corresponding "Steinberg sections" cover $\mathbf{U}_{\alpha_1}n_1 \cdot (\cdots) \cdot \mathbf{U}_{\alpha_r}n_r \cdot (\mathbf{T} \cap \mathscr{D}\mathbf{G})$.) We deduce that

$$\varphi^{-1}(\mathbf{U}_{\alpha_1}n_1\cdot(\cdots)\cdot\mathbf{U}_{\alpha_r}n_r\cdot\mathbf{T})\subset\varphi^{-1}(\mathbf{G}_{\mathrm{reg}}),$$

and then that $\mathbf{U}_{\alpha_1} n_1 \cdot (\cdots) \cdot \mathbf{U}_{\alpha_r} n_r \cdot \mathbf{T} \subset \mathbf{G}_{\text{reg}}$, as desired.

(2) Let us fix, for any $i \in \{1, \dots, r\}$, an isomorphism $u_{\alpha_i} : \mathbb{A}^1_{\Bbbk} \xrightarrow{\sim} \mathbf{U}_{\alpha_i}$. Then we have an isomorphism

$$\mathbf{A} \times \mathbb{A}^r_{\Bbbk} \xrightarrow{\sim} \Sigma,$$

given by $(a, (c_1, \dots, c_r)) \mapsto au_{\alpha_1}(c_1)n_1u_{\alpha_2}(c_2)n_2\cdots u_{\alpha_r}(c_r)n_r$. Using this isomorphism and that of Lemma 2.3, one can consider the morphism of the proposition as a morphism from $\mathbf{A} \times \mathbb{A}^r_{\mathbb{k}}$ to itself. It is clear from definitions that its composition with the projection $\mathbf{A} \times \mathbb{A}^r_{\mathbb{k}} \to \mathbf{A}$ coincides with this projection.

Now we consider the composition of our morphism with projection on $\mathbb{A}_{\mathbb{k}}^r$. For this, we define a partial order \sqsubseteq on $\{1, \dots, r\}$ by declaring that $i \sqsubset j$ if there exists a dominant weight λ for $(\mathscr{D}\mathbf{G}, \mathbf{T} \cap \mathscr{D}\mathbf{G})$ such that $\varpi_j - \lambda$ is a sum of positive roots and $\langle \lambda, \alpha_i^{\vee} \rangle > 0$. (Here, the positive roots for $\mathscr{D}\mathbf{G}$ are taken as the restrictions of those for \mathbf{G} . For an explanation of why this defines an order, see [Hu, §4.16].) For any $i \in \{1, \dots, r\}$, $a \in \mathbf{A}$ and $(c_1, \dots, c_r) \in \mathbb{A}_{\mathbb{k}}^r$, the value of $\chi_{\alpha_i}(au_{\alpha_1}(c_1)n_1u_{\alpha_2}(c_2)n_2\cdots u_{\alpha_r}(c_r)n_r)$ can be computed as the sum (over the weights λ of $\mathsf{L}(\omega_i)$) of the traces of the linear maps

$$\mathsf{L}(\omega_i)_{\lambda} \hookrightarrow \mathsf{L}(\omega_i) \xrightarrow{au_{\alpha_1}(c_1)n_1u_{\alpha_2}(c_2)n_2\cdots u_{\alpha_r}(c_r)n_r} \mathsf{L}(\omega_i) \twoheadrightarrow \mathsf{L}(\omega_i))$$

where the left, resp. right, morphism is the embedding of, resp. projection on, the λ -weight space of $L(\omega_i)$ (parallel to other weight spaces). The discussion in [Hu, §4.17] shows that this morphism vanishes unless λ is dominant, and that

- (1) if $\lambda = \omega_i$, there exists $d_i \in \mathbb{k}^{\times}$ such that the trace is $d_i c_i \omega_i(a)$;
- (2) otherwise, there exists a polynomial $P_{\lambda} \in \mathbb{k}[X_j : j \sqsubset i]$ such that the trace is $\lambda(a)P_{\lambda}((c_j)_{j \sqsubset i})$.

From this analysis we see that the algebra morphism $\mathscr{O}(\mathbf{A} \times \mathbb{A}^r_{\Bbbk}) \to \mathscr{O}(\mathbf{A} \times \mathbb{A}^r_{\Bbbk})$ induced by our morphism of schemes $\mathbf{A} \times \mathbb{A}^r_{\Bbbk} \to \mathbf{A} \times \mathbb{A}^r_{\Bbbk}$ is an isomorphism, so that the latter morphism is an isomorphism too.

2.3. Application to smoothness results. For any closed point $g \in \mathbf{G}$, we will denote by $Z_{\mathbf{G}}(g) \subset \mathbf{G}$ is the scheme-theoretic centralizer of g in \mathbf{G} . Recall (see [BRR, §§2.1–2.2] for references) that the morphism $h \mapsto hgh^{-1}$ factors through a locally closed immersion $\mathbf{G}/Z_{\mathbf{G}}(g) \to \mathbf{G}$, whose image is denoted $\mathcal{O}(g)$ (and called the adjoint orbit of g); it is a smooth locally closed subscheme in \mathbf{G} , whose set of k-points is the conjugacy class of g in the usual sense. We will denote by χ_{reg} the restriction of χ to \mathbf{G}_{reg} . As a first application of the construction of the Steinberg section we prove the following claim.

Proposition 2.5. The morphism χ_{reg} is smooth. Moreover, for any $x \in \mathbf{G}_{\text{reg}}$ we have

$$(\chi_{\operatorname{reg}})^{-1}(\chi(x)) = \mathcal{O}(x).$$

Proof. By a classical characterization of smooth morphisms (see e.g. [Ha, Proposition III.10.4]), to prove smoothness it suffices to prove that the differential $d_g(\chi)$ is surjective for any closed point $g \in \mathbf{G}_{\text{reg}}$. This property is true by Proposition 2.4(2) if $g \in \Sigma$, hence if g is a conjugate of an element of Σ . Now any fiber of χ contains exactly one regular conjugacy class (see [Hu, §4.14]), and it also contains an element of Σ , which is regular by Proposition 2.4(1). It follows that any regular element in **G** is conjugate to an element of Σ , which finishes the proof that χ_{reg} is smooth. Once this is known, we know that $\mathcal{O}(g)$ and $(\chi_{\text{reg}})^{-1}(\chi(x))$ are smooth, and that

the morphism $\mathcal{O}(g) \to (\chi_{\text{reg}})^{-1}(\chi(x))$, which is a locally closed immersion (see [SP, Tag 07RK]) is a bijection on k-points; it is therefore an equality.

We now consider smoothness of centralizers of regular elements in **G**. For this we will have to assume that the (scheme-theoretic) center $Z(\mathbf{G})$ is smooth; by [BRR, Lemma 2.2], this is equivalent to requiring that $X^*(\mathbf{T})/\mathbb{ZR}$ has no ℓ -torsion.

Lemma 2.6. Assume that $Z(\mathbf{G})$ is smooth. Then for any $g \in \mathbf{G}_{reg}$ the centralizer $Z_{\mathbf{G}}(g)$ is smooth.

Proof. By [Co, Theorem 7.13(1)] the quotient $Z_{\mathbf{G}}(g)/Z(\mathbf{G})$ is smooth, which implies that $Z_{\mathbf{G}}(g)$ is smooth under our assumption. (Here the cited theorem applies to "strongly regular" elements; looking at [Co, Definitions 5.2 and 5.11] one sees that this notion is equivalent to regularity for reductive groups over \Bbbk with simply connected derived subgroup.)

Remark 2.7. Under the additional assumption that ℓ is good for **G**, Lemma 2.6 can also be deduced from the results of [He].

The following smoothness result will also be crucial below.

Proposition 2.8. Assume that $Z(\mathbf{G})$ is smooth. Then the morphism

 $\mathbf{G} \times \Sigma \to \mathbf{G}_{\mathrm{reg}}$

defined by $(g, s) \mapsto gsg^{-1}$ is smooth and surjective.

Proof. To prove smoothness of our morphism, as in the proof of Proposition 2.5, what we need to show is that the differential of this morphism at any closed point of $\mathbf{G} \times \Sigma$ is surjective. By **G**-equivariance, it suffices to do so at points of the form (e, s) with $s \in \Sigma$. For such s, from Proposition 2.4(1) and Proposition 2.5 we obtain that the differential $d_s(\chi)$ is surjective, and that its kernel is $T_s(\mathcal{O}(s))$. Now since the composition $\Sigma \to \mathbf{G} \to \mathbf{T}/\mathbf{W}_{\mathbf{f}}$ is an isomorphism (see Proposition 2.4(2)), $T_s(\Sigma)$ is a complement to the kernel of $d_s(\chi)$, which implies that

(2.9)
$$T_s(\mathbf{G}) = T_s(\mathcal{O}(s)) \oplus T_s(\Sigma).$$

The differential of the morphism in the statement is the sum of the differential at e of the morphism $\mathbf{G} \to \mathbf{G}$ given by $g \mapsto gsg^{-1}$ and the embedding $T_s(\Sigma) \to T_s(\mathbf{G})$. The first of these morphisms can be described as the composition

$$T_e(\mathbf{G}) \to T_s(\mathcal{O}(s)) \to T_s(\mathbf{G}),$$

where the first morphism is the differential of the morphism $\mathbf{G} \to \mathcal{O}(s)$ given by $g \mapsto gsg^{-1}$. The latter morphism identifies with the quotient morphism $\mathbf{G} \to \mathbf{G}/Z_{\mathbf{G}}(s)$, which is smooth by Lemma 2.6 and the comments in [BRR, §2.1]. Its differential is therefore surjective, which finishes the proof in view of (2.9).

Once we know that our morphism is smooth, we know that its image is open (see [SP, Tag 01UA]), so that to prove surjectivity it suffices to prove that this image contains all closed points of \mathbf{G}_{reg} . This property was observed in the course of the proof of Proposition 2.5.

Remark 2.9. Proposition 2.8 does not hold in general if $Z(\mathbf{G})$ is not smooth. For instance, explicit computation shows that when $\mathbf{G} = \mathrm{SL}_2(\mathbb{k})$ the morphism under consideration is not smooth in characteristic 2.

Let us note the following consequence of Proposition 2.8, which will be used in Section 10.

Corollary 2.10. Consider the action of **G** on itself by conjugation, and the induced action on the algebra $\mathscr{O}(\mathbf{G})$. For any **G**-equivariant $\mathscr{O}(\mathbf{G})$ -module M and any $n \in \mathbb{Z}_{>0}$ we have $\operatorname{Tor}_{n}^{\mathscr{O}(\mathbf{G})}(M, \mathscr{O}(\Sigma)) = 0$.

Proof. Since **G** is affine, the category of **G**-equivariant $\mathscr{O}(\mathbf{G})$ -modules is equivalent to that of **G**-equivariant quasi-coherent sheaves on **G**. If we denote by $i: \Sigma \to \mathbf{G}$ the embedding, and consider the derived pullback functor

$$Li^*: D^-\mathsf{QCoh}(\mathbf{G}) \to D^-\mathsf{QCoh}(\Sigma),$$

the claim we want to prove is therefore equivalent to the statement that $Li^*(\mathscr{F})$ is concentrated in degree 0 for any \mathscr{F} in $\mathsf{QCoh}^{\mathbf{G}}(\mathbf{G})$. Now the morphism *i* can be written as a composition

$$\Sigma \xrightarrow{\jmath} \mathbf{G} \times \Sigma \to \mathbf{G}$$

where the first morphism is given by j(s) = (e, s) and the second one is the morphism of Proposition 2.8. Since the latter morphism is smooth (hence flat) and **G**-equivariant (for the action on $\mathbf{G} \times \Sigma$ induced by multiplication on the left on the first factor), to prove the desired statement it suffices to prove that for any \mathscr{G} in $\mathsf{QCoh}^{\mathbf{G}}(\mathbf{G} \times \Sigma)$ the complex $Lj^*(\mathscr{G})$ is concentrated in degree 0. Now if $q: \mathbf{G} \times \Sigma \to \Sigma$ is the morphism of projection on the second factor, the functor q^* induces an equivalence of categories $\mathsf{QCoh}(\Sigma) \xrightarrow{\sim} \mathsf{QCoh}^{\mathbf{G}}(\mathbf{G} \times \Sigma)$; in particular, any object \mathscr{G} of $\mathsf{QCoh}^{\mathbf{G}}(\mathbf{G} \times \Sigma)$ is of the form $q^*\mathscr{M}$ for some \mathscr{M} in $\mathsf{QCoh}(\Sigma)$, and moreover since q is flat we have $q^*\mathscr{M} = Lq^*\mathscr{M}$. We deduce that

$$Lj^*(\mathscr{G}) \cong Lj^*Lq^*\mathscr{M} = \mathscr{M}$$

since $q \circ j = id$; in particular, this complex is indeed concentrated in degree 0. \Box

2.4. Multiplicative Grothendieck, Springer and Steinberg varieties. Recall from [BRR, §2.4] that the *multiplicative Springer resolution* is the induced variety

$$\widetilde{\mathcal{U}} := \mathbf{G} \times^{\mathbf{B}} \mathbf{U},$$

where \mathbf{B} acts on \mathbf{U} via the adjoint action. In this paper we will also consider the *multiplicative Grothendieck resolution*

Groth :=
$$\mathbf{G} \times^{\mathbf{B}} \mathbf{B}$$
,

where \mathbf{B} acts on itself by conjugation. We have a natural projective morphism

$$\nu$$
 : Groth \rightarrow **G**,

defined by $\nu([g:b]) = gbg^{-1}$ for $g \in \mathbf{G}$ and $b \in \mathbf{B}$. Using this morphism we can consider the fiber product

$$\mathsf{St} := \mathsf{Groth} \times_{\mathbf{G}} \mathsf{Groth},$$

which we will call the *multiplicative Steinberg variety*. We also have a canonical morphism

$$\eta:\mathsf{Groth} o \mathbf{T}$$

sending a class [g:b] to the image of b in $\mathbf{B}/\mathbf{U} \leftarrow \mathbf{T}$. The embedding $\mathbf{U} \subset \mathbf{B}$ induces a closed embedding $\widetilde{\mathcal{U}} \subset \mathsf{Groth}$, which identifies $\widetilde{\mathcal{U}}$ with $\eta^{-1}(e)$.

If we set

$$\mathsf{Groth}' := \mathbf{G} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbf{T},$$

it is a classical observation that the morphisms ν and η combine to give a morphism of schemes

$$artheta: \mathsf{Groth} o \mathsf{Groth}'.$$

The morphism ν obviously factors through ϑ , so that we can consider the fiber product

$$\mathsf{St}' := \mathsf{Groth}' \times_{\mathbf{G}} \mathsf{Groth}' = \mathbf{G} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbf{D}_{\mathrm{f}}$$

where

$$\mathbf{D} := \mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}.$$

Using the morphism ϑ considered above we obtain a canonical morphism $\mathsf{St} \to \mathsf{St}'$.

2.5. Some coherent sheaves on Groth. Let **H** be an affine k-group scheme of finite type, and consider the adjoint action of **H** on itself. Recall that for any $V \in \text{Rep}(\mathbf{H})$ the **H**-equivariant coherent sheaf $V \otimes \mathcal{O}_{\mathbf{H}}$ on **H** (where the equivariant structure is diagonal) admits a canonical "tautological" automorphism $\mathsf{m}_V^{\text{taut}}$ which can be described as follows. Taking global sections induces an equivalence of categories between $\mathsf{Coh}^{\mathbf{H}}(\mathbf{H})$ and the category of **H**-equivariant finitely generated $\mathcal{O}(\mathbf{H})$ -modules; under this equivalence, $\mathsf{m}_V^{\text{taut}}$ corresponds to the composition

$$V \otimes \mathscr{O}(\mathbf{H}) \xrightarrow{\Delta_V \otimes \mathrm{id}} V \otimes \mathscr{O}(\mathbf{H}) \otimes \mathscr{O}(\mathbf{H}) \xrightarrow{\mathrm{id} \otimes m_{\mathscr{O}(\mathbf{H})}} V \otimes \mathscr{O}(\mathbf{H})$$

where $\Delta_V : V \to V \otimes \mathscr{O}(\mathbf{H})$ is the coaction morphism and $m_{\mathscr{O}(\mathbf{H})}$ is the multiplication morphism in the ring $\mathscr{O}(\mathbf{H})$. This construction is functorial in V, but also in \mathbf{H} , in the sense that if \mathbf{K} is another affine k-group scheme and $f : \mathbf{K} \to \mathbf{H}$ is a morphism of k-group schemes, then the canonical isomorphism $f^*(V \otimes \mathscr{O}_{\mathbf{H}}) \cong (\mathsf{For}_{\mathbf{K}}^{\mathbf{H}}V) \otimes \mathscr{O}_{\mathbf{K}}$ (where $\mathsf{For}_{\mathbf{K}}^{\mathbf{H}} : \mathsf{Rep}(\mathbf{H}) \to \mathsf{Rep}(\mathbf{K})$ is the "restriction" functor associated with f) intertwines the automorphisms $f^*\mathfrak{m}_V^{\mathrm{taut}}$ and $\mathfrak{m}_{\mathsf{For}_{\mathbf{V}}^{\mathrm{taut}}}$.

We will consider in particular this construction in the case of the group schemes **G** and **B**. More specifically, for any V in $\text{Rep}(\mathbf{G})$ we will consider the automorphism $\nu^*(\mathsf{m}_{V}^{\text{taut}})$ of $V \otimes \mathscr{O}_{\text{Groth}}$. It is well known that restriction to

$$\mathbf{B} = \{e\} \times \mathbf{B} \subset \mathbf{G} \times^{\mathbf{B}} \mathbf{B} = \mathsf{Groth}$$

induces an equivalence of categories

(2.10)
$$\operatorname{Coh}^{\mathbf{G}}(\operatorname{Groth}) \xrightarrow{\sim} \operatorname{Coh}^{\mathbf{B}}(\mathbf{B})$$

under this equivalence, $\nu^*(\mathbf{m}_V^{\text{taut}})$ identifies with $\mathbf{m}_{\mathsf{For}_{\mathbf{B}}^{\mathbf{G}}V}^{\text{taut}}$. On the other hand, we have a canonical morphism $\mathsf{Groth} \to \mathbf{G}/\mathbf{B}$, from which $V \otimes \mathscr{O}_{\mathsf{Groth}}$ is obtained by pullback of $V \otimes \mathscr{O}_{\mathbf{G}/\mathbf{B}}$. If for $\lambda \in X^*(\mathbf{T})$ we denote by $\mathscr{O}_{\mathbf{G}/\mathbf{B}}(\lambda)$ the line bundle on \mathbf{G}/\mathbf{B} associated with λ , and after choosing a completion of \preceq to a total order \leq on $X^*(\mathbf{T}), V \otimes \mathscr{O}_{\mathsf{Groth}}$ admits a canonical filtration indexed by $(X^*(\mathbf{T}), \leq)$ with associated graded

$$\bigoplus_{\mu \in X^*(\mathbf{T})} V_{\mu} \otimes \mathscr{O}_{\mathbf{G}/\mathbf{B}}(\mu)$$

(Here, V_{μ} is the **T**-weight space of weight μ in V.) As a consequence, if we denote by $\mathscr{O}_{\mathsf{Groth}}(\lambda)$ the pullback of $\mathscr{O}_{\mathbf{G}/\mathbf{B}}(\lambda)$ to Groth , then $V \otimes \mathscr{O}_{\mathsf{Groth}}$ admits a canonical

filtration indexed by $X^*(\mathbf{T})$ with associated graded

$$\bigoplus_{\in X^*(\mathbf{T})} V_{\mu} \otimes \mathscr{O}_{\mathsf{Groth}}(\mu).$$

Under the equivalence (2.10), $\mathscr{O}_{\mathsf{Groth}}(\mu)$ corresponds to $\Bbbk_{\mathbf{B}}(\mu) \otimes \mathscr{O}_{\mathbf{B}}$, and the filtration above is induced by the obvious filtration on $\mathsf{For}_{\mathbf{B}}^{\mathbf{G}}V$ indexed by $X^*(\mathbf{T})$ and with associated graded

$$\bigoplus_{\in X^*(\mathbf{T})} V_{\mu} \otimes \Bbbk_{\mathbf{B}}(\mu).$$

In particular, this shows that $\nu^*(\mathsf{m}_V^{\text{taut}})$ preserves this filtration, and acts on the subquotient $V_\mu \otimes \mathscr{O}_{\mathsf{Groth}}(\mu)$ by multiplication by the function $(\mu \circ \eta) \in \mathscr{O}(\mathsf{Groth})$.

Given $\lambda \in X^*_+(\mathbf{T})$, we will say that a representation $V \in \mathsf{Rep}(\mathbf{G})$ has highest weight λ if $\dim(V_{\lambda}) = 1$ and moreover all the weights μ appearing in V satisfy $\mu \leq \lambda$.

Lemma 2.11. If $\lambda \in X^*_+(T)$ and if $V \in \mathsf{Rep}(\mathbf{G})$ has highest weight λ , then we have a canonical embedding

 $V_{w_{\circ}(\lambda)} \otimes \mathscr{O}_{\mathsf{Groth}}(w_{\circ}(\lambda)) \hookrightarrow V \otimes \mathscr{O}_{\mathsf{Groth}},$

whose image is $\ker(\mathsf{m}_{V}^{\text{taut}} - (w_{\circ}(\lambda) \circ \eta) \cdot \mathrm{id}).$

Proof. The weight $w_{\circ}(\lambda)$ is minimal among the weights of V (with respect to our choice of order); the desired inclusion is therefore provided by the subobject labelled by $w_{\circ}(\lambda)$ in our filtration on $V \otimes \mathscr{O}_{\mathsf{Groth}}$. As explained above $\mathbf{m}_{V}^{\mathsf{taut}} - (w_{\circ}(\lambda) \circ \eta) \cdot \mathrm{id}$ preserves this filtration, and acts trivially on $V_{w_{\circ}(\lambda)} \otimes \mathscr{O}_{\mathsf{Groth}}(w_{\circ}(\lambda))$. For any weight μ of V the induced action on the subquotient $V_{\mu} \otimes \mathscr{O}_{\mathsf{Groth}}(\mu)$ is multiplication by the function $(\mu - w_{\circ}(\lambda)) \circ \eta$, which is injective if $\mu \neq w_{\circ}(\lambda)$; this implies that $V_{w_{\circ}(\lambda)} \otimes \mathscr{O}_{\mathsf{Groth}}(w_{\circ}(\lambda))$ identifies with ker $(\mathbf{m}_{V}^{\mathsf{taut}} - (w_{\circ}(\lambda) \circ \eta) \cdot \mathrm{id})$.

2.6. **Regular semisimple elements.** We will denote by $\mathbf{G}_{\circ} \subset \mathbf{G}$ the open subscheme of "éléments réguliers" in the sense of [SGA3₂, Exp. XIII, Théorème 2.6]. (In this case the Cartan subgroup attached to **T** is **T** itself.) We will denote by $N_{\mathbf{G}}(\mathbf{T})$ the (scheme-theoretic) normalizer of **T** in **G**, which is smooth by [SGA3₂, Exp. XIII, Lemme 2.0]. One can then consider the scheme $\mathbf{G} \times^{N_{\mathbf{G}}(\mathbf{T})} \mathbf{T}$, and the natural morphism

$$\mathbf{G} \times^{N_{\mathbf{G}}(\mathbf{T})} \mathbf{T} \to \mathbf{G}.$$

By definition this morphism restricts to an isomorphism on the preimage of \mathbf{G}_{\circ} . This preimage is $\mathbf{G} \times^{N_{\mathbf{G}}(\mathbf{T})} \mathbf{T}_{\circ}$, where $\mathbf{T}_{\circ} := \mathbf{G}_{\circ} \cap \mathbf{T}$ is the open subscheme of \mathbf{T} whose k-points are the elements $t \in \mathbf{T}$ such that $\alpha(t) \neq 1$ for any $\alpha \in \mathfrak{R}$. Comparing [SGA3₂, Exp. XIII, Corollaire 2.5] with [Hu, §2.3], one sees that the k-points of \mathbf{G}_{\circ} are the regular semisimple elements in the usual terminology of the algebraic groups literature; in particular we have $\mathbf{G}_{\circ} \subset \mathbf{G}_{\mathrm{reg}}$.

Recall from [SGA3₂, Exp. XIII, Corollaire 2.7] that \mathbf{G}_{\circ} is the open subscheme in \mathbf{G} defined by a certain section in $\mathscr{O}(\mathbf{G})$. This section is clearly \mathbf{G} -invariant, hence determines an open subscheme $(\mathbf{T}/\mathbf{W}_{f})_{\circ}$ in $\mathbf{T}/\mathbf{W}_{f} \cong \mathbf{G}/\mathbf{G}$ (see (2.2)) such that \mathbf{G}_{\circ} is the inverse image of $(\mathbf{T}/\mathbf{W}_{f})_{\circ}$ in \mathbf{G} . The inverse image of $(\mathbf{T}/\mathbf{W}_{f})_{\circ}$ under the quotient map $\mathbf{T} \to \mathbf{T}/\mathbf{W}_{f}$ is \mathbf{T}_{\circ} ; $(\mathbf{T}/\mathbf{W}_{f})_{\circ}$ therefore identifies with the quotient $\mathbf{T}_{\circ}/\mathbf{W}_{f}$ (see [SGA1, Exp. V, Corollaire 1.4]). Note that the inertia group (in the sense of [SGA1, Exp. V, §2]) of each point in \mathbf{T}_{\circ} (with respect to the action of \mathbf{W}_{f}) is trivial; in fact, by the analysis at the beginning of [SGA1, Exp. V, §2], to justify this claim it suffices to prove that $\mathbf{W}_{\rm f}$ has no fixed point in $\mathbf{T}_{\circ}(\mathbb{K})$ for any algebraically closed extension \mathbb{K} of \mathbb{k} , which follows from the description of centralizers of semisimple elements in [Hu, §2.2] together with Steinberg's connected theorem (see [Hu, §2.11]), which applies since $\mathscr{D}\mathbf{G}$ is assumed to be simply connected. From the theory reviewed in [SGA1, Exp. V, §2], it follows that we have a natural isomorphism

(2.11)
$$\mathbf{W}_{\mathrm{f}} \times \mathbf{T}_{\circ} \xrightarrow{\sim} \mathbf{T}_{\circ} \times_{\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}_{\circ}$$

defined by $(w, t) \mapsto (w \cdot t, t)$.

We set $\mathbf{B}_{\circ} := \mathbf{G}_{\circ} \cap \mathbf{B}$.

Lemma 2.12. The morphism defined by $(u,t) \mapsto utu^{-1}$ induces an isomorphism of schemes

$$\mathbf{U} \times \mathbf{T}_{\circ} \xrightarrow{\sim} \mathbf{B}_{\circ}$$

Proof. We claim that \mathbf{B}_{\circ} coincides with the subset of "éléments réguliers" in the sense of [SGA3₂, Exp. XIII, Théorème 2.6] applied to the group **B**. (Here again the Cartan subgroup associated with **T** is **T** itself, but now its normalizer is again **T**.) Indeed, if *b* is a k-point in \mathbf{B}_{\circ} , then *b* is "régulier" in **B** by [SGA3₂, Exp. XIII, Corollaire 2.8]. On the other hand, if *b* is a k-point of **B** which is "régulier" in **B**, then there exists $c \in \mathbf{B}$ such that $cbc^{-1} \in \mathbf{T}$ and $(\mathbf{b}/\mathbf{t})^{cbc^{-1}} = \{0\}$, i.e. $\alpha(cbc^{-1}) \neq 1$ for any $\alpha \in -\Re_+$. Then $cbc^{-1} \in \mathbf{T}_{\circ}$, hence $b \in \mathbf{G}_{\circ}$, which finishes the proof of our claim.

Now that this claim is established, we obtain from the definition that the natural morphism

$$\mathbf{B} \times^{\mathbf{T}} \mathbf{T} \to \mathbf{B}$$

restricts to an isomorphism over the preimage of \mathbf{B}_{\circ} . The natural embedding $\mathbf{U} \times \mathbf{T} \to \mathbf{B} \times^{\mathbf{T}} \mathbf{T}$ is an isomorphism since $\mathbf{B} = \mathbf{U} \times \mathbf{T}$, and this preimage identifies with $\mathbf{B} \times^{\mathbf{T}} \mathbf{T}_{\circ}$, which finishes the proof.

2.7. Restrictions to the regular locus. Recall the schemes and morphisms introduced in $\S2.4$. We set

$$\mathsf{Groth}_\circ := \nu^{-1}(\mathbf{G}_\circ), \quad \mathsf{Groth}_\circ' := \mathbf{G}_\circ \times_{\mathbf{T}/\mathbf{W}_\mathrm{f}} \mathbf{T} \cong \mathbf{G}_\circ \times_{\mathbf{T}_\circ/\mathbf{W}_\mathrm{f}} \mathbf{T}_\circ,$$

and denote by

$$u_\circ: \mathsf{Groth}_\circ o \mathbf{G}_\circ, \quad artheta_\circ: \mathsf{Groth}_\circ o \mathsf{Groth}_\circ$$

the restrictions of ν and ϑ respectively. Similarly, we set

$$\mathsf{Groth}_{\mathrm{reg}} := \nu^{-1}(\mathbf{G}_{\mathrm{reg}}), \quad \mathsf{Groth}'_{\mathrm{reg}} := \mathbf{G}_{\mathrm{reg}} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbf{T},$$

and denote by

$$u_{
m reg}: {\sf Groth}_{
m reg} o {f G}_{
m reg}, \quad artheta_{
m reg}: {\sf Groth}_{
m reg} o {\sf Groth}_{
m reg}'$$

the restrictions of ν and ϑ respectively.

The following claim is somewhat standard, but no proof appears in the literature, to the best of our knowledge.

Proposition 2.13. The morphism ϑ_{reg} : Groth_{reg} \rightarrow Groth'_{reg} is an isomorphism.

As a preparation we prove the following claim.

Lemma 2.14. The morphism ϑ_{\circ} : Groth_{\circ} \rightarrow Groth'_{\circ} is an isomorphism. Moreover, the morphism

$$\mathbf{G}/\mathbf{T} imes \mathbf{T}_{\circ}
ightarrow \mathsf{Groth}_{\circ}$$

defined by $(g\mathbf{T}, t) \mapsto [g:t]$ is an isomorphism.

Proof. Recall from $\S2.6$ that the natural morphism

$$\mathbf{G} \times^{N_{\mathbf{G}}(\mathbf{T})} \mathbf{T}_{\circ} \to \mathbf{G}_{\circ}$$

is an isomorphism. We deduce an isomorphism

$$\mathbf{G}_{\circ} \times_{\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}_{\circ} \cong \mathbf{G} \times^{N_{\mathbf{G}}(\mathbf{T})} (\mathbf{T}_{\circ} \times_{\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}_{\circ})$$

where $N_{\mathbf{G}}(\mathbf{T})$ acts on the first factor in $\mathbf{T}_{\circ} \times_{\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}_{\circ}$. Combining this with the isomorphism (2.11), we deduce an identification

$$\mathbf{G}_{\circ} \times_{\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}_{\circ} \cong \mathbf{G} \times^{N_{\mathbf{G}}(\mathbf{T})} (N_{\mathbf{G}}(\mathbf{T}) \times^{\mathbf{T}} \mathbf{T}_{\circ}) \cong \mathbf{G} \times^{\mathbf{T}} \mathbf{T}_{\circ}.$$

Here in the right-hand side the action of \mathbf{T} on \mathbf{T}_{\circ} is trivial, so that this scheme identifies with $\mathbf{G}/\mathbf{T} \times \mathbf{T}_{\circ}$. On the other hand, using Lemma 2.12 we obtain an identification

$$\mathsf{Groth}_{\circ} \cong \mathbf{G} \times^{\mathbf{B}} \mathbf{B}_{\circ} \cong \mathbf{G} \times^{\mathbf{B}} (\mathbf{B} \times^{\mathbf{T}} \mathbf{T}_{\circ}) \cong \mathbf{G} \times^{\mathbf{T}} \mathbf{T}_{\circ},$$

which finishes the proof.

Remark 2.15. Since \mathbf{G}_{\circ} is an affine open subscheme in \mathbf{G} (or since \mathbf{G}/\mathbf{T} is known to be affine), Lemma 2.14 implies in particular that Groth_{\circ} is an affine scheme.

Proof of Proposition 2.13. We follow the proof of the analogous statement for Lie algebras, see [R2]: we will prove that ϑ_{reg} is finite and birational, and that its codomain is smooth and irreducible, which will imply the claim since a finite birational morphism $f: X \to Y$ of integral schemes with Y normal is an isomorphism, see [SP, Tag 0AB1].

First, since $\chi_{\rm reg}$ is smooth (see Proposition 2.5) the scheme ${\rm Groth}'_{\rm reg}$ is smooth over **T**, hence smooth. Smoothness (hence flatness) of ${\rm Groth}'_{\rm reg}$ over **T** also implies that ${\rm Groth}'_{\rm reg} = {\rm Groth}'_{\rm reg} \times_{\bf T} {\bf T}_{\circ}$ is dense in ${\rm Groth}'_{\rm reg}$, see [SP, Tag 081H]. (The notion of "scheme theoretic density" used in this statement is equivalent to density in our present setting, see [SP, Tag 056D].) Now by Lemma 2.14 ${\rm Groth}'_{\circ}$ is isomorphic to an open subscheme in the irreducible scheme ${\rm Groth}$, hence is itself irreducible; this implies that ${\rm Groth}'_{\rm reg}$ is irreducible (see [SP, Tag 004W]).

By Lemma 2.14 again, the restriction of ϑ_{reg} to the preimage of \mathbf{G}_{\circ} is an isomorphism. Since both its domain and its codomain are irreducible, this shows that this morphism is birational.

Finally we prove that ϑ_{reg} is finite, i.e. that it is proper and quasi-finite (see e.g. [GW, Corollary 12.89]). In fact, this map is proper by [SP, Tag 01W6], since its composition with the (separated) projection $\mathbf{G}_{\text{reg}} \times_{\mathbf{T}/\mathbf{W}_{\text{f}}} \mathbf{T} \to \mathbf{G}_{\text{reg}}$ is proper. To prove that it is quasi-finite, by [GW, Remark 12.16] it suffices to prove that the induced map on k-points (i.e. closed points) has finite fibers. Now the map on k-points induced by ν_{reg} has finite fibers, see [Hu, §4.9], hence the same holds for ϑ_{reg} , which finishes the proof.

We will denote by St_{reg} , resp. St'_{reg} , the inverse image of G_{reg} under the canonical morphism $St \rightarrow G$, resp. $St' \rightarrow G$. By Proposition 2.13 the morphism ϑ_{reg} induces an isomorphism

$$\mathsf{St}_{\operatorname{reg}} \xrightarrow{\sim} \mathsf{St}'_{\operatorname{reg}}$$

below we will identify these two schemes whenever convenient. (The same comment applies to Groth_{reg} and Groth'_{reg} .)

2.8. Universal centralizer and Steinberg section. In the rest of this section we assume that $Z(\mathbf{G})$ is smooth, so that Proposition 2.8 applies.

Recall that for any separated k-scheme X endowed with an action of \mathbf{G} , the associated *universal stabilizer* is the group scheme over X defined as the fiber product

$$\mathfrak{S}_{\mathbf{G},X} := (\mathbf{G} \times X) \times_{X \times X} X,$$

where the morphism $\mathbf{G} \times X \to X \times X$ is defined by $(g, x) \mapsto (g \cdot x, x)$, and the morphism $X \to X \times X$ is the diagonal embedding. The projection $\mathfrak{S}_{\mathbf{G},X} \to \mathbf{G} \times X$ is a closed embedding as a subgroup scheme, so that $\mathfrak{S}_{\mathbf{G},X}$ is affine over X (but not flat in general). Its fiber over $x \in X$ is the scheme-theoretic stabilizer of x in \mathbf{G} . Moreover, for any \mathbf{G} -equivariant coherent sheaf \mathscr{F} on X there exists a canonical action of $\mathfrak{S}_{\mathbf{G},X}$ on (the underlying coherent sheaf of) \mathscr{F} ; see [MR, §2.2] for details.

We will consider in particular this construction in the case $X = \mathbf{G}$ with the adjoint action, and denote by \mathbb{J} the resulting group scheme. (In this case, we will often use the expression "universal centralizer" instead of universal stabilizer, for obvious reasons.) We will also denote by \mathbb{J}_{reg} , resp. \mathbb{J}_{\circ} , resp. \mathbb{J}_{Σ} , the restriction of \mathbb{J} to \mathbf{G}_{reg} , resp. to \mathbf{G}_{\circ} , resp. to Σ (where Σ is the Steinberg section studied in §2.2).

Remark 2.16. The group scheme \mathbb{J} admits a canonical section, induced by the diagonal embedding $\mathbf{G} \to \mathbf{G} \times \mathbf{G}$. (In other words, this section sends $g \in \mathbf{G}$ to the pair (g,g) where the second g is seen in the centralizer of the first g.) The identity functor of the category $\operatorname{Rep}^{\infty}(\mathbb{J})$ (identified with the category of $\mathscr{O}(\mathbb{J})$ -comodules) therefore admits a "tautological" automorphism, defined on an $\mathscr{O}(\mathbb{J})$ -comodule M by the composition

$$M \to M \otimes_{\mathscr{O}(\mathbf{G})} \mathscr{O}(\mathbb{J}) \to M \otimes_{\mathscr{O}(\mathbf{G})} \mathscr{O}(\mathbf{G}) = M$$

where the first morphism is the coaction and the second one is induced by restriction to the canonical section. By restriction, we deduce similar structures for \mathbb{J}_{reg} , \mathbb{J}_{\circ} and \mathbb{J}_{Σ} .

In the following statement, of course (1) is a consequence of (2), but the proof will require to prove this claim first. In fact, this is the only claim that will be used in the rest of the paper; (2) is stated only for completeness.

Lemma 2.17. (1) The group scheme \mathbb{J}_{Σ} is smooth (in particular, flat) over Σ .

(2) The group scheme \mathbb{J}_{reg} is smooth (in particular, flat) over \mathbf{G}_{reg} .

Proof. (1) By definition, we have

$$\mathbb{J}_{\Sigma} = \Sigma \times_{\mathbf{G}_{\mathrm{reg}} \times \Sigma} (\mathbf{G} \times \Sigma)$$

where the morphism $\Sigma \to \mathbf{G}_{reg} \times \Sigma$ is defined by $s \mapsto (s, s)$, and the morphism $\mathbf{G} \times \Sigma \to \mathbf{G}_{reg} \times \Sigma$ is defined by $(g, s) \mapsto (gsg^{-1}, s)$. It is clear that both of these

maps factor through $\mathbf{G}_{\text{reg}} \times_{\mathbf{T}/\mathbf{W}_{\text{f}}} \Sigma$, so that

$$\mathbb{J}_{\Sigma} = \Sigma \times_{\mathbf{G}_{\mathrm{reg}} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \Sigma} (\mathbf{G} \times \Sigma).$$

Now by Proposition 2.4(2) the projection $\mathbf{G}_{reg} \times_{\mathbf{T}/\mathbf{W}_{f}} \Sigma \to \mathbf{G}_{reg}$ is an isomorphism, so that

$$\mathbb{J}_{\Sigma} = \Sigma \times_{\mathbf{G}_{\mathrm{rag}}} (\mathbf{G} \times \Sigma)$$

where the map $\Sigma \to \mathbf{G}_{\text{reg}}$ is the obvious closed embedding and the map $\mathbf{G} \times \Sigma \to \mathbf{G}_{\text{reg}}$ is defined by $(g, s) \mapsto gsg^{-1}$. The latter map is smooth by Proposition 2.8, hence so is the projection $\mathbb{J}_{\Sigma} \to \Sigma$, which finishes the proof of our claim.

(2) Consider the commutative diagram



where the vertical maps are induced by the structure morphisms $\mathbb{J}_{\text{reg}} \to \mathbf{G}_{\text{reg}}$ and $\mathbb{J}_{\Sigma} \to \Sigma$, the lower horizontal arrow is the morphism of Proposition 2.8, and the upper horizontal arrow is defined by $(g, (h, s)) \mapsto (ghg, gsg^{-1})$ (for $g \in \mathbf{G}$, $s \in \Sigma$ and $h \in Z_{\mathbf{G}}(s)$). Using **G**-equivariance and the corresponding functors of points one checks that this diagram is cartesian. Since the lower horizontal arrow is smooth and surjective by Proposition 2.8, so is the upper horizontal arrow. And since the left vertical arrow is smooth by the case treated above, using [SP, Tag 02K5] we obtain that the right vertical arrow is smooth, as desired.

Remark 2.18. Lemma 2.17 can also be deduced from [Co, Theorem 7.13(1)] applied to the group scheme $\mathbf{G} \times \mathbf{G}_{\text{reg}}$ over \mathbf{G}_{reg} and its "diagonal" section.

Given a separated scheme S and a separated scheme $X \to S$ endowed with an action of **G** such that $X \to S$ is **G**-invariant, for any separated scheme S' and any morphism $S' \to S$ we have a canonical identification

$$\mathfrak{S}_{\mathbf{G},X\times_S S'} \xrightarrow{\sim} S' \times_S \mathfrak{S}_{\mathbf{G},X}.$$

Applying this observation in our context, we obtain that the universal stabilizer for the G-action on Groth', resp. on St', identifies with

$$\mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}, \quad \text{resp.} \quad \mathbf{D} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}.$$

In particular, the universal stabilizer for the $G\mbox{-}action$ on ${\sf Groth}'_{\rm reg},$ resp. on ${\sf St}'_{\rm reg},$ identifies with

$$\mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\mathrm{reg}}, \quad \mathrm{resp.} \quad \mathbf{D} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\mathrm{reg}}$$

Similarly, the universal stabilizer for the $\mathbf{G}\text{-}\mathrm{action}$ on Groth_\circ' identifies with

 $\mathbf{T}_{\circ} \times_{\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\circ}.$

Lemma 2.19. There exists a canonical morphism

$$\mathbf{\Gamma} imes_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\mathrm{reg}} o \mathsf{Groth}'_{\mathrm{reg}} imes \mathbf{T}$$

of group schemes over $\mathsf{Groth}'_{\mathrm{reg}},$ which restricts to an isomorphism

$$\mathbf{T}_{\circ} imes_{\mathbf{T}_{\circ} / \mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\circ} \xrightarrow{\sim} \mathsf{Groth}_{\circ}' imes \mathbf{T}$$

over $\operatorname{Groth}_{\circ}'$.

Proof. As explained above, $\mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\mathrm{reg}}$ identifies with the universal stabilizer for the action of \mathbf{G} on $\mathrm{Groth}'_{\mathrm{reg}} = \mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbf{G}_{\mathrm{reg}}$. On the other hand, by Proposition 2.13 we have a \mathbf{G} -equivariant isomorphism $\mathrm{Groth}_{\mathrm{reg}} \xrightarrow{\sim} \mathrm{Groth}'_{\mathrm{reg}}$. In view of the natural closed immersion $\mathrm{Groth} \hookrightarrow \mathbf{G} \times \mathbf{G}/\mathbf{B}$, the universal stabilizer for the \mathbf{G} -action on $\mathrm{Groth}_{\mathrm{reg}}$ is contained in the subgroup of $\mathbf{G} \times \mathrm{Groth}_{\mathrm{reg}}$ whose fiber over a point $[g:u] \in \mathrm{Groth}_{\mathrm{reg}}$ is $g\mathbf{B}g^{-1}$; moreover, since the torus \mathbf{T} identifies canonically with the quotient of any Borel subgroup by its derived subgroup, there exists a canonical morphism from the latter subgroup to $\mathbf{T} \times \mathrm{Groth}_{\mathrm{reg}}$; we deduce the desired morphism.

Over \mathbf{G}_{\circ} , by Lemma 2.14 the natural morphism $\mathbf{G} \times^{\mathbf{T}} \mathbf{T} \to \mathbf{G} \times^{\mathbf{B}} \mathbf{B}$ induces an isomorphism $\mathbf{G}/\mathbf{T} \times \mathbf{T}_{\circ} \xrightarrow{\sim} \mathbf{Groth}_{\circ}$, which is **G**-equivariant if **G** acts on the left-hand side via its action on \mathbf{G}/\mathbf{T} . The universal stabilizer for the action of **G** on $\mathbf{G}/\mathbf{T} \times \mathbf{T}_{\circ}$ identifies naturally with

$$(\mathbf{G} imes^{\mathbf{T}} \mathbf{T}) imes \mathbf{T}_{\circ} \cong \mathbf{G} / \mathbf{T} imes \mathbf{T} imes \mathbf{T}_{\circ}$$

Under this identification, the restriction to \mathbf{G}_{\circ} of the morphism considered above identifies with the natural isomorphism

$$\mathbf{G}/\mathbf{T} \times \mathbf{T} \times \mathbf{T}_{\circ} \xrightarrow{\sim} (\mathbf{G}/\mathbf{T} \times \mathbf{T}_{\circ}) \times \mathbf{T},$$

which finishes the proof.

We will set

$$\Sigma_{\mathbf{T}} := \Sigma \times_{\mathbf{T}/\mathbf{W}_{\mathbf{f}}} \mathbf{T}, \quad \text{resp.} \quad \Sigma_{\mathbf{D}} := \Sigma \times_{\mathbf{T}/\mathbf{W}_{\mathbf{f}}} \mathbf{D},$$

so that we have a closed immersion $\Sigma_{\mathbf{T}} \to \mathsf{Groth}'$, resp. $\Sigma_{\mathbf{D}} \to \mathsf{St}'$, which factors through $\mathsf{Groth}'_{\mathrm{reg}}$, resp. $\mathsf{St}'_{\mathrm{reg}}$, and whose composition with the natural morphism $\mathsf{Groth}' \to \mathbf{T}$, resp. $\mathsf{St}' \to \mathbf{D}$, is an isomorphism. We will also consider the group schemes

$$\mathbb{J}_{\mathbf{T}} := \mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\Sigma}, \quad \mathbb{J}_{\mathbf{D}} := \mathbf{D} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\Sigma}.$$

Here $\mathbb{J}_{\mathbf{T}}$ identifies with the restriction of $\mathbb{J}_{\mathbf{D}}$ to the diagonal copy of \mathbf{T} in \mathbf{D} , and also with the restriction of the universal stabilizer for the **G**-action on **Groth'** to $\Sigma_{\mathbf{T}}$. Restricting the morphism of Lemma 2.19 to the preimage of Σ we obtain a canonical morphism

$$(2.12) \qquad \qquad \mathbb{J}_{\mathbf{T}} \to \Sigma_{\mathbf{T}} \times \mathbf{T}$$

of group schemes over $\Sigma_{\mathbf{T}} \cong \mathbf{T}$, whose restriction to \mathbf{T}_{\circ} is an isomorphism.

2.9. Application to coherent sheaves. The universal stabilizers for the actions of **G** on \mathbf{G}_{reg} , Groth_{reg} and St_{reg} encode the categories of equivariant coherent sheaves on this schemes, as explained in the following proposition.

Proposition 2.20. Restriction to Σ , resp. to $\Sigma_{\mathbf{T}}$, resp. to $\Sigma_{\mathbf{D}}$, induces an equivalence of abelian categories

$$\mathsf{Coh}^{\mathbf{G}}(\mathbf{G}_{\mathrm{reg}}) \xrightarrow{\sim} \mathsf{Rep}(\mathbb{J}_{\Sigma}),$$

resp. $\mathsf{Coh}^{\mathbf{G}}(\mathsf{Groth}_{\mathrm{reg}}) \xrightarrow{\sim} \mathsf{Rep}(\mathbb{J}_{\mathbf{T}}),$
resp. $\mathsf{Coh}^{\mathbf{G}}(\mathsf{St}_{\mathrm{reg}}) \xrightarrow{\sim} \mathsf{Rep}(\mathbb{J}_{\mathbf{D}}).$

Proof. The proof is similar to that of [R2, Proposition 3.3.11]; the equivalences are obtained by applying descent theory to the morphisms

$$\mathbf{G} \times \Sigma \to \mathbf{G}_{\mathrm{reg}}, \quad \mathbf{G} \times \Sigma_{\mathbf{T}} \to \mathsf{Groth}'_{\mathrm{reg}}, \quad \mathbf{G} \times \Sigma_{\mathbf{D}} \to \mathsf{St}'_{\mathrm{reg}},$$

which are smooth and surjective (hence faithfully flat and quasi compact) by Proposition 2.8, and then identifying $\operatorname{Groth}_{\operatorname{reg}}'$ with $\operatorname{Groth}_{\operatorname{reg}}$ and $\operatorname{St}_{\operatorname{reg}}'$ with $\operatorname{St}_{\operatorname{reg}}$, see §2.7.

Lemma 2.21. For any $\lambda \in X^*(\mathbf{T})$, the image of the restriction of $\mathscr{O}_{\mathsf{Groth}}(\lambda)$ to $\mathsf{Groth}_{\mathrm{reg}}$ under the equivalence

$$\mathsf{Coh}^{\mathbf{G}}(\mathsf{Groth}_{\mathrm{reg}}) \xrightarrow{\sim} \mathsf{Rep}(\mathbb{J}_{\mathbf{T}})$$

of Proposition 2.20 is the restriction along (2.12) of the $(\Sigma_{\mathbf{T}} \times \mathbf{T})$ -module $\mathscr{O}_{\Sigma_{\mathbf{T}}} \otimes \mathbb{k}_{\mathbf{T}}(\lambda)$.

Proof. The equivalence under consideration is induced by restriction to $\Sigma_{\mathbf{T}} \subset \text{Groth}_{\text{reg}}$. Now $\Sigma_{\mathbf{T}}$ identifies with \mathbf{T} , hence any line bundle on this scheme is trivial by [SP, Tag 0BDA]. In particular, there exists an isomorphism

$$\mathscr{O}_{\mathsf{Groth}}(\lambda)_{|\Sigma_{\mathbf{T}}} \cong \mathscr{O}_{\Sigma_{\mathbf{T}}};$$

we fix a choice for this isomorphism. Now this line bundle has a canonical structure of representation of $\mathbb{J}_{\mathbf{T}}$; in other words it is endowed with a coaction morphism

$$\Gamma(\Sigma_{\mathbf{T}}, \mathscr{O}_{\mathsf{Groth}}(\lambda)_{|\Sigma_{\mathbf{T}}}) \to \Gamma(\Sigma_{\mathbf{T}}, \mathscr{O}_{\mathsf{Groth}}(\lambda)_{|\Sigma_{\mathbf{T}}}) \otimes_{\mathscr{O}(\Sigma_{\mathbf{T}})} \mathscr{O}(\mathbb{J}_{\mathbf{T}}).$$

In view of our identification above there exists a group-like element $\rho \in \mathscr{O}(\mathbb{J}_{\mathbf{T}})$, or in other words a morphism of group schemes

$$\varrho': \mathbb{J}_{\mathbf{T}} \to \Sigma_{\mathbf{T}} \times \mathbb{G}_{\mathrm{m}},$$

such that this coaction morphism is given by $m \mapsto m \otimes \varrho$. To conclude the proof, we have to show that ϱ' is the composition

$$\mathbb{J}_{\mathbf{T}} \xrightarrow{(2.12)} \Sigma_{\mathbf{T}} \times \mathbf{T} \xrightarrow{\mathrm{id} \times \lambda} \Sigma_{\mathbf{T}} \times \mathbb{G}_{\mathrm{m}}$$

Since \mathbb{J}_{Σ} is flat over Σ , to prove this claim it suffices to prove that the two morphisms under consideration coincide on the open subscheme $\mathbf{T}_{\circ} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\Sigma}$.

Consider the restriction $\mathscr{O}_{\mathsf{Groth}_{\circ}}(\lambda)$ of $\mathscr{O}_{\mathsf{Groth}_{\circ}}(\lambda)$ to Groth_{\circ} . Under the identification $\mathbf{G}/\mathbf{T} \times \mathbf{T}_{\circ} \xrightarrow{\sim} \mathsf{Groth}_{\circ}$ (see Lemma 2.14), this line bundle is the pullback of the line bundle $\mathscr{O}_{\mathbf{G}/\mathbf{T}}(\lambda)$ on the affine scheme \mathbf{G}/\mathbf{T} associated with λ . As in the proof of Lemma 2.19, the universal stabilizer for the action of \mathbf{G} on \mathbf{G}/\mathbf{T} identifies with $\mathbf{G}/\mathbf{T} \times \mathbf{T}$; under this identification, the action of this group scheme on $\mathscr{O}_{\mathbf{G}/\mathbf{T}}(\lambda)$ is via λ , or in other words corresponds to the coaction morphism

$$\Gamma(\mathbf{G}/\mathbf{T}, \mathscr{O}_{\mathbf{G}/\mathbf{T}}(\lambda)) \to \Gamma(\mathbf{G}/\mathbf{T}, \mathscr{O}_{\mathbf{G}/\mathbf{T}}(\lambda)) \otimes_{\Bbbk} \mathscr{O}(\mathbf{T})$$

given by $m \mapsto m \otimes \lambda$. We deduce a similar claim for the pullback of this line bundle to $\mathbf{G}/\mathbf{T} \times \mathbf{T}_{\circ}$, i.e. for $\mathscr{O}_{\mathsf{Groth}_{\circ}}(\lambda)$, and then for its restriction to

$$\mathsf{Groth}_{\circ} \cap \Sigma_{\mathbf{T}} = \mathbf{T}_{\circ} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \Sigma,$$

which finishes the proof.

3. Some Hecke categories

We continue with the setting of Section 2, still assuming that $Z(\mathbf{G})$ is smooth. In §§3.1–3.3 we introduce some basic constructions that will be used in the proofs of our main results. From §3.4 on we treat more technical constructions that will be required only for the applications in Section 13. These subsections might be skipped at first reading.

3.1. Affine and extended affine Weyl groups. The extended affine Weyl group of (\mathbf{G}, \mathbf{T}) is the semidirect product

$$\mathbf{W} := \mathbf{W}_{\mathrm{f}} \ltimes X^*(\mathbf{T}).$$

The affine Weyl group of (\mathbf{G}, \mathbf{T}) is the subgroup

$$\mathbf{W}' := \mathbf{W}_{\mathrm{f}} \ltimes \mathbb{Z}\mathfrak{R}.$$

For $\lambda \in X^*(\mathbf{T})$, we will denote by $\mathbf{t}(\lambda)$ the associated element of \mathbf{W} . It is a standard fact that there exists a natural subset $\mathbf{S} \subset \mathbf{W}'$ containing \mathbf{S}_{f} and such that $(\mathbf{W}', \mathbf{S})$ is a Coxeter system; more precisely \mathbf{S} consists of the elements of \mathbf{S}_{f} together with the products $\mathbf{t}(\beta)s_{\beta}$ where β is a maximal short root. By construction, \mathbf{W}_{f} is then a parabolic subgroup in \mathbf{W}' .

If we set, for $w \in \mathbf{W}_{\mathrm{f}}$ and $\lambda \in X^*(\mathbf{T})$,

(3.1)
$$\ell(wt(\lambda)) = \sum_{\substack{\alpha \in \mathfrak{R}_+ \\ w(\alpha) \in \mathfrak{R}_+}} |\langle \lambda, \alpha^{\vee} \rangle| + \sum_{\substack{\alpha \in \mathfrak{R}_+ \\ w(\alpha) \in -\mathfrak{R}_+}} |\langle \lambda, \alpha^{\vee} \rangle + 1|,$$

then it is well known that the restriction of ℓ to \mathbf{W}' is the length function associated with our Coxeter generators \mathbf{S} , and that if we set $\Omega = \{w \in \mathbf{W} \mid \ell(w) = 0\}$ then the natural morphism

$$\Omega \ltimes \mathbf{W}' \to \mathbf{W}$$

is a group isomorphism. Moreover, in this semidirect product Ω acts on \mathbf{W}' by Coxeter group automorphisms, i.e. it stabilizes \mathbf{S} .

Lemma 3.1. For any $s \in \mathbf{S} \setminus \mathbf{S}_{f}$, there exist $s' \in \mathbf{S}_{f}$ and $w \in \mathbf{W}$ such that $\ell(ws') = \ell(w) + 1$ and $s = ws'w^{-1}$.

Proof. This claim is well known in case **G** is semisimple (and simply connected); see [**R1**, Lemma 6.1.2] or [**BM**, Lemma 2.1.1]. We deduce the general case as follows. Set $\mathbf{W}_{der} := \mathbf{W} \ltimes X^* (\mathbf{T} \cap \mathscr{D}\mathbf{G})$. Then we have a surjective group morphism $\mathbf{W} \twoheadrightarrow \mathbf{W}_{der}$ (induced by restriction of characters) which is injective on \mathbf{W}' , and \mathbf{W}_{der} is the extended affine Weyl group of the semisimple group $\mathscr{D}\mathbf{G}$ (and its maximal torus $\mathbf{T} \cap \mathscr{D}\mathbf{G}$). The formula recalled above for the lengths shows that this morphism is compatible with the length functions. Using the known case of semisimple groups we obtain that there exist $s' \in \mathbf{S}$ and $w \in \mathbf{W}$ such that $\ell(ws') = \ell(w) + 1$ and the images of s and $ws'w^{-1}$ in \mathbf{W}_{der} coincide. Now \mathbf{W}' is normal in \mathbf{W} , hence it contains $ws'w^{-1}$. Since the morphism $\mathbf{W} \to \mathbf{W}_{der}$ is injective on \mathbf{W}' , we deduce that $s = ws'w^{-1}$.

In the rest of the paper we will fix once and for all, for each $s \in \mathbf{S} \setminus \mathbf{S}_{\mathrm{f}}$, elements $s' \in \mathbf{S}_{\mathrm{f}}$ and $w \in \mathbf{W}$ such that $\ell(ws') = \ell(w) + 1$ and $s = ws'w^{-1}$. (The condition on lengths will not be needed in the present section, but will be used later.)

3.2. Some representations of $\mathbb{J}_{\mathbf{D}}$. Consider the group scheme $\mathbb{J}_{\mathbf{D}}$ over $\mathbf{D} = \mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}$, and its category $\mathsf{Rep}(\mathbb{J}_{\mathbf{D}})$ of representations on coherent $\mathscr{O}_{\mathbf{D}}$ -modules, see §2.8. This category identifies with the category of comodules over the $\mathscr{O}(\mathbf{D})$ -Hopf algebra

$$\mathscr{O}(\mathbb{J}_{\mathbf{D}}) = \mathscr{O}(\mathbb{J}_{\Sigma}) \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbf{D})$$

which are finitely generated as $\mathscr{O}(\mathbf{D})$ -modules. Since $\mathscr{O}(\mathbf{D})$ is finite as an $\mathscr{O}(\Sigma)$ -module, it admits a natural monoidal structure defined by

$$M \circledast N = M \otimes_{\mathscr{O}(\mathbf{T})} N.$$

This bifunctor is right exact on each side, and the unit object for this monoidal structure if $\mathscr{O}(\mathbf{T})$, seen as functions on the diagonal copy $\mathbf{T} \subset \mathbf{D}$, and endowed with the trivial structure as a representation of $\mathbb{J}_{\mathbf{D}}$.

We will now define objects $(\mathscr{M}_w : w \in \mathbf{W}_{ext})$ of $\mathsf{Rep}(\mathbb{J}_{\mathbf{D}})$ parametrized by \mathbf{W} as follows. First, if $w \in \mathbf{W}_{f}$ then \mathscr{M}_w is defined as the structure sheaf of the closed subscheme

$$\{(w(t),t):t\in\mathbf{T}\}\subset\mathbf{T}\times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}}\mathbf{T},$$

endowed with the trivial structure as a representation. The projection on the first component induces an isomorphism $\mathscr{M}_w \xrightarrow{\sim} \mathscr{O}(\mathbf{T})$; under this isomorphism, the action of $\mathscr{O}(\mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}) = \mathscr{O}(\mathbf{T}) \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathscr{O}(\mathbf{T})$ on \mathscr{M}_w is given by $(f \otimes g) \cdot m = fw(g)m$ for $f, g, m \in \mathscr{O}(\mathbf{T})$.

If $\lambda \in X^*(\mathbf{T})$, then in Lemma 2.21 we have considered the pullback to $\mathbb{J}_{\mathbf{T}}$ of the representation $\mathscr{O}(\mathbf{T}) \otimes \mathbb{k}_{\mathbf{T}}(\lambda)$. Pushing this representation forward along the diagonal embedding $\mathbf{T} \to \mathbf{D}$ we obtain an object of $\mathsf{Rep}(\mathbb{J}_{\mathbf{D}})$, which will be denoted $\mathscr{M}_{\mathsf{t}(\lambda)}$.

It is clear that for $w, y \in \mathbf{W}_{\mathrm{f}}$ and $\lambda, \mu \in X^*(\mathbf{T})$ we have canonical isomorphisms

$$(3.2) \mathcal{M}_w \circledast \mathcal{M}_y \xrightarrow{\sim} \mathcal{M}_{wy},$$

(3.3)
$$\mathscr{M}_{\mathsf{t}(\lambda)} \circledast \mathscr{M}_{\mathsf{t}(\mu)} \xrightarrow{\sim} \mathscr{M}_{\mathsf{t}(\lambda+\mu)}.$$

Next we need to study the interplay between these two classes of objects.

We have a canonical action of \mathbf{W}_{f} on Groth' induced by the natural action on \mathbf{T} ; this action commutes with the action of \mathbf{G} and stabilizes $\mathsf{Groth}'_{\mathrm{reg}}$; we deduce a canonical action on the universal stabilizer $\mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{f}} \mathbb{J}_{\mathrm{reg}}$, and then (by restriction) on $\mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{f}} \mathbb{J}_{\Sigma}$. (This action is simply induced by the action on \mathbf{T} .)

Lemma 3.2. The morphism (2.12) is W_{f} -equivariant, where W_{f} acts on the righthand side diagonally.

Proof. By flatness it suffices to check this claim over \mathbf{T}_{\circ} . Now, by Lemma 2.14 we have an isomorphism $\mathbf{G}/\mathbf{T} \times \mathbf{T}_{\circ} \xrightarrow{\sim} \mathsf{Groth}'_{\circ}$. Under this isomorphism the action of \mathbf{W}_{f} on $\mathbf{G}/\mathbf{T} \times \mathbf{T}_{\circ}$ is given by $w \cdot (g\mathbf{T}, t) = (gw^{-1}\mathbf{T}, w(t))$, where we write $gw^{-1}\mathbf{T}$ for $g\dot{w}^{-1}\mathbf{T}$ where \dot{w} is any lift of w to $N_{\mathbf{G}}(\mathbf{T})$. From this description the equivariance is clear.

From Lemma 3.2 we deduce that for $w \in \mathbf{W}_{\mathrm{f}}$ and $\lambda \in X^*(\mathbf{T})$ we have a canonical isomorphism

 $\mathscr{M}_w \circledast \mathscr{M}_{\mathsf{t}(\lambda)} \circledast \mathscr{M}_{w^{-1}} \xrightarrow{\sim} \mathscr{M}_{\mathsf{t}(w(\lambda))}.$

Combining this with (3.2)–(3.3) we deduce that if for $w = xt(\lambda) \in \mathbf{W}_{\mathrm{f}} \ltimes X^{*}(\mathbf{T}) = \mathbf{W}$ we set

$$\mathscr{M}_w := \mathscr{M}_x \circledast \mathscr{M}_{\mathsf{t}(\lambda)},$$

then for any $w, y \in \mathbf{W}$ we have a canonical isomorphism

$$\mathscr{M}_w \circledast \mathscr{M}_y \xrightarrow{\sim} \mathscr{M}_{wy}.$$

We next define some objects $(\mathscr{B}_s : s \in \mathbf{S}_{aff})$ associated with simple reflections in **W**. First, if $s \in \mathbf{S}_{\mathbf{f}}$ we define \mathscr{B}_s by

$$\mathscr{B}_s := \mathscr{O}(\mathbf{T} \times_{\mathbf{T}/\{1,s\}} \mathbf{T}),$$

which we view as an $\mathscr{O}(\mathbf{D})$ -module via the closed embedding

$$\mathbf{\Gamma} \times_{\mathbf{T}/\{1,s\}} \mathbf{T} \subset \mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbf{T},$$

and endow with the trivial structure as a representation. If $s \in \mathbf{S} \setminus \mathbf{S}_{f}$, recall that in §3.1 we have fixed $s' \in \mathbf{S}_{f}$ and $w \in \mathbf{W}$ such that $s = ws'w^{-1}$; we then set

(3.4)
$$\mathscr{B}_s := \mathscr{M}_w \circledast \mathscr{B}_{s'} \circledast \mathscr{M}_{w^{-1}}$$

It is easily seen (e.g. by reduction to the case $s \in \mathbf{S}_{f}$) that for any $s \in \mathbf{S}$ there exist exact sequences

$$\mathcal{M}_e \hookrightarrow \mathcal{B}_s \twoheadrightarrow \mathcal{M}_s, \quad \mathcal{M}_s \hookrightarrow \mathcal{B}_s \twoheadrightarrow \mathcal{M}_e.$$

3.3. Completions. We will denote by $\mathcal{I} \subset \mathscr{O}(\mathbf{D})$ the ideal of the point (e, e), and by \mathbf{D}^{\wedge} the spectrum of the completion of $\mathscr{O}(\mathbf{D})$ with respect to \mathcal{I} . Let also $\mathcal{K} \subset \mathscr{O}(\mathbf{T})$ be the ideal of the point $e \in \mathbf{T}$, and \mathbf{T}^{\wedge} be the spectrum of the completion of $\mathscr{O}(\mathbf{T})$ with respect to \mathcal{K} . Note that

$$\mathcal{I} = \mathcal{K} \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathscr{O}(\mathbf{T}) + \mathscr{O}(\mathbf{T}) \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathcal{K}$$

where both summands are ideals in $\mathscr{O}(\mathbf{D})$ since $\mathscr{O}(\mathbf{T})$ is flat over $\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})$ (by Theorem 2.1). Finally, we will denote by $\mathcal{J} \subset \mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})$ the ideal of the image of $e \in \mathbf{T}$ in $\mathbf{T}/\mathbf{W}_{f}$, and by $(\mathbf{T}/\mathbf{W}_{f})^{\wedge}$ the spectrum of the completion of $\mathscr{O}(\mathbf{T}/\mathbf{W}_{f})$ with respect to \mathcal{J} .

Lemma 3.3. (1) There exist canonical isomorphisms of \Bbbk -schemes

$$\mathbf{T}^{\wedge} \cong \mathbf{T} imes_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} (\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\prime}$$

and

$$\mathbf{D}^{\wedge} \cong \mathbf{T}^{\wedge} imes_{\mathbf{T}} \mathbf{D} \cong \mathbf{D} imes_{\mathbf{T}} \mathbf{T}^{\wedge} \cong \mathbf{D} imes_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} (\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge} \cong \mathbf{T}^{\wedge} imes_{(\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge}} \mathbf{T}^{\wedge}$$

where in the first, resp. second, fiber product the morphism $\mathbf{D} \to \mathbf{T}$ is induced by projection on the first, resp. second, factor. Moreover, $\mathscr{O}(\mathbf{T}^{\wedge})$ is finite and free (in particular, flat) over $\mathscr{O}((\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge})$. (2) The natural morphism $\mathscr{O}((\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge}) \to \mathscr{O}(\mathbf{T}^{\wedge})^{\mathbf{W}_{\mathrm{f}}}$ is an isomorphism.

Proof. (1) Since the morphism $\mathbf{T} \to \mathbf{T}/\mathbf{W}_{\mathrm{f}}$ is finite, and since e is the only closed point in the preimage of the point corresponding to \mathcal{J} , by the structure theory of artinian local rings (see in particular [SP, Tag 00J8]) the ideal $\mathcal{J} \cdot \mathscr{O}(\mathbf{T})$ contains a power of \mathcal{K} . On the other hand this ideal is contained in \mathcal{K} ; hence the completions of $\mathscr{O}(\mathbf{T})$ with respect to \mathcal{K} and to $\mathcal{J} \cdot \mathscr{O}(\mathbf{T})$ are canonically isomorphic. By definition the first of these completions is $\mathscr{O}(\mathbf{T}^{\wedge})$, and since \mathbf{T} is finite over $\mathbf{T}/\mathbf{W}_{\mathrm{f}}$ the second completion identifies with $\mathscr{O}(\mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{f}} (\mathbf{T}/\mathbf{W}_{f})^{\wedge})$, proving the first isomorphism. Combined with Theorem 2.1, this implies that $\mathscr{O}(\mathbf{T}^{\wedge})$ is finite and free over $\mathcal{O}((\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge})$.

Similar considerations using the morphism $\mathbf{D} \to \mathbf{T}/\mathbf{W}_{\mathrm{f}}$ prove the isomorphism between the first and fourth schemes in the second series of isomorphisms. Since the ideals $\mathcal{K} \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathscr{O}(\mathbf{T})$ and $\mathscr{O}(\mathbf{T}) \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathcal{K}$ contain $\mathcal{J} \cdot \mathscr{O}(\mathbf{D})$ and are

contained in \mathcal{I} , the completions of $\mathscr{O}(\mathbf{D})$ with respect to these ideals (or, in other words, the algebras of functions on the second and third schemes) also identify with $\mathscr{O}(\mathbf{D}^{\wedge})$. Finally, the last isomorphism follows from the isomorphism $\mathbf{T}^{\wedge} \cong \mathbf{T} \times_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} (\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge}$.

(2) Using the first isomorphism in (1) we see that the canonical embedding $\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}}) \hookrightarrow \mathscr{O}(\mathbf{T})$ induces an embedding

$$\mathscr{O}((\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge}) \hookrightarrow \mathscr{O}(\mathbf{T}^{\wedge}),$$

which of course factors through an embedding

$$(3.5) \qquad \qquad \mathscr{O}((\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge}) \hookrightarrow \mathscr{O}(\mathbf{T}^{\wedge})^{\mathbf{W}_{\mathrm{f}}}$$

As explained above, any basis of $\mathscr{O}(\mathbf{T})$ over $\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})$ provides a basis of $\mathscr{O}(\mathbf{T}^{\wedge})$ over $\mathscr{O}((\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge})$. On the other hand, in [BR1, Theorem 8.1] it is proved that a specific basis of $\mathscr{O}(\mathbf{T})$ over $\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})$ provides a basis of $\mathscr{O}(\mathbf{T}^{\wedge})$ over $\mathscr{O}(\mathbf{T}^{\wedge})^{\mathbf{W}_{\mathrm{f}}}$. The embedding (3.5) is therefore an equality.

We set

$$\mathbb{J}_{\mathbf{D}}^{\wedge} := \mathbf{D}^{\wedge} imes_{\mathbf{D}} \mathbb{J}_{\mathbf{D}} \cong \mathbf{D}^{\wedge} imes_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbb{J}_{\Sigma},$$

a smooth affine group scheme over the affine scheme \mathbf{D}^{\wedge} . We will consider the category $\operatorname{\mathsf{Rep}}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ of representations of this group scheme which are of finite type over $\mathscr{O}(\mathbf{D}^{\wedge})$. The isomorphisms in Lemma 3.3 show that an $\mathscr{O}(\mathbf{D}^{\wedge})$ -module is the same thing as an $\mathscr{O}(\mathbf{T}^{\wedge})$ -bimodule on which the left and right actions of $\mathscr{O}((\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge}) = \mathscr{O}(\mathbf{T}^{\wedge})^{\mathbf{W}_{\mathrm{f}}}$ coincide. (We will use this identification repeatedly and without further notice below.) In particular the category of such modules admits a natural monoidal product, induced by the tensor product for $\mathscr{O}(\mathbf{T}^{\wedge})$ -bimodules; moreover this product stabilizes the subcategory of finitely generated $\mathscr{O}(\mathbf{D}^{\wedge})$ -modules. Since $\mathbb{J}^{\wedge}_{\mathbf{D}}$ is the pullback of a group scheme over $(\mathbf{T}/\mathbf{W}_{\mathrm{f}})^{\wedge}$, this product induces a monoidal product on the category $\operatorname{\mathsf{Rep}}(\mathbb{J}^{\wedge}_{\mathbf{D}})$, which will again be denoted \circledast .

Recall that a category is called *Krull–Schmidt* if any object admits a decomposition as a (finite) direct sum of objects with local endomorphism ring.

Lemma 3.4. The category $\operatorname{Rep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$ is Krull-Schmidt.

Proof. By [CYZ, Theorem A.1], an additive category is Krull–Schmidt iff it is idempotent complete and the endomorphism ring of any object is semiperfect. Here $\operatorname{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ is idempotent complete because it is abelian, and the endomorphism algebra of any object is semiperfect because it is finite as a module over the noetherian complete local ring $\mathscr{O}(\mathbf{D}^{\wedge})$, see [La, Example 23.3].

Pulling back the representations $(\mathscr{M}_w : w \in \mathbf{W})$ and $(\mathscr{B}_s : s \in \mathbf{S})$ introduced in §3.2 along the natural morphism $\mathbf{D}^{\wedge} \to \mathbf{D}$ we obtain objects $(\mathscr{M}_w^{\wedge} : w \in \mathbf{W})$ and $(\mathscr{B}_s^{\wedge} : s \in \mathbf{S})$ in $\mathsf{Rep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$. It is clear that for any $w, y \in \mathbf{W}$ we have a canonical isomorphism

(3.6)
$$\mathscr{M}_w^\wedge \circledast \mathscr{M}_y^\wedge \xrightarrow{\sim} \mathscr{M}_{wy}^\wedge$$

and that for $s \in \mathbf{S}$ we have exact sequences

$$(3.7) \qquad \qquad \mathscr{M}_e^{\wedge} \hookrightarrow \mathscr{B}_s^{\wedge} \twoheadrightarrow \mathscr{M}_s^{\wedge}, \qquad \mathscr{M}_s^{\wedge} \hookrightarrow \mathscr{B}_s^{\wedge} \twoheadrightarrow \mathscr{M}_e^{\wedge}.$$

The following lemma will be proved in §3.5 below, using a different description of (a subcategory of) $\operatorname{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$. (In this statement we use the fact that $\omega s \omega^{-1} \in \mathbf{S}$ for any $s \in \mathbf{S}$ and $\omega \in \Omega$, see §3.1.)

Lemma 3.5. For any $s \in \mathbf{S} \setminus \mathbf{S}_{\mathrm{f}}$ the object \mathscr{B}_{s}^{\wedge} is independent of the choices of w and s' as in §3.1 up to canonical isomorphism. Moreover, for any $\omega \in \Omega$ and $s \in \mathbf{S}$ we have a canonical isomorphism

$$\mathscr{M}^\wedge_\omega \circledast \mathscr{B}^\wedge_s \circledast \mathscr{M}^\wedge_{\omega^{-1}} \cong \mathscr{B}^\wedge_{\omega s \omega^{-1}}$$

We will denote by $\mathsf{BSRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ the category with

- objects the collections $(\omega, s_1, \cdots, s_i)$ with $\omega \in \Omega$ and $s_1, \cdots, s_i \in \mathbf{S}$;
- morphisms from $(\omega, s_1, \cdots, s_i)$ to $(\omega', s_1', \cdots, s_j')$ given by

$$\operatorname{Hom}_{\mathsf{Rep}(\mathbb{J}_{\mathbf{D}}^{\wedge})}(\mathscr{M}_{\omega}^{\wedge} \circledast \mathscr{B}_{s_{1}}^{\wedge} \circledast \cdots \circledast \mathscr{B}_{s_{i}}^{\wedge}, \mathscr{M}_{\omega'}^{\wedge} \circledast \mathscr{B}_{s_{1}'}^{\wedge} \circledast \cdots \circledast \mathscr{B}_{s_{j}'}^{\wedge}).$$

By definition there exists a canonical fully faithful functor

$$(3.8) \qquad \qquad \mathsf{BSRep}(\mathbb{J}^{\wedge}_{\mathbf{D}}) \to \mathsf{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}}).$$

Using Lemma 3.5 we obtain, for any collections $(\omega, s_1, \dots, s_i)$ and $(\omega', s'_1, \dots, s'_j)$ as above, a canonical isomorphism

$$\begin{pmatrix} \mathscr{M}_{\omega}^{\wedge} \circledast \mathscr{B}_{s_{1}}^{\wedge} \circledast \cdots \circledast \mathscr{B}_{s_{i}}^{\wedge} \end{pmatrix} \circledast \begin{pmatrix} \mathscr{M}_{\omega'}^{\wedge} \circledast \mathscr{B}_{s_{1}'}^{\wedge} \circledast \cdots \circledast \mathscr{B}_{s_{j}'}^{\wedge} \end{pmatrix} \\ \cong \mathscr{M}_{\omega\omega'}^{\wedge} \circledast \mathscr{B}_{(\omega')^{-1}s_{1}\omega'}^{\wedge} \circledast \cdots \circledast \mathscr{B}_{(\omega')^{-1}s_{i}\omega'}^{\wedge} \circledast \mathscr{B}_{s_{1}'}^{\wedge} \circledast \cdots \circledast \mathscr{B}_{s_{j}'}^{\wedge}.$$

This allows us to define a monoidal product (again denoted \circledast) on $\mathsf{BSRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ which is defined on objects by

$$(\omega, s_1, \cdots, s_i) \circledast (\omega', s_1', \cdots, s_j') = (\omega \omega', (\omega')^{-1} s_1 \omega', \cdots, (\omega')^{-1} s_i \omega', s_1', \cdots, s_j')$$

and such that (3.8) is monoidal.

We will denote by

 $\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$

the Karoubian closure of the additive hull of the category $\mathsf{BSRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$. By the Krull–Schmidt property (see Lemma 3.4), this category identifies with the (monoidal) full subcategory of $\mathsf{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ whose objects are direct sums of direct summands of objects of the form

$$\mathscr{M}^{\wedge}_{\omega} \circledast \mathscr{B}^{\wedge}_{s_1} \circledast \cdots \circledast \mathscr{B}^{\wedge}_{s_i}$$

with $\omega \in \Omega$ and $s_1, \dots, s_i \in \mathbf{S}$. (In these notations, "BS" stands for "Bott–Samelson," and "S" for "Soergel," since these constructions are very similar to classical constructions related to Bott–Samelson resolutions and Soergel bimodules.)

3.4. Hecke categories "à la Abe". We now explain how to construct some categories by following a pattern initiated by Abe [Ab1]. We consider a noetherian domain R endowed with an action of \mathbf{W} (by ring automorphisms), and denote by Q the fraction field of R. We denote by $\mathsf{K}'(R)$ the category defined as follows. The objects are the R-bimodules M together with a decomposition

$$(3.9) M \otimes_R Q = \bigoplus_{w \in \mathbf{W}} M_Q^u$$

as (R, Q)-bimodules such that:

- there exist only finitely many w's such that $M_Q^w \neq 0$;
- for any $w \in \mathbf{W}$, $r \in R$ and $m \in M_Q^w$ we have $m \cdot r = w(r) \cdot m$.

Morphisms in this category are defined as morphisms of *R*-bimodules respecting the decompositions (3.9). The category $\mathsf{K}'(R)$ has a natural monoidal structure, with product denoted \star and induced by the tensor product over *R*. (To see this one observes that the conditions above imply that the left *R*-action on $M \otimes_R Q$ extends to an action of Q, see [Ab1, Remark 2.2].)

We will also denote by $\mathsf{K}(R)$ the full subcategory in $\mathsf{K}'(R)$ whose objects are those whose underlying *R*-bimodule is finitely generated, and is flat as a right *R*-module. The latter condition implies that the natural morphism $M \to M \otimes_R Q$ is injective, which (in view of the second condition above) implies in particular that the left and right actions of $R^{\mathbf{W}}$ on M coincide. The arguments in [Ab1, Lemma 2.6] show that the underlying bimodule of any object in $\mathsf{K}(R)$ is in fact finitely generated as a left *R*-module and as a right *R*-module. Using this property, it is easily seen that $\mathsf{K}(R)$ is a monoidal subcategory of $\mathsf{K}'(R)$.

Remark 3.6. As explained in [Ab1, §2.2], for any M in $\mathsf{K}'(R)$ there exists a canonical isomorphism $Q \otimes_R M \xrightarrow{\sim} M \otimes_R Q$. (In the examples we will consider below, the action on \mathbf{W} on R will factor through an action of the finite group \mathbf{W}_{f} , so that R will be finite over $R^{\mathbf{W}}$. In this case, both $Q \otimes_R M$ and $M \otimes_R Q$ identify with $M \otimes_{R^{\mathbf{W}}} \operatorname{Frac}(R^{\mathbf{W}})$.) As a consequence, switching the left and right R-actions defines an autoequivalence of the category $\mathsf{K}'(R)$, where the w-graded part in the image of M is $M_Q^{w^{-1}}$ with the actions switched. This equivalence is "antimonoidal" in the sense that it swaps factors in a tensor product. It restricts to an autoequivalence of the subcategory of $\mathsf{K}'(R)$ whose objects are finitely generated (as bimodules) and flat both as a left and as a right R-module.

We have natural objects in $\mathsf{K}(R)$ attached to elements in \mathbf{W} , and constructed as follows. Given $w \in \mathbf{W}$, we denote by F_w the *R*-bimodule which is isomorphic to *R* as an abelian group, and endowed with the structure of *R*-bimodule determined by the rule

$$r \cdot m \cdot r' = rmw(r')$$

for $r, r' \in R$ and $m \in F_w$. If we endow this bimodule with the decomposition of $F_w \otimes_R Q$ such that this module is concentrated in degree w, we obtain an object in $\mathsf{K}(R)$. It is clear that for any $w, y \in \mathbf{W}$ we have a canonical isomorphism

$$F_w \star F_y \xrightarrow{\sim} F_{wy}$$

Next, for $s\in {\bf S}$ we will denote by $R^s\subset R$ the subring of s-invariants. Assume that

(3.10) there exists $\delta_s \in R$ such that $(1, \delta_s)$ is a basis of R as an R^s -module.

Then we set

$$B_s := R \otimes_{R^s} R.$$

Our assumption ensures that B_s is finite and free (in particular, flat) as a right R-module. Moreover this objects admits a canonical decomposition (3.9), hence defines an object in $\mathsf{K}(R)$. In fact, since the action of s on R is nontrivial by our assumption, the decomposition of $B_s \otimes_R Q = R \otimes_{R^s} Q$ is uniquely determined by the fact that it is concentrated in degrees $\{e, s\} \subset \mathbf{W}$. More explicitly, using the formula

$$\delta_s \delta_s = \delta_s (\delta_s + s(\delta_s)) - \delta_s s(\delta_s)$$

one checks that we have

$$(B_s)_Q^e = (\delta_s \otimes 1 - 1 \otimes s(\delta_s)) \cdot Q, \quad (B_s)_Q^s = (\delta_s \otimes 1 - 1 \otimes \delta_s) \cdot Q.$$

The following lemma can be checked by explicit computation. (A similar claim in a slightly different setting is proved in [BR2, Lemma 2.4].)

Lemma 3.7. Let $s, s' \in \mathbf{S}$, and assume that $w \in \mathbf{W}$ satisfies $s' = wsw^{-1}$. If (3.10) holds for s, then it also holds for s', and moreover we have a canonical isomorphism

$$F_w \star B_s \star F_{w^{-1}} \xrightarrow{\sim} B_{s'}.$$

Remark 3.8. Recall that any element in **S** is conjugate (in **W**) to an element in **S**_f, see Lemma 3.1. In view of Lemma 3.7, to check condition (3.10) for all $s \in \mathbf{S}$ it suffices to do so when $s \in \mathbf{S}_{f}$.

We now assume that (3.10) is satisfied for any $s \in \mathbf{S}$. We will then denote by $\mathsf{BSK}(R)$ the category with

- objects the collections $(\omega, s_1, \cdots, s_i)$ with $\omega \in \Omega$ and $s_1, \cdots, s_i \in \mathbf{S}$;
- morphisms from $(\omega, s_1, \cdots, s_i)$ to $(\omega', s'_1, \cdots, s'_i)$ given by

 $\operatorname{Hom}_{\mathsf{K}(R)}(F_{\omega} \star B_{s_1} \star \cdots \star B_{s_i}, F_{\omega'} \star B_{s'_1} \star \cdots \star B_{s'_s}).$

By definition there exists a canonical fully faithful functor

$$(3.11) \qquad \qquad \mathsf{BSK}(R) \to \mathsf{K}(R).$$

Using the isomorphism in Lemma 3.7 (when $w \in \Omega$) one sees that there exists a natural convolution product (still denoted \star) on $\mathsf{BSK}(R)$ which is defined on objects by

$$(\omega, s_1, \cdots, s_i) \star (\omega', s_1', \cdots, s_j') = (\omega\omega', (\omega')^{-1} s_1 \omega', \cdots, (\omega')^{-1} s_i \omega', s_1', \cdots, s_j'),$$

and such that (3.11) is monoidal.

Remark 3.9. Instead of putting the element in Ω to the left, one can also put it to the right, and define the monoidal category $\mathsf{BSK}_r(R)$ with objects the collections $(s_1, \cdots, s_i, \omega)$ and morphisms defined in the obvious way. The equivalence of Remark 3.6 sends each B_s to itself, and each F_w to $F_{w^{-1}}$. It therefore induces an equivalence of categories $\mathsf{BSK}(R) \xrightarrow{\sim} \mathsf{BSK}_r(R)$ which is antimonoidal and is given on objects by

$$(\omega, s_1, \cdots, s_i) \mapsto (s_i, \cdots, s_1, \omega^{-1}).$$

The following lemma is obvious.

Lemma 3.10. Let R and R' be two noetherian domains endowed with actions of \mathbf{W} by ring automorphisms. Assume that condition (3.10) is satisfied for the ring R (for any $s \in \mathbf{S}$), and assume given a \mathbf{W} -equivariant ring isomorphism $R \xrightarrow{\sim} R'$. Then condition (3.10) is satisfied for the ring R' (for any $s \in \mathbf{S}$), and there exists a natural equivalence of monoidal categories

$$\mathsf{K}'(R) \xrightarrow{\sim} \mathsf{K}'(R')$$

which restricts to an equivalence

$$\mathsf{K}(R) \xrightarrow{\sim} \mathsf{K}(R')$$

sending each object F_w ($w \in \mathbf{W}$) or B_s ($s \in \mathbf{S}$) in $\mathsf{K}(R)$ to the corresponding object in $\mathsf{K}(R')$. As a consequence, we deduce an equivalence of categories

$$\mathsf{BSK}(R) \xrightarrow{\sim} \mathsf{BSK}(R')$$

which is the identity on objects.

The construction above admits a "graded variant" as follows. In this setting we assume that R is a graded noetherian domain, and that the action of \mathbf{W} is by graded ring automorphisms. Then we have a "grading shift" functor (1) on graded R-bimodules, defined in such a way that $(M(1))^i = M^{i+1}$. In this setting we define the category $\mathsf{K}'_{\mathrm{gr}}(R)$ as above, but using graded R-bimodules and morphisms of graded bimodules. One then defines the subcategory $\mathsf{K}_{\mathrm{gr}}(R)$ in the same way as above. In order to define B_s we assume that there exists a homogeneous element $\delta_s \in R$ such that $(1, \delta_s)$ is a basis of R as an R^s -module. Moreover, we set

$$B_s := R \otimes_{R^s} R(1)$$

Finally, $\mathsf{BSK}_{gr}(R)$ is defined as the category with

- objects the collections $(\omega, s_1, \cdots, s_i, n)$ with $\omega \in \Omega$ and $s_1, \cdots, s_i \in \mathbf{S}$ and $n \in \mathbb{Z}$;
- morphisms from $(\omega, s_1, \cdots, s_i, n)$ to $(\omega', s_1', \cdots, s_j', n')$ given by

 $\operatorname{Hom}_{\mathsf{K}_{\sigma^{r}}(R)}(F_{\omega} \star B_{s_{1}} \star \cdots \star B_{s_{i}}(n), F_{\omega'} \star B_{s_{1}'} \star \cdots \star B_{s_{i}'}(n')).$

As above we have a canonical fully faithful functor $\mathsf{BSK}_{\mathrm{gr}}(R) \to \mathsf{K}_{\mathrm{gr}}(R)$. Given M, N in $\mathsf{K}_{\mathrm{gr}}(R)$, we will set

$$\operatorname{Hom}_{\mathsf{K}_{\mathrm{gr}}(R)}^{\bullet}(M,N) = \bigoplus_{n \in \mathbb{Z}} \operatorname{Hom}_{\mathsf{K}_{\mathrm{gr}}(R)}(M,N(n)).$$

Again the category $\mathsf{K}'_{\mathrm{gr}}(R)$ admits a natural convolution product \star , which makes it a monoidal category and stabilizes the subcategory $\mathsf{K}_{\mathrm{gr}}(R)$, and which induces a monoidal structure on $\mathsf{BSK}_{\mathrm{gr}}(R)$ given on objects by

$$(\omega, s_1, \cdots, s_i, n) \star (\omega', s_1', \cdots, s_j', n') = (\omega\omega', (\omega')^{-1}s_1\omega', \cdots, (\omega')^{-1}s_i\omega', s_1', \cdots, s_i', n+n').$$

Remark 3.11. In [Ab1], Abe studies an analogue of the category $\mathsf{BSK}_{gr}(R)$ in the setting where **W** is replaced by a Coxeter group (so that there are no nontrivial elements of length 0) and for a specific choice of graded ring R. We do not claim that the results of [Ab1] apply in the generality considered above, but only that the main definition makes sense.

3.5. Completed Hecke category and representations of $\mathbb{J}_{\mathbf{D}}^{\wedge}$. The first ring to which we will apply the construction of §3.4 is $\mathscr{O}(\mathbf{T}^{\wedge})$, with the action of \mathbf{W} obtained from the natural action of \mathbf{W}_{f} by pullback along the projection $\mathbf{W} \rightarrow \mathbf{W}_{\mathrm{f}}$. To check conditions (3.10) in this case, it suffices to do so when $s \in \mathbf{S}_{\mathrm{f}}$ (see Remark 3.8). In this case the condition can be checked explicitly, or deduced from Lemma 3.3 applied to the Levi factor of \mathbf{G} associated with s. The resulting categories $\mathsf{K}(\mathscr{O}(\mathbf{T}^{\wedge}))$ and $\mathsf{BSK}(\mathscr{O}(\mathbf{T}^{\wedge}))$ will be denoted

$$\mathsf{K}^{\wedge}$$
 and BSK^{\wedge}

respectively.

Recall the category $\operatorname{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ considered in §3.3. We will denote by $\operatorname{Rep}_{\mathrm{fl}}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ the full subcategory of representations whose underlying coherent sheaf is flat with respect to the projection $\mathbf{D}^{\wedge} \to \mathbf{T}^{\wedge}$ on the second component. It is not difficult to check that $\operatorname{Rep}_{\mathrm{fl}}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ is a monoidal subcategory in $\operatorname{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$, and that it contains the essential image of (3.8).

The following statement is an analogue of the statements [BR2, Proposition 2.7 and Lemma 2.9], and its proof is very similar.

Proposition 3.12. There exists a canonical fully faithful monoidal functor

 $\mathsf{Rep}_{\mathrm{fl}}(\mathbb{J}^\wedge_{\mathbf{D}}) \to \mathsf{K}^\wedge$

sending \mathscr{M}_w^{\wedge} to F_w for any $w \in \mathbf{W}$ and \mathscr{B}_s^{\wedge} to B_s for any $s \in S_{\mathrm{f}}$.

Proof. We start by constructing a functor

$$(3.12) \qquad \qquad \mathsf{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}}) \to \mathsf{K}'(\mathscr{O}(\mathbf{T}^{\wedge})).$$

Recall the open subscheme $\mathbf{T}_{\circ} \subset \mathbf{T}$, which is defined by the function $\prod_{\alpha} (\alpha - 1)$ where α runs over the roots of (\mathbf{G}, \mathbf{T}) , see §2.6. We have an open embedding $\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}} \subset \mathbf{T}/\mathbf{W}_{\mathrm{f}}$ and an isomorphism

$$(3.13) \mathbf{W}_{\mathrm{f}} \times \mathbf{T}_{\mathrm{o}} \xrightarrow{\prime} \mathbf{T}_{\mathrm{o}} \times_{\mathbf{T}_{\mathrm{o}}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}_{\mathrm{o}},$$

see §2.11. Let us denote by $\mathbb{J}_{\mathbf{T},\circ}$ the restriction of $\mathbb{J}_{\mathbf{T}}$ to \mathbf{T}_{\circ} . Then it follows from Lemma 2.19 that we have a canonical isomorphism of group schemes

$$(3.14) \qquad \qquad \mathbb{J}_{\mathbf{T},\circ} \xrightarrow{\sim} \mathbf{T}_{\circ} \times \mathbf{T}.$$

We are now ready to explain the construction of the functor (3.12). Starting from an object M in $\text{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$, the underlying $\mathscr{O}(\mathbf{T}^{\wedge})$ -bimodule of its image is simply taken as M with its given $\mathscr{O}(\mathbf{D}^{\wedge})$ -module structure. Next, using Lemma 3.3 and (3.13) we obtain a canonical isomorphism

$$\mathbf{D}^\wedge imes_{\mathbf{T}/\mathbf{W}_{\mathrm{f}}} \mathbf{T}_\circ / \mathbf{W}_{\mathrm{f}} \cong \mathbf{W}_{\mathrm{f}} imes ig(\mathbf{T}^\wedge imes_{\mathbf{T}} \mathbf{T}_\circig).$$

This implies that $M \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathscr{O}(\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}})$ has a canonical decomposition as a direct sum parametrized by \mathbf{W}_{f} . Moreover, each graded component has a natural structure of representation of the group scheme

$$\mathbf{T}^{\wedge} \times_{\mathbf{T}} \mathbb{J}_{\mathbf{T},\circ},$$

which by (3.14) identifies with

$$(\mathbf{T}^{\wedge} \times_{\mathbf{T}} \mathbf{T}_{\circ}) \times \mathbf{T}.$$

This component therefore admits a canonical grading by $X^*(\mathbf{T})$. We can then obtain a decomposition of $M \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathscr{O}(\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}})$ parametrized by \mathbf{W} by defining the summand associated with $t(\lambda)w$ ($\lambda \in X^*(\mathbf{T}), w \in \mathbf{W}_{\mathrm{f}}$) as the λ -graded part in the summand associated with w. Now the morphism

$$\mathscr{O}(\mathbf{T}^{\wedge}) \to \operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge}))$$

factors through the morphism $\mathscr{O}(\mathbf{T}^{\wedge}) \to \mathscr{O}(\mathbf{T}^{\wedge}) \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathscr{O}(\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}})$, and we have

$$M \otimes_{\mathscr{O}(\mathbf{T}^{\wedge})} \operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge})) = \\ \left(M \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathscr{O}(\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}}) \right) \otimes_{\mathscr{O}(\mathbf{T}^{\wedge}) \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathscr{O}(\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}})} \operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge})).$$

From the decomposition of $M \otimes_{\mathscr{O}(\mathbf{T}/\mathbf{W}_{\mathrm{f}})} \mathscr{O}(\mathbf{T}_{\circ}/\mathbf{W}_{\mathrm{f}})$ parametrized by \mathbf{W} we therefore obtain a decomposition of $M \otimes_{\mathscr{O}(\mathbf{T}^{\wedge})} \operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge}))$ parametrized by \mathbf{W} , which finishes the construction of the functor (3.12).

It is clear from construction that our functor (3.12) has a canonical monoidal structure, and takes values in objects whose underlying $\mathscr{O}(\mathbf{T}^{\wedge})$ -bimodule is finitely generated and flat as a right $\mathscr{O}(\mathbf{T}^{\wedge})$ -module. It therefore restricts to a monoidal functor

$$\mathsf{Rep}_{\mathrm{fl}}(\mathbb{J}^{\wedge}_{\mathbf{D}}) \to \mathsf{K}^{\wedge}.$$

We now need to prove that this functor is fully faithful. Morphisms in both of these categories are by definition certain morphisms of $\mathscr{O}(\mathbf{D}^{\wedge})$ -modules; the functor is therefore faithful. If M and N are in $\operatorname{\mathsf{Rep}_{fl}}(\mathbb{J}^{\wedge}_{\mathbf{D}})$, a morphism in K^{\wedge} from the image of M to the image of N is a morphism $f: M \to N$ of $\mathscr{O}(\mathbf{D}^{\wedge})$ -modules such that the induced morphism

$$M \otimes_{\mathscr{O}(\mathbf{T}^{\wedge})} \operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge})) \to N \otimes_{\mathscr{O}(\mathbf{T}^{\wedge})} \operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge}))$$

is a morphism of representations of the group scheme

$$\mathbb{J}^{\wedge}_{\mathbf{D}} \times_{\mathbf{T}^{\wedge}} \operatorname{Spec}(\operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge})))$$

over $\mathbf{D}^{\wedge} \times_{\mathbf{T}^{\wedge}} \operatorname{Spec}(\operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge})))$. To check that f is a morphism of representations of $\mathbb{J}^{\wedge}_{\mathbf{D}}$ we need to check that the two natural morphisms

$$M \to N \otimes_{\mathscr{O}(\mathbf{D}^{\wedge})} \mathscr{O}(\mathbb{J}^{\wedge}_{\mathbf{D}})$$

constructed out of it coincide. Now since N is flat over $\mathscr{O}(\mathbf{T}^{\wedge})$ (for the action on the right) and $\mathscr{O}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ is flat over $\mathscr{O}(\mathbf{D}^{\wedge})$, the right-hand side is flat over $\mathscr{O}(\mathbf{T}^{\wedge})$ (for the action on the right), so that to check this condition it suffices to prove that the induced morphisms

$$M \otimes_{\mathscr{O}(\mathbf{T}^{\wedge})} \operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge})) \to \left(N \otimes_{\mathscr{O}(\mathbf{D}^{\wedge})} \mathscr{O}(\mathbb{J}^{\wedge}_{\mathbf{D}})\right) \otimes_{\mathscr{O}(\mathbf{T}^{\wedge})} \operatorname{Frac}(\mathscr{O}(\mathbf{T}^{\wedge}))$$

coincide, which is exactly the condition given by the fact that f is a morphism in K^{\wedge} .

Finally we prove that our functor sends each \mathscr{M}_w^{\wedge} to F_w (for $w \in \mathbf{W}$) and each \mathscr{B}_s^{\wedge} to B_s (for $s \in \mathbf{S}_{\mathrm{f}}$). The case of the objects \mathscr{B}_s^{\wedge} is clear. It is clear also that this functor sends $\mathscr{M}_{\mathsf{t}(\lambda)}^{\wedge}$ to $F_{\mathsf{t}(\lambda)}$ for any $\lambda \in X^*(\mathbf{T})$, and \mathscr{M}_x^{\wedge} to F_x for any $x \in \mathbf{W}_{\mathrm{f}}$. By monoidality, it therefore sends \mathscr{M}_w^{\wedge} to F_w for any $w \in \mathbf{W}$.

We can now give the proof of Lemma 3.5.

Proof of Lemma 3.5. If (w_1, s'_1) and (w_2, s'_2) are two pairs of elements as in §3.1 for the same element $s \in \mathbf{S} \setminus \mathbf{S}_{\mathrm{f}}$, then by Lemma 3.7 the images under the functor of Proposition 3.12 of the objects

$$\mathscr{M}_{w_1}^{\wedge} \circledast \mathscr{B}_{s'_1}^{\wedge} \circledast \mathscr{M}_{w_1^{-1}}^{\wedge} \quad \text{and} \quad \mathscr{M}_{w_2}^{\wedge} \circledast \mathscr{B}_{s'_2}^{\wedge} \circledast \mathscr{M}_{w_2^{-1}}^{\wedge}$$

are canonically isomorphic. By fully faithfulness, this implies that these objects are canonically isomorphic, proving that the definition of \mathscr{B}_s^{\wedge} is independent of the choice of (w, s').

The proof of the second claim is similar.

From the proof of Lemma 3.5 we see that the functor of Proposition 3.12 also sends \mathscr{B}_s^{\wedge} to B_s for any $s \in \mathbf{S} \setminus \mathbf{S}_f$; as a consequence, by monoidality, for any $\omega \in \Omega$

and $s_1, \dots, s_i \in \mathbf{S}$ it sends $\mathscr{M}^{\wedge}_{\omega} \otimes \mathscr{B}^{\wedge}_{s_1} \otimes \dots \otimes \mathscr{B}^{\wedge}_{s_i}$ to $F_{\omega} \star B_{s_1} \star \dots \star B_{s_i}$. It follows that this functor induces an equivalence of monoidal categories

$$(3.15) \qquad \qquad \mathsf{BSRep}(\mathbb{J}^{\wedge}_{\mathbf{D}}) \xrightarrow{\sim} \mathsf{BSK}^{\wedge}$$

which is the identity on objects.

3.6. "Additive" Hecke categories. From now on, in addition to our running assumptions we will assume that ℓ is good for **G** abd that there exists a **G**-equivariant isomorphism $\mathbf{g} \xrightarrow{\sim} \mathbf{g}^*$. (We fix such an identification.) Of course this assumption holds if $\mathbf{G} = \operatorname{GL}_n$. It also holds if ℓ is very good for **G**, see [Let, Proposition 2.5.12].

We consider the ring $\mathscr{O}(\mathbf{t}^*)$, endowed with the grading such that $\mathbf{t} \subset \mathscr{O}(\mathbf{t}^*)$ is placed in degree 2, and with the action of \mathbf{W} obtained from the natural action of \mathbf{W}_{f} by pullback along the projection $\mathbf{W} \to \mathbf{W}_{\mathrm{f}}$. Conditions (3.10) are again satisfied in this case; indeed by Remark 3.8 we can assume that $s \in \mathbf{S}_{\mathrm{f}}$. In this case, if α is the associated simple root, as explained in [EW, Claim 3.11] one can take as δ_s any element $x \in \mathbf{t}$ such that $d(\alpha)(x) = 1$ where $d(\alpha)$ is the differential of α . (Such an element does exist thanks to our assumption that $X^*(\mathbf{T})/\mathbb{ZR}$ has no ℓ -torsion.) The categories $\mathsf{K}_{\mathrm{gr}}(\mathscr{O}(\mathbf{t}^*))$ and $\mathsf{BSK}_{\mathrm{gr}}(\mathscr{O}(\mathbf{t}^*))$ will be denoted

$$K_{add}$$
 and BSK_{add} .

These categories are (up to the subtleties related to length-0 elements) the categories denoted C and \mathcal{BS} in [Ab1], for the following data:

- the underlying \Bbbk -vector space is $V = \mathbf{t}$;
- if $s \in S_{\mathbf{f}}$, and if α is the simple root associated with s, then the "root" $\alpha_s \in \mathbf{t}$ is the differential of α^{\vee} , and the "coroot" $\alpha_s^{\vee} \in \mathbf{t}^*$ is the differential of α ;
- if $\beta \in \mathfrak{R}_+$ is a maximal short root and $s = \mathbf{t}(\beta)s_\beta$, then the "root" $\alpha_s \in \mathbf{t}$ is the opposite of the differential of β^{\vee} , and the "coroot" $\alpha_s^{\vee} \in \mathbf{t}^*$ is the opposite of the differential of β .

(As explained in [BR1, §2.2], these data satisfy the technical assumptions imposed in [Ab1].)

We will now denote by $(\mathbf{t}^*)^{\wedge}$ the spectrum of the completion of $\mathscr{O}(\mathbf{t}^*)$ with respect to the ideal $\mathbf{t} \cdot \mathscr{O}(\mathbf{t}^*)$. We will consider a third family of categories as in §3.4, now associated with the ring $\mathscr{O}((\mathbf{t}^*)^{\wedge})$. To check that conditions (3.8) hold in this case, one can e.g. use the following "additive" variant of Lemma 3.3 (applied to the Levi factor of **G** associated with s). Here we denote by $(\mathbf{t}^*/\mathbf{W}_f)^{\wedge}$ the spectrum of the completion of $\mathscr{O}(\mathbf{t}^*/\mathbf{W}_f)$ with respect to the ideal corresponding to the image of $0 \in \mathbf{t}^*$, we set

$$\mathbf{D}_{\mathrm{add}} := \mathbf{t}^* \times_{\mathbf{t}^*/\mathbf{W}_{\mathrm{f}}} \mathbf{t}^*,$$

and we denote by $\mathbf{D}_{\text{add}}^{\wedge}$ the spectrum of the completion of $\mathscr{O}(\mathbf{D}_{\text{add}})$ with respect to the ideal corresponding to $(0,0) \in \mathbf{D}_{\text{add}}$.

Lemma 3.13. (1) There exist canonical isomorphisms of k-schemes

$$(\mathbf{t}^*)^{\wedge} \cong \mathbf{t}^* \times_{\mathbf{t}^*/\mathbf{W}_{\mathrm{f}}} (\mathbf{t}^*/\mathbf{W}_{\mathrm{f}})^{\wedge}$$

and

$$\begin{split} \mathbf{D}_{\mathrm{add}}^{\wedge} &\cong (\mathbf{t}^*)^{\wedge} \times_{\mathbf{t}^*} \mathbf{D}_{\mathrm{add}} \cong \mathbf{D}_{\mathrm{add}} \times_{\mathbf{t}^*} (\mathbf{t}^*)^{\wedge} \\ &\cong \mathbf{D}_{\mathrm{add}} \times_{\mathbf{t}^*/\mathbf{W}_{\mathrm{f}}} (\mathbf{t}^*/\mathbf{W}_{\mathrm{f}})^{\wedge} \cong (\mathbf{t}^*)^{\wedge} \times_{(\mathbf{t}^*/\mathbf{W}_{\mathrm{f}})^{\wedge}} (\mathbf{t}^*)^{\wedge} \end{split}$$

where in the first, resp. second, fiber product the morphism $\mathbf{D}_{add} \to \mathbf{t}^*$ is induced by projection on the first, resp. second, factor. Moreover, $\mathscr{O}((\mathbf{t}^*)^{\wedge})$ is finite and free (in particular, flat) over $\mathscr{O}((\mathbf{t}^*/\mathbf{W}_f)^{\wedge})$.

(2) The natural morphism $\mathscr{O}((\mathbf{t}^*/\mathbf{W}_f)^{\wedge}) \to \mathscr{O}((\mathbf{t}^*)^{\wedge})^{\mathbf{W}_f}$ is an isomorphism.

Proof. The proof of (1) is similar to that of the corresponding claim in Lemma 3.3, replacing the reference to Theorem 2.1 to a reference to the main result of [Dem] (applied to the "precised" root system \mathfrak{R}^{\vee} in $X_*(\mathbf{T})$; our assumptions guarantee that ℓ is not a torsion prime for this root system).

To prove (2), let us set $\mathcal{K}_{add} := \mathbf{t} \cdot \mathscr{O}(\mathbf{t}^*)$ and $\mathcal{J}_{add} := \mathcal{K}_{add} \cap \mathscr{O}(\mathbf{t}^*/W_f)$, so that $\mathscr{O}((\mathbf{t}^*)^{\wedge})$ is the completion of $\mathscr{O}(\mathbf{t}^*)$ with respect to \mathcal{K}_{add} and $\mathscr{O}((\mathbf{t}^*/\mathbf{W}_f)^{\wedge})$ is the completion of $\mathscr{O}(\mathbf{t}^*/\mathbf{W}_f)$ with respect to \mathcal{J}_{add} . It is easily seen that $\mathscr{O}((\mathbf{t}^*)^{\wedge})^{\mathbf{W}_f}$ is the completion of $\mathscr{O}(\mathbf{t}^*/\mathbf{W}_f)$ with respect to the (decreasing) family of ideals

$$(\mathcal{K}_{\mathrm{add}})^n \cap \mathscr{O}(\mathbf{t}^*/\mathbf{W}_{\mathrm{f}}) : n \in \mathbb{Z}_{\geq 1}).$$

Now for any $n \geq 1$ we have $(\mathcal{J}_{add})^n \subset (\mathcal{K}_{add})^n \cap \mathscr{O}(\mathbf{t}^*/\mathbf{W}_f)$. On the other hand, as in the proof of Lemma 3.3 there exists N such that $(\mathcal{K}_{add})^N \subset \mathcal{J}_{add} \cdot \mathscr{O}(\mathbf{t}^*)$. We deduce that for any $n \geq 1$ we have $(\mathcal{K}_{add})^{nN} \subset (\mathcal{J}_{add})^n \cdot \mathscr{O}(\mathbf{t}^*)$, and then since the embedding $\mathscr{O}(\mathbf{t}^*/\mathbf{W}_f) \hookrightarrow \mathscr{O}(\mathbf{t}^*)$ admits an $\mathscr{O}(\mathbf{t}^*/\mathbf{W}_f)$ -linear retraction (again by the main result of [Dem]), we deduce that $(\mathcal{K}_{add})^{nN} \cap \mathscr{O}(\mathbf{t}^*/\mathbf{W}_f) \subset (\mathcal{J}_{add})^n$, so that our two completions are isomorphic.

The categories $\mathsf{K}(\mathscr{O}((\mathbf{t}^*)^\wedge))$ and $\mathsf{BSK}(\mathscr{O}((\mathbf{t}^*)^\wedge))$ will be denoted

$$\mathsf{K}^\wedge_{\mathrm{add}}$$
 and $\mathsf{BSK}^\wedge_{\mathrm{add}}$

respectively.

3.7. Additive Hecke categories and representations of the (additive) universal centralizer. We begin with some generalities on (affine) group schemes and their categories of representations. Given a commutative finitely generated \mathbb{Z} -graded k-algebra R (or, equivalently, an affine k-scheme $X = \operatorname{Spec}(R)$ of finite type endowed with an action of the multiplicative group \mathbb{G}_m over k) and a \mathbb{Z} -graded R-Hopf algebra A (or, equivalently, an affine group scheme $H = \operatorname{Spec}(A)$ over Xendowed with an action of \mathbb{G}_m such that the structure morphism $H \to X$, the multiplication morphism $H \times_X H \to H$, the inversion morphism $H \to H$ and the unit section $X \to H$ are \mathbb{G}_m -equivariant), we will denote by $\operatorname{Rep}^{\mathbb{G}_m}(H)$ the category of \mathbb{G}_m -equivariant representations of H on coherent \mathscr{O}_X -modules, or in other words the category of \mathbb{Z} -graded A-comodules which are finitely generated as R-modules. This category admits a "shift of grading" functor (1), defined with the same convention as in §3.4.

We can also forget about the \mathbb{G}_{m} -action, and consider simply the category $\mathsf{Rep}(H)$ of A-comodules which are finitely generated as R-modules. We then have a canonical forgetful functor

$$\operatorname{For}^{\mathbb{G}_{\mathrm{m}}} : \operatorname{Rep}^{\mathbb{G}_{\mathrm{m}}}(H) \to \operatorname{Rep}(H)$$

which satisfies $\operatorname{For}^{\mathbb{G}_{m}} \circ (1) = \operatorname{For}^{\mathbb{G}_{m}}$.

Finally, given an algebra morphism $R \to R'$ where R' is also commutative and of finite type as a k-algebra, we can set $X' := \operatorname{Spec}(R')$ and consider the group scheme obtained by base change

$$X' \times_X H = \operatorname{Spec}(R' \otimes_R A)$$
and its category of representations (finite over R') $\operatorname{Rep}(X' \times_X H)$. We then have a canonical functor

$$R' \otimes_R (-) : \operatorname{Rep}(H) \to \operatorname{Rep}(X' \times_X H).$$

Lemma 3.14. (1) For any M, M' in $\operatorname{Rep}^{\mathbb{G}_{m}}(H)$, the functor $\operatorname{For}^{\mathbb{G}_{m}}$ induces an isomorphism

$$\bigoplus_{n \in \mathbb{Z}} \operatorname{Hom}_{\operatorname{Rep}^{\mathbb{G}_{m}}(H)}(M, M'(n)) \xrightarrow{\sim} \operatorname{Hom}_{\operatorname{Rep}(H)}(\operatorname{For}^{\mathbb{G}_{m}}(M), \operatorname{For}^{\mathbb{G}_{m}}(M')).$$

(2) Assume that the morphism $R \to R'$ is flat. Then for any M, M' in $\operatorname{Rep}(H)$, the functor $R' \otimes_R (-)$ induces an isomorphism

$$R' \otimes_R \operatorname{Hom}_{\operatorname{Rep}(H)}(M, M') \xrightarrow{\sim} \operatorname{Hom}_{\operatorname{Rep}(X' \times_X H)}(R' \otimes_R M, R' \otimes_R M').$$

Proof. (1) We will prove this property when M' is more generally a \mathbb{Z} -graded Acomodule which is not necessarily finitely generated over R. (The category of
such objects will be denoted $\operatorname{Rep}_{\infty}^{\mathbb{G}_m}(H)$.) First, assume that $M' = V \otimes_R A$ for
some graded R-module V (with the coaction induced by the comultiplication in A).
Then by Frobenius reciprocity ([Ja, Proposition I.3.4]), for any $n \in \mathbb{Z}$ we have

$$\operatorname{Hom}_{\operatorname{\mathsf{Rep}}_{\operatorname{\mathsf{cen}}}^{\mathbb{G}_{\mathrm{m}}}(H)}(M, M'(n)) \cong \operatorname{Hom}_{\operatorname{\mathsf{Mod}}^{\mathbb{Z}}(R)}(M, V(n))$$

where $\mathsf{Mod}^{\mathbb{Z}}(R)$ is the category of \mathbb{Z} -graded *R*-modules, and

$$\operatorname{Hom}_{\operatorname{Rep}(H)}(\operatorname{For}^{\mathbb{G}_{\mathrm{m}}}(M), \operatorname{For}^{\mathbb{G}_{\mathrm{m}}}(M')) \cong \operatorname{Hom}_{\operatorname{Mod}(R)}(M, V).$$

Now it is a classical fact that since M is finitely generated over R the forgetful functor induces an isomorphism

$$\bigoplus_{n \in \mathbb{Z}} \operatorname{Hom}_{\mathsf{Mod}^{\mathbb{Z}}(R)}(M, V(n)) \xrightarrow{\sim} \operatorname{Hom}_{\mathsf{Mod}(R)}(M, V);$$

the desired claim follows in this case.

The case of a general \mathbb{Z} -graded A-comodule M' follows from this special case using the five lemma and the fact that for any such M' the coaction defines an injective morphism of \mathbb{Z} -graded A-comodules $M' \to M' \otimes_R A$, where in the righthand side M' is regarded as a graded R-module.

(2) As explained e.g. in [BR1, Lemma 3.8(2)], the *R*-module $\operatorname{Hom}_R(M, M')$ admits a natural structure of *A*-comodule, and we have

$$\operatorname{Hom}_{\operatorname{Rep}(H)}(M, M') = \left(\operatorname{Hom}_{R}(M, M')\right)^{H},$$

where $(-)^H$ is the functor of *H*-fixed points. Similarly, $\operatorname{Hom}_{R'}(R' \otimes_R M, R' \otimes_R M')$ admits a natural structure of $R' \otimes_R A$ -comodule, and we have

$$\operatorname{Hom}_{\operatorname{\mathsf{Rep}}(X'\times_X H)}(R'\otimes_R M, R'\otimes_R M') = \left(\operatorname{Hom}_{R'}(R'\otimes_R M, R'\otimes_R M')\right)^{X'\times_X H}$$

where $(-)^{X' \times_X H}$ is the functor of $X' \times_X H$ -fixed points. Now we have

 $\operatorname{Hom}_{R'}(R' \otimes_R M, R' \otimes_R M') = \operatorname{Hom}_R(M, R' \otimes_R M'),$

and since R' is flat over R we have

$$\operatorname{Hom}_R(M, R' \otimes_R M') = R' \otimes_R \operatorname{Hom}_R(M, M')$$

by [BR1, Lemma 3.8(1)]. Finally by [Ja, Equation I.2.10(3)], using again our flatness assumption we have

$$(R' \otimes_R \operatorname{Hom}_R(M, M'))^{X' \times_X H} = R' \otimes_R (\operatorname{Hom}_R(M, M'))^H.$$

Combining these isomorphisms we deduce the desired claim.

From now on we fix a Kostant section $\mathbf{S} \subset \mathbf{g}$ as in [BR1, §2.3], and denote by \mathbf{S}^* its image under our identification $\mathbf{g} \xrightarrow{\sim} \mathbf{g}^*$. (A Kostant section is an "additive" variant of the Steinberg section Σ ; in particular the coadjoint quotient provides an isomorphism $\mathbf{S}^* \xrightarrow{\sim} \mathbf{t}^*/\mathbf{W}_{\mathrm{f}}$.) The same considerations as in §2.8 lead to the definition of the universal centralizer group scheme $\mathbb{I}_{\mathbf{S}^*}$ over \mathbf{S}^* , see [R2] and [BR1, §2.3] for details. This is a smooth affine group scheme over \mathbf{S}^* , endowed with an action of \mathbb{G}_{m} which is compatible (in the sense considered above Lemma 3.14) with the action on $\mathbf{t}^*/\mathbf{W}_{\mathrm{f}}$ such that the quotient morphism $\mathbf{t}^* \to \mathbf{t}^*/\mathbf{W}_{\mathrm{f}}$ is \mathbb{G}_{m} -equivariant, where $t \in \mathbb{G}_{\mathrm{m}}$ acts on \mathbf{t}^* by multiplication by t^{-2} . We can then consider the group scheme

$$\mathbb{I}_{\mathbf{D}_{\mathrm{add}}} := \mathbf{t}^* imes_{\mathbf{t}^*/\mathbf{W}_{\mathrm{f}}} \mathbb{I}_{\mathbf{S}^*} imes_{\mathbf{t}^*/\mathbf{W}_{\mathrm{f}}} \mathbf{t}^*$$

and the associated category

$$\mathsf{Rep}^{\mathbb{G}_m}(\mathbb{I}_{\mathbf{D}_{\mathrm{add}}})$$

of \mathbb{G}_m -equivariant representations on coherent sheaves. This category admits a canonical convolution product defining a monoidal structure. If we denote by

$$\mathsf{Rep}_{\mathrm{fl}}^{\mathbb{G}_{\mathrm{m}}}(\mathbb{I}_{\mathbf{D}_{\mathrm{add}}})$$

the full subcategory whose objects are the representations whose underlying coherent sheaves are flat with respect to the second projection $\mathbf{D}_{add} \rightarrow \mathbf{t}^*$, then this full subcategory is stable under convolution, hence a monoidal category.

On the other hand, set

$$\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathrm{add}}} := \mathbf{D}^{\wedge}_{\mathrm{add}} imes_{\mathbf{D}_{\mathrm{add}}} \mathbb{I}_{\mathbf{D}_{\mathrm{add}}}.$$

Then once again the category $\mathsf{Rep}(\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathrm{add}}})$ admits a canonical convolution product which makes it a monoidal category. If we denote by $\mathsf{Rep}_{\mathrm{fl}}(\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathrm{add}}})$ the full subcategory whose objects are the representations whose underlying coherent sheaves are flat with respect to the second projection $\mathbf{D}^{\wedge}_{\mathrm{add}} \to (\mathbf{t}^*)^{\wedge}$, then this subcategory is stable under convolution, hence a monoidal category.

Proposition 3.15.(1) There exists a canonical fully faithful monoidal functor(3.16) $\mathsf{Rep}_{\mathbb{G}^m}^{\mathbb{G}_m}(\mathbb{I}_{\mathbf{D}_{add}}) \to \mathsf{K}_{add},$

whose essential image contains the objects B_s $(s \in S)$ and F_w $(w \in \mathbf{W})$. (2) There exists a canonical fully faithful monoidal functor

$$(3.17) \qquad \qquad \mathsf{Rep}_{\mathrm{fl}}(\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathrm{add}}}) \to \mathsf{K}^{\wedge}_{\mathrm{add}},$$

whose essential image contains the objects B_s ($s \in S$) and F_w ($w \in \mathbf{W}$).

Proof. (1) This statement is proved in [BR1, Proposition 2.7 and Lemma 2.9]. (2) The proof is similar to that of Proposition 3.12.

More specifically, one can define *canonical* objects in $\operatorname{Rep}_{\mathrm{fl}}^{\mathbb{G}_{\mathrm{m}}}(\mathbb{I}_{\mathbf{D}_{\mathrm{add}}})$, resp. in $\operatorname{Rep}_{\mathrm{fl}}(\mathbb{I}_{\mathbf{D}_{\mathrm{add}}})$, whose image under (3.16), resp. (3.17), are the corresponding objects B_s and F_w . Using these objects one obtains that the functors (3.11) in these two settings factor through (fully faithful) monoidal functors

$$\mathsf{BSK}_{\mathrm{add}} \to \mathsf{Rep}_{\mathrm{fl}}^{\mathbb{G}_{\mathrm{m}}}(\mathbb{I}_{\mathbf{D}_{\mathrm{add}}}), \qquad \mathsf{BSK}_{\mathrm{add}}^{\wedge} \to \mathsf{Rep}_{\mathrm{fl}}(\mathbb{I}_{\mathbf{D}_{\mathrm{add}}}^{\wedge}).$$

Using the second of these functors one can define a category $\mathsf{BSRep}(\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathrm{add}}})$ of "Bott–Samelson type" representations of $\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathrm{add}}}$, with objects the collections $(\omega, s_1, \cdots, s_i)$

with $\omega \in \Omega$ and $s_1, \dots, s_i \in S$, and which is canonically equivalent to $\mathsf{BSK}^{\wedge}_{\mathrm{add}}$. We can define the category $\mathsf{SRep}(\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathrm{add}}})$ of "Soergel type" representations as the Karoubian closure of the additive hull of $\mathsf{BSRep}(\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathrm{add}}})$; equivalently, this category identifies with the full subcategory of $\mathsf{Rep}_{\mathrm{fl}}(\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathrm{add}}})$ whose objects are direct sums of direct summands of objects in the image of $\mathsf{BSK}^{\wedge}_{\mathrm{add}}$.

We deduce from Proposition 3.15 the following property. (We expect this proposition to admit a direct algebraic proof, but the proof given below relies on geometry and Proposition 3.15.)

Proposition 3.16. There exists a monoidal functor

$$F: \mathsf{K}_{\mathrm{add}} \to \mathsf{K}^{\wedge}_{\mathrm{add}}$$

which satisfies $F \circ (1) = F$ and sends each object F_w ($w \in W$) and B_s ($s \in S$) in K_{add} to the corresponding object in K_{add}^{\wedge} , and such that F induces an isomorphism

 $\operatorname{Hom}_{\mathsf{BSK}_{\mathrm{add}}}^{\bullet}(M,M') \otimes_{\mathscr{O}(\mathbf{t}^*)} \mathscr{O}((\mathbf{t}^*)^{\wedge}) \xrightarrow{\sim} \operatorname{Hom}_{\mathsf{BSK}_{\mathrm{add}}^{\wedge}}(\mathsf{F}(M),\mathsf{F}(M'))$

for any M, M' in BSK_{add} .

Proof. The functor F is defined by

$$\mathsf{F}(M) = M \otimes_{\mathscr{O}(\mathbf{t}^*)} \mathscr{O}((\mathbf{t}^*)^{\wedge}),$$

where in the right-hand side we omit the functor forgetting the \mathbb{Z} -grading. Let us first explain why this indeed defines a functor from $\mathsf{K}_{\mathrm{add}}$ to $\mathsf{K}^{\wedge}_{\mathrm{add}}$. Here since M is an object in $\mathsf{K}_{\mathrm{add}}$, it admits in particular an action of $\mathscr{O}(\mathbf{D}_{\mathrm{add}})$. Hence $M \otimes_{\mathscr{O}(\mathbf{t}^*)} \mathscr{O}((\mathbf{t}^*)^{\wedge})$ admits an action of

$$\mathscr{O}(\mathbf{D}_{\mathrm{add}}) \otimes_{\mathscr{O}(\mathbf{t}^*)} \mathscr{O}((\mathbf{t}^*)^{\wedge}),$$

which identifies with $\mathscr{O}(\mathbf{D}^{\wedge}_{\mathrm{add}})$ by Lemma 3.13. This object can therefore be regarded as a (finitely generated) $\mathscr{O}((\mathbf{t}^*)^{\wedge})$ -bimodule. On the other hand, we have

$$\begin{split} \left(M \otimes_{\mathscr{O}(\mathbf{t}^*)} \mathscr{O}((\mathbf{t}^*)^{\wedge}) \right) \otimes_{\mathscr{O}((\mathbf{t}^*)^{\wedge})} \operatorname{Frac}(\mathscr{O}((\mathbf{t}^*)^{\wedge})) \cong \\ \left(M \otimes_{\mathscr{O}(\mathbf{t}^*)} \operatorname{Frac}(\mathscr{O}(\mathbf{t}^*)) \right) \otimes_{\operatorname{Frac}(\mathscr{O}(\mathbf{t}^*))} \operatorname{Frac}(\mathscr{O}((\mathbf{t}^*)^{\wedge})). \end{split}$$

Here we are given a decomposition of $M \otimes_{\mathscr{O}(\mathbf{t}^*)} \operatorname{Frac}(\mathscr{O}(\mathbf{t}^*))$ parametrized by \mathbf{W} , which induces a decomposition of $(M \otimes_{\mathscr{O}(\mathbf{t}^*)} \mathscr{O}((\mathbf{t}^*)^{\wedge})) \otimes_{\mathscr{O}((\mathbf{t}^*)^{\wedge})} \operatorname{Frac}(\mathscr{O}((\mathbf{t}^*)^{\wedge}))$ parametrized by \mathbf{W} . Finally $M \otimes_{\mathscr{O}(\mathbf{t}^*)} \mathscr{O}((\mathbf{t}^*)^{\wedge})$ is flat over $\mathscr{O}((\mathbf{t}^*)^{\wedge})$ for the action on the right, hence it indeed admits a canonical structure of object in $\mathsf{K}^{\wedge}_{\mathrm{add}}$.

It is easily checked that F has a canonical monoidal structure, and the required action on the objects F_w and B_s . To check that this functor has the required property on morphism spaces, we consider the equivalences of Proposition 3.15, and the functor of pullback under the natural morphism $\mathbf{D}_{\mathrm{add}}^{\wedge} \to \mathbf{D}_{\mathrm{add}}$ (and forgetting the grading). This defines a natural monoidal functor

$$(3.18) \qquad \qquad \mathsf{Rep}_{\mathrm{fl}}^{\mathbb{G}_{\mathrm{im}}}(\mathbb{I}_{\mathbf{D}_{\mathrm{add}}}) \to \mathsf{Rep}_{\mathrm{fl}}(\mathbb{I}_{\mathbf{D}_{\mathrm{add}}}^{\wedge})$$

and, in view of the identification $\mathscr{O}(\mathbf{D}^{\wedge}_{\mathrm{add}}) \cong \mathscr{O}(\mathbf{D}_{\mathrm{add}}) \otimes_{\mathscr{O}(\mathbf{t}^*)} \mathscr{O}((\mathbf{t}^*)^{\wedge})$, the diagram



commutes. The desired property of F therefore follows from the corresponding property of the functor (3.18), which itself follows from Lemma 3.14 and the identification $\mathscr{O}(\mathbf{D}^{\wedge}_{\mathrm{add}}) \cong \mathscr{O}(\mathbf{D}_{\mathrm{add}}) \otimes_{\mathscr{O}(\mathbf{t}^*)} \mathscr{O}((\mathbf{t}^*)^{\wedge})$.

3.8. Relation between the "multiplicative" and "additive" Hecke categories. Finally we explain the relation between the (completed) "additive" and "multiplicative" Hecke categories.

Lemma 3.17. Assume that there exists an étale isogeny $\mathbf{G}' \to \mathbf{G}$ and a morphism $\mathbf{G}' \to \operatorname{Lie}(\mathbf{G}')$ which is \mathbf{G}' -equivariant (for the adjoint actions), sends e to 0, and is étale at e. Then there exists a \mathbf{W}_{f} -equivariant isomorphism $\mathscr{O}((\mathbf{t}^*)^{\wedge}) \xrightarrow{\sim} \mathscr{O}(\mathbf{T}^{\wedge})$, from which we obtain an equivalence of monoidal categories

$$\mathsf{K}^\wedge_{\mathrm{add}}\cong\mathsf{K}^\wedge$$

sending each object F_w ($w \in \mathbf{W}$) or B_s ($s \in \mathbf{S}$) in $\mathsf{K}^{\wedge}_{\mathrm{add}}$ to the corresponding object in K^{\wedge} . As a consequence, we obtain an equivalence of monoidal categories

$$\mathsf{BSK}^{\wedge}_{\mathrm{add}} \cong \mathsf{BSK}^{\wedge}$$

which is the identity on objects.

Proof. By assumption there exists a **G**-equivariant isomorphism $\mathbf{g} \xrightarrow{\sim} \mathbf{g}^*$; if one identifies \mathbf{t}^* with the subspace of \mathbf{g}^* consisting of linear forms vanishing on each root subspace, then this isomorphism must restrict to an isomorphism from $\mathbf{t} = \mathbf{g}^{\mathbf{T}}$ to $\mathbf{t}^* = (\mathbf{g}^*)^{\mathbf{T}}$, which is W_{f} -invariant. To construct our isomorphism it therefore suffices to construct a W_{f} -equivariant isomorphism from $\mathscr{O}(\mathbf{T}^{\wedge})$ to the completion of $\mathscr{O}(\mathbf{t})$ with respect to the ideal corresponding to 0.

Consider now an étale isogeny $\mathbf{G}' \to \mathbf{G}$ as in the statement. If $\mathbf{T}' \subset \mathbf{G}'$ is the preimage of \mathbf{T} , then \mathbf{T}' is a maximal torus in \mathbf{G}' , and our isogeny restricts to an étale morphism $\mathbf{T}' \to \mathbf{T}$ sending e to e. It therefore induces an isomorphism between $\mathscr{O}(\mathbf{T}^{\wedge})$ and the completion $\mathscr{O}(\mathbf{T}')^{\wedge}$ of $\mathscr{O}(\mathbf{T}')$ with respect to the ideal corresponding to e, and also an isomorphism from $\text{Lie}(\mathbf{T}')$ to \mathbf{t} . The Weyl group of $(\mathbf{G}', \mathbf{T}')$ canonically identifies with \mathbf{W}_{f} , and both of our isomorphisms are \mathbf{W}_{f} equivariant.

Our morphism $\mathbf{G}' \to \operatorname{Lie}(\mathbf{G}')$ must restrict to a \mathbf{W}_{f} -equivariant morphism from $\mathbf{T}' = (\mathbf{G}')^{\mathbf{T}'}$ to $\operatorname{Lie}(\mathbf{T}') = (\operatorname{Lie}(\mathbf{G}'))^{\mathbf{T}'}$. Moreover, this morphism sends e to 0 and is étale at e (e.g. by consideration of tangent spaces). It therefore induces a \mathbf{W}_{f} -equivariant isomorphism between $\mathscr{O}(\mathbf{T}')^{\wedge}$ and the completion of $\mathscr{O}(\operatorname{Lie}(\mathbf{T}'))$ with respect to the ideal corresponding to 0. Combining these isomorphisms we deduce the desired isomorphism

$$\mathscr{O}((\mathbf{t}^*)^{\wedge}) \xrightarrow{\sim} \mathscr{O}(\mathbf{T}^{\wedge}).$$

Once this isomorphism is constructed, we deduce the desired equivalences using Lemma 3.10. $\hfill \Box$

Lemma 3.17, together with the equivalence (3.15) and its analogue deduced from Proposition 3.15(2), we obtain an equivalence of additive monoidal categories

$$(3.19) \qquad \qquad \mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}}) \xrightarrow{\sim} \mathsf{SRep}(\mathbb{I}^{\wedge}_{\mathbf{D}_{\mathsf{pdd}}})$$

Remark 3.18. The assumption in Lemma 3.17 holds at least in the following cases:

- (1) $\mathbf{G} = \mathrm{GL}_n(\mathbb{k});$
- (2) ℓ is very good.

In fact, in the first case one can take $\mathbf{G}' = \mathbf{G}$, with the morphism $\operatorname{GL}_n(\mathbb{k}) \to \mathfrak{gl}_n(\mathbb{k})$ given by $X \mapsto X - \mathbf{I}_n$. For the second case one observes first that if ℓ is very good and \mathbf{G} is semisimple (and simply connected) then there exists a morphism $\mathbf{G} \to \mathbf{g}$ sending e to 0 and étale at e: if $\mathbf{G} = \operatorname{SL}_n$ on can take $X \mapsto X - \frac{\operatorname{tr}(X)}{n}\mathbf{I}_n$, if \mathbf{G} is quasi-simple and not of type \mathbf{A} this is a standard consequence of results of Springer–Steinberg (see e.g. [AR1, §5.3]), and the general case follows since \mathbf{G} is a product of such groups. The similar claim of course also holds if \mathbf{G} is a torus. Finally, for a general \mathbf{G} , as explained in [Ja, §1.18] there exists a torus \mathbf{H} and an isogeny $\mathscr{D}(\mathbf{G}) \times \mathbf{H} \to \mathbf{G}$ (where $\mathscr{D}(\mathbf{G})$) is the derived subgroup of \mathbf{G}) whose kernel is a subgroup of the center of $\mathscr{D}(\mathbf{G})$. Since ℓ is very good this center is a discrete group, hence this kernel is smooth, proving that the isogeny is étale. One can therefore take $\mathbf{G}' = \mathscr{D}(\mathbf{G}) \times \mathbf{H}$.

4. Constructible sheaves on Affine flag varieties

4.1. Affine flag varieties. Let \mathbb{F} be an algebraically closed field of characteristic p > 0, and let G be a connected reductive algebraic group over \mathbb{F} . We fix a Borel subgroup $B \subset G$, whose unipotent radical will be denoted U, and a maximal torus $T \subset B$.

Recall that the loop group LG, resp. the positive loop group L^+G , is the group ind-scheme, resp. group scheme, over \mathbb{F} which represents the functor

 $R \mapsto G(R((z))), \text{ resp. } R \mapsto G(R[[z]]),$

where z is an indeterminate. By definition L^+G is a subgroup scheme of LG, hence one can define the affine Grassmannian Gr_G as the fppf quotient

$$\operatorname{Gr}_G = \left(LG/L^+G \right)_{\operatorname{fppf}}.$$

It is well known that Gr_G is an ind-projective ind-scheme over \mathbb{F} .

There exists a canonical morphism of group schemes $L^+G \to G$, induced by the assignment $z \mapsto 0$. The Iwahori subgroup $I \subset L^+G$ is defined as the inverse image of B under this morphism. The pro-unipotent radical of I is the subgroup $I_u \subset I$ defined as the preimage of U. We can then define the affine flag variety Fl_G and the canonical T-torsor \widetilde{Fl}_G over Fl_G as the fppf quotients

$$\operatorname{Fl}_G := (LG/I)_{\operatorname{fppf}}, \quad \operatorname{Fl}_G := (LG/I_u)_{\operatorname{fppf}}$$

Once again these are ind-schemes of ind-finite type, and Fl_G is ind-projective. The embeddings $I_u \subset I \subset L^+G$ induce natural morphisms

$$(4.1) Fl_G \to Fl_G \to Gr_G.$$

It is well known that the second morphism is a Zariski locally trivial fibration with fibers G/B, and that the natural action of T on $\widetilde{\mathrm{Fl}}_G$ (induced by right multiplication on LG) exhibits $\widetilde{\mathrm{Fl}}_G$ as a Zariski locally trivial T-torsor over Fl_G ; this map will be denoted $\pi : \widetilde{\mathrm{Fl}}_G \to \mathrm{Fl}_G$.

Consider the coweight lattice $X_*(T)$. The choice of the Borel subgroup B determines a system of positive roots for (G,T) (chosen as the set of T-weights in Lie(G)/Lie(B)), which then define a subset $X^+_*(T) \subset X_*(T)$ of dominant weights. We will denote by \leq the order on $X_*(T)$ such that $\lambda \leq \mu$ if and only if $\mu - \lambda$ is a sum of positive coroots.

Recall that the L^+G -orbits on Gr_G (for the action induced by left multiplication on LG) are parametrized by $X^+_*(T)$. Namely, any $\lambda \in X_*(T)$ determines a point $z^{\lambda} \in LG$, and for $\lambda \in X^+_*(T)$ we denote by $\operatorname{Gr}_G^{\lambda}$ the L^+G -orbit of the image of z^{λ} in Gr_G (with its reduced subscheme structure). We then have

$$(\operatorname{Gr}_G)_{\operatorname{red}} = \bigsqcup_{\lambda \in X^+_*(T)} \operatorname{Gr}_G^{\lambda}.$$

We will denote by $W_{\rm f} = N_G(T)/T$ the Weyl group of (G, T). The choice of B determines a system $S_{\rm f} \subset W_{\rm f}$ of simple reflections, such that $(W_{\rm f}, S_{\rm f})$ is a Coxeter system. The orbits of I on ${\rm Fl}_G$ are naturally parametrized by the extended affine Weyl group

$$W := W_{\mathrm{f}} \ltimes X_*(T).$$

Namely, let us fix for any $v \in W_{\rm f}$ a lift $\dot{v} \in N_G(T)$. Let $w \in W$, and write $w = t(\lambda)v$ with $v \in W_{\rm f}$ and $\lambda \in X_*(T)$. (Here and below, $t(\lambda)$ denotes the image of λ in W.) Then if we denote by $\operatorname{Fl}_{G,w} \subset \operatorname{Fl}_G$ the I-orbit of the image in Fl_G of $z^{\lambda}\dot{v}$ (again with its reduced subscheme structure), we have

(4.2)
$$(\operatorname{Fl}_G)_{\operatorname{red}} = \bigsqcup_{w \in W} \operatorname{Fl}_{G,w}.$$

It is well known also that each ${\rm Fl}_{G,w}$ is an ${\rm I_u}\text{-}{\rm orbit},$ isomorphic to an affine space For $w\in W$ we will set

$$\ell(w) = \dim(\operatorname{Fl}_{G,w}), \quad \widetilde{\operatorname{Fl}}_{G,w} = \pi^{-1}(\operatorname{Fl}_{G,w}).$$

For $w \in W_{\rm f}$, $\ell(w)$ is the length of w for the Coxeter group structure on $W_{\rm f}$ considered above. We will also set $\Omega := \{w \in W \mid \ell(w) = 0\}.$

4.2. I-equivariant sheaves on the affine flag variety and convolution. Let \Bbbk be an algebraic closure of a finite field of characteristic $\ell \neq p$. We can then consider the I-equivariant derived category of étale sheaves on Fl_G with coefficients in \Bbbk , which we will denote by

 $D_{I,I}$.

(The definition of this category requires a little bit of care but is standard; see [BRR, §§4.1–4.2] for some details. The complexes on ind-schemes that we consider will always be supported on a finite-type subscheme. Similar comments apply to several constructions below, where we will "pretend" that some group schemes of infinite type are honest algebraic groups for notational simplicity.) This category admits a natural perverse t-structure, whose heart will be denoted $\mathsf{P}_{\mathrm{I},\mathrm{I}}$. For each $w \in W$, if we denote by $j_w : \mathrm{Fl}_{G,w} \to \mathrm{Fl}_G$ the (locally closed) embedding, then we will set

$$\Delta^{\mathbf{l}}_{w} := (j_{w})_{!} \underline{\Bbbk}_{\mathrm{Fl}_{G,w}}[\ell(w)], \quad \nabla^{\mathbf{l}}_{w} := (j_{w})_{*} \underline{\Bbbk}_{\mathrm{Fl}_{G,w}}[\ell(w)].$$

These define objects in $D_{I,I}$, which are in fact perverse sheaves since j_w is an affine morphism.

The simple objects in the category $\mathsf{P}_{\mathrm{I},\mathrm{I}}$ are naturally labelled by W. Namely, if for $w \in W$ we denote by $\mathscr{I}\mathscr{C}_w$ the intersection cohomology complex associated with the constant local system on $\mathrm{Fl}_{G,w}$ (in other words, the image of the unique—up to scalar—nonzero morphism $\Delta^{\mathrm{I}}_w \to \nabla^{\mathrm{I}}_w$) then the assignment $w \mapsto \mathscr{I}\mathscr{C}_w$ induces a bijection between W and the set of isomorphism classes of simple objects in $\mathsf{P}_{\mathrm{I},\mathrm{I}}$.

The category $D_{I,I}$ also admits a natural convolution product, whose definition we briefly recall. First we consider the ind-scheme $Fl_G \times Fl_G$, defined as the (fppf) quotient of $LG \times \operatorname{Fl}_G$ by the action of I defined by $g \cdot (h, x) = (hg^{-1}, g \cdot x)$. The multiplication map in LG induces a proper morphism $m : \operatorname{Fl}_G \times \operatorname{Fl}_G \to \operatorname{Fl}_G$. Then, given \mathscr{F}, \mathscr{G} in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$, there exists a unique object $\mathscr{F} \boxtimes \mathscr{G}$ in the I-equivariant derived category of $\operatorname{Fl}_G \times \operatorname{Fl}_G$ whose pullback to $LG \times \operatorname{Fl}_G$ is the exterior product of the pullback of \mathscr{F} to LG with \mathscr{G} ; then we set

$$\mathscr{F} \star_{\mathrm{I}} \mathscr{G} := m_! (\mathscr{F} \widetilde{\boxtimes} \mathscr{G}).$$

With this construction the pair $(\mathsf{D}_{\mathrm{I},\mathrm{I}},\star_{\mathrm{I}})$ is a monoidal category, with unit object $\delta := \mathscr{I}\mathscr{C}_e$.

Remark 4.1. Below, given X, Y some I-invariant subschemes in Fl_G , we will also denote by $X \times Y$ the quotient of $X' \times Y$ by the I-action induced by that on $LG \times \operatorname{Fl}_G$ considered above, where X' is the preimage of X in LG.

Similarly one can consider the I_u-equivariant derived category of étale k-sheaves on Fl_G , which will be denoted

 $D_{I_u,I}$.

This category admits a natural perverse t-structure, whose heart will be denoted $P_{I_u,I}$. We have a canonical t-exact "forgetful" functor

$$(4.3) \qquad \qquad \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}:\mathsf{D}_{\mathrm{I},\mathrm{I}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}},$$

and the simple objects in the category $\mathsf{P}_{I_u,I}$ are (up to isomorphism) the objects $\mathsf{For}_{I_u}^{\mathsf{I}}(\mathscr{IC}_w)$. We also have a canonical right action of $\mathsf{D}_{I,I}$ on $\mathsf{D}_{I_u,I}$, defined by a bifunctor

$$(4.4) \qquad \qquad \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \times \mathsf{D}_{\mathrm{I},\mathrm{I}} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

whose construction repeats exactly the definition of \star_{I} ; this bifunctor will also be denoted \star_{I} . With this definition we have

$$\operatorname{\mathsf{For}}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F}\star_{\mathrm{I}}\mathscr{G})\cong\operatorname{\mathsf{For}}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F})\star_{\mathrm{I}}\mathscr{G}$$

for any \mathscr{F}, \mathscr{G} in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$.

Since each I_u -orbit on Fl_G is isomorphic to an affine space, the methods of [BGS, §§3.2–3.3] in the sense considered e.g. in [RW1, §2.1], with underlying poset W (endowed with the Bruhat order) and standard, resp. costandard, object attached to w the perverse sheaf $\mathsf{For}_{I_u}^{I}(\Delta_w^{I})$, resp. $\mathsf{For}_{I_u}^{I}(\nabla_w^{I})$. In particular we have a notion of tilting object in this category (namely, objects which admit both a filtration with subquotients of the form $\mathsf{For}_{I_u}^{I}(\Delta_w^{I})$, and a filtration with subquotients of the form $\mathsf{For}_{I_u}^{I}(\Delta_w^{I})$, and a filtration with subquotients of the form $\mathsf{For}_{I_u}^{I}(\Delta_w^{I})$, and a filtration with subquotients of the form $\mathsf{For}_{I_u}^{I}(\Delta_w^{I})$, and the isomorphism classes of indecomposable tilting objects are in a natural bijection with W. The indecomposable tilting object associated with w will be denoted \mathscr{T}_w .

4.3. Central sheaves – properties. We now consider the action of L^+G on Gr_G , and denote by

$$\mathsf{D}_{L^+G,L^+G}$$

the L^+G -equivariant derived category of étale sheaves on Gr_G with coefficients in \Bbbk . As for $\mathsf{D}_{\mathrm{I},\mathrm{I}}$ we have a convolution bifunctor \star_{L^+G} on this category. We also have a perverse t-structure, whose heart will be denoted

$$\mathsf{P}_{L^+G,L^+G^+}$$

It is a standard but crucial fact that this subcategory is stable under the bifunctor \star_{L+G} ; one can therefore consider the monoidal category ($\mathsf{P}_{L+G,L+G}, \star_{L+G}$). This category is the main ingredient of the *geometric Satake equivalence* of [MV], which provides a canonical connected reductive algebraic group G_{\Bbbk}^{\vee} over \Bbbk with a maximal torus T_{\Bbbk}^{\vee} such that the root datum of $(G_{\Bbbk}^{\vee}, T_{\Bbbk}^{\vee})$ is dual to that of (G, T), and a canonical equivalence of monoidal categories

$$\mathcal{S}: (\mathsf{P}_{L^+G, L^+G}, \star_{L^+G}) \xrightarrow{\sim} (\mathsf{Rep}(G^{\vee}_{\Bbbk}), \otimes).$$

(See [BRR, §4.1] for more precise references.) The unit object in the category P_{L^+G,L^+G} will be denoted δ_{Gr} . (This object is the skyscraper sheaf at the base point of Gr_G .) We will also denote by B^{\vee}_{\Bbbk} the Borel subgroup of G^{\vee}_{\Bbbk} containing T^{\vee}_{\Bbbk} such that the T^{\vee}_{\Bbbk} -weights in the Lie algebra of B^{\vee}_{\Bbbk} are the negative coroots.

Below we will apply the constructions of Sections 2–3 to the group $\mathbf{G} = G_{\Bbbk}^{\vee}$; in that setting, the groups W_{f} and W identify with the groups \mathbf{W}_{f} and \mathbf{W} considered in those sections, and their structures (in particular, the function ℓ) also identify. In this setting we will denote by $S \subset W' \subset W$ the subsets corresponding to $\mathbf{S} \subset \mathbf{W}' \subset \mathbf{W}$.

Recall that the main construction of [Ga1] (reviewed in detail in [AR4]) provides a canonical monoidal functor

$$\mathsf{Z}:\mathsf{D}_{L^+G,L^+G}\to\mathsf{D}_{\mathrm{I},\mathrm{I}}.$$

For \mathscr{A}, \mathscr{B} in D_{L^+G, L^+G} we will denote by

$$\phi_{\mathscr{A},\mathscr{B}}: \mathsf{Z}(\mathscr{A} \star_{L^{+}G} \mathscr{B}) \xrightarrow{\sim} \mathsf{Z}(\mathscr{A}) \star_{\mathrm{I}} \mathsf{Z}(\mathscr{B})$$

the associated "monoidality" isomorphism.

This functor has a number of favorable properties, which are listed in [BRR, §4]. Among these properties, we note the following for later use.

- (1) The functor Z is t-exact with respect to the perverse t-structures.
- (2) For any \mathscr{A} in D_{L^+G,L^+G} and \mathscr{F} in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$, there exists a canonical isomorphism

$$\sigma_{\mathscr{A},\mathscr{F}}:\mathsf{Z}(\mathscr{A})\star_{\mathrm{I}}\mathscr{F}\xrightarrow{\sim}\mathscr{F}\star_{\mathrm{I}}\mathsf{Z}(\mathscr{A}),$$

and Z, together with the isomorphisms ϕ and σ , define a central functor from P_{L^+G,L^+G} to $\mathsf{D}_{\mathrm{I},\mathrm{I}}$ in the sense of [B1]; in other words these data define a braided monoidal functor from P_{L^+G,L^+G} to the Drinfeld center of $\mathsf{D}_{\mathrm{I},\mathrm{I}}$ (with respect to the commutativity constraint on P_{L^+G,L^+G} and the natural braiding on the Drinfeld center).

(3) Since it is defined by nearby cycles, the functor Z comes with a "monodromy" automorphism \mathfrak{m} , such that for \mathscr{A}, \mathscr{B} in D_{L^+G,L^+G} the isomorphism $\phi_{\mathscr{A},\mathscr{B}}$ intertwines $\mathfrak{m}_{\mathscr{A}\star_{L^+G}}\mathscr{G}$ with $\mathfrak{m}_{\mathscr{A}}\star_{\mathrm{I}}\mathfrak{m}_{\mathscr{B}}$.

For simplicity, from now on we fix a total order \leq on $X_*(T)$ compatible with the dominance order, i.e. such that if $\lambda, \mu \in X_*(T)$ are such that $\mu \preceq \lambda$ then $\mu \leq \lambda$. Recall that in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$ we have the *Wakimoto sheaves* ($\mathscr{W}_{\lambda} : \lambda \in X_*(T)$), see [BRR, §4.5], which are perverse sheaves such that for any $\lambda, \mu \in X_*(T)$ we have a canonical isomorphism

(4.5)
$$\mathscr{W}_{\lambda} \star_{\mathrm{I}} \mathscr{W}_{\mu} \cong \mathscr{W}_{\lambda+\mu}$$

(The construction of these objects is due to Mirković, and appears in particular in [AB].) Recall that an object \mathscr{F} of $\mathsf{P}_{I,I}$ (resp. $\mathsf{P}_{I_u,I}$) is said to *admit a Wakimoto filtration* if there exists a finite filtration on \mathscr{F} such that each subquotient is of the

form \mathscr{W}_{λ} (resp. For $I_{u}^{1}(\mathscr{W}_{\lambda})$) for some $\lambda \in X_{*}(T)$. In this case, there exists a unique filtration $(\mathscr{F}_{\leq \lambda} : \lambda \in X_{*}(T))$ on \mathscr{F} such that $\mathscr{F}_{\leq \lambda} = \{0\}$ for some λ , $\mathscr{F}_{\leq \mu} = \mathscr{F}$ for some μ , and $\mathscr{F}_{\leq \lambda}/\mathscr{F}_{<\lambda}$ is a direct sum of copies of \mathscr{W}_{λ} for each $\lambda \in X_{*}(T)$. (Here, $\mathscr{F}_{<\lambda}$ means $\mathscr{F}_{\leq \lambda'}$ where λ' is the predecessor of λ .) Moreover this filtration is functorial: if \mathscr{F}, \mathscr{G} admit Wakimoto filtrations and $f : \mathscr{F} \to \mathscr{G}$ is any morphism, then $f(\mathscr{F}_{\leq \lambda}) \subset \mathscr{G}_{\leq \lambda}$ for any $\lambda \in X_{*}(T)$; this allows to define the functor $\operatorname{gr}_{\lambda}$ sending an object \mathscr{F} which admits a Wakimoto filtration to

$$\operatorname{gr}_{\lambda}(\mathscr{F}) := \mathscr{F}_{<\lambda}/\mathscr{F}_{<\lambda}.$$

This notion is relevant in the present context thanks to a result of Arkhipov and the first author (see [AB, Theorem 4]; see also [AR4, §4.4] for the extension to positive-characteristic coefficients) which claims that $Z(\mathscr{A})$ admits a Wakimoto filtration for any \mathscr{A} in P_{L^+G,L^+G} , and that moreover the multiplicity of \mathscr{W}_{λ} in $\operatorname{gr}_{\lambda}(Z(\mathscr{A}))$ is the dimension of the λ -weight space of $\mathcal{S}(\mathscr{A})$.

4.4. Central sheaves – construction. For later reference, we now briefly recall how the functor Z and the relevant isomorphisms are constructed. This functor is defined using nearby cycles associated with an ind-scheme

 $\operatorname{Gr}_G^{\operatorname{Cen}} \to \mathbb{A}^1$

called the central affine Grassmannian, whose fiber over 0 identifies canonically with Fl_G , and whose restriction to $\mathbb{A}^1 \smallsetminus \{0\}$ identifies (again, canonically) with $\operatorname{Gr}_G \times (\mathbb{A}^1 \smallsetminus \{0\})$. We have a smooth affine group scheme \mathcal{G} over \mathbb{A}^1 whose restriction to $\mathbb{A}^1 \smallsetminus \{0\}$ identifies with $G \times (\mathbb{A}^1 \smallsetminus \{0\})$, and whose group of $\mathbb{F}[[z]]$ -points is I. In [AR4, §2.2.3] the construction of this group scheme is explained (following Zhu) using fpqc descent. Following [MRR], this group scheme also admits another equivalent description, as the Néron blowup of $G \times \mathbb{A}^1$ in B along the divisor $\{0\} \subset \mathbb{A}^1$; see in particular [MRR, Example 3.3]. Then $\operatorname{Gr}_G^{\operatorname{Cen}}$ is defined as the \mathbb{F} -scheme which represents the functor sending an \mathbb{F} -algebra R to the set of isomorphism classes of triples (y, \mathcal{E}, β) where $y \in \mathbb{A}^1(R)$, \mathcal{E} is a principal \mathcal{G} -bundle (over $\mathbb{A}^1_R)$, and β is a trivialization of \mathcal{E} over $\mathbb{A}^1_R \smallsetminus \Gamma_y$ (where $\Gamma_y \subset \mathbb{A}^1_R$ is the graph of y). Using the Beauville–Laszlo descent theorem (see [AR4, Remark 2.2.12] for details) one sees that $\operatorname{Gr}_G^{\operatorname{Cen}}(R)$ also classifies isomorphism classes of triples $(y, \mathcal{E}', \beta')$ where y is a above, \mathcal{E}' is a principal \mathcal{G} -bundle over the completion $\widehat{\Gamma}_y \circ \mathbb{A}^1_R$ along Γ_y , and β' is a trivialization on $\widehat{\Gamma}_y \smallsetminus \Gamma_y$. Using this description, the identification

$$\{0\} \times_{\mathbb{A}^1} \operatorname{Gr}_G^{\operatorname{Cen}} = \operatorname{Fl}_G$$

simply follows from the fact that Fl_G represents the functor sending R to isomorphism classes of pairs consisting of a principal $\mathcal{G}_{|\operatorname{Spec}(\mathbb{F}[[z]])}$ -bundle over $\operatorname{Spec}(R[[z]])$ together with a trivialization over $\operatorname{Spec}(R((z)))$; see [AR4, Proposition 2.2.6] for details. (Here, $\mathcal{G}_{|\operatorname{Spec}(\mathbb{F}[[z]])}$ is the Iwahori group scheme attached to B.) The identification

$$(\mathbb{A}^1 \smallsetminus \{0\}) \times_{\mathbb{A}^1} \operatorname{Gr}_G^{\operatorname{Cen}} = \operatorname{Gr}_G \times (\mathbb{A}^1 \smallsetminus \{0\})$$

is obtained using the similar moduli description of Gr_G (in terms of *G*-bundles, see [AR4, Proposition 2.2.2]) and the additive structure on \mathbb{A}^1 , which allows to identify $\widehat{\Gamma}_u$ with $\operatorname{Spec}(R[\![z]\!])$.

The study of this functor also involves another scheme over \mathbb{A}^1 , denoted $\operatorname{Gr}_G^{\mathrm{BD}}$ and called the *Beilinson–Drinfeld affine Grassmannian*. This ind-scheme represents the functor sending an \mathbb{F} -algebra R to isomorphism classes of triples (y, \mathcal{E}, β) where y and \mathcal{E} are as above, but now β is a trivialization on $\mathbb{A}^1_R \setminus (\Gamma_0 \cup \Gamma_y)$. (Here, $\Gamma_0 = \{0\} \times \operatorname{Spec}(R) \subset \mathbb{A}^1_R$ is the graph of the constant point with value 0.) We still have an identification

$$\{0\} \times_{\mathbb{A}^1} \operatorname{Gr}_G^{\operatorname{BD}} = \operatorname{Fl}_G,$$

but now we have

$$(\mathbb{A}^1 \smallsetminus \{0\}) \times_{\mathbb{A}^1} \operatorname{Gr}_G^{\operatorname{Cen}} = \operatorname{Gr}_G \times \operatorname{Fl}_G \times (\mathbb{A}^1 \smallsetminus \{0\}),$$

see [AR4, Lemma 2.3.16]. As explained in [AR4, §3.2.1], nearby cycles along $\operatorname{Gr}_G^{\operatorname{BD}} \to \mathbb{A}^1$ define a bifunctor

$$\mathsf{Y}:\mathsf{D}_{L^+G,L^+G}\times\mathsf{D}_{\mathrm{I},\mathrm{I}}\to\mathsf{D}_{\mathrm{I},\mathrm{I}}.$$

By [AR4, Theorem 3.2.3], for \mathscr{A} in D_{L^+G,L^+G} and \mathscr{F} in $D_{\mathrm{I},\mathrm{I}}$ we have canonical isomorphisms

(4.6)
$$\mathsf{Z}(\mathscr{A}) \star_{\mathrm{I}} \mathscr{F} \cong \mathsf{Y}(\mathscr{A}, \mathscr{F}) \cong \mathscr{F} \star_{\mathrm{I}} \mathsf{Z}(\mathscr{A});$$

in fact the composition of these isomorphisms is precisely the definition of $\sigma_{\mathscr{A},\mathscr{F}}$.

We will now explain how the functor Z, and its various structures, can be entiredly described in terms of the bifunctor Y and some related structures. First, applying (4.6) in case $\mathscr{F} = \delta$, we see that we have $Z(\mathscr{A}) = Y(\mathscr{A}, \delta)$. This can also be seen more directly (in particular, without using (4.6)) from the compatibility of nearby cycles with proper pushforward, after we remark that there exists a closed embedding $\operatorname{Gr}_{G}^{\operatorname{Cen}} \hookrightarrow \operatorname{Gr}_{G}^{\operatorname{BD}}$: in terms of functors this embedding is obtained by sending a triple (y, \mathcal{E}, β) to the triple (y, \mathcal{E}, β') where β' is the restriction of β to $\mathbb{A}_{R}^{1} \smallsetminus (\Gamma_{0} \cup \Gamma_{y})$. The restriction of this embedding to $\mathbb{A}^{1} \smallsetminus \{0\}$ identifies with the natural embedding

$$\operatorname{Gr}_G \times (\mathbb{A}^1 \setminus \{0\}) = \operatorname{Gr}_G \times \operatorname{Fl}_{G,e} \times (\mathbb{A}^1 \setminus \{0\}) \hookrightarrow \operatorname{Gr}_G \times \operatorname{Fl}_G \times (\mathbb{A}^1 \setminus \{0\}).$$

Now we consider the isomorphism $\phi_{\mathscr{A},\mathscr{B}}$. The same arguments as for the construction of this isomorphism (see [AR4, §3.4.1]) show that for \mathscr{A}, \mathscr{B} in D_{L^+G,L^+G} and \mathscr{F}, \mathscr{G} in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$ we have a canonical isomorphism

(4.7)
$$\mathsf{Y}(\mathscr{A},\mathscr{F})\star_{\mathsf{I}}\mathsf{Y}(\mathscr{B},\mathscr{G})\cong\mathsf{Y}(\mathscr{A}\star_{L^{+}G}\mathscr{B},\mathscr{F}\star_{\mathsf{I}}\mathscr{G}).$$

Using this for $\mathscr{F} = \mathscr{G} = \delta$ and using the identification above we recover the isomorphism $\phi_{\mathscr{A},\mathscr{B}}$.

Finally, we note that the isomorphisms in (4.6) can be reconstructed from (4.7), using the fact that the functor $\mathsf{Y}(\delta_{\operatorname{Gr}}, -)$ is the identity. To justify the latter claim one remarks that there exists a natural closed embedding $\operatorname{Fl}_G \times \mathbb{A}^1 \hookrightarrow \operatorname{Gr}_G^{\operatorname{BD}}$, obtained using restriction of trivializations as above, and the fact that $\operatorname{Fl}_G \times \mathbb{A}^1$ represents the functor sending an \mathbb{F} -algebra R to isomorphism classes of triples (y, \mathcal{E}, β) where $y \in \mathbb{A}^1(R)$, \mathcal{E} is a \mathcal{G} -bundle on \mathbb{A}^1_R , and β is a trivialization on $\mathbb{A}^1_R \setminus \Gamma_0$. The restriction of this embedding to $\mathbb{A}^1 \setminus \{0\}$ identifies with the natural embedding

$$\mathrm{Fl}_G \times (\mathbb{A}^1 \setminus \{0\}) = \mathrm{Gr}_G^0 \times \mathrm{Fl}_G \times (\mathbb{A}^1 \setminus \{0\}) \hookrightarrow \mathrm{Gr}_G \times \mathrm{Fl}_G \times (\mathbb{A}^1 \setminus \{0\}).$$

Since nearby cycles for a constant family identify with the identity functor, one deduces the claim.

Then, applying the isomorphism (4.7) with $\mathscr{F} = \delta$ and $\mathscr{B} = \delta_{Gr}$ one obtains the first isomorphism in (4.6), and applying this isomorphism for $\mathscr{A} = \delta_{Gr}$ and $\mathscr{G} = \delta$ one obtains the second one.

Remark 4.2. In order to define the isomorphism $\sigma_{\mathscr{A},\mathscr{F}}$ we need to consider objects in D_{L^+G,L^+G} and $\mathsf{D}_{\mathrm{I},\mathrm{I}}$. But the definition of Y in terms of nearby cycles makes sense without any equivariant structure. In particular, if we denote by $D_c^{\mathrm{b}}(\mathrm{Fl}_G,\mathbb{k})$ the constructible derived category of \mathbb{k} -sheaves on Fl_G , we have a natural bifunctor

$$\mathsf{D}_{L^+G,L^+G} \times \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \to D_c^{\mathrm{b}}(\mathrm{Fl}_G,\Bbbk),$$

which will again be denoted Y, and which satisfies

$$\mathsf{Y}(\mathscr{A},\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F}))\cong\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathsf{Y}(\mathscr{A},\mathscr{F}))$$

for \mathscr{A} in D_{L^+G,L^+G} and \mathscr{F} in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$. In particular, this bifunctor therefore factors through a bifunctor

$$\mathsf{D}_{L^+G,L^+G} \times \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}},$$

which will again be denoted Y.

Below we will need the following standard property. Recall that we have a "loop rotation" action of \mathbb{G}_{m} on LG. (We normalize this action in such a way that for $t \in \mathbb{k}^{\times}$ we have $t \cdot z^{\lambda} = \lambda(t)^{-1} \cdot z^{\lambda}$.) This action induces actions on $\widetilde{\mathrm{Fl}}_{G}$, Fl_{G} and Gr_{G} such that the morphisms in (4.1) are equivariant.

Lemma 4.3. There exists a canonical action of \mathbb{G}_m on $\mathrm{Gr}_G^{\mathrm{Cen}}$ such that the morphism $\mathrm{Gr}_G^{\mathrm{Cen}} \to \mathbb{A}^1$ is \mathbb{G}_m -equivariant with respect to the standard action on \mathbb{A}^1 (by dilation), and whose restriction to $\mathbb{A}^1 \setminus \{0\}$, resp. $\{0\}$, identifies (via the isomorphisms considered above) with the product of the standard action on $\mathbb{A}^1 \setminus \{0\}$ and the loop rotation action on Gr_G , resp. with the loop rotation action on Fl_G .

Sketch of proof. First we note that there exists an action of \mathbb{G}_{m} on \mathcal{G} , compatible with the group structure in the obvious way, such that the projection $\mathcal{G} \to \mathbb{A}^1$ is \mathbb{G}_{m} -equivariant (with respect to the standard action on \mathbb{A}^1), which restricts over $\mathbb{A}^1 \setminus \{0\}$ to the action on $G \times (\mathbb{A}^1 \setminus \{0\})$ on the second factor, and such that the induced action on $\mathbb{F}[[z]]$ -points is by loop rotation. This action can e.g. by constructed using the formalism of [MRR] as follows. By compatibility of Néron blowups with base change (see [MRR, Theorem 3.2(6)]), the fiber product

$$\mathcal{G} \times_{\mathbb{A}^1} (\mathbb{G}_{\mathrm{m}} \times \mathbb{A}^1),$$

where the morphism $\mathbb{G}_{\mathrm{m}} \times \mathbb{A}^1 \to \mathbb{A}^1$ is the action morphism, is the Néron blowup of $(G \times \mathbb{A}^1) \times_{\mathbb{A}^1} (\mathbb{G}_{\mathrm{m}} \times \mathbb{A}^1)$ in $B \times (\mathbb{G}_{\mathrm{m}} \times \{0\})$ along $\mathbb{G}_{\mathrm{m}} \times \{0\}$. Now using the \mathbb{G}_{m} -action on \mathbb{A}^1 we obtain an identification of $(G \times \mathbb{A}^1) \times_{\mathbb{A}^1} (\mathbb{G}_{\mathrm{m}} \times \mathbb{A}^1)$ with the similar fiber product where the morphism $\mathbb{G}_{\mathrm{m}} \times \mathbb{A}^1 \to \mathbb{A}^1$ is the projection. Again by compatibility of Néron blowups with base change, we deduce an isomorphism

$$\mathcal{G} \times_{\mathbb{A}^1} (\mathbb{G}_m \times \mathbb{A}^1) \xrightarrow{\sim} \mathbb{G}_m \times \mathcal{G}$$

as schemes over $\mathbb{G}_m \times \mathbb{A}^1$. Composing the inverse isomorphism with the natural projection on \mathcal{G} defines the desired action.

Once this action is constructed, the \mathbb{G}_{m} -action on $\operatorname{Gr}_{G}^{\operatorname{Cen}}$ is obtained using pullback of torsors; details are left to the reader.

For simplicity of notation, below we will set

$$\mathscr{Z} := \mathsf{Z} \circ \mathcal{S}^{-1} : \mathsf{Rep}(G^{\vee}_{\Bbbk}) \to \mathsf{D}_{\mathrm{I},\mathrm{I}},$$

and write

$$\mathsf{m}_V := \mathsf{m}_{\mathcal{S}^{-1}(V)} \in \operatorname{End}(\mathscr{Z}(V))$$

for V in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$.

4.5. Extending the functor \mathscr{Z} to coherent sheaves on G_{\Bbbk}^{\vee} . Below we will use the following general construction. Let H be an affine \Bbbk -group scheme of finite type. We consider the category $\mathsf{QCoh}^H(H)$ of H-equivariant quasi-coherent sheaves on H, where H acts on itself via the adjoint action. The latter category identifies with the category of H-equivariant $\mathscr{O}(H)$ -modules. The identity functor of $\mathsf{QCoh}^H(H)$ possesses a canonical automorphism $\mathsf{m}_{(-)}^{\mathrm{taut}}$, which can be described as follows. Any H-equivariant $\mathscr{O}(H)$ -module M admits a canonical automorphism, defined as the composition

$$M \to M \otimes \mathscr{O}(H) \to M$$

where the first morphism is the coaction (with respect to the *H*-module structure on *M*) and the second one is the action morphism. It is easily checked that this morphism is a morphism of *H*-equivariant $\mathscr{O}(H)$ -modules, and defines an automorphism of the object \mathscr{F} corresponding to *M* in $\mathsf{QCoh}^H(H)$, which by definition is $\mathsf{m}^{\mathrm{taut}}_{\mathscr{X}}$.

The category $\mathsf{QCoh}^H(H)$ admits a monoidal structure, given by tensor product of \mathscr{O}_H -modules. It is easily checked that for \mathscr{F}, \mathscr{G} in $\mathsf{QCoh}^H(H)$ we have

$$\mathsf{m}^{\mathrm{taut}}_{\mathscr{F}\otimes_{\mathscr{O}_H}\mathscr{G}} = \mathsf{m}^{\mathrm{taut}}_{\mathscr{F}}\otimes_{\mathscr{O}_H}\mathsf{m}^{\mathrm{taut}}_{\mathscr{G}}$$

Remark 4.4. As in §2.8 one can consider the universal stabiliser $\mathfrak{S}_{H,H}$ associated with the adjoint *H*-action on itself, and we have a canonical (monoidal) functor

(4.8)
$$\operatorname{\mathsf{QCoh}}^H(H) \to \operatorname{\mathsf{Rep}}^\infty(\mathfrak{S}_{H,H}).$$

As in Remark 2.16, any object in $\operatorname{Rep}^{\infty}(\mathfrak{S}_{H,H})$ admits a tautological automorphism. It is easily checked that the functor (4.8) sends $\mathfrak{m}_{\mathscr{F}}^{\operatorname{taut}}$ to the tautological automorphism of its image.

Now, consider the full subcategory $\mathsf{Coh}^{H}(H)$ of *H*-equivariant coherent sheaves on *H*, and the category $\mathsf{Rep}(H)$ of finite-dimensional representations of *H*. We have a canonical monoidal functor

$$i : \operatorname{Rep}(H) \to \operatorname{Coh}^H(H)$$

defined by $V \mapsto V \otimes \mathcal{O}_H$, where the *H*-equivariant structure on $V \otimes \mathcal{O}_H$ is diagonal. For *V* in $\operatorname{Rep}(H)$ we will write $\mathsf{m}_V^{\operatorname{taut}}$ for $\mathsf{m}_{\iota(V)}^{\operatorname{taut}}$, so that $\mathsf{m}_V^{\operatorname{taut}}$ is an automorphism of $V \otimes \mathcal{O}_H$ which satisfies

$$\mathsf{m}_{V_1\otimes V_2}^{\mathrm{taut}} = \mathsf{m}_{V_1}^{\mathrm{taut}}\otimes \mathsf{m}_{V_2}^{\mathrm{taut}}$$

for $V_1, V_2 \in \mathsf{Rep}(H)$. We will denote by $\mathsf{Coh}_{\mathrm{fr}}^H(H)$ the full subcategory of $\mathsf{Coh}^H(H)$ whose objects are the coherent sheaves $V \otimes \mathscr{O}_H$ for V in $\mathsf{Rep}(H)$, so that i factors through a functor $\mathsf{Rep}(H) \to \mathsf{Coh}_{\mathrm{fr}}^H(H)$ (still denoted i) which is the obvious bijection on objects. (Note that $\mathsf{Coh}_{\mathrm{fr}}^H(H)$ is defined as a full, but not strictly full, subcategory of $\mathsf{Coh}^H(H)$.)

The following lemma is a variant of [AB, Proposition 4(a)], and follows from similar arguments.

Lemma 4.5. Let A be an additive k-linear monoidal category, and let

$$F: \mathsf{Rep}(H) \to \mathsf{A}$$

be a k-linear monoidal functor. Let $N_{(-)}$ be an automorphism of F such that for any V_1, V_2 in Rep(H) we have

$$\mathsf{N}_{V_1\otimes V_2}=\mathsf{N}_{V_1}\otimes \mathsf{N}_{V_2},$$

and the diagram

commutes, where the vertical arrows are the images under F of the commutativity isomorphisms in Rep(H). Then there exists a unique k-linear monoidal functor

$$F^{\mathsf{Coh}}: \mathsf{Coh}_{\mathrm{fr}}^{H}(H) \to \mathsf{A}$$

such that $F^{Coh} \circ i = F$, and such that

$$F^{\mathsf{Coh}}(\mathsf{m}_V^{\mathrm{taut}}) = \mathsf{N}_V$$

for any V in $\operatorname{Rep}(H)$.

In more concrete terms, this lemma says that the datum of $\mathsf{N}_{(-)}$ allows to "extend" in a canonical way the morphisms

$$\operatorname{Hom}_{\operatorname{\mathsf{Rep}}(H)}(V_1, V_2) \to \operatorname{Hom}_{\operatorname{\mathsf{A}}}(F(V_1), F(V_2))$$

to morphisms

$$\operatorname{Hom}_{\operatorname{Coh}^{H}(H)}(V_{1} \otimes \mathscr{O}_{H}, V_{2} \otimes \mathscr{O}_{H}) \to \operatorname{Hom}_{A}(F(V_{1}), F(V_{2}))$$

for any $V_1, V_2 \in \mathsf{Rep}(H)$. (Here, the left-hand side identifies with $(V_1^* \otimes V_2)^H$ in the first case, and with $(V_1^* \otimes V_2 \otimes \mathscr{O}(H))^H$ in the second case.)

Remark 4.6. If $K \subset H$ is a closed subgroup scheme and if $h \in H$ commutes with K, then we can apply Lemma 4.5 to the restriction functor $\operatorname{For}_{K}^{H} : \operatorname{Rep}(H) \to \operatorname{Rep}(K)$, and its automorphism induced by h. In this case, the functor

$$\operatorname{Coh}_{\mathrm{fr}}^{H}(H) \to \operatorname{Rep}(K)$$

is induced by restriction of coherent sheaves to $h \in H$.

Applying Lemma 4.5 to the monoidal functor $\mathscr{Z} : \mathsf{Rep}(G_{\Bbbk}^{\vee}) \to \mathsf{D}_{\mathrm{I},\mathrm{I}}$ and its automorphism $\mathsf{m}_{(-)}$, we obtain a canonical monoidal functor

$$\mathscr{Z}^{\mathsf{Coh}}: \mathsf{Coh}_{\mathrm{fr}}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk}) \to \mathsf{D}_{\mathrm{I},\mathrm{I}}.$$

In particular, this provides for any V in $\operatorname{\mathsf{Rep}}(G_{\Bbbk}^{\vee})$ a canonical algebra morphism

(4.9)
$$\operatorname{End}_{\operatorname{Coh}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee})}(V \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}) \to \operatorname{End}_{\mathsf{P}_{\mathrm{I},\mathrm{I}}}(\mathscr{Z}(V)).$$

The G_{\Bbbk}^{\vee} -module $\mathscr{O}(G_{\Bbbk}^{\vee})$ (endowed with the action induced by left multiplication of G_{\Bbbk}^{\vee} on itself) defines an ind-object in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$ (namely, the functor $V \mapsto$ $\operatorname{Hom}_{G_{\Bbbk}^{\vee}}(V, \mathscr{O}(G_{\Bbbk}^{\vee})))$; therefore, applying \mathscr{Z} we deduce an ind-object $\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ in $\mathsf{P}_{\mathrm{I},\mathrm{I}}$. As a special case of (4.9) we have a canonical algebra morphism

$$\operatorname{End}_{\operatorname{Ind-Coh}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee})}(\mathscr{O}(G_{\Bbbk}^{\vee})\otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}) \to \operatorname{End}_{\operatorname{Ind-P}_{\mathrm{I},\mathrm{I}}}(\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee}))).$$

Note that given a ring object X in a k-linear monoidal category (A, \odot) , with unit object 1, the vector space

 $\operatorname{Hom}_{\mathsf{A}}(\mathbf{1}, X)$

admits a natural structure of k-algebra, where the product of two morphisms $f,g:\mathbf{1}\to X$ is the composition

$$\mathbf{1} = \mathbf{1} \odot \mathbf{1} \xrightarrow{f \odot g} X \odot X \to X,$$

where the right morphism is the multiplication map for X. Moreover, there exists an algebra morphism

$$\operatorname{Hom}_{\mathsf{A}}(\mathbf{1}, X) \to \operatorname{End}_{\mathsf{A}}(X)^{\operatorname{op}}$$

sending a morphism $f: \mathbf{1} \to X$ to the morphism

$$X = X \odot \mathbf{1} \xrightarrow{\mathrm{id} \odot a} X \odot X \to X,$$

where again the rightmost morphism is induced by multiplication in X; this morphism in fact takes values in endomorphisms of X seen as a left module over itself.

Let us apply this construction to the ring-object $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}$ in Ind-Coh^{G_{\Bbbk}^{\vee}} (G_{\Bbbk}^{\vee}). Then we have

$$\operatorname{Hom}(\mathscr{O}_{G_{\Bbbk}^{\vee}}, \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}) \cong \mathscr{O}(G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee})^{G_{\Bbbk}^{\vee}}$$

where G_{\Bbbk}^{\vee} acts on $G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee}$ via $g \cdot (h_1, h_2) = (gh_1, gh_2g^{-1})$. Now the morphism $G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee} \to G_{\Bbbk}^{\vee}$ defined by $(g, h) \mapsto g^{-1}hg$ defines an algebra isomorphism

$$\mathscr{O}(G^{\vee}_{\Bbbk}) \xrightarrow{\sim} \mathscr{O}(G^{\vee}_{\Bbbk} \times G^{\vee}_{\Bbbk})^{G^{\vee}_{\Bbbk}}$$

which therefore provides a canonical algebra morphism

$$\mathscr{O}(G^{\vee}_{\Bbbk}) \to \operatorname{End}(\mathscr{O}(G^{\vee}_{\Bbbk}) \otimes \mathscr{O}_{G^{\vee}_{\Bbbk}}),$$

hence finally an algebra morphism

$$(4.10) \qquad \qquad \mathscr{O}(G_{\Bbbk}^{\vee}) \to \operatorname{End}_{\operatorname{Ind}-\mathsf{P}_{\mathrm{I},\mathrm{I}}}(\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee})))$$

In this way, $\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ becomes an $\mathscr{O}(G_{\Bbbk}^{\vee})$ -module in the category Ind- $\mathsf{P}_{\mathrm{I},\mathrm{I}}$, in the sense recalled in §B.1.

4.6. The regular quotient: definition. The next considerations will make intensive use of the notions of Serre quotient of an abelian category and of Verdier quotient of a triangulated category; for a brief reminder on these notions, and references, see §A.1.

The main player of [BRR] is the abelian category

 $\mathsf{P}_{\mathrm{I}\mathrm{I}}^{0}$

defined as the Serre quotient of the abelian category $\mathsf{P}_{I,I}$ by the Serre subcategory $\mathsf{P}_{I,I}^+$ generated by the simple objects \mathscr{IC}_w for $w \in W$ such that $\ell(w) > 0$. If we denote by $\mathsf{D}_{I,I}^+$ the full triangulated subcategory of $\mathsf{D}_{I,I}$ generated by $\mathsf{P}_{I,I}^+$, and by $\mathsf{D}_{I,I}^0$ the Verdier quotient of $\mathsf{D}_{I,I}$ by $\mathsf{D}_{I,I}^+$, then by Lemma A.2 (applied using the perverse t-structure on $\mathsf{D}_{I,I}$) there exists a unique t-structure on $\mathsf{D}_{I,I}^0$ such that the quotient functor $\Pi_{I,I}^0:\mathsf{D}_{I,I}\to\mathsf{D}_{I,I}^0$ is t-exact, and moreover this t-structure is bounded, and its heart identifies canonically with $\mathsf{P}_{I,I}^0$. This t-structure will be called the perverse t-structure, and the associated cohomology functors will be denoted ${}^{p}\mathscr{H}^n(-)$. Since $\mathsf{P}_{I,I}$ is a finite-length category, so is $\mathsf{P}_{I,I}^0$, and its simple objects are the objects

$$\delta^0_\omega := \Pi^0_{\mathrm{I},\mathrm{I}}(\mathscr{I}\mathscr{C}_\omega)$$

for $\omega \in \Omega$. (In case $\omega = e$ is the unit, we will write δ^0 for δ_e^0 .)

Consider the bifunctor $\mathsf{D}_{\mathrm{I},\mathrm{I}} \times \mathsf{D}_{\mathrm{I},\mathrm{I}} \to \mathsf{D}^{0}_{\mathrm{I},\mathrm{I}}$ sending a pair $(\mathscr{F},\mathscr{G})$ to $\Pi^{0}_{\mathrm{I},\mathrm{I}}(\mathscr{F} \star_{\mathrm{I}} \mathscr{G})$. By [BRR, Lemma 5.1(1)] we have $\Pi^{0}_{\mathrm{I},\mathrm{I}}(\mathscr{F} \star_{\mathrm{I}} \mathscr{G}) = 0$ if either \mathscr{F} or \mathscr{G} belongs to $D_{I,I}^+$. By the general properties of Verdier quotients (see §A.1), it follows that there exists a unique bifunctor

$$\star^0_{\mathrm{I}}:\mathsf{D}^0_{\mathrm{I},\mathrm{I}}\times\mathsf{D}^0_{\mathrm{I},\mathrm{I}}\to\mathsf{D}^0_{\mathrm{I},\mathrm{I}}$$

such that

$$\Pi^0_{\mathrm{I},\mathrm{I}}(\mathscr{F})\star^0_{\mathrm{I}}\Pi^0_{\mathrm{I},\mathrm{I}}(\mathscr{G})=\Pi^0_{\mathrm{I},\mathrm{I}}(\mathscr{F}\star_{\mathrm{I}}\mathscr{G})$$

for any \mathscr{F}, \mathscr{G} in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$. It is easily seen that this bifunctor equips $\mathsf{D}_{\mathrm{I},\mathrm{I}}^0$ with the structure of a monoidal category, with monoidal unit δ^0 .

From the fact that $\mathscr{IC}_{\omega} \star_{I} \mathscr{IC}_{\omega'} \cong \mathscr{IC}_{\omega\omega'}$ for $\omega, \omega' \in \Omega$ one sees that \star_{I}^{0} is t-exact on both sides with respect to the perverse t-structure; it therefore restricts to a bifunctor $\mathsf{P}_{I,I}^{0} \times \mathsf{P}_{I,I}^{0} \to \mathsf{P}_{I,I}^{0}$ which equips $\mathsf{P}_{I,I}^{0}$ with a monoidal structure. It is clear that the functor

$$\mathscr{Z}^0 := \Pi^0_{\mathrm{I},\mathrm{I}} \circ \mathscr{Z} : \mathsf{Rep}(G^{\vee}_{\Bbbk}) \to \mathsf{P}^0_{\mathrm{I},\mathrm{I}}$$

has a canonical monoidal structure.

4.7. Another convolution bifunctor. For later use, we now explain a variant of the constructions of $\S4.6$ where I-equivariance is replaced by I_u -equivariance.

Recall the category $D_{I_u,I}$, its perverse t-structure, and the heart $P_{I_u,I}$ of this t-structure (see §4.2). Let us denote by $P_{I_u,I}^+$, resp. $D_{I_u,I}^+$, the Serre subcategory of $P_{I_u,I}$, resp. the full triangulated subcategory of $D_{I_u,I}$, generated by the simple perverse sheaves $\text{For}_{I_u}^I(\mathscr{IC}_w)$ with $w \in W$ such that $\ell(w) > 0$. We will denote by $P_{I_u,I}^0$ the Serre quotient of $P_{I_u,I}$ by $P_{I_u,I}^+$, and by $D_{I_u,I}^0$ the Verdier quotient of $D_{I_u,I}$ by $D_{I_u,I}^+$. Then by Lemma A.2 there exists a unique t-structure on $D_{I_u,I}^0$ such that the quotient functor

$$\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}:\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\to\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

is t-exact; moreover, this t-structure is bounded, and its heart identifies with $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$. This t-structure will be called the perverse t-structure, and the associated cohomology functors will be denoted ${}^{\mathrm{p}}\mathcal{H}^n(-)$.

By the universal property of the Verdier quotient, the composition

$$\mathsf{P}_{I,I} \xrightarrow{\mathsf{For}_{I_{\mathrm{u}}}^{\mathrm{I}}} \mathsf{P}_{I_{\mathrm{u}},I} \xrightarrow{\Pi^{0}_{I_{\mathrm{u}},I}} \mathsf{P}^{0}_{I_{\mathrm{u}},I}$$

factors through a triangulated functor

$$\mathsf{For}_{I_u}^{I,0}:\mathsf{D}_{I,I}^0\to\mathsf{D}_{I_u,I}^0.$$

This functor is easily seen to be t-exact with respect to the perverse t-structures, and its restriction to the hearts identifies with the functor provided by the universal property of the Serre quotient. Recall that $\mathsf{For}_{I_u}^{I}$ is fully faithful on perverse sheaves; from the standard description of morphisms in a Serre quotient category (see [Gab]), and since $\mathsf{P}_{I,I}$ is closed under subquotients in $\mathsf{P}_{I_u,I}$, one sees that the restriction of $\mathsf{For}_{I_u}^{I,0}$ to the heart of the perverse t-structure is fully faithful.

Consider the bifunctor

$$\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \times \mathsf{D}_{\mathrm{I},\mathrm{I}} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}^{0}$$

sending a pair $(\mathscr{F}, \mathscr{G})$ to $\Pi^0_{\mathrm{I}_u,\mathrm{I}}(\mathscr{F} \star_{\mathrm{I}} \mathscr{G})$, where \star_{I} is as in (4.4). It follows again from [BRR, Lemma 5.1(1)] and the general properties of the Verdier quotient (see §A.1) that there exists a unique triangulated bifunctor

$$\star^0_{\mathrm{I}}:\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\times\mathsf{D}^0_{\mathrm{I},\mathrm{I}}\to\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

such that for \mathscr{F} in $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ and \mathscr{G} in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$ we have

$$\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\mathscr{F})\star^0_{\mathrm{I}}\Pi^0_{\mathrm{I},\mathrm{I}}(\mathscr{G})=\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\mathscr{F}\star^0_{\mathrm{I}}\mathscr{G})$$

This bifunctor defines a right action of the monoidal category $(\mathsf{D}^0_{\mathrm{I},\mathrm{I}},\star^0_{\mathrm{I}})$ on $\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$; for \mathscr{F},\mathscr{G} in $\mathsf{D}^0_{\mathrm{L}}$ we also have a canonical isomorphism

(4.11)
$$\operatorname{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{F}\star_{\mathrm{I}}^{0}\mathscr{G})\cong\operatorname{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{F})\star_{\mathrm{I}}^{0}\mathscr{G}.$$

As in the I-equivariant setting in §4.6 one sees that the bifunctor \star^0_I is t-exact on both sides; its restriction to the hearts of the perverse t-structures defines a right action of the monoidal category $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$ on $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$, by exact autoequivalences.

4.8. The regular quotient: coherent description. In this subsection we make the following assumptions:

- (1) the quotient of $X^*(T)$ by the root lattice of (G, T) is free;
- (2) the quotient of $X_*(T)$ by the coroot lattice of (G, T) has no ℓ -torsion;
- (3) for any indecomposable factor in the root system of (G, T), ℓ is strictly bigger than the corresponding value in Figure 1.1.

Here the first assumption is equivalent to requiring that G_{\Bbbk}^{\vee} has simply connected derived subgroup, and the second one that its scheme-theoretic center is smooth. The third assumption can most probably be weakened; it implies in particular that ℓ is good for G.

By [BRR, Theorem 5.4], there exists a regular unipotent element $u \in G_{\mathbb{k}}^{\vee}$ and an equivalence of monoidal categories

$$\Phi_{\mathrm{I},\mathrm{I}}:(\mathsf{P}^{0}_{\mathrm{LI}},\star^{0}_{\mathrm{I}})\xrightarrow{\sim}(\mathsf{Rep}(Z_{G^{\vee}_{\mathrm{L}}}(\mathsf{u})),\otimes)$$

such that

$$\mathscr{Z}^0 \circ \Phi_{\mathrm{I},\mathrm{I}} \cong \mathsf{For}_{Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})}^{G_{\Bbbk}^{\vee}}$$

as monoidal functors, where $Z_{G_k^{\vee}}(u)$ is the centralizer of u and

$$\mathsf{For}_{Z_{G^{\vee}_{\Bbbk}}(\mathsf{u})}^{G^{\vee}_{\Bbbk}}:\mathsf{Rep}(G^{\vee}_{\Bbbk})\to\mathsf{Rep}(Z_{G^{\vee}_{\Bbbk}}(\mathsf{u}))$$

is the restriction functor. (Note that by Lemma 2.6 the scheme-theoretic centralizer of \mathbf{u} is smooth, so the structure we consider on $Z_{G_{\mathbf{k}}^{\vee}}(\mathbf{u})$ is unambiguous.) This equivalence furthermore satisfies the property that the automorphism

$$\Phi_{\mathrm{I},\mathrm{I}}(\Pi^0_{\mathrm{I},\mathrm{I}}(\mathsf{m}_V))$$

identifies with the action of \mathbf{u} on V, for any V in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$.

Below we will need a more explicit description on this equivalence than what is provided in [BRR], which we now explain. This description will make use of the notion of tensor product with an R-module in a category, whose definition is recalled in §B.1.

We first recall the structure of the main construction in [BRR]. As explained in §4.5 the G_{\Bbbk}^{\vee} -module $\mathscr{O}(G_{\Bbbk}^{\vee})$ defines an ind-object in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$, which is moreover a ring ind-object. The image $\mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee}))$ therefore defines a ring ind-object in the category $\mathsf{P}_{\mathrm{I},\mathrm{I}}^0$. As explained in [BRR, §3.3] (following [B1]) any left ideal subobject in $\mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee}))$ is automatically a two-sided ideal; in particular if we fix a maximal left ideal subobject $\mathscr{J} \subset \mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee}))$ then the quotient

$$\mathscr{R}^0 := \mathscr{Z}^0(\mathscr{O}(G^{ee}_\Bbbk))/\mathscr{J}$$

has a canonical structure of ring ind-object such that the surjection $\mathscr{Z}^0(\mathscr{O}(G^{\vee}_{\Bbbk})) \to \mathscr{R}^0$ is a ring morphism. One then checks that the assignment

$$\Psi_{\mathrm{I},\mathrm{I}}:\mathscr{F}\mapsto\mathrm{Hom}_{\mathrm{Ind}-\mathsf{P}^0_{\mathrm{I},\mathrm{I}}}(\delta^0,\mathscr{R}^0\star^0_{\mathrm{I}}\mathscr{F})$$

defines a functor from $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$ to the category Vect_{\Bbbk} of finite-dimensional \Bbbk -vector spaces, that this functor admits a canonical monoidal structure (induced in an appropriate way by the ring structure on \mathscr{R}^0), and that its composition with \mathscr{Z}^0 identifies with the forgetful functor $\mathsf{For}^{G_{\Bbbk}^{\vee}} : \mathsf{Rep}(G_{\Bbbk}^{\vee}) \to \mathrm{Vect}_{\Bbbk}$. Using this functor we invoke Tannakian formalism to obtain a closed subgroup scheme $H \subset G_{\Bbbk}^{\vee}$ and an equivalence of monoidal categories

$$(\mathsf{P}^0_{\mathrm{I},\mathrm{I}},\star^0_{\mathrm{I}}) \xrightarrow{\sim} (\mathsf{Rep}(H),\otimes)$$

whose pre-composition with \mathscr{Z}^0 is the restriction functor $\mathsf{For}_H^{G_k^{\vee}}$, and whose postcomposition with the forgetful functor For^H is $\Psi_{\mathrm{I,I}}$. From the automorphism m of the functor \mathscr{Z}^0 we obtain an automorphism of the functor $\mathsf{For}^{G_k^{\vee}}$, which defines an element $\mathsf{u} \in G_k^{\vee}$. Most of the content of [BRR] is then devoted to showing that u is unipotent regular, and that $H = Z_{G_k^{\vee}}(\mathsf{u}).^2$

A posteriori, the ind-object $\mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee}))$ identifies with $\mathscr{O}(G_{\Bbbk}^{\vee})$ seen as a $Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})$ -representation; its maximal left ideals are therefore parametrized by the cosets in $Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})\backslash G_{\Bbbk}^{\vee}$. We claim that, if we still denote by $\Phi_{\mathrm{I},\mathrm{I}}$ the induced equivalence on ind-objects, we have a canonical identification

$$\Phi_{\mathrm{I},\mathrm{I}}(\mathscr{R}^0) \cong \mathscr{O}(Z_{G_{*}^{\vee}}(\mathsf{u})).$$

In fact the ring surjection $\mathscr{Z}^0(\mathscr{O}(G^{\vee}_{\Bbbk})) \to \mathscr{R}^0$ induces a ring map

$$(4.12) \qquad \qquad \mathscr{O}(G_{\Bbbk}^{\vee}) = \Phi_{\mathrm{I},\mathrm{I}}\big(\mathscr{Z}^{0}(\mathscr{O}(G_{\Bbbk}^{\vee}))\big) \to \Phi_{\mathrm{I},\mathrm{I}}(\mathscr{R}^{0}),$$

which as explained above identifies the right-hand side with functions on a certain coset in $Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})\backslash G_{\Bbbk}^{\vee}$; what remains to be justified is that this coset is $Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})$. To check this it suffices to prove that our coset contains the unit element e, i.e. that the augmentation morphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \to \Bbbk$ factors through our morphism (4.12). For that we consider the commutative diagram

where the upper vertical arrows are induced by \mathscr{Z}^0 , the lower vertical arrow by the quotient morphism $\mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee})) \to \mathscr{R}^0$, and the horizontal ones by multiplication in $\mathscr{O}(G_{\Bbbk}^{\vee})$. (The G_{\Bbbk}^{\vee} -action on each copy of $\mathscr{O}(G_{\Bbbk}^{\vee})$ is the left regular action.) Here the diagonal arrow factors through $\Phi_{\mathrm{I},\mathrm{I}}(\mathscr{R}^0)$, and the upper line identifies with the augmentation morphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathbb{K}$, via the morphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathscr{O}(G_{\Bbbk}^{\vee})$

²In fact the equivalence we initially obtain concerns only a full subcategory of $P_{I,I}^0$, which is then shown to coincide with $P_{I,I}^0$. This subtlety is irrelevant for the present discussion.

induced by $f \otimes g \mapsto f(e)g$ and Frobenius reciprocity. Moreover the composition of the right vertical arrows is an isomorphism in view of the isomorphism $\operatorname{Hom}(\delta^0, \mathscr{R}^0) \cong \Phi_{\mathrm{I},\mathrm{I}}(\delta^0) = \Bbbk$. The desired claim follows.

Recall from (4.10) that $\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ is an $\mathscr{O}(G_{\Bbbk}^{\vee})$ -module; hence $\mathscr{Z}^{0}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ has the same structure, and the corresponding morphism

$$\mathscr{O}(G^{\vee}_{\Bbbk}) \to \operatorname{End}_{\operatorname{Ind-Rep}(Z_{G^{\vee}}(\mathsf{u}))}(\mathscr{O}(G^{\vee}_{\Bbbk}))$$

(obtained by applying $\Phi_{I,I}$) is induced by the morphism $G_{\Bbbk}^{\vee} \to G_{\Bbbk}^{\vee}$ given by $g \mapsto g^{-1} \mathfrak{u} g$. It is easily seen from definitions that restriction induces an isomorphism

$$\mathscr{O}(G^{\vee}_{\Bbbk}) \otimes_{\mathscr{O}(G^{\vee}_{\Bbbk})} \mathscr{O}(\{\mathsf{u}\}) \xrightarrow{\sim} \mathscr{O}(Z_{G^{\vee}_{\Bbbk}}(\mathsf{u})),$$

from which we deduce a canonical isomorphism

$$\mathscr{Z}^0(\mathscr{O}(G^{\vee}_{\Bbbk}))\otimes_{\mathscr{O}(G^{\vee}_{\Bbbk})}\mathscr{O}(\{\mathsf{u}\})\xrightarrow{\sim}\mathscr{R}^0$$

Here, since $\mathscr{O}(G_{\Bbbk}^{\vee})$ acts on $\mathscr{Z}^{0}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ by endomorphisms of left modules, the left-hand side is naturally a left $\mathscr{Z}^{0}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ -module, and this identification is compatible with this structure; it follows that the multiplication map on \mathscr{R}^{0} can be recovered from this description.

The comultiplication morphism of $\mathscr{O}(G_{\Bbbk}^{\vee})$ defines a morphism of G_{\Bbbk}^{\vee} -modules

$$(4.13) \qquad \qquad \mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}(G_{\Bbbk}^{\vee})$$

(where the action on the right term in the tensor product is trivial) which we use to obtain a morphism

$$\mathscr{O}(G^{\vee}_{\Bbbk}) \otimes_{\Bbbk} \mathscr{O}_{G^{\vee}_{\Bbbk}} \to \left(\mathscr{O}(G^{\vee}_{\Bbbk}) \otimes_{\Bbbk} \mathscr{O}_{G^{\vee}_{\Bbbk}} \right) \otimes \mathscr{O}(G^{\vee}_{\Bbbk})$$

in Ind-Coh $_{\mathrm{fr}}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee})$. Here the right-hand side has an action of $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}(G_{\Bbbk}^{\vee})$ obtained from the $\mathscr{O}(G_{\Bbbk}^{\vee})$ -action on the first term (as in §4.5) and the obvious $\mathscr{O}(G_{\Bbbk}^{\vee})$ -action on the second term. If we restrict this action to $\mathscr{O}(G_{\Bbbk}^{\vee})$ via the morphism induced by $(g,h) \mapsto h^{-1}gh$, then explicit computation shows that our morphism is $\mathscr{O}(G_{\Bbbk}^{\vee})$ -linear. We now consider the morphism

$$\mathscr{Z}(\mathscr{O}(G^{\vee}_{\Bbbk})) \to \mathscr{Z}(\mathscr{O}(G^{\vee}_{\Bbbk})) \otimes \mathscr{O}(G^{\vee}_{\Bbbk})$$

in Ind- $\mathsf{P}_{\mathrm{I},\mathrm{I}}$ obtained by applying (the extension to ind-objects of) $\mathscr{Z}^{\mathsf{Coh}}$. This morphism is again $\mathscr{O}(G^{\vee}_{\Bbbk})$ -linear; as a consequence the composition

$$\begin{aligned} \mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee})) \to \mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes \mathscr{O}(G_{\Bbbk}^{\vee}) \\ \to \left(\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\{\mathsf{u}\})\right) \otimes \mathscr{O}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})) \end{aligned}$$

factors (uniquely) through a morphism

$$\left(\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee}))\otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})}\mathscr{O}(\{\mathsf{u}\})\right)\to \left(\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee}))\otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})}\mathscr{O}(\{\mathsf{u}\})\right)\otimes\mathscr{O}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})).$$

Applying $\Pi^0_{I,I}$, we deduce a morphism

$$(4.14) \ \left(\mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\{\mathsf{u}\})\right) \to \left(\mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\{\mathsf{u}\})\right) \otimes \mathscr{O}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})).$$

These considerations show that the functor

$$(4.15) \qquad \Phi_{\mathrm{I},\mathrm{I}}: \mathsf{P}^{0}_{\mathrm{I},\mathrm{I}} \xrightarrow{\sim} \mathsf{Rep}(Z_{G^{\vee}_{\Bbbk}}(\mathsf{u}))$$

can be reconstructed a posteriori as the functor

 $\mathscr{F} \mapsto \operatorname{Hom}(\delta^{0}, (\mathscr{Z}^{0}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\{\mathsf{u}\})) \star^{0}_{\mathrm{I}} \mathscr{F}),$

with the monoidal structure induced by the product on $\mathscr{Z}^{0}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\{\mathsf{u}\})$ induced by the product on $\mathscr{Z}^{0}(\mathscr{O}(G_{\Bbbk}^{\vee}))$, and the $Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})$ -action defined by the coaction induced by (4.14). With this description, the regular unipotent element u can in fact be chosen a priori, and arbitrarily, and the induced functor (4.15) will be an equivalence in all cases. (As explained above, this choice is equivalent to the choice of a left ideal ideal subobject in $\mathscr{Z}^{0}(\mathscr{O}(G_{\Bbbk}^{\vee}))$.)

- Remark 4.7. (1) We do not claim that we know how to prove that $\Phi_{I,I}$ is an equivalence using the description as above, but only that we can give this description a posteriori, once we know that it provides an equivalence.
 - (2) These considerations show that the ind-object \mathscr{R}^0 is in fact the image under Π^0_{LI} of a canonical ind-object in $\mathsf{P}_{I,I}$, namely the tensor product

$$(4.16) \qquad \qquad \mathscr{R} := \mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\{\mathsf{u}\}).$$

Moreover, the same arguments as for \mathscr{R}^0 show that \mathscr{R} has a natural structure of ring ind-object. As explained above, the morphism (4.14) defining the $Z_{G_{\nu}^{\vee}}(\mathsf{u})$ -action is also defined at the level of this object.

(3) One can also describe a variant of the equivalence $\Phi_{I,I}$ which does not require any choice; namely, if \mathcal{U}_{reg} denotes the unique open orbit in the unipotent cone \mathcal{U} of G_{\Bbbk}^{\vee} , then as explained in [BRR, §2.4] the map $hZ_{G_{\Bbbk}^{\vee}}(\mathsf{u}) \mapsto$ $h\mathsf{u}h^{-1}$ induces an isomorphism of varieties

$$G^{\vee}_{\Bbbk}/Z_{G^{\vee}_{\Bbbk}}(\mathsf{u}) \xrightarrow{\sim} \mathcal{U}_{\mathrm{reg}},$$

hence an equivalence of categories $\mathsf{Coh}^{G_{\Bbbk}^{\vee}}(\mathcal{U}_{\mathrm{reg}}) \xrightarrow{\sim} \mathsf{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))$. One can check that the composition of $\Phi_{\mathrm{I},\mathrm{I}}$ with the inverse of this equivalence defines an equivalence

$$\mathsf{P}^0_{\mathrm{I},\mathrm{I}} \xrightarrow{\sim} \mathsf{Coh}^{G^\vee_\Bbbk}(\mathcal{U}_{\mathrm{reg}})$$

which is independent of the initial choice of the ideal \mathscr{J} (or, equivalently, of the element u).

5. Construction of the monodromic regular quotient

In this section we provisionally come back to the general setting of $\S4.1$.

5.1. I_u-monodromic sheaves on the extended affine flag variety, convolution, and monodromy. Recall the ind-scheme \widetilde{Fl}_G defined in §4.1. This indscheme admits an action of I_u, and we can consider the associated equivariant derived category $D_{I_u}^{\rm b}(\widetilde{Fl}_G, \Bbbk)$. We will denote by

 $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$

the full triangulated subcategory of $D_{I_u}^b(\widetilde{Fl}_G, \Bbbk)$ generated by the essential image of the pullback functor $\pi^* : D_{I_u}^b(Fl_G, \Bbbk) \to D_{I_u}^b(\widetilde{Fl}_G, \Bbbk)$. (In other words D_{I_u,I_u} is the I_u -equivariant derived category of sheaves on \widetilde{Fl}_G , constructible with respect to the stratification given by inverse images of I-orbits on Fl_G .) The perverse t-structure on $D_{I_u}^b(\widetilde{Fl}_G, \Bbbk)$ restricts to a t-structure $({}^{\mathsf{P}}\mathsf{D}_{\overline{I_u},I_u}^{\geq 0})$ on D_{I_u,I_u} , whose heart will be denoted P_{I_u,I_u} . Since π is smooth with connected fibers the functor

$$\pi^{\dagger} := \pi^*[\dim(T)] \cong \pi^![-\dim(T)] : \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

is t-exact, and the simple objects in the category $\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ are the object $\pi^{\dagger}\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{IC}_{w})$ with $w \in W$.

We define a natural convolution product $-\star_{I_u}$ – on the category $D_{I_u}^b(\widetilde{Fl}_G, \Bbbk)$ as follows. We denote by $\widetilde{Fl}_G \times \widetilde{Fl}_G$ the fppf quotient of $LG \times \widetilde{Fl}_G$ by the action of I_u defined by $g \cdot (h, x) = (hg^{-1}, g \cdot x)$; it is easily seen that this functor is represented by an ind-scheme, which we denote in the same way. The multiplication map in LG defines a (non proper!) morphism $\widetilde{m} : \widetilde{Fl}_G \times \widetilde{Fl}_G \to \widetilde{Fl}_G$. Then, given \mathscr{F}, \mathscr{G} in $D_{I_u}^b(\widetilde{Fl}_G, \Bbbk)$, there exists a unique complex $\mathscr{F} \boxtimes \mathscr{G}$ in $D_{I_u}^b(\widetilde{Fl}_G \times \widetilde{Fl}_G, \Bbbk)$ whose pullback to $LG \times \widetilde{Fl}_G$ is the exterior product of the pullback of \mathscr{F} to LG with \mathscr{G} . We set

$$\mathscr{F} \star_{\mathbf{I}_{u}} \mathscr{G} := \widetilde{m}_{!}(\mathscr{F} \boxtimes \mathscr{G})[\dim(T)].$$

It is not difficult to check that this operation admits a natural associativity constraint. (In this definition we use the !-pushforward, which differs from the *pushforward since \tilde{m} is not proper.)

Remark 5.1. As in Remark 4.1, given I_u -stable subschemes X, Y in \widetilde{Fl}_G , we will also denote by $X \times Y$ the quotient of $X' \times Y$ by the I_u -action induced by that on $LG \times \widetilde{Fl}_G$, where X' is the preimage of X in LG.

Lemma 5.2. For any \mathscr{F}, \mathscr{G} in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$ we have

$$\left(\pi^{\dagger}\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F})\right)\star_{\mathrm{I}_{\mathrm{u}}}\left(\pi^{\dagger}\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{G})\right)\cong\left(\pi^{\dagger}\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F}\star_{\mathrm{I}}\mathscr{G})\right)\otimes_{\Bbbk}\mathsf{H}^{[\bullet]}_{c}(T;\Bbbk)[2\dim(T)],$$

where we write $\mathsf{H}_{c}^{[\bullet]}(T; \Bbbk)$ for $\bigoplus_{i \in \mathbb{Z}} \mathsf{H}_{c}^{i}(T, \Bbbk)[-i]$.

Proof. If we denote by $D_{\mathrm{I}}^{\mathrm{b}}(\widetilde{\mathrm{Fl}}_{G}, \Bbbk)$ the I-equivariant derived category of $\widetilde{\mathrm{Fl}}_{G}$, the same definition as for the convolution product \star_{I} defines a canonical bifunctor

$$\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \times D^{\mathrm{b}}_{\mathrm{I}}(\widetilde{\mathrm{Fl}}_{G}, \Bbbk) \to D^{\mathrm{b}}_{\mathrm{I}_{\mathrm{u}}}(\widetilde{\mathrm{Fl}}_{G}, \Bbbk),$$

which will also be denoted \star_I . Let us again denote by

$$\operatorname{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}: D^{\mathrm{b}}_{\mathrm{I}}(\widetilde{\mathrm{Fl}}_{G}, \Bbbk) \to D^{\mathrm{b}}_{\mathrm{I}_{\mathrm{u}}}(\widetilde{\mathrm{Fl}}_{G}, \Bbbk)$$

the natural forgetful functor; then by the same considerations as for [BGMRR, Lemma 2.5], for \mathscr{F}' in $D^{\mathrm{b}}_{\mathrm{I}_{\mathrm{u}}}(\widetilde{\mathrm{Fl}}_{G},\Bbbk)$ and \mathscr{G}' in $D^{\mathrm{b}}_{\mathrm{I}}(\widetilde{\mathrm{Fl}}_{G},\Bbbk)$ we have a canonical isomorphism

$$\mathscr{F}' \star_{\mathrm{I}_{\mathrm{u}}} \mathrm{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{G}') \cong (\pi_{!}\mathscr{F}') \star_{\mathrm{I}} \mathscr{G}'[\dim(T)].$$

With \mathscr{F}, \mathscr{G} as in the statement, we deduce an isomorphism

$$\begin{aligned} \left(\pi^{\dagger}\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F})\right)\star_{\mathrm{I}_{\mathrm{u}}}\left(\pi^{\dagger}\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{G})\right) &\cong \left(\pi^{*}\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F})\right)\star_{\mathrm{I}_{\mathrm{u}}}\left(\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}\circ\pi^{*}(\mathscr{G})\right)[2\dim(T)] \\ &\cong \left(\pi_{!}\pi^{*}\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F})\right)\star_{\mathrm{I}}\left(\pi^{*}\mathscr{G}\right)[3\dim(T)]. \end{aligned}$$

Now we have

$$\pi_! \pi^* \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F}) \big) \star_{\mathrm{I}} \big(\pi^* \mathscr{G} \big) \cong \pi^* \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}} \big((\pi_! \pi^* \mathscr{F}) \star_{\mathrm{I}} \mathscr{G} \big).$$

If we denote by $X \subset \operatorname{Fl}_G$ a closed union of I-orbits over which \mathscr{F} is supported, and by \widetilde{X} its preimage in $\widetilde{\operatorname{Fl}}_G$, then by the projection formula we have $\pi_!\pi^*\mathscr{F} \cong \mathscr{F} \otimes_{\Bbbk}$ $\pi_!\underline{\Bbbk}_{\widetilde{X}}$. Since \widetilde{X} is a *T*-torsor over *X*, we have a canonical isomorphism $\widetilde{X} \times_X \widetilde{X} \cong$ $T \times \widetilde{X}$; the base change theorem then implies that $\pi^*\pi_!\underline{\Bbbk}_{\widetilde{X}} \cong \underline{\Bbbk}_{\widetilde{X}} \otimes_{\Bbbk} \operatorname{H}_c^{[\bullet]}(T; \Bbbk)$, and using this (and the definition of the convolution product) we obtain an isomorphism

$$(\pi_! \pi^* \mathscr{F}) \star_{\mathrm{I}} \mathscr{G} \cong (\mathscr{F} \star_{\mathrm{I}} \mathscr{G}) \otimes_{\Bbbk} \mathsf{H}_c^{[\bullet]}(T; \Bbbk).$$

The desired isomorphism follows.

The formula in Lemma 5.2 shows in particular that the bifunctor \star_{I_u} restricts to a bifunctor

$$\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

A similar construction provides a bifunctor

$$\star_{\mathrm{I}_{\mathrm{u}}}:\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

such that

$$\pi^{\dagger}(\mathscr{F}\star_{\mathrm{I}_{\mathrm{u}}}\mathscr{G})\cong\mathscr{F}\star_{\mathrm{I}_{\mathrm{u}}}\pi^{\dagger}(\mathscr{G})$$

for \mathscr{F} in $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and \mathscr{G} in $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$.

Verdier's monodromy construction (see [Ve]; see also [BR1, Go1] for additional comments) with respect to the action of $T \times T$ on \widetilde{Fl}_G via $(t_1, t_2) \cdot gI_u = t_1gt_2I_u$ provides, for any \mathscr{F} in D_{I_u,I_u} , a canonical algebra morphism

$$\mu_{\mathscr{F}}: \mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk}) \to \operatorname{End}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{F}),$$

which is unipotent in the sense that it vanishes on a power of the kernel of the natural augmentation morphism $\mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk}) \to \Bbbk$. (Here we use the identification $\mathscr{O}(T^{\vee}_{\Bbbk}) = \Bbbk[X_*(T)]$.) This construction satisfies various forms of functoriality; in particular, for any \mathscr{F}, \mathscr{G} in D_{I_u, I_u} and any morphism $f : \mathscr{F} \to \mathscr{G}$ we have

$$f \circ \mu_{\mathscr{F}}(x) = \mu_{\mathscr{G}}(x) \circ f$$

for all $x \in \mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk})$. With this structure, $\mathsf{D}_{I_{u},I_{u}}$ becomes an $\mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk})$ -linear category.

Remark 5.3. Recall that each I-orbit on Fl_G is stable under the loop rotation action, as well as each pullback of such an orbit to $\widetilde{\operatorname{Fl}}_G$. As a consequence, for every object \mathscr{F} in $\mathsf{D}_{\mathrm{I}_{\mathrm{U}},\mathrm{I}_{\mathrm{U}}}$, we have a canonical monodromy morphism

$$\overline{\mu}^{\mathrm{rot}}_{\mathscr{F}}: \Bbbk[x, x^{-1}] \to \mathrm{End}_{\mathsf{D}_{\mathrm{I},\mathrm{I}}}(\mathscr{F}), \quad \mathrm{resp.} \quad \mu^{\mathrm{rot}}_{\mathscr{G}}: \Bbbk[x, x^{-1}] \to \mathrm{End}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{G}).$$

These morphisms possess the same functoriality properties as those considered above.

Recall also that any object of P_{L^+G,L^+G} is automatically equivariant with respect to the loop rotation action, see [MV, Proposition 2.2]. As explained in [AB, §5.2] (see also [AR4, Proposition 2.4.6 and its proof]), as a consequence of Lemma 4.3 and this property, for any \mathscr{A} in P_{L^+G,L^+G} we have

(5.1)
$$\mathbf{m}_{\mathscr{A}} = \overline{\mu}_{\mathsf{Z}(\mathscr{A})}^{\mathrm{rot}}(x^{-1}).$$

5.2. The monodromic regular quotient. The main player in this paper will be the abelian category

 $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$

defined as the Serre quotient of $\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ by the Serre subcategory $\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{+}$ generated by the simple objects $\pi^{\dagger}\mathsf{For}_{\mathrm{I}_{u}}^{\mathrm{I}}(\mathscr{IC}_{w})$ for $w \in W$ such that $\ell(w) > 0$. Let us also denote by $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{+}$ the full triangulated subcategory of $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ generated by the perverse sheaves $\pi^{\dagger}\mathsf{For}_{\mathrm{I}_{u}}^{\mathrm{I}}(\mathscr{IC}_{w})$ for $w \in W$ such that $\ell(w) > 0$, and by $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{0}$ the Verdier quotient of $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ by $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{+}$. By Lemma A.2 once again, there exists a unique t-structure on $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{0}$ such that the quotient functor

$$\Pi^0_{I_u,I_u}:\mathsf{D}_{I_u,I_u}\to\mathsf{D}^0_{I_u,I_u}$$

is t-exact; moreover this t-structure is bounded, and its heart identifies with $\mathsf{P}^{0}_{\mathrm{I}_{u},\mathrm{I}_{u}}$. This t-structure will be called the perverse t-structure, and the corresponding cohomology functors will be denoted ${}^{p}\mathscr{H}^{n}(-)$.

(2) For
$$\mathscr{F}, \mathscr{G}$$
 in ${}^{\mathrm{p}}\mathsf{D}^{\leq 0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and $n \in \mathbb{Z}_{>0}$, the object ${}^{\mathrm{p}}\mathscr{H}^{n}(\mathscr{F}\star_{\mathrm{I}_{\mathrm{u}}}\mathscr{G})$ belongs to $\mathsf{P}^{+}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$

Sketch of proof. The proof reduces to the case \mathscr{F} and \mathscr{G} are simple perverse sheaves, which reduces to the claims in [BRR, Lemma 5.1] using Lemma 5.2.

As a consequence of Lemma 5.4 we obtain that the bifunctor

$$\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{0}$$

defined by

$$(\mathscr{F},\mathscr{G})\mapsto \Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}\star_{\mathrm{I}_{\mathrm{u}}}\mathscr{G})$$

factors through a bifunctor

$$\star^0_{I_u}:\mathsf{D}^0_{I_u,I_u}\times\mathsf{D}^0_{I_u,I_u}\to\mathsf{D}^0_{I_u,I_u},$$

which is triangulated and defines a structure of monoidal category (without unit object) on $\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. This bifunctor is moreover "right t-exact" in the sense that if \mathscr{F},\mathscr{G} belong to the nonpositive part of the perverse t-structure on $\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, then so does $\mathscr{F}\star^0_{\mathrm{I}_{\mathrm{u}}}\mathscr{G}$. (This bifunctor is *not* t-exact if G is nontrivial, contrary to the situation for $\mathsf{D}^0_{\mathrm{L}}$.)

We also define the bifunctor

$$(-) \ {}^{\mathbf{p}}\!\!\star^{0}_{\mathrm{I}_{\mathrm{u}}} (-) : \mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \times \mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

by setting, for \mathscr{F}, \mathscr{G} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$,

$$\mathscr{F} \, {}^{\mathrm{p}} \! \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, \mathscr{G} := {}^{\mathrm{p}} \! \mathscr{H}^{0} (\mathscr{F} \, \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, \mathscr{G}).$$

The right t-exactness of $\star^0_{I_u}$ implies that the bifunctor $(-) {}^{p} \star^0_{I_u} (-)$ is right exact on both sides, and also that for $\mathscr{F}, \mathscr{G}, \mathscr{H}$ in $\mathsf{P}^0_{I_u, I_u}$ we have

$$\begin{split} & (\mathscr{F} \, {}^{\mathbf{p}} \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, \mathscr{G}) \, {}^{\mathbf{p}} \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, \mathscr{H} \cong {}^{\mathbf{p}} \mathscr{H}^{0} \big((\mathscr{F} \, \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, \mathscr{G}) \, \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, \mathscr{H} \big), \\ & \mathscr{F} \, {}^{\mathbf{p}} \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, (\mathscr{G} \, {}^{\mathbf{p}} \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, \mathscr{H}) \cong {}^{\mathbf{p}} \mathscr{H}^{0} \big(\mathscr{F} \, \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, (\mathscr{G} \, \star^{0}_{\mathrm{I}_{\mathrm{u}}} \, \mathscr{H}) \big); \end{split}$$

in particular, the associativity constraint on $\star^0_{I_u}$ induces an associativity constraint on ${}^{p}\!\star^0_{I_u}$, so that we obtain a monoidal category

$$(\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}, {}^{\mathrm{p}} \star^0_{\mathrm{I}_{\mathrm{u}}})$$

(again, without unit object).

5.3. Relation with the regular quotient. It follows from the definitions and the universal property of the Verdier quotient that the composition

$$\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \circ \pi^{\dagger} \circ \mathsf{For}^1_{\mathrm{I}_{\mathrm{u}}} : \mathsf{D}_{\mathrm{I},\mathrm{I}} \to \mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

factors through a triangulated functor

$$\pi_0^{\dagger}:\mathsf{D}^0_{\mathrm{I},\mathrm{I}}\to\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

From the t-exactness of π^{\dagger} we deduce that π_0^{\dagger} is t-exact with respect to the perverse t-structures.

Lemma 5.5. The restriction of π_0^{\dagger} to the hearts of the perverse t-structures admits a natural structure of monoidal functor

$$\pi_0^{\dagger}: (\mathsf{P}_{\mathrm{I},\mathrm{I}}^0, \star_{\mathrm{I}}^0) \to (\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^0, {}^{\mathrm{p}}\!\star_{\mathrm{I}_{\mathrm{u}}}^0).$$

Proof. From Lemma 5.2 we obtain for \mathscr{F}, \mathscr{G} in $\mathsf{D}^0_{\mathrm{I},\mathrm{I}}$ a canonical isomorphism

$$\pi_0^{\dagger}(\mathscr{F}) \star_{\mathrm{I}_{\mathrm{u}}}^{\dagger} \pi_0^{\dagger}(\mathscr{G}) \cong \pi_0^{\dagger}(\mathscr{F} \star_{\mathrm{I}}^0 \mathscr{G}) \otimes_{\Bbbk} \mathsf{H}_c^{[\bullet]}(T; \Bbbk)[2 \dim(T)].$$

If ${\mathscr F}$ and ${\mathscr G}$ belong to the heart of the perverse t-structure, we deduce a canonical isomorphism

$$\pi_0^{\dagger}(\mathscr{F}) \,{}^{\mathrm{p}}\!\star^0_{\mathrm{I}_{\mathrm{u}}} \,\pi_0^{\dagger}(\mathscr{G}) \cong \pi_0^{\dagger}(\mathscr{F} \star^0_{\mathrm{I}} \mathscr{G}) \otimes \mathsf{H}^{2\dim(T)}_c(T; \Bbbk).$$

Now the vector space $\mathsf{H}^{2\dim(T)}_{c}(T; \Bbbk)$ is canonically isomorphic to \Bbbk since T is connected, which provides an isomorphism of bifunctors defining a monoidal structure on our functor.

Similar considerations show that the composition

$$\Pi^0_{I_u,I_u} \circ \pi^\dagger: \mathsf{D}_{I_u,I} \to \mathsf{D}^0_{I_u,I_u}$$

factors through a triangulated functor

$$\pi^{\dagger,0}:\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\to\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

which is t-exact for the perverse t-structures and satisfies

(5.2)
$$\pi_0^{\dagger} \cong \pi^{\dagger,0} \circ \operatorname{For}_{I_{u}}^{I,0}$$

where $\mathsf{For}_{I_{u}}^{I,0}$ is defined in §4.7.

6. Free-monodromic perverse sheaves

We continue to consider the general setting of **§4.1**.

6.1. **Yun's completed category.** Following Yun (see [BY, Appendix A]; see also [BR1] for additional comments) we consider the "completed" category

$$\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

associated with the *T*-torsor $\pi : \widetilde{\mathrm{Fl}}_G \to \mathrm{Fl}_G$ and the I_u-action on these ind-schemes. Recall that this category is defined as the full subcategory of the category of proobjects in $\mathsf{D}_{\mathrm{I}_u,\mathrm{I}_u}$ whose objects are the systems

$$\lim_{n \in \mathbb{Z}_{\geq 0}} \mathscr{F}_n$$

which are

- π -constant, in the sense that the pro-object " \varprojlim " $\pi_!(\mathscr{F}_n)$ is isomorphic to an object in $\mathsf{D}_{\mathrm{I}_u,\mathrm{I}}$;
- uniformly bounded in degrees, in the sense that we have an isomorphism " $\lim \mathscr{F}_n \cong \lim \mathscr{F}'_n$ and some $N \in \mathbb{Z}_{\geq 0}$ such that each \mathscr{F}'_n satisfies $\mathscr{P}\mathscr{H}^i(\mathscr{F}'_n) = 0$ unless $i \in [-N, N]$.

It is proved in [BY, Appendix A] that this category admits a triangulated structure, for which the distinguished triangles are the diagrams isomorphic to one of the form

$$\underset{}{\overset{\text{``}}\varprojlim} \mathscr{F}_n \xrightarrow{\overset{\text{``}}\amalg} \mathscr{F}_n \xrightarrow{\overset{\text{``}}\amalg} \mathscr{G}_n \xrightarrow{\overset{\text{``}}\amalg} \mathscr{G}_n \xrightarrow{\overset{\text{``}}\amalg} \overset{\text{``}}\longleftrightarrow} \mathscr{F}_n \xrightarrow{\overset{\text{``}}\amalg} \mathscr{F}_n \xrightarrow{\overset{\text{``}}\amalg} \mathscr{F}_n[1]$$

where each

$$\mathscr{F}_n \xrightarrow{\alpha_n} \mathscr{G}_n \xrightarrow{\beta_n} \mathscr{H}_n \xrightarrow{\gamma_n} \mathscr{F}_n[1]$$

is a distinguished triangle in $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}.$ In particular we have a canonical fully faithful triangulated functor

$$(6.1) D_{I_u,I_u} \to D^{\wedge}_{I_u,I_u}$$

sending an object \mathscr{F} to the constant projective system with value \mathscr{F} . The functor $\pi_{\dagger} = \pi_![\dim(T)]$ defines a functor $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$, which will also be denoted π_{\dagger} , and which can be shown to be triangulated. This functor is conservative by [BY, Lemma A.3.5]. By [BR1, Corollary 5.5], the category $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is Krull–Schmidt.

It is clear that in this definition one can replace the ind-scheme Fl_G by the inverse image of any locally closed union X of I-orbits in Fl_G ; in this setting the completed category will be denoted $\mathsf{D}_{\operatorname{I}_n}^{\wedge}(X,\Bbbk)$.

Recall the local systems $\mathscr{L}_{T,n}$ on T considered in [BR1, §10.1], and the associated pro-object

$$\mathscr{L}_T^\wedge := \underset{n}{\overset{\text{\tiny{(in)}}}{\underset{n}{\underset{n}{\overset{}}{\overset{}}}}} \mathscr{L}_{T,n}$$

on T. (Here, each $\mathscr{L}_{T,n}$ is an extension of copies of the constant local system $\underline{\Bbbk}_T$.) We have $\widetilde{\mathrm{Fl}}_{G,e} = B/U \cong T$. The pro-object \mathscr{L}_T^{\wedge} therefore defines a pro-local system on $\widetilde{\mathrm{Fl}}_{G,e}$; taking the pushforward under the embedding in $\widetilde{\mathrm{Fl}}_G$ of its shift by dim(T)we obtain an object in $\mathsf{D}_{\mathrm{I}_u,\mathrm{I}_u}^{\wedge}$, which will be denoted δ^{\wedge} . By [BR1, Equation (3.3)], we have a canonical isomorphism

$$\pi_{\dagger}(\delta^{\wedge}) \cong \mathsf{For}^{\mathsf{I}}_{\mathsf{I}_{\mathsf{u}}}(\delta).$$

The arguments of [BY, §4.3] (see also [BR1, §7.3] for the analoguous case of G/U) show that the monoidal product \star_{I_u} extends to a bifunctor

$$\widehat{\star}:\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

which is triangulated on both sides. More specifically, the bifunctor \star_{I_u} extends in a canonical way to a bifunctor $\widehat{\star}$ on pro-objects, so that for pro-objects " $\varprojlim_{i \in I}$ " \mathscr{F}_i and " $\varprojlim_{i \in J}$ " \mathscr{G}_j we have

$$\left(\underset{i\in I}{``\varprojlim_i}"\mathscr{F}_i\right)\widehat{\star}\left(\underset{j\in J}{``\varprojlim_j}"\mathscr{G}_j\right)=\underset{(i,j)\in I\times J}{``\varprojlim_i}"\mathscr{F}_i\star_{\mathrm{I}_{\mathrm{u}}}\mathscr{G}_j.$$

Now if these pro-objects belong to $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ (so that, in particular, we can assume $I = J = \mathbb{Z}_{\geq 0}$) we have

$$\left(\underbrace{``\lim_{n} "\mathscr{F}_n}_{n}\right)\widehat{\star}\left(\underbrace{``\lim_{m} "\mathscr{G}_m}_{m}\right) = \underbrace{``\lim_{m} "}_{m}\left(\underbrace{``\lim_{n} "\mathscr{F}_n \star_{\mathrm{I}_u} \mathscr{G}_m}_{n}\right)$$

where for each *m* the pro-object " \varprojlim_n " $\mathscr{F}_n \star_{I_u} \mathscr{G}_m$ is representable by an object of D_{I_u,I_u} . It is explained in [BY, §4.3] that this formal inverse limit of objects in D_{I_u,I_u}

belongs to $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. In fact we also have

$$\left(\underbrace{``\lim_{n} "\mathscr{F}_n}_{n} \right) \widehat{\star} \left(\underbrace{``\lim_{m} "\mathscr{G}_m}_{m} \right) = \underbrace{``\lim_{n} "}_{n} \left(\underbrace{``\lim_{m} "\mathscr{F}_n \star_{\mathrm{I}_u} \mathscr{G}_m}_{m} \right)$$

where for each *n* the pro-object " \varprojlim_{m} " $\mathscr{F}_{n} \star_{\mathrm{I}_{\mathrm{u}}} \mathscr{G}_{m}$ belongs to $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

The bifunctor $\hat{\star}$ on $\mathsf{D}^{\wedge}_{\mathbf{I}_u,\mathbf{I}_u}$ admits a natural associativity constraint and a unit object (namely, δ^{\wedge}), which equips $\mathsf{D}^{\wedge}_{\mathbf{I}_u,\mathbf{I}_u}$ with the structure of a monoidal category. It restricts to (triangulated) bifunctors

$$\widehat{\star}:\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},\quad \widehat{\star}:\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

which define left and right actions of the monoidal category $(\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge},\widehat{\star})$ on the category $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ respectively. These two actions commute, and are compatible in various ways; in particular, for $\mathscr{F}_1,\mathscr{F}_2$ in $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$ and $\mathscr{G}_1,\mathscr{G}_2$ in $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ we have canonical isomorphisms

(6.2)

$$(\mathscr{G}_1 \mathbin{\widehat{\star}} \mathscr{F}_1) \star_{\mathrm{I}_{\mathrm{u}}} (\mathscr{F}_2 \mathbin{\widehat{\star}} \mathscr{G}_2) \cong \mathscr{G}_1 \star_{\mathrm{I}_{\mathrm{u}}} ((\mathscr{F}_1 \mathbin{\widehat{\star}} \mathscr{F}_2) \mathbin{\widehat{\star}} \mathscr{G}_2) \cong (\mathscr{G}_1 \mathbin{\widehat{\star}} (\mathscr{F}_1 \mathbin{\widehat{\star}} \mathscr{F}_2)) \star_{\mathrm{I}_{\mathrm{u}}} \mathscr{G}_2,$$

$$(6.3) \qquad \qquad (\mathscr{F}_1 \mathbin{\widehat{\star}} \mathscr{G}_1) \star_{\mathrm{I}_{\mathrm{u}}} \mathscr{G}_2 \cong \mathscr{F}_1 \mathbin{\widehat{\star}} (\mathscr{G}_1 \star_{\mathrm{I}_{\mathrm{u}}} \mathscr{G}_2).$$

Remark 6.1. As explained above, the convolution product on $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$ extends that on $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, and there exists a unit object for this product. The construction of $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$ can be seen as a way to "complete" the category $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ so that it admits a unit object, while staying in the world of *triangulated* monoidal categories.

We similarly have an action on the category $D_{I_u,I}$, defined by a bifunctor

$$\widehat{\star}:\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} imes\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} o\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

whose definition is similar to that considered above, and which is also triangulated on both sides. These functors satisfy the following relations (see [BR1, Equations (7.2) and (7.6), Lemma 7.2]):

(6.4)
$$\pi_{\dagger}(\mathscr{F} \widehat{\star} \mathscr{G}) \cong \mathscr{F} \widehat{\star} \pi_{\dagger}(\mathscr{G})$$

(6.5)
$$\mathscr{F} \mathbin{\widehat{\star}} \operatorname{For}_{\mathrm{L}}^{\mathrm{I}}(\mathscr{H}) \cong \pi_{\dagger}(\mathscr{F}) \star_{\mathrm{I}} \mathscr{H}$$

(6.6)
$$\mathscr{F} \mathbin{\widehat{\star}} \pi^{\dagger}(\mathscr{K}) \cong \pi^{\dagger}(\mathscr{F} \mathbin{\widehat{\star}} \mathscr{K})$$

 $\mathrm{for}\ \mathscr{F}, \mathscr{G}\ \mathrm{in}\ \mathsf{D}^\wedge_{\mathrm{I}_\mathrm{u},\mathrm{I}_\mathrm{u}},\ \mathscr{H}\ \mathrm{in}\ \mathsf{D}_{\mathrm{I},\mathrm{I}}\ \mathrm{and}\ \mathscr{K}\ \mathrm{in}\ \mathsf{D}_{\mathrm{I}_\mathrm{u},\mathrm{I}}.$

The monodromy constructions of §5.1 pass to the completion, and provide for any \mathscr{F} in $\mathsf{D}^{\wedge}_{\mathrm{I}_n,\mathrm{I}_n}$ canonical algebra morphisms

$$\mu_{\mathscr{F}}:\mathscr{O}(T^{\vee}_{\Bbbk}\times T^{\vee}_{\Bbbk})\to \mathrm{End}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{F}), \quad \mu^{\mathrm{rot}}_{\mathscr{F}}:\Bbbk[x,x^{-1}]\to \mathrm{End}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{F})$$

which commute with all morphisms. In particular, with the morphisms $\mu_{\mathscr{F}}$, $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ becomes an $\mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk})$ -linear category. These operations are compatible with convolution, in the sense that for \mathscr{F}, \mathscr{G} in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and $f, g \in \mathscr{O}(T^{\vee}_{\Bbbk})$ we have

(6.7)
$$\mu_{\mathscr{F}\widehat{\star}\mathscr{G}}(f\otimes g) = \mu_{\mathscr{F}}(f\otimes 1)\widehat{\star}\,\mu_{\mathscr{G}}(1\otimes g)$$

(6.8)
$$\mu_{\mathscr{F}}(1 \otimes f) \widehat{\star} \operatorname{id}_{\mathscr{G}} = \operatorname{id}_{\mathscr{F}} \widehat{\star} \mu_{\mathscr{G}}(f \otimes 1).$$

(The proof is similar to that given for sheaves on G/U in [BR1, Lemma 7.3].) It is not difficult to check that for \mathscr{F}, \mathscr{G} in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ we also have

(6.9)
$$\mu_{\mathscr{F}\widehat{\star}\mathscr{G}}^{\mathrm{rot}}(x) = \mu_{\mathscr{F}}^{\mathrm{rot}}(x) \widehat{\star} \, \mu_{\mathscr{G}}^{\mathrm{rot}}(x).$$

The following claim follows from [BR1, Remark 5.1].

Lemma 6.2. Let \mathscr{F} in $\mathsf{D}^{\wedge}_{\mathbf{I}_{\mathrm{u}},\mathbf{I}_{\mathrm{u}}}$. Then \mathscr{F} belongs to the essential image of the functor (6.1) iff the restriction of $\mu_{\mathscr{F}}$ to $\Bbbk \otimes_{\Bbbk} \mathscr{O}(T^{\vee}_{\Bbbk}) \subset \mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk})$ vanishes on some power of the maximal ideal corresponding to $e \in T_{\Bbbk}^{\vee}$.

6.2. The perverse t-structure. Another important feature of the "completed" (or "free monodromic") category $\mathsf{D}^\wedge_{\mathrm{I}_u,\mathrm{I}_u}$ is that it admits a "perverse" t-structure $({}^{p}\mathsf{D}_{I_{u},I_{u}}^{\wedge,\leq 0}, {}^{p}\mathsf{D}_{I_{u},I_{u}}^{\wedge,\geq 0})$, whose heart will be denoted $\mathsf{P}_{I_{u},I_{u}}^{\wedge}$. (For the definition of this t-structure, see [BR1, §5.2]; for an earlier and slightly different construction of this t-structure, see [BY, (6.1)].) From the construction it is clear that the functor (6.1)is t-exact. One can check also that the functor π_{\dagger} of §6.1 is right t-exact, see [BR1, Corollary 5.8].

For any $w \in W$, the quotient $I_u \setminus \widetilde{Fl}_{G,w}$ is a *T*-torsor over $\operatorname{Spec}(\Bbbk)$ for the action induced by right multiplication on LG (but this torsor does not admit any canonical trivialization in general). After choosing a T-equivariant isomorphism $I_u \setminus \widetilde{Fl}_{G,w} \xrightarrow{\sim}$ T, we can then define

where p_w is the composition $\widetilde{\mathrm{Fl}}_{G,w} \to \mathrm{I}_{\mathrm{u}} \setminus \widetilde{\mathrm{Fl}}_{G,w} \xrightarrow{\sim} T$, and $\widetilde{j}_w : \widetilde{\mathrm{Fl}}_{G,w} \to \widetilde{\mathrm{Fl}}_G$ is the embedding. These objects are perverse sheaves, and do not depend on the choice of trivialization up to (noncanonical) isomorphism. They also satisfy

(6.10)
$$\pi_{\dagger}(\Delta_w^{\wedge}) \cong \operatorname{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\Delta_w^{\mathrm{I}}), \quad \pi_{\dagger}(\nabla_w^{\wedge}) \cong \operatorname{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\nabla_w^{\mathrm{I}}),$$

see [BR1, Equation (5.3)]. With these objects at hand, one can describe the nonpositive part ${}^{p}\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge,\leq 0}$ of the perverse t-structure on $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$ as the subcategory generated under extensions by the objects $\Delta_{w}^{\wedge}[n]$ with $w \in W$ and $n \geq 0$, see [BR1, Lemma 5.6]. (The similar statement for the nonnegative part of the t-structure does *not* hold.)

Lemma 6.3. (1) For any $w \in W$, there exist isomorphisms

$$\Delta^{\wedge}_{w} \mathbin{\widehat{\star}} \nabla^{\wedge}_{w^{-1}} \cong \delta^{\wedge}, \quad \nabla^{\wedge}_{w^{-1}} \mathbin{\widehat{\star}} \Delta^{\wedge}_{w} \cong \delta^{\wedge}.$$

(2) If $w, y \in W$ are such that $\ell(wy) = \ell(w) + \ell(y)$, then there exist isomorphisms $\Delta_w^{\wedge} \widehat{\star} \Delta_y^{\wedge} \cong \Delta_{wy}^{\wedge}, \quad \nabla_w^{\wedge} \widehat{\star} \nabla_y^{\wedge} \cong \nabla_{wy}^{\wedge}.$

$$\Delta_w^{\wedge} \star \Delta_y^{\wedge} \cong \Delta_{wy}^{\wedge}, \quad \nabla_w^{\wedge} \star \nabla_y^{\wedge} \cong \nabla_{wy}^{\wedge},$$

(3) For any $w, y \in W$, the objects

$$\Delta_w^\wedge \mathbin{\widehat{\star}} \nabla_y^\wedge \quad and \quad \nabla_w^\wedge \mathbin{\widehat{\star}} \Delta_y^\wedge$$

belong to $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

Proof. For (1)-(2), the proof is similar to that of the corresponding statements on G/U, treated in detail in [BR1, Lemma 7.7]. For (3), the proof is again based on the same idea: by (6.4)-(6.5) and (6.10) we have

$$\pi_{\dagger}(\Delta_w^{\wedge} \mathbin{\widehat{\star}} \nabla_y^{\wedge}) \cong \Delta_w^{\wedge} \mathbin{\widehat{\star}} \operatorname{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\nabla_y^{\mathrm{I}}) \cong \operatorname{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\Delta_w^{\mathrm{I}}) \star_{\mathrm{I}} \nabla_y^{\mathrm{I}}$$

Now it is well known that the right-hand side is a perverse sheaf, see e.g. [AR4, Lemma 4.1.7]. Since an object whose image under π_{\dagger} is perverse is itself perverse (see [BR1, Lemma 5.3(1)]), the claim follows. Remark 6.4. Lemma 6.3(1) implies in particular that the functor of left (resp. right) convolution with Δ_w^{\wedge} is an equivalence of categories, with quasi-inverse given by left (resp. right) convolution with $\nabla_{w^{-1}}^{\wedge}$.

Corollary 6.5. For any $w \in W$:

(1) the functors

$$\nabla_{w}^{\wedge} \widehat{\star} (-), \, (-) \, \widehat{\star} \, \nabla_{w}^{\wedge} : \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$$

- are right t-exact;
- (2) the functors

$$\Delta_w^{\wedge} \mathbin{\widehat{\star}} (-), \, (-) \mathbin{\widehat{\star}} \Delta_w^{\wedge} : \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$$

are left t-exact

Proof. Since the nonpositive part of the perverse t-structure on $\mathsf{D}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ is generated under extensions by the objects $\Delta^{\wedge}_{y}[n]$ for $y \in W$ and $n \geq 0$ (see above), (1) is a consequence of Lemma 6.3(3). We deduce (2) using the fact that the functor $\nabla^{\wedge}_{w^{-1}} \widehat{\star}(-)$, resp. $(-) \widehat{\star} \nabla^{\wedge}_{w^{-1}}$ is left adjoint to $\Delta^{\wedge}_{w} \widehat{\star}(-)$, resp. $(-) \widehat{\star} \Delta^{\wedge}_{w}$, see Remark 6.4.

Below we will also need to consider the monodromy morphisms for the objects Δ_w^{\wedge} and ∇_w^{\wedge} . The following lemma is the analogue in our present setting of [BR1, Lemma 5.4, Lemma 6.1]. Here we denote by $(T_{\Bbbk}^{\vee})^{\wedge}$ the spectrum of the completion of $\mathscr{O}(T_{\Bbbk}^{\vee})$ with respect to the ideal corresponding to $e \in T_{\Bbbk}^{\vee}$. For $\lambda \in X_*(T)$, we will also denote by e^{λ} the corresponding morphism $T_{\Bbbk}^{\vee} \to \mathbb{G}_m$, seen as an element in $\mathscr{O}(T_{\Bbbk}^{\vee})$.

Lemma 6.6. Let $w \in W$, and write $w = vt(\lambda)$ with $v \in W_f$ and $\lambda \in X_*(T)$.

(1) The restriction of $\mu_{\Delta_w^{\wedge}}$ to the subalgebra

$$\mathscr{O}(T^{\vee}_{\Bbbk}) = \Bbbk \otimes \mathscr{O}(T^{\vee}_{\Bbbk}) \subset \mathscr{O}(T^{\vee}_{\Bbbk}) \otimes \mathscr{O}(T^{\vee}_{\Bbbk}) = \mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk})$$

factors (in the natural way) through an isomorphism

$$\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge}) \xrightarrow{\sim} \operatorname{End}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}}},\mathrm{I}_{\mathrm{u}}}(\Delta^{\wedge}_{w}),$$

and any nonzero endomorphism of Δ_w^{\wedge} is injective. Moreover, for any $f \in \mathscr{O}(T_{\Bbbk}^{\vee})$ we have

$$\mu_{\Delta_{w}^{\wedge}}(f \otimes 1) = \mu_{\Delta_{w}^{\wedge}}(1 \otimes v^{-1}(f)),$$

and we have

$$\mu_{\Delta_w^{\wedge}}^{\mathrm{rot}}(x) = \mu_{\Delta_w^{\wedge}}(1 \otimes e^{-\lambda}).$$

(2) The restriction of $\mu_{\nabla_w^{\wedge}}$ to the subalgebra

$$\mathscr{O}(T^{\vee}_{\Bbbk}) = \Bbbk \otimes \mathscr{O}(T^{\vee}_{\Bbbk}) \subset \mathscr{O}(T^{\vee}_{\Bbbk}) \otimes \mathscr{O}(T^{\vee}_{\Bbbk}) = \mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk})$$

factors (in the natural way) through an isomorphism

$$\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge}) \xrightarrow{\sim} \operatorname{End}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{u}},\mathrm{I}_{u}}(\nabla^{\wedge}_{w}).$$

Moreover, for any $f \in \mathscr{O}(T^{\vee}_{\Bbbk})$ we have

$$\mu_{\nabla_{w}^{\wedge}}(f \otimes 1) = \mu_{\nabla_{w}^{\wedge}}(1 \otimes v^{-1}(f)),$$

and we have

$$\mu_{\nabla_w^{\wedge}}^{\mathrm{rot}}(x) = \mu_{\nabla_w^{\wedge}}(1 \otimes e^{-\lambda}).$$

Proof. The claims about the right monodromy and the injectivity of nonzero morphisms are general facts in completed derived categories, see [BR1, Lemma 5.4]. Since I_u is normal in I, the left action of T on $\widetilde{Fl}_{G,w}$ induces an action on the quotient $I_u \setminus \widetilde{Fl}_{G,w}$, and it is easily seen that this action is the twist of the action considered above on $I_u \setminus \widetilde{Fl}_{G,w}$ by v^{-1} , which implies the claim about left monodromy by basic properties of the monodromy construction, see [BR1, Lemma 2.5]. Similarly, the loop rotation action on $\widetilde{Fl}_{G,w}$ induces an action on the quotient $I_u \setminus \widetilde{Fl}_{G,w}$, which is deduced from the T-action via $-\lambda : \mathbb{G}_m \to T$; this implies the claims about the loop rotation monodromy.

6.3. Tilting perverse sheaves. Recall that an object \mathscr{F} in $\mathsf{P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ is said to be tilting if it admits a filtration (in the abelian category $\mathsf{P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$) with subquotients of the form Δ^{\wedge}_{w} ($w \in W$) and a filtration with subquotients of the form ∇^{\wedge}_{w} ($w \in W$). The full subcategory of $\mathsf{P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ whose objects are the tilting perverse sheaves will be denoted $\mathsf{T}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$. As explained in [BY, §A.7] (see also [BR1, §5.5]), the isomorphism classes of indecomposable objects in $\mathsf{T}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ are in a canonical bijection with W; more specifically, for any $w \in W$ there exists a unique (up to isomorphism) object \mathscr{T}^{\wedge}_{w} in $\mathsf{D}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ such that $\pi_{\dagger}(\mathscr{T}^{\wedge}_{w}) \cong \mathscr{T}_{w}$ (where the right-hand side is defined in §4.2); then \mathscr{T}^{\wedge}_{w} is an indecomposable tilting object in $\mathsf{P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$, and the assignment $w \mapsto \mathscr{T}^{\wedge}_{w}$ provides the desired bijection. (We insist that \mathscr{T}^{\wedge}_{w} is defined only up to isomorphism.)

The same arguments as in [BR1, Remark 7.9] show that $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is a monoidal subcategory in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. Note also that we have natural equivalences of categories

(6.11)
$$K^{\mathbf{b}}\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \xrightarrow{\sim} D^{\mathbf{b}}\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \xrightarrow{\sim} \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

see [BR1, Proposition 5.11], and that the composition of these two equivalences has a natural monoidal structure.

Using tilting objects one obtains the following property of monodromy.

Lemma 6.7. For any \mathscr{F} in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, the monodromy morphism $\mu_{\mathscr{F}}$ factors through the surjection

$$\mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk}) \to \mathscr{O}(T^{\vee}_{\Bbbk} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} T^{\vee}_{\Bbbk}).$$

Proof. Let us first note that for $w, y \in W$ we have

(6.12)
$$\operatorname{Hom}_{\mathsf{P}_{\operatorname{In},\operatorname{In}}^{\wedge}}(\Delta_w^{\wedge}, \Delta_y^{\wedge}) = 0 \quad \text{if } w \neq y$$

In fact, write $w = vt(\lambda)$ and $y = v't(\lambda')$ with $v, v' \in W_f$ and $\lambda, \lambda' \in X_*(T)$. If $f : \Delta_w^{\wedge} \to \Delta_y^{\wedge}$ is a nonzero morphism and \mathscr{F} is its image, then for any $r \in \mathscr{O}(T_k^{\vee})$ we have $\mu_{\mathscr{F}}(r \otimes 1) = \mu_{\mathscr{F}}(1 \otimes v^{-1}(r))$ by Lemma 6.6(1) and the fact that \mathscr{F} is a quotient of Δ_w^{\wedge} , and $\mu_{\mathscr{F}}(r \otimes 1) = \mu_{\mathscr{F}}(1 \otimes (v')^{-1}(r))$ by the same lemma and the fact that \mathscr{F} is a subobject of Δ_y^{\wedge} . Hence $\mu_{\mathscr{F}}(1 \otimes (v^{-1}(r) - (v')^{-1}(r))) = 0$ for any r. The injectivity claim in Lemma 6.6(1) then implies that $v^{-1}(r) = (v')^{-1}(r)$ for any r, so that v = v'. The same considerations using $\mu_{\mathscr{F}}^{\mathrm{rot}}$ show that $\lambda = \lambda'$, which finishes the proof of (6.12).

Once this claim is proved, using the fact that $\mu_{\Delta_w^{\wedge}}$ factors through the surjection $\mathscr{O}(T_{\Bbbk}^{\vee} \times T_{\Bbbk}^{\vee}) \to \mathscr{O}(T_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} T_{\Bbbk}^{\vee})$ (see once again Lemma 6.6(1)) one obtains as in [BR1, §§6.3–6.4] or in §7.3 below (using a faithful associated graded functor) that this property holds for any object of $\mathsf{T}_{\mathrm{I}_{w},\mathrm{I}_{w}}^{\wedge}$. Finally, the first equivalence in (6.11)

shows that any object in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is a subquotient of a tilting object, which implies our claim. \Box

We will still denote by $\mu_{\mathscr{F}}$ the morphism

 $\mathscr{O}(T^{\vee}_{\Bbbk} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} T^{\vee}_{\Bbbk}) \to \mathrm{End}(\mathscr{F})$

induced by the map previously denoted $\mu_{\mathscr{F}}$. In concrete terms, Lemma 6.7 means that the actions of $\mathscr{O}(T^{\vee}_{\Bbbk})$ on an object of $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ defined by monodromy for the left and right actions of T on $\widetilde{\mathrm{Fl}}_{G}$ coincide on the subalgebra $\mathscr{O}(T^{\vee}_{\Bbbk}/W_{\mathrm{f}})$. We can therefore speak unambiguously of the monodromy action of $\mathscr{O}(T^{\vee}_{\Bbbk}/W_{\mathrm{f}})$ on such an object. In view of (6.7) the same comment will apply to any object of the form $\mathscr{F} \widehat{\star} \mathscr{G}$ with \mathscr{F}, \mathscr{G} in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, and moreover the action on $\mathscr{F} \widehat{\star} \mathscr{G}$ identifies with both the action induced by that on \mathscr{F} and the action induced by that on \mathscr{G} .

6.4. Actions of $\mathsf{D}_{I_u,I_u}^{\wedge}$ on D_{I_u,I_u}^0 . Recall from §6.1 that the monoidal category $(\mathsf{D}_{I_u,I_u}^{\wedge}, \widehat{\star})$ acts on D_{I_u,I_u} , via a bifunctor

$$\widehat{\star}:\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}
ightarrow\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

Lemma 6.8. For $\mathscr{F} \in \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and $\mathscr{G} \in \mathsf{D}^{+}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, the object $\mathscr{F} \mathbin{\widehat{\star}} \mathscr{G}$ belongs to $\mathsf{D}^{+}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

Proof. As for Lemma 5.4, it suffices to prove the claim when $\mathscr{G} = \pi^{\dagger} \mathsf{For}_{\mathrm{I}_{u}}^{\mathrm{I}}(\mathscr{IC}_{w})$ for some $w \in W$ with $\ell(w) > 0$. Now using (6.5)–(6.6) we see that

$$\mathscr{F} \mathbin{\widehat{\star}} \pi^{\dagger} \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{I}\mathscr{C}_w) \cong \pi^{\dagger}((\pi_{\dagger}\mathscr{F}) \star_{\mathrm{I}} \mathscr{I}\mathscr{C}_w).$$

Here $\pi_{\dagger}\mathscr{F}$ is an object of $\mathsf{D}_{I_u,I}$, and the bifunctor we consider is that of (4.4). Then the claim follows once again from [BRR, Lemma 5.1].

With this lemma at hand, the universal property of the Verdier quotient implies that for any \mathscr{F} in $\mathsf{D}^\wedge_{\mathrm{I}_u,\mathrm{I}_u}$, the functor

$$\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{0}$$

defined by $\mathscr{G} \mapsto \Pi^0_{I_u,I_u}(\mathscr{F} \widehat{\star} \mathscr{G})$ factors canonically through a functor $\mathsf{D}^0_{I_u,I_u} \to \mathsf{D}^0_{I_u,I_u}$, and one checks easily that this operation defines a triangulated bifunctor

$$\widehat{\star}^0:\mathsf{D}^\wedge_{\mathrm{I}_\mathrm{u},\mathrm{I}_\mathrm{u}}\times\mathsf{D}^0_{\mathrm{I}_\mathrm{u},\mathrm{I}_\mathrm{u}}\to\mathsf{D}^0_{\mathrm{I}_\mathrm{u},\mathrm{I}_\mathrm{u}}$$

defining a left action of the monoidal category $(D^{\wedge}_{I_u,I_u}, \widehat{\star})$ on $D^{0}_{I_u,I_u}$. Similar considerations starting with the right action of $(D^{\wedge}_{I_u,I_u}, \widehat{\star})$ on D_{I_u,I_u} (by convolution on the right) leads to the construction of a triangulated bifunctor

$$\widehat{\star}^0:\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{D}^\wedge_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

defining a right action of the monoidal category $(\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}, \widehat{\star})$ on $\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. These two actions commute, and are related in various ways; in particular, from (6.2)–(6.3) we deduce that for $\mathscr{F}_{1}, \mathscr{F}_{2}$ in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and $\mathscr{G}_{1}, \mathscr{G}_{2}$ in $\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ we have canonical isomorphisms (6.13)

$$(\mathfrak{G}_{1} \overset{(0.13)}{\star^{0}} \mathscr{F}_{1}) \star^{0}_{\mathrm{I}_{\mathrm{u}}} (\mathscr{F}_{2} \overset{()}{\star^{0}} \mathscr{G}_{2}) \cong \mathscr{G}_{1} \star^{0}_{\mathrm{I}_{\mathrm{u}}} ((\mathscr{F}_{1} \overset{()}{\star} \mathscr{F}_{2}) \overset{()}{\star^{0}} \mathscr{G}_{2}) \cong (\mathscr{G}_{1} \overset{()}{\star^{0}} (\mathscr{F}_{1} \overset{()}{\star} \mathscr{F}_{2})) \star^{0}_{\mathrm{I}_{\mathrm{u}}} \mathscr{G}_{2},$$

$$(\mathfrak{6}.14) \qquad (\mathscr{F}_{1} \overset{()}{\star^{0}} \mathscr{G}_{1}) \star^{0}_{\mathrm{I}_{\mathrm{u}}} \mathscr{G}_{2} \cong \mathscr{F}_{1} \overset{()}{\star^{0}} (\mathscr{G}_{1} \star^{0}_{\mathrm{I}_{\mathrm{u}}} \mathscr{G}_{2}).$$

We will also consider the bifunctors

 ${}^{p}\widehat{\star}^{0}:\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},\quad {}^{p}\widehat{\star}^{0}:\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\times\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$

defined by

$$\mathscr{F}^{\mathrm{p}}\widehat{\star}^{0}\mathscr{G} := {}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F}\widehat{\star}^{0}\mathscr{G}).$$

Lemma 6.9. If $\mathscr{F} \in \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and $\mathscr{G} \in \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ belong to the nonpositive parts of the perverse t-structures, then so does $\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F} \mathbin{\widehat{\star}} \mathscr{G})$. In particular, for \mathscr{F} in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ the functor

$$\mathscr{F}^{p} \widehat{\star}^{0}(-) : \mathsf{P}^{0}_{\mathrm{I}_{u},\mathrm{I}_{u}} \to \mathsf{P}^{0}_{\mathrm{I}_{u},\mathrm{I}_{u}}$$

is right exact, and for \mathscr{G} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ the functor

$$(-) \stackrel{\mathrm{p}}{\star}^0 \mathscr{G} : \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

is right exact.

Proof. The first claim will follow in general if we prove it when \mathscr{G} is a simple perverse sheaf, i.e. is of the form $\pi^{\dagger}(\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{IC}_w))$ for some $w \in W$. In this case, as in the proof of Lemma 6.8 we have

$$\mathscr{F} \mathbin{\widehat{\star}} \pi^{\dagger}(\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{I}\mathscr{C}_{w})) \cong \pi^{\dagger}(\pi_{\dagger}(\mathscr{F}) \star_{\mathrm{I}} \mathscr{I}\mathscr{C}_{w}).$$

Now if $\ell(w) > 0$, Lemma 6.8 implies that $\Pi^0_{I_u,I_u}(\pi^{\dagger}(\pi_{\dagger}(\mathscr{F}) \star_I \mathscr{IC}_w)) = 0$, and if $\ell(w) = 0$ the object $\pi_{\dagger}(\mathscr{F}) \star_I \mathscr{IC}_w$ belongs to the nonpositive part of the perverse t-structure by right exactness of π_{\dagger} (see §6.2) and exactness of $(-) \star_I \mathscr{IC}_w$.

To prove the second claim, we fix \mathscr{F} in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. An exact sequence in $\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is given by a distinguished triangle

$$\mathscr{G}_1 \to \mathscr{G}_2 \to \mathscr{G}_3 \xrightarrow{[1]}$$

in $\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ where each \mathscr{G}_i belongs to $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. Applying the triangulated functor $\mathscr{F}^0(-)$ we deduce a distinguished triangle

$$\mathscr{F} \stackrel{\sim}{\star}^0 \mathscr{G}_1 \to \mathscr{F} \stackrel{\sim}{\star}^0 \mathscr{G}_2 \to \mathscr{F} \stackrel{\sim}{\star}^0 \mathscr{G}_3 \xrightarrow{[1]} .$$

Now $\mathscr{G}_1 = \Pi^0_{I_u,I_u}(\mathscr{G}'_1)$ for some \mathscr{G}'_1 in P_{I_u,I_u} (by essential surjectivity and t-exactness of $\Pi^0_{I_u,I_u}$), and then $\mathscr{F}^0 \mathscr{G}_1 = \Pi^0_{I_u,I_u}(\mathscr{F} \widehat{\star} \mathscr{G}'_1)$ is concentrated in nonpositive perverse degrees by the first claim; taking the long exact sequence of perverse cohomology associated with the above distinguished triangle we deduce an exact sequence

$${}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}}^{0} \mathscr{G}_{1}) \to {}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}}^{0} \mathscr{G}_{2}) \to {}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}}^{0} \mathscr{G}_{3}) \to 0,$$

showing the right exactness of $\mathscr{F}_{p^{\star}}^{p^{\star}}(-)$. The proof that $(-)^{p^{\star}}\mathscr{G}$ is right exact is similar, and left to the reader

6.5. Truncation of completed perverse sheaves. Recall that the monodromy morphism for the right action of T on $\widetilde{\mathrm{Fl}}_G$ (see §6.1) provides for any \mathscr{F} in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ a canonical \Bbbk -algebra morphism

$$\mathscr{O}(T^{\vee}_{\Bbbk}) \to \operatorname{End}_{\mathsf{P}^{\wedge}_{\operatorname{In},\operatorname{In}}}(\mathscr{F}),$$

which is compatible in the obvious way with all morphisms in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$; in the language of §B.1, this means that the monodromy constructions provides a functor

$$(6.15) \qquad \qquad \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{Mod}\left(\mathscr{O}(T^{\vee}_{\Bbbk}),\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\right)$$

whose composition with the obvious forgetful functor

$$\mathsf{Mod}\left(\mathscr{O}(T^{\vee}_{\Bbbk}),\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\right)\to\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

is the identity. Composing (6.15) with the bifunctor (B.2), we therefore obtain a bifunctor

$$(-) \otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})} (-) : \mathsf{Mod}^{\mathrm{fg}}(\mathscr{O}(T^{\vee}_{\Bbbk})) \times \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}.$$

We will denote by $\mathcal{J} \subset \mathscr{O}(T^{\vee}_{\Bbbk}/W_{\mathrm{f}})$ the ideal of the image of $e \in T^{\vee}_{\Bbbk}$ in $T^{\vee}_{\Bbbk}/W_{\mathrm{f}}$, and for $m \geq 1$ we set

$$(T^{\vee}_{\Bbbk})^{(m)} := \operatorname{Spec}(\mathscr{O}(T^{\vee}_{\Bbbk})/\mathcal{J}^m \cdot \mathscr{O}(T^{\vee}_{\Bbbk})).$$

Below we will make use of the "truncation" functor

$$\mathsf{C}_m := \mathscr{O}((T^{\vee}_{\Bbbk})^{(m)}) \otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})} (-) : \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}.$$

In view of Lemma 6.2, this functor in fact takes values in P_{I_u,I_u} , which allows us to also consider the composition

$$\mathsf{C}_m^0 := \Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \circ \mathsf{C}_m : \mathsf{P}^\wedge_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

6.6. The case of the finite flag variety. Recall that the action of G on the base point of $\widetilde{\mathrm{Fl}}_G$ provides a canonical closed embedding $G/U \hookrightarrow \widetilde{\mathrm{Fl}}_G$, which identifies G/U with the closure of $\widetilde{\mathrm{Fl}}_{G,w_o}$, where $w_o \in W_{\mathrm{f}}$ is the longest element (or, in other words, the union of the orbits $\widetilde{\mathrm{Fl}}_{G,w}$ with $w \in W_{\mathrm{f}}$). Under this identification, the action of I_{u} on the closure of $\widetilde{\mathrm{Fl}}_{G,w_o}$ corresponds to the action on G/U induced by the natural U-action (by left multiplication) via the projection $\mathrm{I}_{\mathrm{u}} \to U$.

The same construction as for D_{I_u,I_u} and $D^{\wedge}_{I_u,I_u}$ above provides a category $D_{U,U}$ of sheaves on G/U, and a "completed" triangulated category $D^{\wedge}_{U,U}$, with a monoidal product $\hat{\star}_U$. There is also a perverse t-structure on $D^{\wedge}_{U,U}$, whose heart will be denoted $P^{\wedge}_{U,U}$, and a notion of tilting perverse sheaves; the full subcategory of $P^{\wedge}_{U,U}$ whose objects are the tilting perverse sheaves will be denoted $T^{\wedge}_{U,U}$. The pushforward functor associated with the closed embedding $G/U \hookrightarrow \widetilde{Fl}_G$ provides a t-exact, monoidal, fully faithful functor

$$(\mathsf{D}^{\wedge}_{U,U}, \widehat{\star}_U) \to (\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}, \widehat{\star}),$$

which sends tilting perverse sheaves to tilting perverse sheaves. (This functor will usually be omitted from notation; similarly, objects of $\mathsf{D}^{\wedge}_{\mathbf{I}_u,\mathbf{I}_u}$ supported on G/U will be considered as objects in $\mathsf{D}^{\wedge}_{U,U}$ whenever convenient.) The tilting perverse sheaves in the essential image of this functor are those which are direct sums of objects \mathscr{T}^{\wedge}_w with $w \in W_{\mathrm{f}}$.

The study of the category $\mathsf{T}^{\wedge}_{U,U}$ is the main subject of [BR1]. In the course of this study, we in particular construct explicit representatives for the objects $\mathscr{T}^{\wedge}_{w_{\circ}}$ and \mathscr{T}^{\wedge}_{s} $(s \in S_{\mathrm{f}})$, as follows. Let us denote by U^{+} the unipotent radical of the Borel subgroup of G opposite to B with respect to T, so that the T-weights in the Lie algebra of U^{+} are the positive roots. Fix, for any $s \in S_{\mathrm{f}}$, a trivialization of the root subgroup $U_{s} \subset U^{+}$ associated with the simple root corresponding to s. Then we obtain a group morphism $\chi: U^{+} \to \mathbb{G}_{\mathrm{a}}$ as the following composition:

$$U^+ \to U^+ / \mathscr{D}(U^+) \xleftarrow{\sim} \prod_{s \in S_{\mathrm{f}}} U_s \cong \prod_{s \in S_{\mathrm{f}}} \mathbb{G}_{\mathrm{a}} \xrightarrow{+} \mathbb{G}_{\mathrm{a}}.$$

(Here we denote by $\mathscr{D}(U^+)$ the derived subgroup of U^+ .) Let us also fix a nontrivial *p*-th root of unity in \Bbbk , and denote by \mathscr{L}_{AS} the associated Artin–Schreier local system on \mathbb{G}_a . Then as in [BR1, §10.3] one can consider the $(U^+, \chi^*(\mathscr{L}_{AS}))$ equivariant derived category of \Bbbk -sheaves on G/B, which will be denoted $\mathsf{D}_{\mathrm{Wh},B}$,

and the full triangulated subcategory $\mathsf{D}_{\mathrm{Wh},U}$ in the $(U^+, \chi^*(\mathscr{L}_{\mathrm{AS}}))$ -equivariant derived category of k-sheaves on G/U generated by complexes obtain by pullback from $\mathsf{D}_{\mathrm{Wh},B}$. We have "averaging" functors

 $\mathsf{Av}_{\mathrm{Wh}}:\mathsf{D}_{U,U}\to\mathsf{D}_{\mathrm{Wh},U},\qquad\mathsf{Av}_{U,*},\mathsf{Av}_{U,!}:\mathsf{D}_{\mathrm{Wh},U}\to\mathsf{D}_{U,U},$

which form adjoint pairs $(Av_{U,!}, Av_{Wh})$ and $(Av_{Wh}, Av_{U,*})$. One can also consider a "completed" category $D^{\wedge}_{Wh,U}$ using the same procedure as above (based on [BY]), and the averaging functors induce triangulated functors

$$\mathsf{Av}_{\mathrm{Wh}}:\mathsf{D}^{\wedge}_{U,U}\to\mathsf{D}^{\wedge}_{\mathrm{Wh},U},\qquad\mathsf{Av}_{U,*},\mathsf{Av}_{U,!}:\mathsf{D}^{\wedge}_{\mathrm{Wh},U}\to\mathsf{D}^{\wedge}_{U,U},$$

which have the same adjointness properties as above.

We set

$$\Xi_!^{\wedge} := \mathsf{Av}_{U,!} \circ \mathsf{Av}_{\mathrm{Wh}}(\delta_e^{\wedge}), \qquad \Xi_*^{\wedge} := \mathsf{Av}_{U,*} \circ \mathsf{Av}_{\mathrm{Wh}}(\delta_e^{\wedge}).$$

It is proved in [BR1, Lemma 10.1] (following standard arguments taken from [BBM, BY]) that $\Xi_{!}^{\wedge}$ and Ξ_{*}^{\wedge} are (noncanonically) isomorphic, and representatives for the object $\mathscr{T}_{w_{\circ}}^{\wedge}$. We will also set

$$\Xi_! := \pi_{\dagger}(\Xi_!^{\wedge}), \quad \Xi_* := \pi_{\dagger}(\Xi_*^{\wedge}),$$

so that we have $\Xi_! \cong \Xi_* \cong \mathscr{T}_{w_{\circ}}$.

In the following lemma we gather some standard properties of this object that will be used later.

Lemma 6.10. For $w \in W_f$ and $n \in \mathbb{Z}$ we have

$$\operatorname{Hom}_{\mathsf{D}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge}, \pi^{\dagger}\mathsf{For}_{\mathrm{I}_{u}}^{\mathrm{I}}(\mathscr{I}_{w})[n]) \cong \begin{cases} \mathbb{k} & \text{if } w = e \text{ and } n = 0; \\ 0 & \text{otherwise.} \end{cases}$$

In particular, the object Ξ_1^{\wedge} is projective in $\mathsf{P}_{U,U}^{\wedge}$.

Proof. Let us denote by $\mathsf{D}_{U,B}$ the *U*-equivariant derived category of constructible \Bbbk -sheaves on G/B, and by $\mathsf{P}_{U,B}$ the heart of its perverse t-structure. Then the realization functor $D^{\mathrm{b}}\mathsf{P}_{U,B} \to \mathsf{D}_{U,B}$ is an equivalence by the formalism of highest weight categories, see [BGS], and $\pi_{\dagger}(\Xi_{!}^{\wedge}) \cong \mathscr{T}_{w_{\circ}}$ is the projective cover of $\mathsf{For}_{\mathrm{I}_{u}}^{\mathrm{I}}(\mathscr{I}_{e})$ in $\mathsf{P}_{U,B}$. We deduce the desired isomorphism using adjunction. This implies in particular that

$$\operatorname{Hom}_{\mathsf{D}_{U}^{\wedge}}(\Xi^{\wedge}_{!},\mathscr{F})=0$$

for any \mathscr{F} in ${}^{\mathrm{p}}\mathsf{D}_{U,U}^{\leq -1}$. By Proposition D.4, any object \mathscr{F} in ${}^{\mathrm{p}}\mathsf{D}_{U,U}^{\wedge,\leq -1}$ can be written as " \lim_{n} " \mathscr{F}_n where each \mathscr{F}_n belongs to ${}^{\mathrm{p}}\mathsf{D}_{U,U}^{\leq -1}$; we then have

$$\operatorname{Hom}_{\mathsf{D}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge},\mathscr{F}) = \varprojlim_{n} \operatorname{Hom}_{\mathsf{D}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge},\mathscr{F}_{n}) = 0,$$

which proves that $\Xi_{!}^{\wedge}$ is projective in $\mathsf{P}_{U,U}^{\wedge}$.

Using the fact that convolution on the left and on the right commute with each other, one sees that the functors

$$\Xi^{\wedge}_{!} \widehat{\star}_{U} (-), \Xi^{\wedge}_{*} \widehat{\star}_{U} (-) : \mathsf{D}^{\wedge}_{U,U} \to \mathsf{D}^{\wedge}_{U,U}$$

are canonically isomorphic to the functors $Av_{U,!} \circ Av_{Wh}$ and $Av_{U,*} \circ Av_{Wh}$ respectively. In particular, the adjunction morphisms for the pairs $(Av_{U,!}, Av_{Wh})$ and $(Av_{Wh}, Av_{U,*})$ provide canonical morphisms

(6.16)
$$\Xi_{!}^{\wedge} \widehat{\star}_{U} \Xi_{*}^{\wedge} \to \delta_{e}^{\wedge}, \qquad \delta_{e}^{\wedge} \to \Xi_{*}^{\wedge} \widehat{\star}_{U} \Xi_{!}^{\wedge}$$

which satisfy the appropriate zigzag relations, so that the functor $\Xi_{!}^{\wedge} \widehat{\star}_{U}(-)$ is left adjoint to $\Xi_{*}^{\wedge} \widehat{\star}_{U}(-)$. The same holds for any category which admits an action of $\mathsf{D}_{U,U}^{\wedge}$, e.g. $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\wedge}$.

Lemma 6.11. The functor

$$\Xi^{\wedge}_{!} \mathbin{\widehat{\star}} (-) : \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

is t-exact for the perverse t-structure.

Proof. The subcategory ${}^{\mathrm{p}}\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\wedge,\leq 0}$ is generated under extensions by the objects $\Delta_{w}^{\wedge}[n]$ for $w \in W$ and $n \in \mathbb{Z}_{\geq 0}$, see [BR1, Lemma 5.6]. Now $\Xi_{1}^{\wedge} \widehat{\star} \Delta_{w}^{\wedge}$ is perverse for any w by Lemma 6.3(3), since Ξ_{1}^{\wedge} admits a costandard filtration. This implies that our functor is right t-exact. As explained above this functor is isomorphic to $\Xi_{*}^{\wedge} \widehat{\star}(-)$, which admits as a left adjoint the right t-exact functor $\Xi_{1}^{\wedge} \widehat{\star}(-)$. It is therefore left t-exact, hence finally t-exact.

Remark 6.12. One can check (using essentially the same arguments) that in fact the functor $\mathscr{F} \stackrel{\sim}{\star} (-)$ is t-exact for any \mathscr{F} in $\mathsf{T}^{\wedge}_{\mathrm{I}_{n},\mathrm{I}_{n}}$.

If now $s \in S_{\mathbf{f}}$ is a simple reflection in $W_{\mathbf{f}}$, let us denote by $\bar{\imath}_s$ the embedding of the closure of $\widetilde{\mathrm{Fl}}_{G,s}$ in $\widetilde{\mathrm{Fl}}_G$. We set

$$\Xi_{s,!}^{\wedge} := (\bar{\imath}_s)_* (\bar{\imath}_s)^* \Xi_!^{\wedge}, \qquad \Xi_{s,*}^{\wedge} := (\bar{\imath}_s)_* (\bar{\imath}_s)^! \Xi_*^{\wedge}.$$

As explained in [BR1, Remark 9.5], these objects are (noncanonically) isomorphic, and are representatives for the object \mathscr{T}_s^{\wedge} . Moreover, there exists a surjection $\Xi_{!}^{\wedge} \to \Xi_{s,!}^{\wedge}$ whose kernel admits a standard filtration, and there exists an embedding $\Xi_{s,*}^{\wedge} \to \Xi_{*}^{\wedge}$ whose cokernel admits a costandard filtration.

7. Free-monodromic central sheaves and Wakimoto sheaves

As in Sections 5-6 we continue with the general setting of $\S4.1$.

7.1. Free-monodromic Wakimoto sheaves – definition. Recall the Wakimoto sheaves mentioned in §4.3. In this subsection we explain (following [B2]) how these objects can be "lifted" to the completed category $D^{\wedge}_{I_{I_1},I_{I_2}}$.

Recall that for $\lambda \in X_*(T)$ we have a point $z^{\lambda} \in LG(\mathbb{F})$ whose image in $\widetilde{\mathrm{Fl}}_G$ belongs to $\widetilde{\mathrm{Fl}}_{G,\mathfrak{t}(\lambda)}$. This point defines a point in the quotient $\mathrm{I_u}\setminus\widetilde{\mathrm{Fl}}_{G,w}$, hence a trivialization of this *T*-torsor. In case $w \in X_*(T) \subset W$, the objects Δ_w^{\wedge} and ∇_w^{\wedge} therefore admit *canonical* representatives $\Delta_w^{\wedge,\mathrm{can}}$ and $\nabla_w^{\wedge,\mathrm{can}}$, defined using this trivialization. With this definition, we have canonical isomorphisms

(7.1)
$$\pi_{\dagger}(\Delta_{w}^{\wedge,\operatorname{can}}) \cong \operatorname{For}_{\operatorname{I}_{\mathrm{u}}}^{\mathrm{I}}(\Delta_{w}^{\mathrm{I}}), \quad \pi_{\dagger}(\nabla_{w}^{\wedge,\operatorname{can}}) \cong \operatorname{For}_{\operatorname{I}_{\mathrm{u}}}^{\mathrm{I}}(\nabla_{w}^{\mathrm{I}}).$$

We will also denote by $p_w^{\operatorname{can}} : \operatorname{Fl}_{G,w} \to T$ the associated morphism (for $w \in X_*(T)$).

Lemma 7.1. If $\lambda, \mu \in X^+_*(T)$, then there exist canonical isomorphisms

$$\Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda)} \mathbin{\widehat{\star}} \Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)} \xrightarrow{\sim} \Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\mu)}, \quad \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda)} \mathbin{\widehat{\star}} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)} \xrightarrow{\sim} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\mu)}.$$

Moreover these isomorphisms are compatible with associativity, in the sense that for $\lambda, \mu, \nu \in X^+_*(T)$ the two natural isomorphisms between

$$\Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda)} \mathbin{\widehat{\star}} \Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)} \mathbin{\widehat{\star}} \Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(\nu)} \quad and \quad \Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\mu+\nu)},$$

resp. between

$$\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda)} \mathbin{\widehat{\star}} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)} \mathbin{\widehat{\star}} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\nu)} \quad and \quad \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\mu+\nu)},$$

which can be constructed by combining these isomorphisms coincide.

Proof. The proof is similar to that of [B2, Lemma 4].

Namely, let us first consider the case of the objects $\Delta_{t(\lambda)}^{\wedge, \text{can}}$. Recall that for $\lambda, \mu \in X^+_*(T)$ we have $\ell(t(\lambda)) + \ell(t(\mu)) = \ell(t(\lambda + \mu))$; the morphism *m* therefore induces an isomorphism

(7.2)
$$\operatorname{Fl}_{G,t(\lambda)} \approx \operatorname{Fl}_{G,t(\mu)} \xrightarrow{\sim} \operatorname{Fl}_{G,t(\lambda+\mu)}$$

Let us denote by

$$\widetilde{m}_{\lambda,\mu}:\widetilde{\mathrm{Fl}}_{G,\mathsf{t}(\lambda)}\times\widetilde{\mathrm{Fl}}_{G,\mathsf{t}(\mu)}\to\widetilde{\mathrm{Fl}}_{G,\mathsf{t}(\lambda+\mu)}$$

the morphism induced by \widetilde{m} , by

$$\widetilde{\jmath}_{\mathsf{t}(\lambda)} \widetilde{\times} \widetilde{\jmath}_{\mathsf{t}(\mu)} : \widetilde{\mathrm{Fl}}_{G,\mathsf{t}(\lambda)} \widetilde{\times} \widetilde{\mathrm{Fl}}_{G,\mathsf{t}(\mu)} \to \widetilde{\mathrm{Fl}}_G \widetilde{\times} \widetilde{\mathrm{Fl}}_G$$

the (locally closed) embedding, and by

$$p_{\mathsf{t}(\lambda)}^{\mathrm{can}} \widetilde{\times} p_{\mathsf{t}(\mu)}^{\mathrm{can}} : \mathrm{Fl}_{G,\mathsf{t}(\lambda)} \widetilde{\times} \mathrm{Fl}_{G,\mathsf{t}(\mu)} \to T \times T$$

the morphism induced by $p_{t(\lambda)}^{can}$ and $p_{t(\mu)}^{can}$. These morphisms fit in a diagram

$$\begin{array}{c|c} \widetilde{\mathrm{Fl}}_{G, \mathfrak{t}(\lambda)} \widetilde{\times} \widetilde{\mathrm{Fl}}_{G, \mathfrak{t}(\mu)} & \xrightarrow{\widetilde{m}_{\lambda, \mu}} & \widetilde{\mathrm{Fl}}_{G, \mathfrak{t}(\lambda + \mu)} \\ p_{\mathfrak{t}(\lambda)}^{\mathrm{can}} \widetilde{\times} p_{\mathfrak{t}(\mu)}^{\mathrm{can}} & \bigvee \\ T \times T & \xrightarrow{m_T} & T \end{array}$$

(where the lower horizontal arrow is multiplication in T), which is easily seen to be commutative. Moreover, the isomorphism (7.2) implies that this diagram is cartesian.

Using the isomorphism

$$\widetilde{m}_! \circ (\widetilde{\jmath}_{\mathsf{t}(\lambda)} \times \widetilde{\jmath}_{\mathsf{t}(\mu)})_! \cong (\widetilde{\jmath}_{\mathsf{t}(\lambda+\mu)})_! \circ (\widetilde{m}_{\lambda,\mu})_!$$

and the definition of $\widehat{\star}$ we obtain an isomorphism

$$\Delta_{\mathsf{t}(\lambda)}^{\wedge,\operatorname{can}} \,\widehat{\star}\, \Delta_{\mathsf{t}(\mu)}^{\wedge,\operatorname{can}} \cong \\ \underset{m}{\overset{\text{"lim}}{\longleftarrow}} \, \overset{\text{"lim}}{\underset{n}{\overset{\text{"lim}}{\longrightarrow}}} \, \overset{\text{"lim}}{(\widetilde{\jmath}_{\mathsf{t}(\lambda+\mu)})_{!}} (\widetilde{m}_{\lambda,\mu})_{!} (p_{\mathsf{t}(\lambda)}^{\operatorname{can}} \,\widetilde{\times}\, p_{\mathsf{t}(\mu)}^{\operatorname{can}})^{*} (\mathscr{L}_{T,n} \boxtimes \mathscr{L}_{T,m}) [\ell(\mathsf{t}(\lambda+\mu)) + 3\dim(T)].$$

Using the base change theorem we deduce an isomorphism

$$\Delta_{\mathsf{t}(\lambda)}^{\wedge,\mathrm{can}} \mathbin{\widehat{\star}} \Delta_{\mathsf{t}(\mu)}^{\wedge,\mathrm{can}} \cong \underset{m}{\overset{\underset{m}{\leftarrow}}{\underset{n}{\underset{n}{\overset{m}{\leftarrow}}}} (\widetilde{\jmath}_{\mathsf{t}(\lambda+\mu)})_! (p_{\mathsf{t}(\lambda+\mu)}^{\mathrm{can}})^* (m_T)_! (\mathscr{L}_{T,n} \boxtimes \mathscr{L}_{T,m}) [\ell(\mathsf{t}(\lambda+\mu)) + 3\dim(T)].$$

Now by [BR1, Lemma 3.4], for any m there is a canonical isomorphism

$$\lim_{n \to \infty} (m_T)!(\mathscr{L}_{T,n} \boxtimes \mathscr{L}_{T,m}) \cong \mathscr{L}_{T,m}[-2\dim(T)],$$

which provides the desired isomorphism. The verification that this isomorphism is compatible with associativity in the sense above is straightforward, and left to the reader.

The proof for the objects $\nabla_{t(\lambda)}^{\wedge,can}$ will be similar, once we construct a canonical isomorphism

$$\widetilde{m}_! \circ (\widetilde{\jmath}_{\mathsf{t}(\lambda)} \times \widetilde{\jmath}_{\mathsf{t}(\mu)})_* \mathscr{F} \xrightarrow{\sim} (\widetilde{\jmath}_{\mathsf{t}(\lambda+\mu)})_* \circ (\widetilde{m}_{\lambda,\mu})_! \mathscr{F}$$

for each \mathscr{F} of the form $(p_{t(\lambda)}^{\operatorname{can}} \times p_{t(\mu)}^{\operatorname{can}})^* (\mathscr{L}_{T,n} \boxtimes \mathscr{L}_{T,m})$ with $n, m \geq 0$. We obtain a morphism from the left-hand side to the right-hand side as the composition

$$\begin{split} \widetilde{m}_!(\widetilde{\jmath}_{\mathsf{t}(\lambda)} \widetilde{\times} \widetilde{\jmath}_{\mathsf{t}(\mu)})_* \mathscr{F} &\to (\widetilde{\jmath}_{\mathsf{t}(\lambda+\mu)})_* (\widetilde{\jmath}_{\mathsf{t}(\lambda+\mu)})^* \widetilde{m}_! (\widetilde{\jmath}_{\mathsf{t}(\lambda)} \widetilde{\times} \widetilde{\jmath}_{\mathsf{t}(\mu)})_* \mathscr{F} \\ & \xrightarrow{\sim} (\widetilde{\jmath}_{\mathsf{t}(\lambda+\mu)})_* (\widetilde{m}_{\lambda,\mu})_! (\widetilde{\jmath}_{\mathsf{t}(\lambda)} \widetilde{\times} \widetilde{\jmath}_{\mathsf{t}(\mu)})^* (\widetilde{\jmath}_{\mathsf{t}(\lambda)} \widetilde{\times} \widetilde{\jmath}_{\mathsf{t}(\mu)})_* \mathscr{F} \to (\widetilde{\jmath}_{\mathsf{t}(\lambda+\mu)})_* (\widetilde{m}_{\lambda,\mu})_! \mathscr{F} \end{split}$$

where the first and third morphisms are induced by adjunction, and the middle isomorphism is given by the base change theorem. Since the objects we have to consider are all extensions of constant local systems, to prove that this morphism is invertible on these objects it suffices to prove this property when \mathscr{F} is constant. In this case, Lemma 5.2 shows that the left-hand side identifies with

$$(\widetilde{j}_{\mathsf{t}(\lambda+\mu)})_* \underline{\Bbbk} \otimes_{\Bbbk} \mathsf{H}^{\bullet}_c(T, \Bbbk),$$

and the same holds for the right-hand side since $\tilde{m}_{\lambda,\mu}$ is a trivial *T*-torsor. It is easily seen that our morphism identifies with the identity of this objects, which concludes the proof.

This lemma will allow us to define the *free-monodromic Wakimoto sheaves* as follows. Given $\lambda \in X_*(T)$, for \mathscr{F} in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and $\mu, \nu \in X^+_*(T)$ and such that $\lambda = \mu - \nu$ we consider the k-vector space

$$\operatorname{Hom}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\nabla^{\wedge,\operatorname{can}}_{\mathsf{t}(\mu)},\nabla^{\wedge,\operatorname{can}}_{\mathsf{t}(\nu)}\mathbin{\widehat{\star}}\mathscr{F}).$$

If (μ', ν') is another pair of elements of $X^+_*(T)$ such that $\lambda = \mu' - \nu'$, and if $\mu' - \mu \in X^+_*(T)$, then there is a canonical isomorphism

$$\operatorname{Hom}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)},\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\nu)}\mathbin{\widehat{\star}}\mathscr{F})\xrightarrow{\sim}\operatorname{Hom}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu')},\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\nu')}\mathbin{\widehat{\star}}\mathscr{F})$$

induced by left convolution with $\nabla_{t(\mu'-\mu)}^{\wedge, \operatorname{can}}$ and the isomorphisms of Lemma 7.1. Moreover, the compatibility with associativity implies that this collection of spaces and isomorphisms is an inductive system for the order on pairs (μ, ν) of elements of $X_*^+(T)$ such that $\lambda = \mu - \nu$ given by $(\mu, \nu) \leq (\mu', \nu')$ iff $\mu' - \mu \in X_*^+(T)$. One can therefore consider the functor

$$\mathscr{F} \mapsto \varinjlim_{(\mu,\nu)} \operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\nabla_{\mathsf{t}(\mu)}^{\wedge,\mathrm{can}},\nabla_{\mathsf{t}(\nu)}^{\wedge,\mathrm{can}}\,\widehat{\star}\,\mathscr{F}).$$

This functor is representable, since each transition morphism is an isomorphism and for any given choice of pair (μ, ν) we have an isomorphism

$$\operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\nabla_{\mathsf{t}(\mu)}^{\wedge,\mathrm{can}},\nabla_{\mathsf{t}(\nu)}^{\wedge,\mathrm{can}}\,\widehat{\star}\,\mathscr{F})\cong\operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\Delta_{\mathsf{t}(-\nu)}^{\wedge,\mathrm{can}}\,\widehat{\star}\,\nabla_{\mathsf{t}(\mu)}^{\wedge,\mathrm{can}},\mathscr{F})$$

see Lemma 6.3. One can therefore define $\mathscr{W}^{\wedge}_{\lambda}$ as the object representing this functor. From the construction, we see that for any pair $(\mu, \nu) \in (X^+_*(T))^2$ such that $\lambda = \mu - \nu$ we have a *noncanonical* isomorphism

$$\mathscr{W}^\wedge_\lambda\cong\Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(-\nu)}\mathbin{\widehat{\star}}\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)}.$$

Using Lemma 6.3, it is also not difficult to check that the right-hand side is (again, noncanonically) isomorphic to $\nabla_{t(\mu)}^{\wedge, can} \stackrel{\sim}{\star} \Delta_{t(-\nu)}^{\wedge, can}$.

7.2. Free-monodromic Wakimoto sheaves – properties. The following statement gathers the main properties of free-monodromic Wakimoto sheaves that we will need below.

Lemma 7.2. (1) If $\lambda \in X_*(T)$, and if $\mu \in X_*^+(T)$ is such that $\lambda + \mu \in X_*^+(T)$, then there exists a canonical isomorphism

$$\mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)} \cong \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\mu)}.$$

(2) For $\lambda, \mu \in X_*(T)$ and $n \in \mathbb{Z}$, we have

$$\operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\mathscr{W}_{\lambda}^{\wedge},\mathscr{W}_{\mu}^{\wedge}[n])=0$$

unless $\mu \leq \lambda$.

(3) For any $\lambda \in X_*(T)$, we have a canonical isomorphism

$$\pi_{\dagger}(\mathscr{W}_{\lambda}^{\wedge}) \cong \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{W}_{\lambda}).$$

Moreover, $\mathscr{W}^{\wedge}_{\lambda}$ is a perverse sheaf. (4) For $\lambda, \mu \in X_*(T)$ there exists a canonical isomorphism

$$\mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \mathscr{W}^{\wedge}_{\mu} \cong \mathscr{W}^{\wedge}_{\lambda+\mu}$$

(5) For any $\lambda \in X_*(T)$, the restriction of $\mu_{\mathscr{W}^{\wedge}_{\lambda}}$ to

$$\mathscr{O}(T^{\vee}_{\Bbbk}) = \Bbbk \otimes \mathscr{O}(T^{\vee}_{\Bbbk}) \subset \mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk})$$

factors (in the canonical way) through an isomorphism

$$\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge}) \xrightarrow{\sim} \operatorname{End}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{W}^{\wedge}_{\lambda}),$$

and we have

$$\operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{W}_{\lambda}^{\wedge},\mathscr{W}_{\lambda}^{\wedge}[1])=0.$$

(6) For any $\lambda \in X_*(T)$ and $f \in \mathscr{O}(T^{\vee}_{\Bbbk})$ we have

$$\mu_{\mathscr{W}^{\wedge}}(f \otimes 1) = \mu_{\mathscr{W}^{\wedge}}(1 \otimes f),$$

and moreover

$$\mu^{\mathrm{rot}}_{\mathscr{W}^{\wedge}_{\lambda}}(x) = \mu_{\mathscr{W}^{\wedge}_{\lambda}}(1 \otimes e^{-\lambda}).$$

Proof. (1) The proof is identical to that of its counterpart in $D_{I,I}$, see [AR4, Lemma 4.2.5].

(2) Let $\nu \in X^+_*(T)$ be such that $\lambda + \nu \in X^+_*(T)$ and $\mu + \nu \in X^+_*(T)$. Since the functor $(-) \hat{\star} \nabla^{\wedge, \text{can}}_{\mathsf{t}(\nu)}$ is an equivalence of categories (see Remark 6.4), using (1) we obtain an isomorphism

$$\operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{W}^{\wedge}_{\lambda},\mathscr{W}^{\wedge}_{\mu}[n]) \cong \operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\nu)},\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu+\nu)}[n]).$$

Now the right-hand side vanishes unless $\operatorname{Fl}_{G,\mathfrak{t}(\mu+\nu)} \subset \overline{\operatorname{Fl}_{G,\mathfrak{t}(\lambda+\nu)}}$. It is well known that this condition is satisfied if and only if $\mu + \nu \preceq \lambda + \nu$, which implies the claim.

(3) If $\mu \in X^+_*(T)$ is such that $\lambda + \mu \in X^+_*(T)$, then by (1) we have a canonical isomorphism

$$\mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)} \cong \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\mu)}.$$

Now we have

$$\pi_{\dagger}(\mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \nabla^{\wedge, \mathrm{can}}_{\mathsf{t}(\mu)}) \stackrel{(\boldsymbol{6}, \boldsymbol{4})}{\cong} \mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \pi_{\dagger}(\nabla^{\wedge, \mathrm{can}}_{\mathsf{t}(\mu)}) \stackrel{(\boldsymbol{7}, 1)}{\cong} \mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\nabla^{\mathrm{I}}_{\mathsf{t}(\mu)})$$
$$\stackrel{(\boldsymbol{6}, \boldsymbol{5})}{\cong} \pi_{\dagger}(\mathscr{W}^{\wedge}_{\lambda}) \star_{\mathrm{I}} \nabla^{\mathrm{I}}_{\mathsf{t}(\mu)}.$$
Using again (7.1), we deduce a canonical isomorphism

$$\pi_{\dagger}(\mathscr{W}^{\wedge}_{\lambda}) \star_{\mathrm{I}} \nabla^{\mathrm{I}}_{\mathsf{t}(\mu)} \cong \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\nabla^{\mathrm{I}}_{\mathsf{t}(\lambda+\mu)}).$$

By [AR4, Lemma 4.2.7] the right-hand side is canonically isomorphic to $\mathsf{For}_{I_u}^{I}(\mathscr{W}_{\lambda})\star_{I}$ $\nabla_{t(\mu)}^{I}$. Since the functor $(-)\star_{I}\nabla_{t(\mu)}^{I}$ is an equivalence of categories (with quasiinverse $(-)\star_{I}\Delta_{t(-\mu)}^{I}$), this defines an isomorphism

$$\pi_{\dagger}(\mathscr{W}^{\wedge}_{\lambda}) \cong \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{W}_{\lambda}).$$

One can easily check that this isomorphism does not depend on the choice of μ , hence is indeed canonical. Finally, since \mathscr{W}_{λ} is perverse this isomorphism implies that $\mathscr{W}_{\lambda}^{\wedge}$ is perverse by [BR1, Lemma 5.3(1)].

(4) If $\nu \in X^+_*(T)$ is such that $\mu + \nu \in X^+_*(T)$ and $\lambda + \mu + \nu \in X^+_*(T)$, then using (1) we obtain canonical isomorphisms

$$\mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \mathscr{W}^{\wedge}_{\mu} \mathbin{\widehat{\star}} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\nu)} \cong \mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu+\nu)} \cong \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\mu+\nu)} \cong \mathscr{W}^{\wedge}_{\lambda+\mu} \mathbin{\widehat{\star}} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\nu)}$$

Since the functor $(-) \stackrel{\sim}{\star} \nabla^{\wedge, \text{can}}_{t(\nu)}$ is an equivalence of categories (see Remark 6.4), we deduce an isomorphism

$$\mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \mathscr{W}^{\wedge}_{\mu} \cong \mathscr{W}^{\wedge}_{\lambda+\mu}$$

It can be checked that this isomorphism does not depend on ν , hence is indeed canonical.

(5) Let $\mu \in X^+_*(T)$ be such that $\lambda + \mu \in X^+_*(T)$. Then as above, using (1) we obtain that the functor $(-) \stackrel{\sim}{\star} \nabla^{\wedge, \text{can}}_{\mathsf{t}(\mu)}$ induces an isomorphism

$$\operatorname{Hom}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{W}^{\wedge}_{\lambda},\mathscr{W}^{\wedge}_{\lambda}[n]) \xrightarrow{\sim} \operatorname{Hom}_{\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\mu)},\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\mu)}[n])$$

for any $n \in \mathbb{Z}$. When n = 1, it follows from [BR1, Corollary 4.6] and adjunction that the right-hand side vanishes, so that the left-hand side vanishes as well. For n = 0, it follows from (6.7)–(6.8) and Lemma 6.6 that the composition of this isomorphism with $\mu_{\mathcal{W}^{\wedge}_{\lambda}}$ is $\mu_{\nabla^{\wedge, \operatorname{can}}_{t(\lambda+\mu)}}$; the desired claim therefore follows from Lemma 6.6.

(6) Let us fix $\mu, \nu \in X^+_*(T)$ such that $\lambda = \mu - \nu$, and an isomorphism $\mathscr{W}^{\wedge}_{\lambda} \cong \Delta^{\wedge, \operatorname{can}}_{\mathfrak{t}(-\nu)} \widehat{\star} \nabla^{\wedge, \operatorname{can}}_{\mathfrak{t}(\mu)}$. Identifying these objects via this isomorphism and using (6.7)–(6.8) and Lemma 6.6 we obtain that for $f \in \mathscr{O}(T^{\vee}_{\Bbbk})$ we have

$$\begin{split} \mu_{\mathscr{W}^{\wedge}_{\lambda}}(f\otimes 1) &= \mu_{\Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(-\nu)}}(f\otimes 1)\,\widehat{\star}\,\mathrm{id} = \mu_{\Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(-\nu)}}(1\otimes f)\,\widehat{\star}\,\mathrm{id} \\ &= \mathrm{id}\,\widehat{\star}\,\mu_{\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)}}(f\otimes 1) = \mathrm{id}\,\widehat{\star}\,\mu_{\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)}}(1\otimes f) = \mu_{\mathscr{W}^{\wedge}_{\lambda}}(1\otimes f), \end{split}$$

which proves the first claim. To prove the second claim, we observe that by (6.7)-(6.8)-(6.9) and Lemma 6.6 we have

$$\begin{split} \mu^{\mathrm{rot}}_{\mathscr{W}^{\wedge}_{\lambda}}(x) &= \mu^{\mathrm{rot}}_{\Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(-\nu)}}(x) \mathbin{\widehat{\star}} \mu^{\mathrm{rot}}_{\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)}}(x) = \mu_{\Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(-\nu)}}(1 \otimes e^{\nu}) \mathbin{\widehat{\star}} \mu_{\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)}}(1 \otimes e^{-\mu}) \\ &= \mathrm{id} \mathbin{\widehat{\star}} \mu_{\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)}}(1 \otimes e^{-\lambda}) = \mu_{\mathscr{W}^{\wedge}_{\lambda}}(1 \otimes e^{-\lambda}), \end{split}$$

which finishes the proof.

Below we will also need the following property, which has no counterpart in the nonmonodromic setting.

Lemma 7.3. If $\lambda, \mu \in X_*(T)$ are distinct, then we have $\operatorname{Hom}_{\mathsf{P}^{\wedge}_{\mathrm{In},\mathrm{In}}}(\mathscr{W}^{\wedge}_{\lambda}, \mathscr{W}^{\wedge}_{\mu}) = 0.$

Proof. Let $\nu \in -X^+_*(T)$ be such that $\lambda + \nu$ and $\mu + \nu$ belong to $-X^+_*(T)$. Then we have

$$\mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(\nu)} \cong \mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \mathscr{W}^{\wedge}_{\nu} \cong \mathscr{W}^{\wedge}_{\lambda+\nu} \cong \Delta^{\wedge,\mathrm{can}}_{\mathsf{t}(\lambda+\nu)},$$

and similarly for μ . Since the functor $(-) \widehat{\star} \Delta_{t(\nu)}^{\wedge, can}$ is an equivalence of categories (see Remark 6.4), we deduce an isomorphism

$$\operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\mathscr{W}_{\lambda}^{\wedge},\mathscr{W}_{\mu}^{\wedge}) \cong \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\Delta_{\mathsf{t}(\lambda+\nu)}^{\wedge,\mathrm{can}},\Delta_{\mathsf{t}(\mu+\nu)}^{\wedge,\mathrm{can}}).$$

The claim then follows from (6.12).

7.3. Wakimoto filtrations of free-monodromic perverse sheaves. The notion of objects admitting a Wakimoto filtration (see §4.3) has an obvious analogue in the category $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$: we will say that an object \mathscr{F} admits a Wakimoto filtration if there exists a finite filtration on \mathscr{F} such that each subquotient is a freemonodromic Wakimoto sheaf $\mathscr{W}^{\wedge}_{\lambda}$ with $\lambda \in X_{*}(T)$. As in the case of $\mathsf{P}_{\mathrm{I},\mathrm{I}}$ and $\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, the properties of free-monodromic Wakimoto sheaves stated in Lemma 7.2(2)–(5) imply that if \mathscr{F} admits a Wakimoto filtration, then there exists a unique filtration $(\mathscr{F}_{\leq \lambda} : \lambda \in X_{*}(T))$ on \mathscr{F} such that $\mathscr{F}_{\leq \lambda} = \{0\}$ for some λ , $\mathscr{F}_{\leq \mu} = \mathscr{F}$ for some μ , and $\mathscr{F}_{\leq \lambda}/\mathscr{F}_{<\lambda}$ is a direct sum of copies of $\mathscr{W}^{\wedge}_{\lambda}$ for each $\lambda \in X_{*}(T)$. Moreover, this filtration is functorial in the same sense as for its "traditional" counterpart in §4.3, which allows to define the functor $\mathrm{gr}^{\wedge}_{\lambda}$ sending an object \mathscr{F} which admits a free-monodromic Wakimoto filtration to

$$\operatorname{gr}_{\lambda}^{\wedge}(\mathscr{F}) := \mathscr{F}_{\leq \lambda}/\mathscr{F}_{<\lambda}.$$

For such \mathscr{F} , we will also set

$$\operatorname{gr}^{\wedge}_{\bullet}(\mathscr{F}) = \bigoplus_{\lambda \in X_{*}(T)} \operatorname{gr}^{\wedge}_{\lambda}(\mathscr{F}).$$

Proposition 7.4. If \mathscr{F}, \mathscr{G} in $\mathsf{P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ admit Wakimoto filtrations, then the morphism

$$\operatorname{Hom}_{\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{F},\mathscr{G}) \to \operatorname{Hom}_{\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\operatorname{gr}^{\wedge}_{\bullet}(\mathscr{F}), \operatorname{gr}^{\wedge}_{\bullet}(\mathscr{G}))$$

induced by the functor gr_{\bullet}^{\wedge} is injective.

Proof. The claim is an easy consequence of Lemma 7.3, following the same arguments as in [BR1, Corollary 6.3]. \Box

From this proposition we deduce in particular the following claim.

Corollary 7.5. For any \mathscr{F} in $\mathsf{P}^\wedge_{I_u,I_u}$ which admits a Wakimoto filtration, the morphism

$$\mu_{\mathscr{F}}:\mathscr{O}(T^{\vee}_{\Bbbk}\times T^{\vee}_{\Bbbk})\to \operatorname{End}(\mathscr{F})$$

factors through the morphism $\mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk}) \to \mathscr{O}(T^{\vee}_{\Bbbk})$ induced by the diagonal embedding $T^{\vee}_{\Bbbk} \hookrightarrow T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk}$; in other words, for any f in $\mathscr{O}(T^{\vee}_{\Bbbk})$ we have

$$\mu_{\mathscr{F}}(f\otimes 1) = \mu_{\mathscr{F}}(1\otimes f).$$

Proof. The functoriality of monodromy implies that the composition

$$\mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk}) \xrightarrow{\mu_{\mathscr{F}}} \operatorname{End}(\mathscr{F}) \to \operatorname{End}(\operatorname{gr}^{\wedge}_{\bullet}(\mathscr{F}))$$

(where the second morphism is induced by the functor $\operatorname{gr}^{\wedge}_{\bullet}$) coincides with $\mu_{\operatorname{gr}^{\wedge}_{\bullet}(\mathscr{F})}$. This reduces the proof to the case \mathscr{F} is a free-monodromic Wakimoto sheaf, in which case the claim was proved in Lemma 7.2(6).

We finish this subsection with a criterion for the existence of Wakimoto filtrations.

Lemma 7.6. Let \mathscr{F} in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. Then \mathscr{F} belongs to $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and admits a Wakimoto filtration iff $\pi_{\dagger}(\mathscr{F})$ belongs to $\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ and admits a Wakimoto filtration. Moreover, in this case the multiplicity of $\mathscr{W}^{\wedge}_{\lambda}$ in $\mathrm{gr}^{\wedge}_{\lambda}(\mathscr{F})$ equals the multiplicity of \mathscr{W}_{λ} in $\mathrm{gr}_{\lambda}(\mathscr{F})$.

Proof. If \mathscr{F} belongs to $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and admits a Wakimoto filtration, then Lemma 7.2(3) implies that $\pi_{\dagger}(\mathscr{F})$ belongs to $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ and admits a Wakimoto filtration.

Now, assume that $\pi_{\dagger}(\mathscr{F})$ belongs to $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ and admits a Wakimoto filtration. Choose $\mu \in X^+_*(T)$ such that $\lambda + \mu$ is dominant for any $\lambda \in X_*(T)$ such that $\mathrm{gr}_{\lambda}(\pi_{\dagger}(\mathscr{F})) \neq 0$. We have

$$\pi_{\dagger}(\mathscr{F} \mathbin{\widehat{\star}} \nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)}) \stackrel{(\mathbf{6}.4)}{\cong} \mathscr{F} \mathbin{\widehat{\star}} \pi_{\dagger}(\nabla^{\wedge,\mathrm{can}}_{\mathsf{t}(\mu)}) \stackrel{(\mathbf{7}.1)}{\cong} \mathscr{F} \mathbin{\widehat{\star}} \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\nabla^{\mathrm{I}}_{\mathsf{t}(\mu)}) \stackrel{(\mathbf{6}.5)}{\cong} \pi_{\dagger}(\mathscr{F}) \star_{\mathrm{I}} \nabla^{\mathrm{I}}_{\mathsf{t}(\mu)}.$$

Our assumption, (4.5) and the choice of μ imply that $\pi_{\dagger}(\mathscr{F}) \star_{\mathrm{I}} \nabla_{\mathrm{t}(\mu)}^{\mathrm{I}}$ is perverse and admits a filtration with subquotients of the form $\nabla_{\nu}^{\mathrm{I}}$ with $\nu \in X_{*}^{+}(T)$. The arguments in the proof of [BR1, Lemma 5.9(1)] (see also [B2, Proposition 9]) show that this condition implies that $\mathscr{F} \star \nabla_{\mathrm{t}(\mu)}^{\wedge,\mathrm{can}}$ is perverse and admits a filtration with subquotients of the form $\nabla_{\mathrm{t}(\nu)}^{\wedge,\mathrm{can}}$ with $\nu \in X_{*}(T)$. Applying the functor $(-) \star \Delta_{\mathrm{t}(-\mu)}^{\wedge,\mathrm{can}}$ we deduce that \mathscr{F} is perverse and admits a Wakimoto filtration, as desired.

The proof of the claim about multiplicities is immediate for these considerations. $\hfill \square$

7.4. Free-monodromic central sheaves. We now explain how to "upgrade" Gaitsgory's functor Z (see §4.3) to a functor with values in $D^{\wedge}_{I_u,I_u}$, following the case of $\overline{\mathbb{Q}}_{\ell}$ -coefficients treated in [B2, §3.5]. (This construction is based on the point of view explained in §4.4).

We start with the group scheme \mathcal{H} defined as the Néron blowup of $G \times \mathbb{A}^1$ in U along the divisor $\{0\} \subset \mathbb{A}^1$, in the sense of [MRR]. Then \mathcal{H} is a smooth group scheme over \mathbb{A}^1 , and the ind-scheme $\widetilde{\mathrm{Fl}}_G$ represents the functor sending an \mathbb{F} -algebra R to isomorphism classes of pairs consisting of a principal \mathcal{H} -bundle over $\mathrm{Spec}(R[[z]])$ together with a trivialization over $\mathrm{Spec}(R((z)))$. One can then define the ind-scheme $\widetilde{\mathrm{Gr}}_G^{\mathrm{BD}}$ by simply replacing \mathcal{G} by \mathcal{H} in the definition of $\mathrm{Gr}_G^{\mathrm{BD}}$. In this way we have canonical identifications

$$\{0\} \times_{\mathbb{A}^1} \widetilde{\operatorname{Gr}}_G^{\operatorname{BD}} = \widetilde{\operatorname{Fl}}_G,$$

and

$$(\mathbb{A}^1 \smallsetminus \{0\}) \times_{\mathbb{A}^1} \widetilde{\operatorname{Gr}}_G^{\operatorname{BD}} = \operatorname{Gr}_G \times \widetilde{\operatorname{Fl}}_G \times (\mathbb{A}^1 \smallsetminus \{0\}).$$

Using nearby cycles we therefore obtain a bifunctor

$$\mathbf{Y}: \mathsf{D}_{L^+G, L^+G} \times \mathsf{D}_{\mathrm{I}_{\mathrm{u}}, \mathrm{I}_{\mathrm{u}}} \to D_c^{\mathrm{b}}(\mathrm{Fl}_G, \mathbb{k}).$$

We have a smooth morphism $\widetilde{\operatorname{Gr}}_{G}^{\operatorname{BD}} \to \operatorname{Gr}_{G}^{\operatorname{BD}}$ which restricts to the morphism $\pi : \widetilde{\operatorname{Fl}}_{G} \to \operatorname{Fl}_{G}$ over 0, and to the induced morphism $\operatorname{Gr}_{G} \times \widetilde{\operatorname{Fl}}_{G} \times (\mathbb{A}^{1} \setminus \{0\}) \to \operatorname{Gr}_{G} \times \operatorname{Fl}_{G} \times (\mathbb{A}^{1} \setminus \{0\})$ over $\mathbb{A}^{1} \setminus \{0\}$ under the identifications above; by compatibility of nearby cycles with smooth pullback we deduce a canonical isomorphism

(7.3)
$$\widetilde{\mathsf{Y}}(\mathscr{A}, \pi^* \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{F})) \cong \pi^* \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathsf{Y}(\mathscr{A}, \mathscr{F}))$$

for any \mathscr{A} in D_{L^+G,L^+G} and \mathscr{F} in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$. In particular, this shows that the functor $\widetilde{\mathsf{Y}}$ factors through a bifunctor

$$\mathsf{D}_{L^+G,L^+G} \times \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},$$

which will also be denoted Y.

In the following statement, we use the variant of the bifunctor Y considered in Remark 4.2.

Lemma 7.7. For any \mathscr{A} in D_{L^+G,L^+G} and \mathscr{F} in D_{I_u,I_u} , we have a canonical isomorphism

$$\pi_! \mathsf{Y}(\mathscr{A}, \mathscr{F}) \cong \mathsf{Y}(\mathscr{A}, \pi_! \mathscr{F}).$$

Proof. By general properties of nearby cycles (see $[Del, \S2.1.7]$), there exists a canonical morphism

$$\pi_! \mathbf{Y}(\mathscr{A}, \mathscr{F}) \to \mathbf{Y}(\mathscr{A}, \pi_! \mathscr{F}).$$

Since D_{I_u,I_u} is generated, as a triangulated category, by the objects of the form $\pi^*\mathsf{For}^{I}_{I_u}(\mathscr{G})$ with \mathscr{G} in $\mathsf{D}_{I,I}$, to prove that our morphism is an isomorphism it suffices to do so for such objects. Now by the projection formula we have

$$\pi_! \pi^* \mathscr{G} \cong \mathscr{G} \otimes_{\Bbbk} \pi_! \underline{\Bbbk}_{\widetilde{\mathrm{Fl}}_{C}},$$

so that

$$\mathsf{Y}(\mathscr{A},\pi_!\pi^*\mathscr{G})\cong\mathsf{Y}(\mathscr{A},\mathscr{G})\otimes_{\Bbbk}\pi_!\underline{\Bbbk}_{\widetilde{\mathsf{Fl}}_{\mathscr{A}}}$$

On the other hand, using (7.3) and again the projection formula we see that

$$\pi_! \mathsf{Y}(\mathscr{A}, \pi^* \mathscr{G}) \cong \pi_! \pi^* \mathsf{Y}(\mathscr{A}, \mathscr{G}) \cong \mathsf{Y}(\mathscr{A}, \mathscr{G}) \otimes_{\Bbbk} \pi_! \underline{\Bbbk}_{\widetilde{\mathsf{Fl}}_{\mathcal{C}}}.$$

One can check that under these identifications our morphism identifies with the identity morphism of $\mathsf{Y}(\mathscr{A},\mathscr{G}) \otimes_{\Bbbk} \pi_! \underline{\Bbbk}_{\widetilde{Fl}_G}$; in particular it is indeed an isomorphism, which finishes the proof.

For \mathscr{A} in D_{L^+G,L^+G} , the functor $\mathsf{Y}(\mathscr{A},-)$ extends to a functor from $\mathsf{D}^{\wedge}_{\mathrm{I}_u,\mathrm{I}_u}$ to pro-objects in $\mathsf{D}_{\mathrm{I}_u,\mathrm{I}_u}$. Using Lemma 7.7 one sees that this functor in fact takes values in $\mathsf{D}^{\wedge}_{\mathrm{I}_u,\mathrm{I}_u}$; we have therefore obtained a bifunctor

$$\widetilde{\mathsf{Y}}:\mathsf{D}_{L^+G,L^+G}\times\mathsf{D}^\wedge_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{D}^\wedge_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

which satisfies

(7.4)

$$\pi_{\dagger}\widetilde{\mathsf{Y}}(\mathscr{A},\mathscr{F})\cong\mathsf{Y}(\mathscr{A},\pi_{\dagger}\mathscr{F})$$

for any \mathscr{A} in D_{L^+G,L^+G} and \mathscr{F} in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

The same considerations as in §4.4 show that for \mathscr{A}, \mathscr{B} in D_{L^+G, L^+G} and \mathscr{F}, \mathscr{G} in $\mathsf{D}_{\mathrm{I}_{\mathrm{u}}, \mathrm{I}_{\mathrm{u}}}$ we have a canonical isomorphism

$$\widetilde{\mathsf{Y}}(\mathscr{A},\mathscr{F})\star_{\mathrm{I}_{\mathrm{u}}}\widetilde{\mathsf{Y}}(\mathscr{B},\mathscr{G})\cong \widetilde{\mathsf{Y}}(\mathscr{A}\star_{L^{+}G}\mathscr{B},\mathscr{F}\star_{\mathrm{I}_{\mathrm{u}}}\mathscr{G}).$$

(The proof of this isomorphism involves the compatibility of nearby cycles with respect to pushforward along a nonproper map; this compatibility does not follow from a general result, but can be obtained using considerations similar to those encountered in the proof of Lemma 7.7.) Once this is proved, one obtains more generally that for \mathscr{A}, \mathscr{B} in D_{L^+G,L^+G} and \mathscr{F}, \mathscr{G} in $\mathsf{D}^{\wedge}_{\mathrm{I_u},\mathrm{I_u}}$ we have a canonical isomorphism

(7.5)
$$\widetilde{\mathsf{Y}}(\mathscr{A},\mathscr{F}) \mathbin{\widehat{\star}} \widetilde{\mathsf{Y}}(\mathscr{B},\mathscr{G}) \cong \widetilde{\mathsf{Y}}(\mathscr{A} \star_{L^+G} \mathscr{B}, \mathscr{F} \mathbin{\widehat{\star}} \mathscr{G}).$$

We now define the functor

$$\widehat{\mathsf{Z}}:\mathsf{D}_{L^+G,L^+G}\to\mathsf{D}^\wedge_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

by setting

$$\widehat{\mathsf{Z}}(\mathscr{A}) := \widetilde{\mathsf{Y}}(\mathscr{A}, \delta^{\wedge}).$$

Since this morphism is defined by nearby cycles, it comes with a canonical monodromy automorphism $\widehat{\mathfrak{m}}$. These data possess properties similar to those of the "traditional" functor Z of §4.3, as explained in the following statement.

Theorem 7.8. (1) There exists a canonical isomorphism of functors

$$\pi_{\dagger} \circ \widehat{\mathsf{Z}} \cong \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}} \circ \mathsf{Z}$$

which identifies $\pi_{\dagger}\widehat{\mathsf{m}}$ with $\mathsf{For}_{\mathrm{I}_{u}}^{\mathrm{I}}(\mathsf{m})$.

- (2) The functor $\widehat{\mathsf{Z}}$ restricts to an exact functor from P_{L^+G,L^+G} to $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.
- (3) There exist canonical isomorphisms

$$\widehat{\phi}_{\mathscr{A},\mathscr{B}}: \widehat{\mathsf{Z}}(\mathscr{A} \star_{L^+G} \mathscr{B}) \xrightarrow{\sim} \widehat{\mathsf{Z}}(\mathscr{A}) \mathbin{\widehat{\star}} \widehat{\mathsf{Z}}(\mathscr{B})$$

for \mathscr{A}, \mathscr{B} in D_{L^+G, L^+G} and $\widehat{\mathsf{Z}}(\delta_{\mathrm{Gr}}) \cong \delta^{\wedge}$ which endow $\widehat{\mathsf{Z}}$ with a monoidal structure. Moreover, via the identification

$$\pi_{\dagger}(\widehat{\mathsf{Z}}(\mathscr{A}) \mathbin{\widehat{\star}} \widehat{\mathsf{Z}}(\mathscr{B})) \stackrel{(\mathbf{6.4})}{\cong} \widehat{\mathsf{Z}}(\mathscr{A}) \mathbin{\widehat{\star}} \pi_{\dagger}(\widehat{\mathsf{Z}}(\mathscr{B})) \stackrel{(\mathbf{1})}{\cong} \widehat{\mathsf{Z}}(\mathscr{A}) \mathbin{\widehat{\star}} \operatorname{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathsf{Z}(\mathscr{B})) \\ \stackrel{(\mathbf{6.5})}{\cong} \pi_{\dagger}(\widehat{\mathsf{Z}}(\mathscr{A})) \star_{\mathrm{I}} \mathsf{Z}(\mathscr{B}) \stackrel{(\mathbf{1})}{\cong} \mathsf{Z}(\mathscr{A}) \star_{\mathrm{I}} \mathsf{Z}(\mathscr{B})$$

we have $\pi_{\dagger}(\widehat{\phi}_{\mathscr{A},\mathscr{B}}) = \phi_{\mathscr{A},\mathscr{B}}$, and $\widehat{\phi}_{\mathscr{A},\mathscr{B}}$ intertwines the automorphisms $\widehat{\mathfrak{m}}_{\mathscr{A}\star_{L^{+}G}\mathscr{B}} \text{ of } \widehat{\mathsf{Z}}(\mathscr{A}\star_{L^{+}G}\mathscr{B}) \text{ and } \widehat{\mathfrak{m}}_{\mathscr{A}} \widehat{\star} \widehat{\mathfrak{m}}_{\mathscr{B}} \text{ of } \widehat{\mathsf{Z}}(\mathscr{A}) \widehat{\star} \widehat{\mathsf{Z}}(\mathscr{B}).$ (4) For any \mathscr{A} in $\mathsf{D}_{L^{+}G,L^{+}G}$ and \mathscr{F} in $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\wedge}$, there exists a canonical isomor-

phism

$$\widehat{\sigma}_{\mathscr{A},\mathscr{F}}:\widehat{\mathsf{Z}}(\mathscr{A})\mathbin{\widehat{\star}}\mathscr{F}\xrightarrow{\sim}\mathscr{F}\mathbin{\widehat{\star}}\widehat{\mathsf{Z}}(\mathscr{A})$$

which identifies $\widehat{\mathfrak{m}}_{\mathscr{A}} \stackrel{\star}{\star} \operatorname{id}_{\mathscr{F}}$ and $\operatorname{id}_{\mathscr{F}} \stackrel{\star}{\star} \widehat{\mathfrak{m}}_{\mathscr{A}}$. Moreover the functor Z, together with the isomorphisms $\widehat{\sigma}_{\mathscr{A},\mathscr{F}}$ and $\widehat{\phi}_{\mathscr{A},\mathscr{B}}$, define a central functor from P_{L^+G,L^+G} to $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ in the sense of [B1]; in other words these data define a braided monoidal functor from P_{L^+G,L^+G} to the Drinfeld center of $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}.$

(5) For any \mathscr{A} in P_{L^+G,L^+G} , the functor

$$\widehat{\mathsf{Z}}(\mathscr{A}) \mathbin{\widehat{\star}} (-) : \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

is t-exact.

(6) For any \mathscr{A} in P_{L^+G,L^+G} we have

$$\widehat{\mathsf{m}}_{\mathscr{A}} = \mu_{\widehat{\mathsf{Z}}(\mathscr{A})}^{\mathrm{rot}}(x^{-1}).$$

(7) For any \mathscr{A} in P_{L^+G,L^+G} , the perverse sheaf $\widehat{\mathsf{Z}}(\mathscr{A})$ admits a Wakimoto filtration. Moreover, for any $\lambda \in X_*(T)$ the multiplicity of $\mathscr{W}^{\wedge}_{\lambda}$ in $\operatorname{gr}^{\wedge}_{\lambda}(\widehat{\mathsf{Z}}(\mathscr{A}))$ equals the dimension of the λ -weight space of $\mathcal{S}(\mathscr{A})$.

Proof. (1) Using (7.4), for \mathscr{A} in D_{L^+G,L^+G} we obtain a canonical isomorphism

$$\pi_{\dagger}\widehat{\mathsf{Z}}(\mathscr{A}) = \pi_{\dagger}\widetilde{\mathsf{Y}}(\mathscr{A},\delta^{\wedge}) \cong \mathsf{Y}(\mathscr{A},\pi_{\dagger}\delta^{\wedge}) \cong \mathsf{Y}(\mathscr{A},\delta).$$

Now by (4.6) the right-hand side identifies with $Z(\mathscr{A})$, which provides the desired isomorphism. The compatibility with monodromy automorphisms follows from general properties of nearby cycles functors.

(2) The isomorphism in (1) and the exactness of Z show that $\pi_{\dagger}\widehat{Z}(\mathscr{A})$ is perverse for any \mathscr{A} in P_{L^+G,L^+G} . By [BR1, Lemma 5.3], this implies that $\widehat{Z}(\mathscr{A})$ is perverse. Hence \widehat{Z} sends P_{L^+G,L^+G} to $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. Exactness of the restriction is automatic since this functor is obtained from a triangulated functor.

(3) The isomorphism $\phi_{\mathscr{A},\mathscr{B}}$ is obtained by applying (7.5) in the case $\mathscr{F} = \mathscr{G} = \delta^{\wedge}$ and using the canonical isomorphism $\delta^{\wedge} \stackrel{\sim}{\star} \delta^{\wedge} \cong \delta^{\wedge}$. The compatibility with $\phi_{\mathscr{A},\mathscr{B}}$ follows from the comments in §4.4. The proof of the compatibility with monodromy is similar to that in the case of Z.

To justify that $Z(\delta_{Gr}) = \delta^{\wedge}$, we remark that more generally we have

(7.6)
$$\widetilde{\mathsf{Y}}(\delta_{\mathrm{Gr}},\mathscr{F})\cong\mathscr{F}$$

for any \mathscr{F} in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. In fact this follows from considerations analogous to those for the isomorphism $\mathsf{Y}(\delta_{\mathrm{Gr}},-)\cong$ id in §4.4, using the natural closed embedding $\widetilde{\mathrm{Fl}}_G \times \mathbb{A}^1 \hookrightarrow \widetilde{\mathrm{Gr}}_G^{\mathrm{BD}}$.

The fact that these data define a monoidal structure on $\widehat{\mathsf{Z}}$ is easy, and left to the reader.

(4) As in the case of the functor Z in §4.4, the isomorphism $\hat{\sigma}_{\mathscr{A},\mathscr{F}}$ is constructed from (7.5) and the isomorphism (7.6). Namely, these isomorphisms imply that we have

$$\widehat{\mathsf{Z}}(\mathscr{A}) \mathbin{\widehat{\star}} \mathscr{F} \cong \widetilde{\mathsf{Y}}(\mathscr{A}, \delta^{\wedge}) \mathbin{\widehat{\star}} \widetilde{\mathsf{Y}}(\delta_{\mathrm{Gr}}, \mathscr{F}) \cong \widetilde{\mathsf{Y}}(\mathscr{A} \star_{L^{+}G} \delta_{\mathrm{Gr}}, \delta^{\wedge} \mathbin{\widehat{\star}} \mathscr{F}) \cong \widetilde{\mathsf{Y}}(\mathscr{A}, \mathscr{F})$$

on the one hand, and that

$$\mathscr{F} \mathbin{\widehat{\star}} \widehat{\mathsf{Z}}(\mathscr{A}) \cong \widetilde{\mathsf{Y}}(\delta_{\mathrm{Gr}}, \mathscr{F}) \mathbin{\widehat{\star}} \widetilde{\mathsf{Y}}(\mathscr{A}, \delta^{\wedge}) \cong \widetilde{\mathsf{Y}}(\delta_{\mathrm{Gr}} \star_{L^{+}G} \mathscr{A}, \mathscr{F} \mathbin{\widehat{\star}} \delta^{\wedge}) \cong \widetilde{\mathsf{Y}}(\mathscr{A}, \mathscr{F})$$

on the other hand. The proof that these data define a central functor can be copied from the case of Z (see [Ga2] or [AR4]).

(5) First we prove that our functor is right exact. For that, since the nonpositive part of the perverse t-structure is generated under extensions by the objects Δ_w^{\wedge} $(w \in W)$, it suffices to prove that for any such w the object $\widehat{Z}(\mathscr{A}) \stackrel{\star}{\star} \Delta_w^{\wedge}$ has no perverse cohomology in positive degrees. In fact we will prove that this object is perverse. Indeed, using (6.4)–(6.5) and (1) we obtain that

$$\pi_{\dagger}(\widehat{\mathsf{Z}}(\mathscr{A}) \mathbin{\widehat{\star}} \Delta_w^{\wedge}) \cong \widehat{\mathsf{Z}}(\mathscr{A}) \mathbin{\widehat{\star}} \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\Delta_w^{\mathrm{I}}) \cong \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathsf{Z}(\mathscr{A}) \star_{\mathrm{I}} \Delta_w^{\mathrm{I}}).$$

Now $Z(\mathscr{A}) \star_I \Delta_w^I$ is perverse by "convolution exactness" of usual central sheaves (see [BRR, Theorem 4.2(2)]). Using [BR1, Lemma 5.3(1)] we deduce that $\widehat{Z}(\mathscr{A}) \widehat{\star} \Delta_w^{\wedge}$ is perverse, as desired.

To prove left exactness of our functor, we consider the rigid dual \mathscr{A}^{\vee} of \mathscr{A} in the rigid tensor category $\operatorname{Perv}_{L^+G}(\operatorname{Gr}_G, \mathbb{k})$. By monoidality of $\widehat{\mathsf{Z}}$ (see (3)), the functor $\widehat{\mathsf{Z}}(\mathscr{A}^{\vee}) \widehat{\star}(-)$ is left adjoint to $\widehat{\mathsf{Z}}(\mathscr{A}) \widehat{\star}(-)$. Since the former functor is right exact, we deduce that the latter is left exact, which finishes the proof.

(6) The proof is similar to that of the corresponding claim for Z, see (5.1).

(7) The claim follows from Lemma 7.6, in view of (1) and the property that $Z(\mathscr{A})$ admits a Wakimoto filtration, see §4.3.

For simplicity of notation, below we will set

$$\widehat{\mathscr{Z}} := \widehat{\mathsf{Z}} \circ \mathcal{S}^{-1} : \mathsf{Rep}(G^{\vee}_{\Bbbk}) \to \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},$$

and write

$$\widehat{\mathsf{m}}_V := \widehat{\mathsf{m}}_{\mathcal{S}^{-1}(V)} \in \operatorname{End}(\widehat{\mathscr{Z}}(V)).$$

Theorem 7.8(7) and Corollary 7.5 imply that for any V in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$ the morphism $\mu_{\widehat{\mathscr{R}}(V)}$ factors through a canonical morphism

(7.7)
$$\mu_V : \mathscr{O}(T^{\vee}_{\Bbbk}) \to \operatorname{End}(\widehat{\mathscr{Z}}(V)).$$

Let us note for later use that from Theorem 7.8(4)–(5) we also deduce, for any $V \in \mathsf{Rep}(G^{\vee}_{\Bbbk})$ and \mathscr{F} in $\mathsf{D}^{0}_{\mathrm{In},\mathrm{In}}$, a canonical isomorphism

(7.8)
$$\widehat{\mathscr{Z}}(V)\,\widehat{\star}^0\,\mathscr{F}\xrightarrow{\sim}\mathscr{F}\,\widehat{\star}^0\,\widehat{\mathscr{Z}}(V),$$

and that these objects belong to the heart of the perverse t-structure if \mathscr{F} does.

7.5. Some tilting perverse sheaves. In this subsection we assume that the conditions in §4.8 are satisfied. Under this assumption, the free-monodromic central sheaves, together with the object Ξ_1^{\wedge} introduced in §6.6, allow to describe a family of tilting objects in P_{I_n,I_n}^{\wedge} , as follows.

Proposition 7.9. For any $V \in \mathsf{Rep}(G_{\Bbbk}^{\vee})$ which is tilting, the perverse sheaf $\Xi_{!}^{\wedge} \widehat{\star} \widehat{\mathscr{L}}(V)$ is tilting.

Proof. In view of [BR1, Lemma 5.9], to prove the claim it suffices to prove that the object $\pi_{\dagger}(\Xi_{!}^{\wedge} \widehat{\mathscr{L}}(V))$ of $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ is a tilting perverse sheaf. Now, using Theorem 7.8(1) we have

$$\pi_{\dagger} \left(\Xi_{!}^{\wedge} \mathbin{\widehat{\mathscr{X}}}(V) \right) \stackrel{\text{(6.4)}}{\cong} \Xi_{!}^{\wedge} \mathbin{\widehat{\star}} \pi_{\dagger} \left(\widehat{\mathscr{X}}(V) \right) \cong \Xi_{!}^{\wedge} \mathbin{\widehat{\star}} \operatorname{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{Z}(V))$$
$$\stackrel{\text{(6.5)}}{\cong} \pi_{\dagger}(\Xi_{!}^{\wedge}) \star_{\mathrm{I}} \mathscr{Z}(V) \cong \Xi_{!} \star_{\mathrm{I}} \mathscr{Z}(V).$$

Let I_u^+ be the inverse image of U^+ under the evaluation morphism $L^+G \to G$, and consider the composition

$$\chi_{\mathrm{I}}: \mathrm{I}_{\mathrm{u}}^{+} \to U^{+} \xrightarrow{\chi} \mathbb{G}_{\mathrm{a}},$$

where the first morphism is the obvious projection adn χ is as in §6.6. Let us denote by $\mathsf{D}_{\mathcal{IW},\mathrm{I}}$ the $(\mathrm{I}^+_\mathrm{u}, \chi^*_\mathrm{I}(\mathscr{L}_{\mathrm{AS}}))$ -equivariant derived category of k-sheaves on Fl_G . (This category is denoted $D^{\mathrm{b}}_{\mathcal{IW}}(\mathrm{Fl}_G, \Bbbk)$ in [BRR].) This category has a natural perverse t-structure, whose heart $\mathsf{P}_{\mathcal{IW},\mathrm{I}}$ has a canonical structure of highest weight category.

As in the case of G/U in §6.6, standard constructions provide t-exact "averaging" functors

$$\mathsf{Av}_{\mathcal{I}\mathcal{W}}:\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\to\mathsf{D}_{\mathcal{I}\mathcal{W},\mathrm{I}},\quad\mathsf{Av}_{\mathrm{I}_{\mathrm{u}},\mathrm{!}}:\mathsf{D}_{\mathcal{I}\mathcal{W},\mathrm{I}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}},\qquad\mathsf{Av}_{\mathrm{I}_{\mathrm{u}},*}:\mathsf{D}_{\mathcal{I}\mathcal{W},\mathrm{I}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

such that $Av_{I_u,!}$ is left adjoint to $Av_{\mathcal{IW}}$ and $Av_{I_u,*}$ is right adjoint to $Av_{\mathcal{IW}}$. Standard considerations (see e.g. [AR6, Lemma 3.6]) show that $Av_{I_u,!}$, resp. $Av_{I_u,*}$, sends objects admitting a standard, resp. costandard, filtration to objects admitting a standard, filtration.

It follows from the definition of $\Xi_{!}$ that we have a canonical isomorphism of functors

$$\Xi_{!} \star_{I} (-) \cong \mathsf{Av}_{I_{u}, !} \circ \mathsf{Av}_{\mathcal{I}\mathcal{W}} \circ \mathsf{For}^{I}_{I_{u}};$$

in particular we deduce that

$$\pi_{\dagger}\big(\Xi^{\wedge}_{!} \mathbin{\widehat{\star}} \widehat{\mathscr{Z}}(V)\big) \cong \mathsf{Av}_{\mathrm{I}_{\mathrm{u}},!} \circ \mathsf{Av}_{\mathcal{I}\mathcal{W}} \circ \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{Z}(V)).$$

Now by [BRR, Theorem 8.1] the perverse sheaf $\mathsf{Av}_{\mathcal{IW}} \circ \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{Z}(V))$ is tilting, from which we deduce that $\mathsf{Av}_{\mathrm{I}_{\mathrm{u}}}, \circ \mathsf{Av}_{\mathcal{IW}} \circ \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{Z}(V))$ is a perverse sheaf admitting a standard filtration.

Using the isomorphism $\Xi_! \cong \Xi_*$ we similarly obtain that

$$\pi_{\dagger}\big(\Xi^{\wedge}_{!} \mathbin{\widehat{\star}} \mathscr{\widehat{Z}}(V)\big) \cong \mathsf{Av}_{\mathrm{I}_{\mathrm{u}},*} \circ \mathsf{Av}_{\mathcal{I}\mathcal{W}} \circ \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{Z}(V)),$$

which implies that $\pi_{\dagger}(\Xi_{!}^{\wedge} \widehat{\mathscr{Z}}(V))$ admits a costandard filtration and finishes the proof.

7.6. Quantum trace of monodromy. Recall (see e.g. [EGNO, §2.10]) that if (A, \odot) is a monoidal category with unit object 1, and if X is an object of A, a *left dual* of X is the data of an object X^{\vee} together with morphisms

$$\operatorname{ev}_X : X^{\vee} \odot X \to \mathbf{1}, \quad \operatorname{coev}_X : \mathbf{1} \to X \odot X^{\vee}$$

such that the compositions

$$X \xrightarrow{\operatorname{coev}_X \odot \operatorname{id}} X \odot X^{\vee} \odot X \xrightarrow{\operatorname{id} \odot \operatorname{ev}_X} X,$$
$$X^{\vee} \xrightarrow{\operatorname{id} \odot \operatorname{coev}_X} X^{\vee} \odot X \odot X^{\vee} \xrightarrow{\operatorname{ev}_X \odot \operatorname{id}} X^{\vee}$$

are the identity morphisms of X and X^{\vee} , respectively. (Here, we omit the unit and associativity isomorphisms.) Similarly, a *right dual* of X is the data of an object $^{\vee}X$ together with morphisms

$$\operatorname{ev}'_X : X \odot {}^{\vee}X \to \mathbf{1}, \quad \operatorname{coev}'_X : \mathbf{1} \to {}^{\vee}X \odot X$$

such that the compositions

$$\begin{array}{c} {}^{\vee}X \xrightarrow{\operatorname{coev}'_X \odot \operatorname{id}} {}^{\vee}X \odot X \odot {}^{\vee}X \xrightarrow{\operatorname{id} \odot \operatorname{ev}'_X} {}^{\vee}X, \\ X \xrightarrow{\operatorname{id} \odot \operatorname{coev}'_X} X \odot {}^{\vee}X \odot X \xrightarrow{\operatorname{ev}'_X \odot \operatorname{id}} X \end{array}$$

are the identity morphisms of $\forall X$ and X, respectively. The object X is called *left dualizable*, resp. *right dualizable*, if a left dual, resp. right dual, exists; in this case such a dual is unique up to unique isomorphism, see [EGNO, Proposition 2.10.5]. This notion is functorial is the following sense: if X, Y are left, resp. right, dualizable, then there exists a canonical isomorphism

 $\operatorname{Hom}_{\mathsf{A}}(X,Y) \xrightarrow{\sim} \operatorname{Hom}_{\mathsf{A}}(Y^{\vee},X^{\vee}), \quad \operatorname{resp.} \quad \operatorname{Hom}_{\mathsf{A}}(X,Y) \xrightarrow{\sim} \operatorname{Hom}_{\mathsf{A}}({}^{\vee}Y,{}^{\vee}X),$

denoted $f \mapsto f^{\vee}$, resp. $f \mapsto {}^{\vee}f$. Here, given $f : X \to Y$, the morphism f^{\vee} is the composition

$$Y^{\vee} \xrightarrow{\operatorname{id} \odot \operatorname{coev}_X} Y^{\vee} \odot X \odot X^{\vee} \xrightarrow{\operatorname{id} \odot f \odot \operatorname{id}} Y^{\vee} \odot Y \odot X^{\vee} \xrightarrow{\operatorname{ev}_Y \odot \operatorname{id}} X^{\vee},$$

and the morphism $\forall f$ is the composition

$$^{\vee}Y \xrightarrow{\operatorname{coev}'_X \odot \operatorname{id}} {}^{\vee}X \odot X \odot {}^{\vee}Y \xrightarrow{\operatorname{id} \odot f \odot \operatorname{id}} {}^{\vee}X \odot Y \odot {}^{\vee}Y \xrightarrow{\operatorname{id} \odot \operatorname{ev}'_Y} {}^{\vee}X.$$

Below we will also use the fact that if X is left dualizable, resp. right dualizable, then the functor $X^{\vee} \odot (-)$ is left adjoint to $X \odot (-)$ and the functor $(-) \odot X$ is left adjoint to $(-) \odot X^{\vee}$, resp. the functor $X \odot (-)$ is left adjoint to ${}^{\vee}X \odot (-)$ and the functor $(-) \odot {}^{\vee}X$ is left adjoint to $(-) \odot X$; see [EGNO, Proposition 2.10.8].

The application of this notion that will be relevant for us is to the definition of *quantum traces*, see [EGNO, §4.7]. Namely, consider an object X which is left dualizable, and assume that X^{\vee} is itself left dualizable (with left dual denoted $X^{\vee\vee}$). Then for any $a \in \operatorname{Hom}_{A}(X, X^{\vee\vee})$ the *left quantum trace* $\operatorname{tr}^{L}(a)$ of a is defined as the endomorphism of 1 obtained as the composition

$$\mathbf{1} \xrightarrow{\operatorname{coev}_X} X \odot X^{\vee} \xrightarrow{a \odot \operatorname{id}} X^{\vee \vee} \odot X^{\vee} \xrightarrow{\operatorname{ev}_X \vee} \mathbf{1}.$$

A similar definition leads to the notion of the *right quantum trace* of a morphism $a: X \to {}^{\vee \vee}X$, in case X and ${}^{\vee}X$ are right dualizable.

In our present setting, since a monoidal functor sends dualizable objects to dualizable objects, and their duals to the corresponding duals (see [EGNO, Exercise 2.10.6]), and since every object V in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$ is left and right dualizable with left and right duals V^{*} (together with the obvious evaluation and coevaluation maps), for any V the object $\mathscr{Z}(V)$, resp. $\widehat{\mathscr{Z}}(V)$, is left and right dualizable in $\mathsf{D}_{\mathrm{I},\mathrm{I}}$, resp. $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$, with left and right dual $\mathscr{Z}(V^*)$, resp. $\widehat{\mathscr{Z}}(V^*)$. Hence the (left) quantum trace $\operatorname{tr}^{\mathrm{L}}(a)$ is defined for any $a \in \operatorname{End}_{\mathsf{D}_{\mathrm{I},\mathrm{I}}}(\mathscr{Z}(V))$, resp. $a \in \operatorname{End}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\widehat{\mathscr{Z}}(V))$. The case of $\mathsf{D}_{\mathrm{I},\mathrm{I}}$ is not very rich, since the endomorphisms of δ are k. But in $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$ we have $\operatorname{End}(\delta^{\wedge}) = \mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})$ (see Lemma 6.6); the left quantum trace of a morphism is therefore an element in $\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})$.

The following lemma will play a technical role in the construction of a functor in Section 10. Its proof will occupy the rest of the section. (No detail of this proof will be used in later sections, so that these subsections can be safely skipped.)

Lemma 7.10. For any V in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$ we have

$$\operatorname{tr}^{\mathcal{L}}(\widehat{\mathsf{m}}_{V}) = \sum_{\mu \in X_{*}(T)} \dim(V_{\mu}) \cdot e^{\mu}$$

where V_{μ} is the μ -weight space of V. In other words, $\operatorname{tr}^{L}(\widehat{\mathfrak{m}}_{V})$ is the image of the character of V (seen as a function on T_{\Bbbk}^{\vee}) in $\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})$.

7.7. **Description of duals.** The proof of Lemma 7.10 will use the free-monodromic Wakimoto filtration on $\widehat{\mathscr{L}}(V)$. For this we will need to show that each subquotient in this filtration is dualizable, and describe its dual.

For the next lemma, we will have to assume that the order \leq on $X_*(T)$ chosen in §4.3 satisfies the following property:

for $\lambda, \mu \in X_*(T)$, $\lambda \leq \mu$ if and only if $-\mu \leq -\lambda$.

(Of course, there exists an order with the desired properties.)

Lemma 7.11. Consider some V in $\text{Rep}(G_{\Bbbk}^{\vee})$, and let $\lambda, \mu \in X_*(T)$ be such that $\lambda \leq \mu$.

- (1) The object $\mathscr{Z}(V)_{\leq \mu}/\mathscr{Z}(V)_{\leq \lambda}$ is left and right dualizable, with left and right dual $\mathscr{Z}(V^*)_{<-\lambda}/\mathscr{Z}(V^*)_{<-\mu}$.
- (2) The object $\widehat{\mathscr{Z}}(V)_{\leq \mu}/\widehat{\mathscr{Z}}(V)_{\leq \lambda}$ is left and right dualizable, with left and right dual $\widehat{\mathscr{Z}}(V^*)_{<-\lambda}/\widehat{\mathscr{Z}}(V^*)_{<-\mu}$.

Remark 7.12. It can be checked that in a monoidal category (A, \odot) where A is abelian and \odot is exact, the kernel (resp. cokernel) of a morphism between dualizable objects is dualizable, with dual the cokernel (resp. kernel) of the dual morphism. This statement does not apply here, since we do not have an obvious abelian subcategory of D_{I_u,I_u}^{\wedge} or $D_{I,I}$ containing the central sheaves and stable under the convolution product; however the proof below repeats arguments close to those required to prove this property.

Proof of Lemma 7.11. We will treat the two cases in parallel, and work with left duals; the proof for right duals is similar. First, let us assume that λ satisfies $\lambda < \nu$ for any $\nu \in X_*(T)$ such that $V_{\nu} \neq 0$, so that $\mathscr{Z}(V)_{\leq \lambda} = 0$ and $\mathscr{Z}(V^*)_{<-\lambda} = \mathscr{Z}(V^*)$ (and similarly for $\widehat{\mathscr{Z}}(V)$ and $\widehat{\mathscr{Z}}(V^*)$). In this case we will proceed by downward induction on μ , and prove (in addition to the fact that $\mathscr{Z}(V)_{\leq \mu}$ and $\widehat{\mathscr{Z}}(V)_{\leq \mu}$ are dualizable with the duals given in the statement) that the dual of the embedding $\mathscr{Z}(V)_{\leq \mu} \hookrightarrow \mathscr{Z}(V)$, resp. $\widehat{\mathscr{Z}}(V)_{\leq \mu} \hookrightarrow \widehat{\mathscr{Z}}(V)$, is the projection $\mathscr{Z}(V^*) \twoheadrightarrow \mathscr{Z}(V^*)_{<-\mu}$. If $\mu \geq \nu$ for any $\nu \in X_*(T)$ such that $V_{\nu} \neq 0$ we have $\mathscr{Z}(V)_{\leq \mu} = \mathscr{Z}(V)$ and

If $\mu \geq \nu$ for any $\nu \in X_*(T)$ such that $V_{\nu} \neq 0$ we have $\mathscr{Z}(V)_{\leq \mu} = \mathscr{Z}(V)$ and $\mathscr{Z}(V^*)_{<-\mu} = 0$ (and similarly for $\widehat{\mathscr{Z}}(V)$ and $\widehat{\mathscr{Z}}(V^*)$); in this case the claim has already been justified above Lemma 7.10. Now we fix $\mu \in X_*(T)$, and assume the claim is known for the successor μ' of μ , i.e. that $\mathscr{Z}(V)_{\leq \mu'}$ and $\widehat{\mathscr{Z}}(V)_{\leq \mu'}$ are left dualizable, with left duals

$$\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu'}$$
 and $\widehat{\mathscr{Z}}(V^*)/\widehat{\mathscr{Z}}(V^*)_{<-\mu'}$

respectively, and that the dual of the embedding in $\mathscr{Z}(V)$, resp. $\widehat{\mathscr{Z}}(V)$, is the projection from $\mathscr{Z}(V^*)$, resp. $\widehat{\mathscr{Z}}(V^*)$. We now consider the exact sequence

(7.9)
$$\mathscr{Z}(V)_{\leq \mu} \hookrightarrow \mathscr{Z}(V)_{\leq \mu'} \twoheadrightarrow \operatorname{gr}_{\mu'}(\mathscr{Z}(V)).$$

Here the right-hand side is isomorphic to $\mathscr{W}_{\mu'}^{\oplus r}$ for some $r \geq 0$; we fix an isomorphism $\operatorname{gr}_{\mu'}(\mathscr{Z}(V)) \cong \mathscr{W}_{\mu'}^{\oplus r}$ and therefore identify the second morphism in (7.9) with a surjection $f: \mathscr{Z}(V)_{\leq \mu'} \twoheadrightarrow \mathscr{W}_{\mu'}^{\oplus r}$. By assumption $\mathscr{Z}(V)_{\leq \mu'}$ is left dualizable, and since $\mathscr{W}_{\mu'}$ is invertible it is also left dualizable (with left dual $\mathscr{W}_{-\mu'}$); hence so is $\mathscr{W}_{\mu'}^{\oplus r}$. We can therefore consider the dual morphism

$$f^{\vee}: (\mathscr{W}_{\mu'}^{\oplus r})^{\vee} \to (\mathscr{Z}(V)_{\leq \mu'})^{\vee},$$

which we interpret as a morphism from $\mathscr{W}_{-\mu'}^{\oplus r}$ to $\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu'}$. By functoriality of Wakimoto filtrations this morphism factors through a morphism

(7.10)
$$\widetilde{f^{\vee}}: \mathscr{W}_{-\mu'}^{\oplus r} \to \operatorname{gr}_{-\mu'}(\mathscr{Z}(V^*))$$

We claim that $\widetilde{f^{\vee}}$ is an isomorphism. In fact we have $\operatorname{gr}_{-\mu'}(\mathscr{Z}(V^*)) \cong \mathscr{W}_{-\mu'}^{\oplus r}$ since $\operatorname{dim}((V^*)_{-\mu'}) = \operatorname{dim}(V_{\mu'})$. Since $\operatorname{End}(\mathscr{W}_{-\mu'}) \cong \Bbbk$ (see [BRR, §4.5]), the morphism $\widetilde{f^{\vee}}$ can be represented by an $r \times r$ -matrix, and saying that it is invertible is equivalent to this matrix being invertible. If this were not the case, then there would exist an embedding $\mathscr{W}_{-\mu'} \hookrightarrow \mathscr{W}_{-\mu'}^{\oplus r}$ as a direct summand such that the composition

$$\mathscr{W}_{-\mu'} \to \mathscr{W}_{-\mu'}^{\oplus r} \xrightarrow{f^{\vee}} \mathscr{Z}(V^*) / \mathscr{Z}(V^*)_{<-\mu'}$$

vanishes. However, by adjunction we have

$$\operatorname{Hom}(\mathscr{W}_{-\mu'}, \mathscr{W}_{-\mu'}^{\oplus r}) \cong \operatorname{Hom}(\mathscr{W}_{-\mu'} \star_{\mathrm{I}} \mathscr{W}_{\mu'}^{\oplus r}, \delta),$$
$$\operatorname{Hom}(\mathscr{W}_{-\mu'}, \mathscr{Z}(V^*) / \mathscr{Z}(V^*)_{<-\mu'}) \cong \operatorname{Hom}(\mathscr{W}_{-\mu'} \star_{\mathrm{I}} \mathscr{Z}(V)_{\leq \mu'}, \delta),$$

and through this identification the morphism $f^{\vee} \circ (-)$ corresponds to the morphism

$$(-) \circ (\mathrm{id} \star f) : \mathrm{Hom}(\mathscr{W}_{-\mu'} \star_{\mathrm{I}} \mathscr{W}_{\mu'}^{\oplus r}, \delta) \to \mathrm{Hom}(\mathscr{W}_{-\mu'} \star_{\mathrm{I}} \mathscr{Z}(V)_{\leq \mu'}, \delta),$$

which is injective since $\operatorname{id} \star f$ is surjective. This provides a contradiction, proving therefore that $\widetilde{f^{\vee}}$ indeed is an isomorphism.

We have now obtained an isomorphism

$$\left(\mathrm{gr}_{\mu'}(\mathscr{Z}(V))\right)^{\vee} \xrightarrow{\sim} (\mathscr{W}_{\mu'}^{\oplus r})^{\vee} \xrightarrow{\sim} \mathscr{W}_{-\mu'}^{\oplus r} \xrightarrow{\sim} \mathrm{gr}_{-\mu'}(\mathscr{Z}(V^*)),$$

which is easily seen not to depend on our initial choice of isomorphism $\operatorname{gr}_{\mu'}(\mathscr{Z}(V)) \cong \mathscr{W}_{\mu'}^{\oplus r}$; it is therefore canonical. Moreover, through this identification the dual of the projection $\mathscr{Z}(V)_{\leq \mu'} \twoheadrightarrow \operatorname{gr}_{\mu'}(\mathscr{Z}(V))$ is the embedding $\operatorname{gr}_{-\mu'}(\mathscr{Z}(V^*)) \hookrightarrow \mathscr{Z}(V^*) / \mathscr{Z}(V^*)_{<-\mu'}$.

Let us consider the composition

$$\delta \xrightarrow{\operatorname{coev}} \mathscr{Z}(V)_{\leq \mu'} \star_{\mathrm{I}} (\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu'}) \twoheadrightarrow \mathscr{Z}(V)_{\leq \mu'} \star_{\mathrm{I}} (\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu}).$$

The preceding considerations show that its composition with the surjection

$$\mathscr{Z}(V)_{\leq \mu'} \star_{\mathrm{I}} (\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu}) \twoheadrightarrow \mathrm{gr}_{\mu'}(\mathscr{Z}(V)) \star_{\mathrm{I}} (\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu})$$

vanishes; this morphism therefore factors through a morphism

(7.11)
$$\delta \xrightarrow{\text{coev}} \mathscr{Z}(V)_{\leq \mu} \star_{\mathrm{I}} (\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu})$$

Similarly, the composition

$$(\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu'}) \star_{\mathrm{I}} \mathscr{Z}(V)_{\leq \mu} \hookrightarrow (\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu'}) \star_{\mathrm{I}} \mathscr{Z}(V)_{\leq \mu'} \xrightarrow{\mathrm{ev}} \delta$$

factors through a morphism

(7.12)
$$(\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu}) \star_{\mathrm{I}} \mathscr{Z}(V)_{\leq \mu} \to \delta.$$

It is then not difficult to check that (7.11) and (7.12) exhibit $\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu}$ as the left dual of $\mathscr{Z}(V)_{\leq\mu}$. Moreover from the construction of the evaluation and coevaluation morphisms one sees that the dual of the embedding $\mathscr{Z}(V)_{\leq\mu} \hookrightarrow \mathscr{Z}(V)_{\leq\mu'}$ is the projection $\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu'} \twoheadrightarrow \mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu}$; by compability of duality with composition and the induction hypothesis, it follows that the dual of the embedding $\mathscr{Z}(V)_{\leq\mu} \hookrightarrow \mathscr{Z}(V)$ is the natural projection $\mathscr{Z}(V^*) \twoheadrightarrow \mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu}$.

Next, we consider the free-monodromic setting, and more specifically the exact sequence

$$\widehat{\mathscr{Z}}(V)_{\leq \mu} \hookrightarrow \widehat{\mathscr{Z}}(V)_{\leq \mu'} \twoheadrightarrow \mathrm{gr}^\wedge_{\mu'}(\widehat{\mathscr{Z}}(V)).$$

Here again, by induction the middle term is left dualizable, and the right-hand side is dualizable because it is isomorphic to a direct sum of invertible objects. If we fix an isomorphism $\operatorname{gr}_{\mu'}^{\wedge}(\widehat{\mathscr{Z}}(V)) \cong (\mathscr{W}_{\mu'}^{\wedge})^{\oplus r}$, the dual of the surjection $f: \widehat{\mathscr{Z}}(V)_{\leq \mu'} \twoheadrightarrow (\mathscr{W}_{\mu'}^{\wedge})^{\oplus r}$ is a morphism

$$f^{\vee}: (\mathscr{W}^{\wedge}_{-\mu'})^{\oplus r} \to \widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\mu'}$$

which has to factor through a morphism

$$(\mathscr{W}^{\wedge}_{-\mu'})^{\oplus r} \to \operatorname{gr}^{\wedge}_{-\mu'}(\widehat{\mathscr{Z}}(V^*)).$$

It is clear that the image of this morphism under π_{\dagger} is the isomorphism

$$\mathscr{W}_{-\mu'}^{\oplus r} \to \operatorname{gr}_{-\mu'}(\mathscr{Z}(V^*))$$

considered in (7.10); since the functor π_{\dagger} is conservative (see §6.1) this implies that our morphism is also invertible, and as in the I-equivariant setting we deduce a canonical isomorphism

$$\left(\mathrm{gr}_{\mu'}^{\wedge}(\widehat{\mathscr{Z}}(V))\right)^{\vee}\cong \mathrm{gr}_{-\mu'}(\widehat{\mathscr{Z}}(V^*)).$$

Once this is established, the same arguments as above allow to prove that $\widehat{\mathscr{Z}}(V)_{<\mu}$ is left dualizable, with left dual $\widehat{\mathscr{Z}}(V^*)/\widehat{\mathscr{Z}}(V^*)_{<-\mu}$, and that the dual of the embedding $\widehat{\mathscr{Z}}(V)_{\leq \mu} \hookrightarrow \widehat{\mathscr{Z}}(V)$ is the surjection $\widehat{\mathscr{Z}}(V^*) \twoheadrightarrow \widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\mu}$. Now we fix μ , and prove by upward induction on λ that the object

$$\mathscr{Z}(V)_{\leq \mu}/\mathscr{Z}(V)_{\leq \lambda}$$
, resp. $\widehat{\mathscr{Z}}(V)_{\leq \mu}/\widehat{\mathscr{Z}}(V)_{\leq \lambda}$,

is left dualizable, with left dual

$$\mathscr{Z}(V^*)_{<-\lambda}/\mathscr{Z}(V^*)_{<-\mu}, \quad \text{resp.} \quad \widehat{\mathscr{Z}}(V^*)_{<-\lambda}/\widehat{\mathscr{Z}}(V^*)_{<-\mu}.$$

The two cases are similar, so we only treat the second one. We consider the exact sequence

$$\widehat{\mathscr{Z}}(V)_{\leq \lambda} \hookrightarrow \widehat{\mathscr{Z}}(V)_{\leq \mu} \twoheadrightarrow \widehat{\mathscr{Z}}(V)_{\leq \mu} / \widehat{\mathscr{Z}}(V)_{\leq \lambda}.$$

We now know that the first two terms here are left dualizable; moreover the dual of the composition of the first map with the embedding $\widehat{\mathscr{Z}}(V)_{<\mu} \hookrightarrow \widehat{\mathscr{Z}}(V)$ is the surjection $\widehat{\mathscr{Z}}(V^*) \twoheadrightarrow \mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\lambda}$, with the dual of the latter map being the surjection $\widehat{\mathscr{Z}}(V^*) \twoheadrightarrow \mathscr{Z}(V^*) / \mathscr{Z}(V^*)_{<-\mu}$; by compatibility of duality with composition this implies that the dual of this first map is the surjection $\mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\mu} \twoheadrightarrow \mathscr{Z}(V^*)/\mathscr{Z}(V^*)_{<-\lambda}$. From this claim we deduce that the composition

$$\begin{split} \delta^{\wedge} \xrightarrow{\operatorname{coev}} \widehat{\mathscr{Z}}(V)_{\leq \mu} \widehat{\star} \left(\widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\mu} \right) \\ \to \left(\widehat{\mathscr{Z}}(V)_{\leq \mu} / \widehat{\mathscr{Z}}(V)_{\leq \lambda} \right) \widehat{\star} \left(\widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\mu} \right) \\ \to \left(\widehat{\mathscr{Z}}(V)_{\leq \mu} / \widehat{\mathscr{Z}}(V)_{\leq \lambda} \right) \widehat{\star} \left(\widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\lambda} \right) \end{split}$$

vanishes; it follows that the composition of the first two maps factors through a morphism

$$\delta^{\wedge} \to \left(\widehat{\mathscr{X}}(V)_{\leq \mu}/\widehat{\mathscr{X}}(V)_{\leq \lambda}\right) \widehat{\star} \left(\widehat{\mathscr{X}}(V^*)_{<-\lambda}/\widehat{\mathscr{X}}(V^*)_{<-\mu}\right).$$

Similarly, from the evaluation morphism

$$\left(\widehat{\mathscr{Z}}(V^*)/\widehat{\mathscr{Z}}(V^*)_{<-\mu}\right)\widehat{\star}\widehat{\mathscr{Z}}(V)_{\leq\mu}\to\delta^{\wedge}$$

we obtain a morphism

$$\left(\widehat{\mathscr{Z}}(V^*)_{<-\lambda}/\widehat{\mathscr{Z}}(V^*)_{<-\mu}\right)\widehat{\star}\left(\widehat{\mathscr{Z}}(V)_{\leq\mu}/\widehat{\mathscr{Z}}(V)_{\leq\lambda}\right)\to\delta^{\wedge}.$$

It is easily seen that, taken together, these maps exhibit $\widehat{\mathscr{Z}}(V^*)_{<-\lambda}/\widehat{\mathscr{Z}}(V^*)_{<-\mu}$ as the left dual of $\widehat{\mathscr{Z}}(V)_{\leq \mu}/\widehat{\mathscr{Z}}(V)_{\leq \lambda}$, which finishes the proof. \Box

Remark 7.13. From the proof of Lemma 7.11 we see that if $\lambda \leq \lambda' \leq \mu' \leq \mu$, the dual of the embedding

$$\widehat{\mathscr{Z}}(V)_{\leq \mu'}/\widehat{\mathscr{Z}}(V)_{\leq \lambda} \hookrightarrow \widehat{\mathscr{Z}}(V)_{\leq \mu}/\widehat{\mathscr{Z}}(V)_{\leq \lambda}$$

is the projection

$$\widehat{\mathscr{Z}}(V^*)_{<-\lambda}/\widehat{\mathscr{Z}}(V^*)_{<-\mu} \twoheadrightarrow \widehat{\mathscr{Z}}(V^*)_{<-\lambda}/\widehat{\mathscr{Z}}(V^*)_{<-\mu'},$$

and the dual of the projection

$$\widehat{\mathscr{Z}}(V)_{\leq \mu}/\widehat{\mathscr{Z}}(V)_{\leq \lambda}\twoheadrightarrow \widehat{\mathscr{Z}}(V)_{\leq \mu}/\widehat{\mathscr{Z}}(V)_{\leq \lambda'}$$

is the embedding

$$\widehat{\mathscr{Z}}(V^*)_{<-\lambda'}/\widehat{\mathscr{Z}}(V^*)_{<-\mu} \hookrightarrow \widehat{\mathscr{Z}}(V^*)_{<-\lambda}/\widehat{\mathscr{Z}}(V^*)_{<-\mu'}$$

7.8. **Proof of Lemma 7.10.** In order to give the proof of Lemma 7.10 we need another lemma.

Lemma 7.14. For any $\lambda \in X_*(T)$, we have

$$\mathbf{r}^{\mathrm{L}}(\mu_{\mathscr{W}^{\wedge}}^{\mathrm{rot}}(x^{-1})) = e^{\lambda}.$$

Proof. By definition, $\operatorname{tr}^{\mathrm{L}}(\mu_{\mathscr{W}_{\lambda}}^{\mathrm{rot}}(x^{-1}))$ is the composition

$$\delta^{\wedge} \xrightarrow[]{\operatorname{coev}} \mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \mathscr{W}^{\wedge}_{-\lambda} \xrightarrow[]{\mathscr{W}^{\wedge}_{\lambda}(x^{-1}) \mathbin{\widehat{\star}} \operatorname{id}} \mathscr{W}^{\wedge}_{\lambda} \mathbin{\widehat{\star}} \mathscr{W}^{\wedge}_{-\lambda} \xrightarrow[]{\operatorname{ev}} \delta^{\wedge}.$$

Here by Lemma 7.2(6) we have $\mu_{\mathscr{W}^{\wedge}_{\lambda}}^{\text{rot}}(x^{-1}) = \mu_{\mathscr{W}^{\wedge}_{\lambda}}(e^{\lambda} \otimes 1)$, so that the middle map is $\mu_{\mathscr{W}^{\wedge}_{\lambda} \widehat{\star} \mathscr{W}^{\wedge}_{-\lambda}}(e^{\lambda} \otimes 1)$ by (6.7). By functoriality of monodromy this implies that the composition above is $\mu_{\delta^{\wedge}}(e^{\lambda} \otimes 1)$, which implies the desired claim.

A general result about monoidal categories states that in an abelian monoidal category with exact monoidal product, the quantum trace is additive on short exact sequences (for morphisms compatible with the exact sequence), see [EGNO, Proposition 4.7.5]. As in Remark 7.12 this statement does not apply directly in our setting, but our proof of Lemma 7.10 will consist of repeating its proof³ and using Lemma 7.14 to compute the appropriate trace by induction.

Proof of Lemma 7.10. By Theorem 7.8(6) we have

$$\widehat{\mathsf{m}}_V = \mu_{\widehat{\mathscr{Z}}(V)}^{\mathrm{rot}}(x^{-1}).$$

We will prove by induction on λ that

$$\operatorname{tr}^{\mathcal{L}}\left(\mu_{\widehat{\mathscr{Z}}(V)\leq\lambda}^{\operatorname{rot}}(x^{-1})\right) = \sum_{\mu\leq\lambda} \dim(V_{\mu}) \cdot e^{\mu};$$

this will imply the desired equality by taking λ such that $\nu \leq \lambda$ for any ν such that $V_{\nu} \neq 0$.

If λ satisfies $\lambda < \nu$ for any ν such that $V_{\nu} \neq 0$, then this equality holds since both sides vanish. Now let $\lambda \in X_*(T)$, and assume the equality is known for the predecessor λ' of λ . We consider the exact sequence

$$\widehat{\mathscr{Z}}(V)_{\leq \lambda'} \hookrightarrow \widehat{\mathscr{Z}}(V)_{\leq \lambda} \twoheadrightarrow \mathrm{gr}_{\lambda}(\widehat{\mathscr{Z}}(V)).$$

³This proof was kindly explained to us by P. Etingof.

By functoriality of monodromy, the automorphism $\mu_{\widehat{\mathscr{Z}}(V)\leq\lambda}^{\operatorname{rot}}(x^{-1})$ of $\widehat{\mathscr{Z}}(V)_{\leq\lambda}$ preserves $\widehat{\mathscr{Z}}(V)_{\leq\lambda'}$, and restricts to $\mu_{\widehat{\mathscr{Z}}(V)\leq\lambda'}^{\operatorname{rot}}(x^{-1})$ on this subobject. Moreover, the induced automorphism of $\operatorname{gr}_{\lambda}(\widehat{\mathscr{Z}}(V))$ is $\mu_{\operatorname{gr}_{\lambda}(\widehat{\mathscr{Z}}(V))}^{\operatorname{rot}}(x^{-1})$.

The object

$$\widehat{\mathscr{Z}}(V)_{\leq\lambda} \mathbin{\widehat{\star}} (\widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\lambda}) = \widehat{\mathscr{Z}}(V)_{\leq\lambda} \mathbin{\widehat{\star}} (\widehat{\mathscr{Z}}(V)_{\leq\lambda})^{\vee}$$

admits a canonical 3-step filtration

$$\mathscr{M}_1 \subset \mathscr{M}_2 \subset \widehat{\mathscr{Z}}(V)_{\leq \lambda} \mathbin{\widehat{\star}} (\widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\lambda})$$

with successive associated subquotients given by

$$\begin{split} \widehat{\mathscr{X}}(V)_{\leq\lambda'} &\stackrel{\scriptstyle \widehat{\star}}{\Rightarrow} \operatorname{gr}_{-\lambda}(\widehat{\mathscr{X}}(V^*)), \\ & \widehat{\mathscr{X}}(V)_{\leq\lambda'} \stackrel{\scriptstyle \widehat{\star}}{\star} (\widehat{\mathscr{X}}(V^*)/\widehat{\mathscr{X}}(V^*)_{<-\lambda'}) \oplus \operatorname{gr}_{\lambda}(\widehat{\mathscr{X}}(V)) \stackrel{\scriptstyle \widehat{\star}}{\star} \operatorname{gr}_{-\lambda}(\widehat{\mathscr{X}}(V^*)) \\ & \text{ and } \operatorname{gr}_{\lambda}(\widehat{\mathscr{X}}(V)) \stackrel{\scriptstyle \widehat{\star}}{\star} (\widehat{\mathscr{X}}(V^*)/\widehat{\mathscr{X}}(V^*)_{<-\lambda'}). \end{split}$$

We have

$$\begin{split} \operatorname{Hom}(\delta^{\wedge}, \operatorname{gr}_{\lambda}(\widehat{\mathscr{Z}}(V)) \mathbin{\widehat{\star}} (\widehat{\mathscr{Z}}(V^{*}) / \widehat{\mathscr{Z}}(V^{*})_{<-\lambda'})) \\ &\cong \operatorname{Hom}(\operatorname{gr}_{-\lambda}(\widehat{\mathscr{Z}}(V^{*})), \widehat{\mathscr{Z}}(V^{*}) / \widehat{\mathscr{Z}}(V^{*})_{<-\lambda'}) = 0, \end{split}$$

since $(\widehat{\mathscr{Z}}(V^*)/\widehat{\mathscr{Z}}(V^*)_{\leq -\lambda'})_{\leq -\lambda} = 0$; it follows that the coevaluation map

$$\delta^{\wedge} \to \widehat{\mathscr{Z}}(V)_{\leq \lambda} \mathbin{\widehat{\star}} (\widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\lambda})$$

factors through a map $\delta^{\wedge} \to \mathscr{M}_2$. For similar reasons, the evaluation map

$$\widehat{\mathscr{Z}}(V)_{\leq \lambda} \mathbin{\widehat{\star}} (\widehat{\mathscr{Z}}(V^*)/\widehat{\mathscr{Z}}(V^*)_{<-\lambda}) \to \delta^{\wedge}$$

vanishes on \mathcal{M}_1 , hence factors through a morphism

$$\left(\widehat{\mathscr{Z}}(V)_{\leq\lambda} \mathbin{\widehat{\star}} (\widehat{\mathscr{Z}}(V^*)/\widehat{\mathscr{Z}}(V^*)_{<-\lambda})\right)/\mathscr{M}_1 \to \delta^{\wedge}.$$

It follows that our trace is the composition

$$\begin{split} \delta^{\wedge} &\to \widehat{\mathscr{Z}}(V)_{\leq \lambda'} \mathbin{\widehat{\star}} (\widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\lambda'}) \oplus \operatorname{gr}_{\lambda}(\widehat{\mathscr{Z}}(V)) \mathbin{\widehat{\star}} \operatorname{gr}_{-\lambda}(\widehat{\mathscr{Z}}(V^*)) \\ &\to \widehat{\mathscr{Z}}(V)_{\leq \lambda'} \mathbin{\widehat{\star}} (\widehat{\mathscr{Z}}(V^*) / \widehat{\mathscr{Z}}(V^*)_{<-\lambda'}) \oplus \operatorname{gr}_{\lambda}(\widehat{\mathscr{Z}}(V)) \mathbin{\widehat{\star}} \operatorname{gr}_{-\lambda}(\widehat{\mathscr{Z}}(V^*)) \to \delta^{\wedge} \end{split}$$

where the first, resp. third, map is the sum of the coevaluation, resp. evaluation, morphisms for $\widehat{\mathscr{X}}(V)_{\leq \lambda'}$ and $\operatorname{gr}_{\lambda}(\widehat{\mathscr{X}}(V))$, and the middle arrow is the map induced by $\mu^{\operatorname{rot}}_{\widehat{\mathscr{X}}(V)\leq \lambda}(x^{-1}) \stackrel{\sim}{\star} \operatorname{id}$, i.e. the direct sum

$$\mu^{\mathrm{rot}}_{\widehat{\mathscr{Z}}(V)_{\leq \lambda'}}(x^{-1}) \mathbin{\widehat{\star}} \mathrm{id} \oplus \mu^{\mathrm{rot}}_{\mathrm{gr}_{\lambda}(\widehat{\mathscr{Z}}(V))}(x^{-1}) \mathbin{\widehat{\star}} \mathrm{id}.$$

We deduce that

$$\operatorname{tr}^{\mathcal{L}}(\mu_{\widehat{\mathscr{Z}}(V)_{\leq \lambda}}^{\operatorname{rot}}(x^{-1})) = \operatorname{tr}^{\mathcal{L}}(\mu_{\widehat{\mathscr{Z}}(V)_{\leq \lambda'}}^{\operatorname{rot}}(x^{-1})) + \operatorname{tr}^{\mathcal{L}}(\mu_{\operatorname{gr}_{\lambda}}^{\operatorname{rot}}(\widehat{\mathscr{Z}}(V))}(x^{-1})),$$

which implies the desired formula by the induction hypothesis and Lemma 7.14. \Box

R. BEZRUKAVNIKOV AND S. RICHE

8. Perverse sheaves on G/U

We continue with the setting of Sections 4–7, and consider also the constructions of Section 2 in the case $\mathbf{G} = G_{\Bbbk}^{\vee}$ (with the Borel subgroup B_{\Bbbk}^{\vee} and the maximal torus T_{\Bbbk}^{\vee}). It is clear that in this case the Coxeter system ($\mathbf{W}_{\mathrm{f}}, \mathbf{S}_{\mathrm{f}}$) of Section 2 identifies with the Coxeter system ($W_{\mathrm{f}}, S_{\mathrm{f}}$) of Section 4. We will assume in this section that G_{\Bbbk}^{\vee} has simply connected derived subgroup, or in other words that the quotient of $X^*(T)$ by the root lattice is free.

Before constructing the main equivalence of the paper, we explain a similar construction for perverse sheaves on the "finite" flag variety G/B (or, in fact, on the basic affine space G/U). This construction is essentially a reinterpretation of the main result of [BR1]; it will serve as a "toy example" to illustrate our methods, but will also play a role in the proof of the theorem.

Remark 8.1. As explained above, the proofs in this section rely on the results of [BR1]. In this reference it is assumed that the group G is semisimple of adjoint type, but all the proofs apply more generally under the present assumption that the dual group has simply connected derived subgroup. (In fact, the main ingredient that requires some assumption is Theorem 2.1, which holds under our assumption by Remark 2.1.)

8.1. Categories of sheaves on G/U. Recall the categories $\mathsf{D}_{U,U}$ and $\mathsf{D}_{U,U}^{\wedge}$ considered in §6.6. These categories admit perverse t-structures, whose hearts are denoted $\mathsf{P}_{U,U}$ and $\mathsf{P}_{U,U}^{\wedge}$ respectively. In fact, pushforward along the closed embedding $G/U \hookrightarrow \widetilde{\mathsf{Fl}}_G$ identifies $\mathsf{P}_{U,U}$, resp. $\mathsf{D}_{U,U}$ with the Serre subcategory of $\mathsf{P}_{\mathsf{I}_u,\mathsf{I}_u}$, resp. the full triangulated subcategory of $\mathsf{D}_{\mathsf{I}_u,\mathsf{I}_u}$, generated by the simple objects $\pi^{\dagger}\mathsf{For}_{\mathsf{I}_u}(\mathscr{I}_w)$ with $w \in W_{\mathsf{f}}$; it also provides a t-exact fully faithful functor $\mathsf{D}_{U,U}^{\wedge} \to \mathsf{D}_{\mathsf{f}_u,\mathsf{I}_u}^{\wedge}$. If we denote by $\mathsf{P}_{U,U}^+$, resp. $\mathsf{D}_{U,U}^+$, the Serre subcategory of $\mathsf{P}_{U,U}$, resp. the full triangulated subcategory of $\mathsf{D}_{U,U}$, generated by the objects $\pi^{\dagger}\mathsf{For}_{\mathsf{I}_u}(\mathscr{I}_w)$ with $w \in W_{\mathsf{f}} \setminus \{e\}$, then we can consider the quotient categories

$$\mathsf{P}^0_{U,U} := \mathsf{P}_{U,U} / \mathsf{P}^+_{U,U}, \quad \mathsf{D}^0_{U,U} := \mathsf{D}_{U,U} / \mathsf{D}^+_{U,U}.$$

By Lemma A.2, there exists a unique t-structure on $\mathsf{D}^0_{U,U}$ such that the quotient functor

$$\Pi^0_{U,U}: \mathsf{D}_{U,U} \to \mathsf{D}^0_{U,U}$$

is t-exact. This t-structure is bounded, and its heart identifies with $\mathsf{P}^0_{U,U}$; it will be called the perverse t-structure, and the associated cohomology functors will once again be denoted ${}^{\mathrm{p}}\mathscr{H}^n(-)$. We have a canonical t-exact functor

$$(8.1) D^0_{U,U} \to D^0_{I_u,I_u},$$

whose restriction to the heart of the perverse t-structure is fully faithful.

As for $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, the category $\mathsf{D}_{U,U}$ admits a natural convolution product \star_U which equips it with the structure of a monoidal category (without unit object) such that the embedding $\mathsf{D}_{U,U} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is monoidal, and which induces (in the appropriate sense) the product $\hat{\star}_U$. We also have a canonical bifunctor

$$\widehat{\star}_U : \mathsf{D}_{U,U}^\wedge \times \mathsf{D}_{U,U} \to \mathsf{D}_{U,U}$$

which defines an action of $(\mathsf{D}_{U,U}^{\wedge}, \widehat{\star}_U)$ on $\mathsf{D}_{U,U}$.

As in §5.2, the bifunctor $(\mathscr{F}, \mathscr{G}) \mapsto \Pi^0_{U,U}(\mathscr{F} \star_U \mathscr{G})$ factors through a triangulated bifunctor

$$\star^0_U:\mathsf{D}^0_{U,U}\times\mathsf{D}^0_{U,U}\to\mathsf{D}^0_{U,U}$$

which defines a monoidal structure (without unit object) on $\mathsf{D}_{U,U}^0$ so that (8.1) is monoidal. Moreover, \star_U^0 is "right t-exact" in the sense that if \mathscr{F}, \mathscr{G} belong to the nonpositive part of the perverse t-structure on $\mathsf{D}_{U,U}^0$ then so does $\mathscr{F} \star_U^0 \mathscr{G}$. We therefore obtain a monoidal structure (without unit object) on $\mathsf{P}_{U,U}^0$ by setting

$$\mathscr{F}^{\mathrm{p}}\star^{0}_{U}\mathscr{G}:={}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F}\star^{0}_{U}\mathscr{G})$$

for \mathscr{F}, \mathscr{G} in $\mathsf{P}^0_{U,U}$; then we have a fully faithful exact monoidal functor

(8.2)
$$(\mathsf{P}^{0}_{U,U}, {}^{\mathsf{p}}\!\star^{0}_{U}) \to (\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}, {}^{\mathsf{p}}\!\star^{0}_{\mathrm{I}_{\mathrm{u}}})$$

As in $\S6.4$ we have a canonical bifunctor

$$\widehat{\star}^0_U:\mathsf{D}^\wedge_{U,U} imes\mathsf{D}^0_{U,U} o\mathsf{D}^0_{U,U}$$

compatible with $\widehat{\star}^0$ in the obvious way, and which defines an action of $(\mathsf{D}_{U,U}^{\wedge}, \widehat{\star}_U)$ on the category $\mathsf{D}_{U,U}^0$. For \mathscr{F} in $\mathsf{P}_{U,U}^{\wedge}$ and \mathscr{G} in $\mathsf{P}_{U,U}^0$ we then set

$$\mathscr{F}^{\mathrm{p}}\widehat{\star}^{0}_{U}\mathscr{G}:={}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F}\widehat{\star}^{0}_{U}\mathscr{G}).$$

As in Lemma 6.9, this bifunctor is right exact on each side.

Recall also the functor C_m considered in §6.5. It is clear that this functor restricts to a functor from $\mathsf{P}_{U,U}^{\wedge}$ to $\mathsf{P}_{U,U}$, which will again be denoted C_m .

8.2. Morphisms from $\Pi^0_{U,U}(\Xi^{\wedge}_!)$. Recall the object $\Xi^{\wedge}_!$ defined in §6.6. Considering this object as a pro-object in $\mathsf{D}_{U,U}$, and applying the extension to pro-objects of the functor $\Pi^0_{U,U}$ we obtain a pro-object $\Pi^0_{U,U}(\Xi^{\wedge}_!)$ in $\mathsf{D}^0_{U,U}$. We can then consider the functor from $\mathsf{P}^0_{U,U}$ to the category of k-vector spaces given by

$$\mathscr{F} \mapsto \operatorname{Hom}_{\mathsf{D}^0_{U\,U}}(\Pi^0_{U,U}(\Xi^\wedge_!), \mathscr{F}),$$

where in the right-hand side we mean morphisms in the category of pro-objects in $\mathsf{D}^0_{U,U}$. Concretely, if we write $\Xi^{\wedge}_! = \underset{n}{\overset{\circ}{\longleftarrow}} \mathscr{A}_n$ for some objects \mathscr{A}_n in $\mathsf{D}_{U,U}$, then we have

$$\operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\Xi^{\wedge}_{!}),\mathscr{F}) = \varinjlim_{n} \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\mathscr{A}_{n}),\mathscr{F}).$$

Similarly, given \mathscr{G} in $\mathsf{P}_{U,U}$ we can consider the vector space

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi^{\wedge}_{!},\mathscr{G}) = \varinjlim_{n} \operatorname{Hom}_{\mathsf{D}_{U,U}}(\mathscr{A}_{n},\mathscr{G}).$$

Lemma 8.2. For any \mathscr{G} in $\mathsf{P}_{U,U}$, the canonical morphism

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi^{\wedge}_{!},\mathscr{G}) \to \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi^{0}_{U,U}(\Xi^{\wedge}_{!}),\Pi^{0}_{U,U}(\mathscr{G}))$$

is an isomorphism.

Proof. Fix \mathscr{G} in $\mathsf{P}_{U,U}$. As explained above, writing $\Xi_!^{\wedge} = \underset{n}{``} \mathscr{A}_n$ for some \mathscr{A}_n in $\mathsf{D}_{U,U}$, our morphism can be written more concretely as the morphism

(8.3)
$$\varinjlim_{n} \operatorname{Hom}_{\mathsf{D}_{U,U}}(\mathscr{A}_{n},\mathscr{G}) \to \varinjlim_{n} \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\mathscr{A}_{n}), \Pi_{U,U}^{0}(\mathscr{G}))$$

induced by the functor $\Pi^0_{U,U}$.

Let us first show that (8.3) is surjective. A morphism in the right-hand side is represented by a morphism $\Pi^0_{U,U}(\mathscr{A}_n) \to \Pi^0_{U,U}(\mathscr{G})$ in $\mathsf{D}^0_{U,U}$ for some n, i.e. by a diagram

$$\mathscr{A}_n \xleftarrow{s} \mathscr{X} \xrightarrow{f} \mathscr{G}$$

where \mathscr{X} is an object in $\mathsf{D}_{U,U}$ and s, f are morphisms in this category such that the cone \mathscr{C} of s belongs to $\mathsf{D}^+_{U,U}$. Now, by Lemma 6.10 we have

$$\operatorname{Hom}_{\mathsf{D}_{U_{U}}^{\wedge}}(\Xi_{!}^{\wedge},\pi^{\dagger}(\mathscr{I}\mathscr{C}_{w})[i])=0$$

for any $w \in W_{\mathbf{f}} \setminus \{e\}$ and $i \in \mathbb{Z}$; it follows that

$$\operatorname{Hom}_{\mathsf{D}_{UU}^{\wedge}}(\Xi^{\wedge}_{!},\mathscr{C})=0,$$

or in other words that

$$\varinjlim_{m} \operatorname{Hom}_{\mathsf{D}_{U,U}}(\mathscr{A}_{m}, \mathscr{C}) = 0.$$

We deduce that for $m \gg n$ the composition $\mathscr{A}_m \to \mathscr{A}_n \to \mathscr{C}$ vanishes. Fix such an m, and denote by h the structure morphism $\mathscr{A}_m \to \mathscr{A}_n$. If we complete the morphisms s and h to a commutative diagram



in $\mathsf{D}_{U,U}$ such that the cone of t belongs to $\mathsf{D}^+_{U,U}$, then the image of our morphism in $\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\mathscr{A}_m), \Pi^0_{U,U}(\mathscr{G}))$ is represented by the diagram

$$\mathscr{A}_m \xleftarrow{t} \mathscr{Y} \xrightarrow{f \circ g} \mathscr{G}.$$

Now since the composition $\mathscr{A}_m \to \mathscr{A}_n \to \mathscr{C}$ vanishes, there exists $k : \mathscr{A}_m \to \mathscr{X}$ in $\mathsf{D}_{U,U}$ such that $s \circ k = h$. Then in $\mathsf{D}_{U,U}^0$ we have

$$f \circ g \circ t^{-1} = f \circ s^{-1} \circ h = f \circ k.$$

This shows that the image of our morphism in $\operatorname{Hom}_{\mathsf{D}_{U,U}^0}(\Pi^0_{U,U}(\mathscr{A}_m), \Pi^0_{U,U}(\mathscr{G}))$ is the image of a morphism in $\operatorname{Hom}_{\mathsf{D}_{U,U}}(\mathscr{A}_m, \mathscr{G})$, which finishes the proof of surjectivity.

The proof of injectivity is similar. If a morphism $f : \mathscr{A}_n \to \mathscr{G}$ has trivial image in $\varinjlim_{m} \operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\mathscr{A}_m), \Pi^0_{U,U}(\mathscr{G}))$, then composing with the morphism $\mathscr{A}_m \to \mathscr{A}_n$ for some $m \ge n$ we can assume that $\Pi^0_{U,U}(f) = 0$. This means that there exists a morphism $g : \mathscr{G} \to \mathscr{H}$ whose cone \mathscr{C} belongs to $\mathsf{D}^+_{U,U}$ such that $g \circ f = 0$. Then f factors through a morphism $\mathscr{A}_n \to \mathscr{C}[-1]$. Replacing again n by a larger morphism we can assume that this morphism vanishes, so that f = 0 in the inductive limit, which finishes the proof. \Box

8.3. Towards a monoidal structure. Recall the scheme D from §2.4; in our present setting we have

$$\mathbf{D} := T^{\vee}_{\Bbbk} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} T^{\vee}_{\Bbbk}.$$

For any \mathscr{F} in $\mathsf{P}^0_{U,U}$, monodromy for the action of T on the left and on the right on G/U equips $\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\Xi^\wedge_!),\mathscr{F})$ with the structure of an $\mathscr{O}(T^\vee_{\Bbbk})$ -bimodule. In fact, it follows from Lemma 6.7 that these actions factor through a structure of $\mathscr{O}(\mathbf{D})$ -module. Our goal in this subsection is to explain how to define, for \mathscr{F}, \mathscr{G} in $\mathsf{P}^0_{U,U}$, a canonical morphism

$$(8.4) \quad \operatorname{Hom}_{\mathsf{D}^{0}_{U,U}}(\Pi^{0}_{U,U}(\Xi^{\wedge}_{!}),\mathscr{F}) \otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})} \operatorname{Hom}_{\mathsf{D}^{0}_{U,U}}(\Pi^{0}_{U,U}(\Xi^{\wedge}_{!}),\mathscr{G}) \\ \rightarrow \operatorname{Hom}_{\mathsf{D}^{0}_{U,U}}(\Pi^{0}_{U,U}(\Xi^{\wedge}_{!}),\mathscr{F}^{\mathsf{p}} \star^{0}_{U} \mathscr{G}),$$

which will eventually be shown to define a monoidal structure on the functor $\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\Xi^\wedge_1), -)$. (Here $\mathscr{O}(T^\vee_{\Bbbk})$ acts on $\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\Xi^\wedge_1), \mathscr{F})$ via the projection $\mathbf{D} \to T^\vee_{\Bbbk}$ on the second factor, and on $\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\Xi^\wedge_1), \mathscr{F})$ via the projection $\mathbf{D} \to T^\vee_{\Bbbk}$ on the first factor.) To explain this construction we first need to recall a similar construction from [BR1].

First, consider the scheme \mathbf{D}^{\wedge} considered in §3.3. By Lemma 3.3(2) this scheme is the spectrum of the algebra

$$\mathscr{O}((T^{\vee}_{\Bbbk})^{\vee}) \otimes_{\mathscr{O}((T^{\vee}_{\Bbbk})^{\vee})^{W_{\mathrm{f}}}} \mathscr{O}((T^{\vee}_{\Bbbk})^{\vee})$$

that appears e.g. in [BR1, Theorem 9.1]. (Section 3 is written under the running assumption that $Z(\mathbf{G})$ is smooth, but this condition is not required for this specific lemma.)

Recall the category $\mathsf{T}_{U,U}^{\wedge}$ of tilting objects in $\mathsf{P}_{U,U}^{\wedge}$. As explained in [BR1, Remark 7.9], this subcategory is closed under the convolution product $\widehat{\star}_U$. For \mathscr{F} in $\mathsf{P}_{U,U}^{\wedge}$, in [BR1] we explain that monodromy defines on $\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_!^{\wedge},\mathscr{F})$ the structure of a finitely generated $\mathscr{O}(\mathbf{D}^{\wedge})$ -module. Moreover, in [BR1, §11.3] we construct a monoidal structure on the functor

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi^{\wedge}_{!},-):\mathsf{T}_{U,U}^{\wedge}\to\mathsf{Mod}^{\operatorname{fg}}(\mathscr{O}(\mathbf{D}^{\wedge})),$$

for the monoidal product on $\mathsf{T}_{U,U}^{\wedge}$ given by $\widehat{\star}_U$, and that on $\mathscr{O}(\mathbf{D}^{\wedge})$ -modules given by tensor product over $\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})$. Using this structure we obtain an isomorphism (8.5) $\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge},\Xi_{!}^{\wedge},\Xi_{!}^{\wedge}) \cong \operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge},\Xi_{!}^{\wedge}) \otimes_{\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})} \operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge},\Xi_{!}^{\wedge})$. Here the right-hand side has a canonical element, given by $\operatorname{id}_{\Xi_{1}^{\wedge}} \otimes \operatorname{id}_{\Xi_{1}^{\wedge}}$, which then defines a canonical morphism $\xi : \Xi_{!}^{\wedge} \to \Xi_{!}^{\wedge} \widehat{\star}_{U} \Xi_{!}^{\wedge}$. Concretely, writing $\Xi_{!}^{\wedge} =$ " \varprojlim_{n} " \mathscr{A}_{n} for some \mathscr{A}_{n} in $\mathsf{D}_{U,U}$ as in §8.2, we have

$$\Xi_{!}^{\wedge} \,\widehat{\star}_{U} \,\Xi_{!}^{\wedge} = " \varprojlim_{n, m \geq 0} " \,\mathscr{A}_{n} \,\star_{U} \,\mathscr{A}_{m},$$

so that

F

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge},\Xi_{!}^{\wedge}\widehat{\star}_{U}\Xi_{!}^{\wedge}) = \varprojlim_{n,m} \varinjlim_{q} \operatorname{Hom}_{\mathsf{D}_{U,U}}(\mathscr{A}_{q},\mathscr{A}_{n}\star_{U}\mathscr{A}_{m});$$

our morphism is therefore defined by a collection of morphisms $\xi_{n,m} : \mathscr{A}_{q(n,m)} \to \mathscr{A}_n \star_U \mathscr{A}_m$ for some function $q : (\mathbb{Z}_{\geq 0})^2 \to \mathbb{Z}_{\geq 0}$, which we fix from now on.

Finally we can explain the construction of (8.4). Consider some elements f in $\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\Xi^\wedge_!),\mathscr{F})$ and g in $\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\Xi^\wedge_!),\mathscr{G})$, represented by morphisms $f: \Pi^0_{U,U}(\mathscr{A}_n) \to \mathscr{F}$ and $g: \Pi^0_{U,U}(\mathscr{A}_m) \to \mathscr{G}$. Then the image of $f \otimes g$ under (8.4) is the composition

$$\begin{split} \Pi^{0}_{U,U}(\mathscr{A}_{q(n,m)}) \xrightarrow{\Pi^{0}_{U,U}(\xi_{n,m})} \Pi^{0}_{U,U}(\mathscr{A}_{n} \star_{U} \mathscr{A}_{m}) &= \Pi^{0}_{U,U}(\mathscr{A}_{n}) \star^{0}_{U} \Pi^{0}_{U,U}(\mathscr{A}_{m}) \\ \xrightarrow{f \star^{0}_{U} \mathscr{G}} \mathscr{F} \star^{0}_{U} \mathscr{G} \to \mathscr{F}^{\mathsf{P}} \star^{0}_{U} \mathscr{G} \end{split}$$

where the last morphism is the natural truncation morphism. (Recall that $\mathscr{F} \star^0_U \mathscr{G}$ belongs to the nonpositive part of the perverse t-structure, and $\mathscr{F} {}^{\mathrm{p}} \star^0_U \mathscr{G}$ is its degree-0 cohomology.)

8.4. **Statement.** We will denote by $Coh_0(\mathbf{D})$ the category of coherent sheaves on \mathbf{D} which are supported set-theoretically on the closed subscheme $\{(e, e)\}$, which we identify with the category of finitely generated $\mathscr{O}(\mathbf{D})$ -modules on which a power of the ideal \mathcal{I} (see §3.3), or equivalently a power of the ideal \mathcal{J} , acts trivially. This category is monoidal (without unit object) for the product \circledast defined by

$$M \circledast N := M \otimes_{\mathscr{O}(T_*^{\vee})} N.$$

(Here in the tensor product the action on M is induced by the projection $\mathbf{D} \to T_{\Bbbk}^{\vee}$ on the second factor, and the action on N is induced by projection on the first factor; the action of $\mathscr{O}(\mathbf{D})$ on the tensor product is the obvious one, defined in terms of the remaining actions.)

The following proposition is the promised "finite variant" of our main result. Its proof will be explained in the next subsection.

Theorem 8.3. The functor $\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\Xi^\wedge_!), -)$ induces an equivalence of monoidal abelian categories

$$\Phi_{U,U}: \left(\mathsf{P}^0_{U,U}, \star^0_U\right) \xrightarrow{\sim} \left(\mathsf{Coh}_0(\mathbf{D}), \circledast\right).$$

8.5. "Truncated" version. For $m \ge 1$ we set

$$\mathbf{D}^{(m)} := \operatorname{Spec}(\mathscr{O}(\mathbf{D})/\mathcal{J}^m \cdot \mathscr{O}(\mathbf{D})).$$

Pushforward along the closed embedding $\mathbf{D}^{(m)} \to \mathbf{D}$ provides a fully faithful functor

$$\mathsf{Coh}(\mathbf{D}^{(m)}) \to \mathsf{Coh}_0(\mathbf{D})$$

and it is clear from definitions that the product \circledast restricts to a monoidal product on $\mathsf{Coh}(\mathbf{D}^{(m)})$. In fact, this collection of functors realizes $\mathsf{Coh}_0(\mathbf{D})$ as the direct limit of its subcategories $\mathsf{Coh}(\mathbf{D}^{(m)})$, in a way compatible with the monoidal product.

On the constructible side, again for $m \geq 1$ we will denote by $\mathsf{P}_{U,U}^{(m)}$ the full abelian subcategory of $\mathsf{P}_{U,U}$ whose objects are the perverse sheaves such that the monodromy action of $\mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})$ (see §6.3) vanishes on \mathcal{J}^m . This subcategory contains all the simple objects of $\mathsf{P}_{U,U}$, and is stable under subquotients (but not under extensions). If we denote by $\mathsf{P}_{U,U}^{(m),0}$ the Serre quotient of $\mathsf{P}_{U,U}^{(m)}$ by the Serre subcategory generated by the simple objects $\pi^{\dagger}\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{I}\mathscr{C}_w)$ with $w \in W_{\mathrm{f}} \setminus \{e\}$, then we have a natural fully faithful functor

$$\mathsf{P}_{U,U}^{(m),0} \to \mathsf{P}_{U,U}^{0}$$

The essential image of this functor can be described as follows. It is clear that monodromy induces, for any \mathscr{F} in $\mathsf{P}^0_{U,U}$, a canonical morphism $\mu^0_{\mathscr{F}} : \mathscr{O}(\mathbf{D}) \to$ $\operatorname{End}_{\mathsf{P}^0_{U,U}}(\mathscr{F})$. Then the essential image of $\mathsf{P}^{(m),0}_{U,U}$ in $\mathsf{P}^0_{U,U}$ identifies with the subcategory consisting of objects such that $\mu^0_{\mathscr{F}}$ vanishes on \mathscr{J}^m . Indeed, any object in this essential image clearly satisfies this property. On the other hand, if $\mu^0_{\mathscr{F}}$ vanishes on \mathscr{J}^m , writing $\mathscr{F} = \Pi^0_{U,U}(\mathscr{G})$ for some \mathscr{G} in $\mathsf{P}_{U,U}$, we see that the surjection $\mathscr{G} \to \mathscr{G}/\mathscr{J}^m \cdot \mathscr{G}$ becomes an isomorphism after application of $\Pi^0_{U,U}$, and obviously $\mathscr{G}/\mathscr{J}^m \cdot \mathscr{G}$ belongs to $\mathsf{P}^{(m)}_{UU}$. Using (6.7)–(6.8) one sees that the convolution product \star^0_U restricts to a monoidal product on $\mathsf{P}^{(m),0}_{U,U}$. In this way we realize the category $\mathsf{P}^0_{U,U}$ as the direct limit of its subcategories $\mathsf{P}^{(m),0}_{U,U}$, in a way compatible with the monoidal product. It is clear that the restriction of the quotient functor $\Pi^0_{U,U}$ to $\mathsf{P}^{(m)}_{U,U}$ takes values in $\mathsf{P}^{(m),0}_{U,U}$, and identifies with the quotient functor $\mathsf{P}^{(m)}_{U,U} \to \mathsf{P}^{(m),0}_{U,U}$.

From these considerations we see that Theorem 8.3 is a corollary of the following statement.

Proposition 8.4. For any $m \ge 1$, the functor $\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\Xi^\wedge_!), -)$ induces an equivalence of abelian categories

$$\mathsf{P}_{U,U}^{(m),0} \xrightarrow{\sim} \mathsf{Coh}(\mathbf{D}^{(m)}).$$

Moreover these equivalences admit structures of monoidal functors compatible in the obvious way with the natural embeddings

$$\mathsf{P}_{U,U}^{(m),0} \to \mathsf{P}_{U,U}^{(m'),0}, \qquad \mathsf{Coh}(\mathbf{D}^{(m)}) \to \mathsf{Coh}(\mathbf{D}^{(m')})$$

when $m \leq m'$.

In the proof of this proposition we will consider, for $m \ge 1$, the object

$$\Xi_!^{(m)} := \mathsf{C}_m(\Xi_!^\wedge) \quad \in \mathsf{P}_{U,U}$$

It is clear that this object belongs to the subcategory $\mathsf{P}_{IIII}^{(m)}$.

Lemma 8.5. For any $m \ge 1$, there exists a canonical isomorphism

$$\Pi^0_{U,U} \left(\mathsf{C}_m(\Xi^\wedge_! \,\widehat{\star}_U \,\Xi^\wedge_!) \right) \cong \Pi^0_{U,U}(\Xi^{(m)}_!) \,{}^{\mathsf{p}} \star^0_U \,\Pi^0_{U,U}(\Xi^{(m)}_!)$$

Proof. Since the functor $\Xi_{1}^{\wedge} \widehat{\star}_{U}(-)$ is t-exact (see Lemma 6.11) the morphism

$$\Xi_{!}^{\wedge} \widehat{\star}_{U} \Xi_{!}^{\wedge} \to \Xi_{!}^{\wedge} \widehat{\star}_{U} \Xi_{!}^{(m)}$$

is surjective, and identifies the right-hand side with $C_m(\Xi_!^{\wedge} \widehat{\star}_U \Xi_!^{\wedge})$. On the other hand, by definition we have

$$\Pi^{0}_{U,U}(\Xi^{(m)}_{!}) \,{}^{\mathbf{p}} \star^{0}_{U} \,\Pi^{0}_{U,U}(\Xi^{(m)}_{!}) \cong {}^{\mathbf{p}} \mathscr{H}^{0}(\Pi^{0}_{U,U}(\Xi^{(m)}_{!} \star_{U} \Xi^{(m)}_{!})).$$

What we have to construct is therefore a canonical isomorphism

$$\Pi^0_{U,U}(\Xi^{\wedge}_! \widehat{\star}_U \Xi^{(m)}_!) \cong {}^{\mathbf{p}} \mathscr{H}^0(\Pi^0_{U,U}(\Xi^{(m)}_! \star_U \Xi^{(m)}_!)).$$

A choice of a family of r generators of the ideal \mathcal{J}^m defines an exact sequence

$$(\Xi^{\wedge}_{!})^{\oplus r} \to \Xi^{\wedge}_{!} \to \Xi^{(m)}_{!} \to 0.$$

Now the functor

$${}^{\mathbf{p}}\mathscr{H}^{0}(\Pi^{0}_{U,U}((-) \widehat{\star}_{U} \Xi^{(m)}_{!})) = (-) {}^{\mathbf{p}} \widehat{\star}^{0}_{U} \Pi^{0}_{U,U}(\Xi^{(m)}_{!}) : \mathsf{P}^{\wedge}_{U,U} \to \mathsf{P}^{0}_{U,U}$$

is right exact (see $\S8.1$); we therefore deduce an exact squence

$$\Pi^{0}_{U,U}((\Xi^{\wedge}_{!})^{\oplus r} \widehat{\star}_{U} \Xi^{(m)}_{!}) \to \Pi^{0}_{U,U}(\Xi^{\wedge}_{!} \widehat{\star}_{U} \Xi^{(m)}_{!}) \to {}^{\mathrm{p}}\mathscr{H}^{0}(\Pi^{0}_{U,U}(\Xi^{(m)}_{!} \star_{U} \Xi^{(m)}_{!})) \to 0.$$

Using (6.7)–(6.8) one sees that the first morphism in this sequence vanishes, which shows that the second morphism is an isomorphism and finishes the proof. \Box

Proof of Proposition 8.4. We claim that for $m \ge 1$ the morphism $\mu_{\Xi_{l}^{(m)}}$ factors through an isomorphism

(8.6)
$$\mathscr{O}(\mathbf{D}^{(m)}) \xrightarrow{\sim} \operatorname{End}_{\mathsf{P}_{U,U}}(\Xi_!^{(m)}).$$

In fact, since by definition the action of $\mathscr{O}(\mathbf{D})$ on $\Xi_1^{(m)}$ vanishes on \mathcal{J}^m we have

 $\operatorname{End}_{\mathsf{P}_{U,U}}(\Xi_!^{(m)}) \cong \operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_!^{\wedge},\Xi_!^{(m)}).$

Now by projectivity of $\Xi_{!}^{\wedge}$ (see Lemma 6.10) we have

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge},\Xi_{!}^{(m)})\cong\operatorname{End}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge})\otimes_{\mathscr{O}(\mathbf{D})}\mathscr{O}(\mathbf{D}^{(m)}).$$

Finally, by [BR1, Theorem 9.1] and the comments in §8.3, $\mu_{\Xi_1^{\wedge}}$ induces an algebra isomorphism

(8.7)
$$\mathscr{O}(\mathbf{D}^{\wedge}) \xrightarrow{\sim} \operatorname{End}(\Xi_{!}^{\wedge})$$

We deduce that (8.6) is an isomorphism, as desired.

For any \mathscr{F} in $\mathsf{P}_{U,U}^{(m)}$ we have

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi^{\wedge}_{!},\mathscr{F}) = \operatorname{Hom}_{\mathsf{P}_{U,U}^{(m)}}(\Xi^{(m)}_{!},\mathscr{F}).$$

From this and Lemma 6.10 we deduce that $\Xi_{!}^{(m)}$ is the projective cover of the simple object $\pi^{\dagger} \operatorname{For}_{I_u}^{I}(\mathscr{IC}_e)$ in $\mathsf{P}_{U,U}^{(m)}$, which implies that the image $\Pi_{U,U}^{0}(\Xi_{!}^{(m)})$ of $\Xi_{!}^{(m)}$ in $\mathsf{P}_{U,U}^{(m),0}$ is the projective cover of the unique simple object in this category; by standard arguments this implies that the functor

$$\mathscr{F} \mapsto \operatorname{Hom}_{\mathsf{P}_{U,U}^{(m),0}}(\Pi^0_{U,U}(\Xi^{(m)}_!),\mathscr{F})$$

induces an equivalence of categories

(8.8)
$$\mathsf{P}_{U,U}^{(m),0} \xrightarrow{\sim} \mathsf{Mod}_{\mathbf{r}}^{\mathrm{fg}} \big(\mathrm{End}_{\mathsf{P}_{U,U}^{(m),0}}(\Pi_{U,U}^{0}(\Xi_{!}^{(m)})) \big).$$

Now since $\Xi_{!}^{(m)}$ is projective in $\mathsf{P}_{U,U}^{(m)}$ with unique simple quotient $\pi^{\dagger}\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{IC}_{e})$, for any \mathscr{F} in $\mathsf{P}_{U,U}^{(m)}$ the morphism

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{(m)}}(\Xi^{(m)}_{!},\mathscr{F}) \to \operatorname{Hom}_{\mathsf{P}_{U,U}^{(m),0}}(\Pi^{0}_{U,U}(\Xi^{(m)}_{!}),\Pi^{0}_{U,U}(\mathscr{F}))$$

induced by $\Pi^0_{U,U}$ is an isomorphism. Using the isomorphism (8.6) this allows to identify the algebra $\operatorname{End}_{\mathsf{P}^{(m),0}_{U,U}}(\Pi^0_{U,U}(\Xi^{(m)}_!))$ with $\mathscr{O}(\mathbf{D}^{(m)})$, hence the category $\mathsf{Mod}^{\mathrm{fg}}_{\mathrm{r}}(\operatorname{End}_{\mathsf{P}^{(m),0}_{U,U}}(\Xi^{(m)}_!))$ with $\mathsf{Coh}(\mathbf{D}^{(m)})$.

We claim that the functor

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{(m),0}}(\Pi_{U,U}^{0}(\Xi_{!}^{(m)}),-):\mathsf{P}_{U,U}^{(m),0}\to\mathsf{Mod}(\mathscr{O}(\mathbf{D}))$$

identifies canonically with the restriction of

$$\operatorname{Hom}_{\mathsf{D}^0_{U,U}}(\Pi^0_{U,U}(\Xi^\wedge_!),-):\mathsf{P}^0_{U,U}\to\mathsf{Mod}(\mathscr{O}(\mathbf{D}))$$

to $\mathsf{P}_{U,U}^{(m),0}$. In fact, to prove this claim it suffices to construct, for \mathscr{F} in $\mathsf{P}_{U,U}^{(m)}$, a functorial isomorphism

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{(m),0}}(\Pi_{U,U}^{0}(\Xi_{!}^{(m)}),\Pi_{U,U}^{0}(\mathscr{F})) \cong \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\Xi_{!}^{\wedge}),\Pi_{U,U}^{0}(\mathscr{F})).$$

And for this, as explained above the left-hand side identifies canonically with $\operatorname{Hom}_{\mathsf{P}_{U,U}^{(m)}}(\Xi_{!}^{(m)},\mathscr{F})$, and then with $\operatorname{Hom}_{\mathsf{P}_{U,U}^{^{(m)}}}(\Xi_{!}^{^{(n)}},\mathscr{F})$. The desired identification is therefore provided by Lemma 8.2. This isomorphism and the considerations above show that the restriction of the functor $\operatorname{Hom}_{\mathsf{D}_{U,U}^{^{(n)}}}(\Pi_{U,U}^{^{(0)}}(\Xi_{!}^{^{(n)}}), -)$ to $\mathsf{P}_{U,U}^{(m),0}$ takes values in $\mathsf{Coh}(\mathbf{D}^{(m)})$, and induces an equivalence between these categories.

To conclude the proof, we have to construct compatible monoidal structures on our equivalences, which will be done if we prove that (8.4) is an isomorphism for any \mathscr{F}, \mathscr{G} in $\mathsf{P}_{U,U}^{(m),0}$.

First, consider the case $\mathscr{F} = \mathscr{G} = \Pi^0_{U,U}(\Xi^{(m)}_!)$. In this case we have seen that

(8.9)
$$\operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\Xi_{!}^{\wedge}),\Pi_{U,U}^{0}(\Xi_{!}^{(m)})) \cong \mathscr{O}(\mathbf{D}^{(m)}).$$

On the other hand, by Lemma 8.5 we have

$$\begin{split} \operatorname{Hom}_{\mathsf{D}^{0}_{U,U}}(\Pi^{0}_{U,U}(\Xi^{\wedge}_{!}),\Pi^{0}_{U,U}(\Xi^{(m)}_{!}) \stackrel{\mathrm{p}}{\star^{0}_{U}} \Pi^{0}_{U,U}(\Xi^{(m)}_{!})) \\ & \cong \operatorname{Hom}_{\mathsf{D}^{0}_{U,U}}(\Pi^{0}_{U,U}(\Xi^{\wedge}_{!}),\Pi^{0}_{U,U}(\mathsf{C}_{m}(\Xi^{\wedge}_{!} \stackrel{\widehat{\star}_{U}}{\star^{0}_{U}}\Xi^{\wedge}_{!}))), \end{split}$$

and by Lemma 8.2 the right-hand side identifies with

$$\operatorname{Hom}_{\mathsf{P}_{U}^{\wedge}}(\Xi_{!}^{\wedge},\mathsf{C}_{m}((\Xi_{!}^{\wedge} \widehat{\star}_{U} \Xi_{!}^{\wedge}))).$$

By projectivity of Ξ_1^{\wedge} (see Lemma 6.10) this space identifies with

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge},\Xi_{!}^{\wedge}\,\widehat{\star}_{U}\,\Xi_{!}^{\wedge})\otimes_{\mathscr{O}(\mathbf{D})}\mathscr{O}(\mathbf{D}^{(m)}),$$

which itself, in view of (8.5) and (8.7), identifies with

$$(\mathscr{O}(\mathbf{D})\otimes_{\mathscr{O}(T_{\flat}^{\vee})}\mathscr{O}(\mathbf{D}))\otimes_{\mathscr{O}(\mathbf{D})}\mathscr{O}(\mathbf{D}^{(m)}).$$

It is easily seen that under these identifications the morphism (8.4) identifies with the natural isomorphism

$$\mathscr{O}(\mathbf{D}^{(m)}) \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \mathscr{O}(\mathbf{D}^{(m)}) \cong \left(\mathscr{O}(\mathbf{D}) \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \mathscr{O}(\mathbf{D}) \right) \otimes_{\mathscr{O}(\mathbf{D})} \mathscr{O}(\mathbf{D}^{(m)})$$

and is therefore an isomorphism.

Now we prove that (8.4) is an isomorphism in case $\mathscr{F} = \Pi^0_{U,U}(\Xi^{(m)}_!)$ and \mathscr{G} is arbitrary. For this, since we already know the equivalence

$$\mathsf{P}_{U,U}^{(m),0} \xrightarrow{\sim} \mathsf{Coh}(\mathbf{D}^{(m)})$$

we know that there exists a presentation

$$\left(\Pi^0_{U,U}(\Xi^{(m)}_!)\right)^{\oplus r} \to \left(\Pi^0_{U,U}(\Xi^{(m)}_!)\right)^{\oplus s} \to \mathscr{G} \to 0$$

for some $r, s \in \mathbb{Z}_{\geq 0}$. Then we have exact sequences

$$\begin{split} & \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\Xi_{!}^{\wedge}), \Pi_{U,U}^{0}(\Xi_{!}^{(m)})) \otimes_{\mathscr{O}(T_{k}^{\vee})} \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\Xi_{!}^{\wedge}), \Pi_{U,U}^{0}(\Xi_{!}^{(m)})^{\oplus r}) \to \\ & \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\Xi_{!}^{\wedge}), \Pi_{U,U}^{0}(\Xi_{!}^{(m)})) \otimes_{\mathscr{O}(T_{k}^{\vee})} \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\Xi_{!}^{\wedge}), \Pi_{U,U}^{0}(\Xi_{!}^{(m)})^{\oplus s}) \to \\ & \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\Xi_{!}^{\wedge}), \Pi_{U,U}^{0}(\Xi_{!}^{(m)})) \otimes_{\mathscr{O}(T_{k}^{\vee})} \operatorname{Hom}_{\mathsf{D}_{U,U}^{0}}(\Pi_{U,U}^{0}(\Xi_{!}^{\wedge}), \mathscr{G}) \to 0 \end{split}$$

and

94

$$\begin{split} \operatorname{Hom}_{\mathsf{D}^{0}_{U,U}}(\Pi^{0}_{U,U}(\Xi^{\wedge}_{!}),\Pi^{0}_{U,U}(\Xi^{(m)}_{!}) \overset{\mathrm{p}_{\star^{0}}}{\to} \Pi^{0}_{U,U}(\Xi^{(m)}_{!})^{\oplus r}) \to \\ \operatorname{Hom}_{\mathsf{D}^{0}_{U,U}}(\Pi^{0}_{U,U}(\Xi^{\wedge}_{!}),\Pi^{0}_{U,U}(\Xi^{(m)}_{!}) \overset{\mathrm{p}_{\star^{0}}}{\to} \Pi^{0}_{U,U}(\Xi^{(m)}_{!})^{\oplus s}) \to \\ \operatorname{Hom}_{\mathsf{D}^{0}_{U,U}}(\Pi^{0}_{U,U}(\Xi^{\wedge}_{!}),\Pi^{0}_{U,U}(\Xi^{(m)}_{!}) \overset{\mathrm{p}_{\star^{0}}}{\to} \Omega) \to 0, \end{split}$$

which are related by the corresponding morphisms (8.4). The morphisms relating the first two terms are isomorphisms by the case treated above, hence the one relating the third terms is also an isomorphism, which finishes the proof of the case under consideration.

Finally, one passes from this case to the case of general \mathscr{F}, \mathscr{G} using similar arguments, which finishes the proof of the proposition.

8.6. Images of truncated costandard objects. Recall that in §3.2 we have defined some representations $(\mathscr{M}_w : w \in W)$ of $\mathbb{J}_{\mathbf{D}}$. By definition, in case $w \in W_f$, the representation \mathscr{M}_w is a coherent sheaf on \mathbf{D} , endowed with the trivial structure as a representation of $\mathbb{J}_{\mathbf{D}}$. In this subsection these representations will be simply considered as coherent sheaves on \mathbf{D} , or equivalently as finitely generated $\mathscr{O}(\mathbf{D})$ -modules.

Recall the functors C_m^0 of §6.5. The comments in §8.1 show that if we consider an object $\mathscr{F} \in \mathsf{P}_{U,U}^{\wedge}$, seen as an object in $\mathsf{P}_{\mathrm{I}_u,\mathrm{I}_u}^{\wedge}$, for any $m \geq 1$ the object $\mathsf{C}_m^0(\mathscr{F})$ belongs to the full subcategory $\mathsf{P}_{U,U}^0 \subset \mathsf{P}_{\mathrm{I}_u,\mathrm{I}_u}^0$. In this way we obtain a projective system $(\mathsf{C}_m^0(\mathscr{F}): m \geq 1)$ of objects in $\mathsf{P}_{U,U}^0$.

Lemma 8.6. For any $w \in W_f$, we have an isomorphism of projective systems

$$(\Phi_{U,U}(\mathsf{C}_m^0(\nabla_w^\wedge)): m \ge 1) \cong (\mathscr{M}_w/\mathcal{J}^m \cdot \mathscr{M}_w: m \ge 1)$$

Proof. For any $m \ge 1$, by Lemma 8.2 there exists a canonical isomorphism

$$\Phi_{U,U}(\mathsf{C}_m^0(\nabla_w^\wedge)) \cong \operatorname{Hom}_{\mathsf{P}_{U,U}^\wedge}(\Xi^\wedge_!, \mathsf{C}_m(\nabla_w^\wedge)).$$

By projectivity of Ξ_1^{\wedge} (see Lemma 6.10), the right-hand side identifies with

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi^{\wedge}_{!},\nabla^{\wedge}_{w}) \otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})} (\mathscr{O}(T^{\vee}_{\Bbbk})/\mathcal{J}^{m} \cdot \mathscr{O}(T^{\vee}_{\Bbbk})).$$

Finally, it is known that the standard object Δ_w^{\wedge} appears with multiplicity 1 in $\Xi_!^{\wedge}$, see [BR1, §9.1]; we deduce an isomorphism of $\mathscr{O}(\mathbf{D}^{\wedge})$ -modules $\operatorname{Hom}_{\mathbf{P}_{U,U}^{\wedge}}(\Xi_!^{\wedge}, \nabla_w^{\wedge}) \cong \mathscr{M}_w^{\wedge}$, which finishes the proof. \Box

9. Truncation of perverse sheaves

We continue with the setting and assumptions of Section 8. In this section we prove a number of technical statements regarding the functors C_m and C_m^0 introduced in §6.5. These results will be used in the next section in our study of the category P_{I_u,I_u}^0 .

9.1. Flatness of standard, costandard and Wakimoto sheaves. In §B.2 we recall what it means for a module in a category to be flat. Here we will be more specifically interested in the case of $\mathscr{O}(T_{\Bbbk}^{\vee})$ -modules in $\mathsf{P}_{\mathbf{I}_{\mathrm{u}},\mathbf{I}_{\mathrm{u}}}^{\wedge}$. We will say that an object of $\mathsf{P}_{\mathbf{I}_{\mathrm{u}},\mathbf{I}_{\mathrm{u}}}^{\wedge}$ is $\mathscr{O}(T_{\Bbbk}^{\vee})$ -flat if its image under (6.15) is flat; in other words, $\mathscr{F} \in \mathsf{P}_{\mathbf{I}_{\mathrm{u}},\mathbf{I}_{\mathrm{u}}}^{\wedge}$ is $\mathscr{O}(T_{\Bbbk}^{\vee})$ -flat iff the functor

$$(-) \otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})} \mathscr{F} : \mathsf{Mod}^{\mathrm{fg}}(\mathscr{O}(T^{\vee}_{\Bbbk})) \to \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

is exact.

Our goal in this subsection is to show that standard perverse sheaves, costandard perverse sheaves and free-monodromic Wakimoto sheaves are $\mathscr{O}(T_{\Bbbk}^{\vee})$ -flat. We start with standard and costandard sheaves.

Lemma 9.1. For any $w \in W$, the objects Δ_w^{\wedge} and ∇_w^{\wedge} are $\mathscr{O}(T_{\Bbbk}^{\vee})$ -flat. As a consequence, every object in $\mathsf{P}_{\mathrm{I}_u,\mathrm{I}_u}^{\wedge}$ which admits a standard or a costandard filtration (in particular, any tilting object) is $\mathscr{O}(T_{\Bbbk}^{\vee})$ -flat.

Proof. We will prove the claim for the objects Δ_w^{\wedge} ; the case of the objects ∇_w^{\wedge} can be treated similarly, and the claim about objects with a standard or costandard filtration then follows in view of Lemma B.4(2).

By Proposition D.1, the functor

$$(\widetilde{j}_w)_! : \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}}}(\mathrm{Fl}_{G,w}, \Bbbk) \to \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

(where $(\tilde{j}_w)_!$ is as in §6.2) is t-exact, and by [BR1, Proposition 4.5] we have a canonical equivalence of triangulated categories

$$\phi_w: D^{\mathrm{b}}\mathsf{Mod}^{\mathrm{fg}}(\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge})) \xrightarrow{\sim} \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}}}(\mathrm{Fl}_{G,w}, \Bbbk).$$

By definition, under this identification the perverse t-structure on $\mathsf{D}^{\wedge}_{\mathrm{I}_{u}}(\widetilde{\mathrm{Fl}}_{G,w},\Bbbk)$ corresponds to the tautological t-structure on $D^{\mathrm{b}}\mathsf{Mod}^{\mathrm{fg}}(\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge}))$. The monodromy construction also provides an $\mathscr{O}(T^{\vee}_{\Bbbk})$ -module structure on any object in $\mathsf{D}^{\wedge}_{\mathrm{I}_{u}}(\widetilde{\mathrm{Fl}}_{G,w},\Bbbk)$, which under the equivalence ϕ_{w} corresponds to the obvious $\mathscr{O}(T^{\vee}_{\Bbbk})$ module structure on a complex of $\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge})$ -modules. From these remarks we obtain that for any N in $\mathsf{Mod}^{\mathrm{fg}}(\mathscr{O}(T^{\vee}_{\Bbbk}))$ we have a canonical isomorphism

$$N \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \Delta_w^{\wedge} \cong (\widetilde{j}_w)_! \phi_w(N \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})).$$

The functor on the right-hand side is t-exact by flatness of $\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge})$ over $\mathscr{O}(T^{\vee}_{\Bbbk})$ and t-exactness of $(\tilde{j}_w)_!$ and ϕ_w , which proves that Δ^{\wedge}_w is flat.

Lemma 9.2. For any finitely generated $\mathscr{O}(T^{\vee}_{\Bbbk})$ -module M and any $w, y \in W$, the convolution

$$\Delta^\wedge_w \,\widehat{\star}\, (M \otimes_{\mathscr{O}(T^\vee_\Bbbk)} \nabla^\wedge_y)$$

belongs to the heart of the perverse t-structure on $\mathsf{D}^\wedge_{I_u,I_u}.$

Proof. As in the proof of Lemma 9.1 we have

$$M \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \nabla_{y}^{\wedge} \cong (\widetilde{j}_{y})_{*} \phi_{y}(N \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge}))$$

By construction of the equivalence ϕ_y in [BR1, Proposition 4.5] and Lemma D.3, if for $m \geq 1$ we denote by \mathscr{F}_m the k-local system on $\widetilde{\mathrm{Fl}}_{G,y}$ corresponding the $\mathscr{O}(T_k^{\vee})$ -module $N/\mathcal{K}^m N$, then we deduce that

$$M \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \nabla_{y}^{\wedge} \cong \lim_{m} \widetilde{\mathcal{J}}_{y})_{*}\mathscr{F}_{m}[\ell(y) + \dim(T)],$$

hence that

$$\Delta_w^{\wedge} \,\widehat{\star} \, (M \otimes_{\mathscr{O}(T_k^{\vee})} \nabla_y^{\wedge}) \cong \underset{m}{\overset{\text{lim}}{\longleftarrow}} \, {}^{n} \Delta_w^{\wedge} \,\widehat{\star} \, \big((\widetilde{\jmath}_y)_* \mathscr{F}_m[\ell(y) + \dim(T)] \big).$$

Each \mathscr{F}_m is an extension of copies of the constant local system, so that $\Delta_w^{\wedge} \hat{\star} ((\tilde{j}_y)_* \mathscr{F}_m[\ell(y) + \dim(T)])$ is an extension (in the sense of triangulated categories) of copies of

$$\Delta_w^{\wedge} \mathbin{\widehat{\star}} \pi^{\dagger} \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\nabla_y^{\mathrm{I}}) \cong \pi^{\dagger}(\Delta_w^{\mathrm{I}} \star_{\mathrm{I}} \nabla_y^{\mathrm{I}}),$$

where the isomorphism follows from (6.5)–(6.6). Here the right-hand side is perverse (by [AR4, Lemma 4.1.7] and t-exactness of π^{\dagger}), hence each $\Delta_w^{\wedge} \hat{\star} ((\tilde{\jmath}_y)_* \mathscr{F}_m[\ell(y) + \dim(T)])$ is perverse. By Proposition D.4, this implies the lemma.

Corollary 9.3. For any $\lambda \in X_*(T)$, the free-monodromic Wakimoto sheaf $\mathscr{W}^{\wedge}_{\lambda}$ is $\mathscr{O}(T^{\vee}_{\Bbbk})$ -flat. As a consequence, every object of $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ which admits a Wakimoto filtration is flat.

Proof. Once again, using Lemma B.4(2) it suffices to prove the first claim. Let $\lambda \in X_*(T)$, and choose $\mu, \nu \in X^+_*(T)$ such that $\lambda = \mu - \nu$, so that $\mathscr{W}^{\wedge}_{\lambda} \cong \nabla^{\wedge}_{\mathfrak{t}(\mu)} \widehat{\star} \Delta^{\wedge}_{\mathfrak{t}(-\nu)}$. First, we claim that for any finitely generated $\mathscr{O}(T^{\vee}_{\Bbbk})$ -module M we have

(9.1)
$$M \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \mathscr{W}_{\lambda}^{\wedge} \cong \nabla_{\mathsf{t}(\mu)}^{\wedge} \widehat{\star} (M \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \Delta_{\mathsf{t}(-\nu)}^{\wedge}).$$

Indeed, choose a presentation $\mathscr{O}(T^{\vee}_{\Bbbk})^{\oplus r} \to \mathscr{O}(T^{\vee}_{\Bbbk})^{\oplus s} \to M \to 0$. We deduce an exact sequence

$$(\Delta^{\wedge}_{\mathsf{t}(-\nu)})^{\oplus r} \to (\Delta^{\wedge}_{\mathsf{t}(-\nu)})^{\oplus s} \to M \otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})} \Delta^{\wedge}_{\mathsf{t}(-\nu)} \to 0.$$

By right exactness of the functor $\nabla^{\wedge}_{t(\mu)} \stackrel{p}{\star} (-)$ (see Corollary 6.5(1)), we deduce an exact sequence

$$(\nabla^{\wedge}_{\mathfrak{t}(\mu)} \stackrel{p_{\widehat{\star}}}{\to} \Delta^{\wedge}_{\mathfrak{t}(-\nu)})^{\oplus r} \to (\nabla^{\wedge}_{\mathfrak{t}(\mu)} \stackrel{p_{\widehat{\star}}}{\to} \Delta^{\wedge}_{\mathfrak{t}(-\nu)})^{\oplus s} \to \nabla^{\wedge}_{\mathfrak{t}(\mu)} \stackrel{p_{\widehat{\star}}}{\to} (M \otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})} \Delta^{\wedge}_{\mathfrak{t}(-\nu)}) \to 0.$$

Here the first two terms identify with $(\mathscr{W}_{\lambda}^{\wedge})^{\oplus r}$ and $(\mathscr{W}_{\lambda}^{\wedge})^{\oplus s}$ respectively, and by Lemma 9.2 one can replace $p_{\widehat{\star}}$ by $\widehat{\star}$ in the third term. We deduce (9.1).

Now, consider an exact sequence $M_1 \hookrightarrow M_2 \twoheadrightarrow M_3$ of finitely generated $\mathscr{O}(T^{\vee}_{\Bbbk})$ -module. By Lemma 9.1, the sequence

$$0 \to M_1 \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \nabla_{\mathsf{t}(\mu)}^{\wedge} \to M_2 \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \nabla_{\mathsf{t}(\mu)}^{\wedge} \to M_3 \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \nabla_{\mathsf{t}(\mu)}^{\wedge} \to 0$$

is exact. Applying the triangulated functor $\nabla^{\wedge}_{\mathsf{t}(\mu)} \mathbin{\widehat{\star}} (-)$ we deduce a distinguished triangle

$$\begin{split} \nabla^{\wedge}_{\mathsf{t}(\mu)} \,\widehat{\star} \left(M_1 \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \nabla^{\wedge}_{\mathsf{t}(\mu)} \right) &\to \nabla^{\wedge}_{\mathsf{t}(\mu)} \,\widehat{\star} \left(M_2 \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \nabla^{\wedge}_{\mathsf{t}(\mu)} \right) \\ &\to \nabla^{\wedge}_{\mathsf{t}(\mu)} \,\widehat{\star} \left(M_3 \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \nabla^{\wedge}_{\mathsf{t}(\mu)} \right) \xrightarrow{[1]} . \end{split}$$

By (9.1) all terms here are in the heart of the perverse t-structure, and this triangle corresponds to an exact sequence

$$0 \to M_1 \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \mathscr{W}_{\lambda}^{\wedge} \to M_2 \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \mathscr{W}_{\lambda}^{\wedge} \to M_3 \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \mathscr{W}_{\lambda}^{\wedge} \to 0$$

in $\mathsf{P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$, which finishes the proof.

The main consequence of these results that we will use below is the following.

Proposition 9.4. Consider an exact sequence

$$0 \to \mathscr{F}_1 \to \mathscr{F}_2 \to \mathscr{F}_3 \to 0$$

in $\mathsf{P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$. If \mathscr{F}_{3} admits either a standard filtration, or a costandard filtration, or a Wakimoto filtration, then for any $m \geq 1$ the induced sequence

$$0 \to \mathsf{C}_m(\mathscr{F}_1) \to \mathsf{C}_m(\mathscr{F}_2) \to \mathsf{C}_m(\mathscr{F}_3) \to 0$$

 $is \ exact.$

Proof. In view of the definition of C_m , the claim follows from Lemma B.4(1) and either Lemma 9.1 (in the first two cases) or Corollary 9.3 (in the third case). \Box

9.2. Truncation functors for perverse sheaves: monoidality. We will now study some monoidality properties of the functors C_m .

Given \mathscr{F}, \mathscr{G} in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ we have canonical maps

$$\mathscr{F} \to \mathsf{C}_m(\mathscr{F}), \qquad \mathscr{G} \to \mathsf{C}_m(\mathscr{G}),$$

which give rise to a natural morphism

$${}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}} \mathscr{G}) o {}^{\mathrm{p}}\mathscr{H}^{0}(\mathsf{C}_{m}(\mathscr{F}) \star_{\mathrm{I}_{\mathrm{u}}} \mathsf{C}_{m}(\mathscr{G})).$$

It follows from (6.7) that the action of $\mathscr{O}(T_{\Bbbk}^{\vee})$ on the right-hand side vanishes on \mathcal{J}^m ; this morphism therefore factors uniquely though a morphism

$$\mathsf{C}_m({}^{\mathrm{p}}\mathscr{H}^0(\mathscr{F} \mathbin{\widehat{\star}} \mathscr{G})) \to {}^{\mathrm{p}}\mathscr{H}^0(\mathsf{C}_m(\mathscr{F}) \star_{\mathrm{I}_{\mathrm{u}}} \mathsf{C}_m(\mathscr{G})).$$

Finally, applying the functor $\Pi^0_{I_u,I_u}$ and using the definition of the bifunctor ${}^{p}\star^0_{I_u}$ (see §5.2), we obtain a canonical morphism

(9.2)
$$\mathsf{C}^{0}_{m}(^{\mathsf{p}}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}} \mathscr{G})) \to \mathsf{C}^{0}_{m}(\mathscr{F}) \,^{\mathsf{p}} \mathsf{\star}^{0}_{\mathsf{I}_{u}} \, \mathsf{C}^{0}_{m}(\mathscr{G}).$$

Lemma 9.5. Assume that one of functors

$${}^{p}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}} (-)), \, {}^{p}\mathscr{H}^{0}((-) \mathbin{\widehat{\star}} \mathscr{G}) : \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

is right exact. Then (9.2) is an isomorphism.

Proof. We will write the proof in case the functor ${}^{p}\mathcal{H}^{0}(\mathscr{F} \widehat{\star}(-))$ is right exact; the other case is similar. Choosing a family of r generators of the ideal \mathcal{J}^{m} , we obtain an exact sequence

$$\mathscr{G}^{\oplus r} \to \mathscr{G} \to \mathsf{C}^m(\mathscr{G}) \to 0$$

Applying the right-exact functor ${}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F} \widehat{\star} (-))$, we deduce an exact sequence

$${}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}} \mathscr{G})^{\oplus r} \to {}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}} \mathscr{G}) \to {}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}} \mathsf{C}^{m}(\mathscr{G})) \to 0,$$

which provides a canonical isomorphism

$$\mathcal{L}_m(^{\mathbf{p}}\mathscr{H}^0(\mathscr{F} \mathbin{\widehat{\star}} \mathscr{G})) \cong {}^{\mathbf{p}}\mathscr{H}^0(\mathscr{F} \mathbin{\widehat{\star}} \mathsf{C}_m(\mathscr{G})).$$

Applying $\Pi^0_{I_n,I_n}$, we deduce a canonical isomorphism

$$\mathsf{C}^0_m({}^{\mathrm{p}}\mathscr{H}^0(\mathscr{F} \mathbin{\widehat{\star}} \mathscr{G})) \cong \mathscr{F} \, {}^{\mathrm{p}} \mathbin{\widehat{\star}}^0 \, \mathsf{C}^0_m(\mathscr{G}).$$

At this point, to conclude it suffices to show that the morphism

$$\mathscr{F}^{\mathrm{p}} \widehat{\star}^{0} \mathsf{C}^{0}_{m}(\mathscr{G}) \to \mathsf{C}^{0}_{m}(\mathscr{F})^{\mathrm{p}} \star^{0}_{\mathrm{I}_{u}} \mathsf{C}^{0}_{m}(\mathscr{G})$$

induced by the morphism $\mathscr{F} \to \mathsf{C}^m(\mathscr{F})$ is an isomorphism.

Our choice of generators for \mathcal{J}^m also provides an exact sequence

$$\mathscr{F}^{\oplus r} \to \mathscr{F} \to \mathsf{C}^m(\mathscr{F}) \to 0.$$

By Lemma 6.9 the functor $(-)^{p} \widehat{\star}^{0} C_{m}^{0}(\mathscr{G}) : \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is right exact; we therefore deduce an exact sequence

$$(\mathscr{F} \stackrel{\mathrm{p} \widehat{\star}^0}{\to} \mathsf{C}^0_m(\mathscr{G}))^{\oplus r} \to \mathscr{F} \stackrel{\mathrm{p} \widehat{\star}^0}{\to} \mathsf{C}^0_m(\mathscr{G}) \to \mathsf{C}^m(\mathscr{F}) \stackrel{\mathrm{p} _{\mathsf{T}_{\mathrm{lu}}}^0}{\to} \mathsf{C}^0_m(\mathscr{G}) \to 0.$$

By (6.7), \mathcal{J}^m acts trivially on $\mathscr{F}^{p_{\star}^0} C^0_m(\mathscr{G})$, hence the first map in this sequence vanishes; we deduce that the second arrow is an isomorphism, as desired.

Proposition 9.6. Assume that we are in one of the following settings:

- (1) either \mathscr{F} or \mathscr{G} admits a costandard filtration;
- (2) $\mathscr{F} = \Delta_w^{\wedge}$ and $\mathscr{G} = \Delta_y^{\wedge}$ for some $w, y \in W$ such that $\ell(wy) = \ell(w) + \ell(y)$.

Then (9.2) is an isomorphism.

Proof. To treat the first case, in view of Lemma 9.5 it suffices to show that if \mathscr{F} admits a costandard filtration the functors

$${}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{F} \mathbin{\widehat{\star}} (-)), \, {}^{\mathrm{p}}\mathscr{H}^{0}((-) \mathbin{\widehat{\star}} \mathscr{F}) : \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

are right exact. For that, it suffices to remark that the functors

$$\mathscr{F} \mathbin{\widehat{\star}} (-), (-) \mathbin{\widehat{\star}} \mathscr{F} : \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

are right t-exact, as follows from Corollary 6.5(1).

Now, let us assume that $\mathscr{F} = \Delta_w^{\wedge}$ and $\mathscr{G} = \Delta_y^{\wedge}$ for some $w, y \in W$ such that $\ell(wy) = \ell(w) + \ell(y)$. Then by Lemma 6.3(2) we have $\mathscr{F} \stackrel{\star}{\star} \mathscr{G} \cong \Delta_{wy}^{\wedge}$. Recall from the proof of Lemma 9.1 that for any $x \in W$ we have

$$\mathsf{C}_m(\Delta_x^{\wedge}) \cong (\widetilde{\jmath}_x)_! \phi_x(\mathscr{O}(T_{\Bbbk}^{\vee})/\mathcal{J}^m \mathscr{O}(T_{\Bbbk}^{\vee}))$$

Now, using [BR1, Lemma 3.4] and considerations similar to those encountered in the proof of Lemma 7.1, it is not difficult to check that we have an

$$\Delta^{\wedge}_{w} \widehat{\star} \left((\widetilde{j}_{y})_{!} \phi_{y}(\mathscr{O}(T^{\vee}_{\Bbbk}) / \mathcal{J}^{m} \mathscr{O}(T^{\vee}_{\Bbbk})) \right) \xrightarrow{\sim} (\widetilde{j}_{wy})_{!} \phi_{wy}(\mathscr{O}(T^{\vee}_{\Bbbk}) / \mathcal{J}^{m} \mathscr{O}(T^{\vee}_{\Bbbk})),$$

i.e. an isomorphism

$$\Delta_w^{\wedge} \mathbin{\widehat{\star}} \mathsf{C}_m(\Delta_y^{\wedge}) \cong \mathsf{C}_m(\Delta_{wy}^{\wedge}).$$

Once this is known, the same arguments as in the final part of the proof of Lemma 9.5 show the desired claim. $\hfill \Box$

Recall from §6.3 that the subcategory $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \subset \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is closed under the convolution product $\widehat{\star}$. Proposition 9.6 implies in particular that for any $m \geq 1$, the functor C^{0}_{m} induces a monoidal functor

$$(\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},\widehat{\star}) \to (\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},{}^{\mathrm{p}}\!\star^{0}_{\mathrm{I}_{\mathrm{u}}}).$$

9.3. Truncation functors for perverse sheaves: fully faithfulness. We now prove a statement that will allow us to describe tilting objects from their images under the functors C_m^0 .

Lemma 9.7. Let $\mathscr{T}, \mathscr{T}'$ in $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

(1) The functors C_m induce an isomorphism

$$\operatorname{Hom}_{\mathsf{P}_{\operatorname{I}_{\operatorname{u}},\operatorname{I}_{\operatorname{u}}}^{\wedge}}(\mathscr{T},\mathscr{T}') \xrightarrow{\sim} \varprojlim_{m} \operatorname{Hom}_{\mathsf{P}_{\operatorname{I}_{\operatorname{u}},\operatorname{I}_{\operatorname{u}}}}(\mathsf{C}_{m}(\mathscr{T}),\mathsf{C}_{m}(\mathscr{T}'))$$

(2) For any $m \geq 1$, the functor $\Pi^0_{I_u,I_u}$ induces an isomorphism

$$\operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathsf{C}_{m}(\mathscr{T}),\mathsf{C}_{m}(\mathscr{T}'))\xrightarrow{\sim}\operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathsf{C}_{m}^{0}(\mathscr{T}),\mathsf{C}_{m}^{0}(\mathscr{T}'))$$

Proof. (1) By definition of the tensor product (and functoriality of monodromy), for any $m \ge 1$ we have

$$\operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}}(\mathsf{C}_{m}(\mathscr{T}),\mathsf{C}_{m}(\mathscr{T}'))\cong\operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}}(\mathscr{T},\mathsf{C}_{m}(\mathscr{T}')).$$

The isomorphism we have to prove can therefore by written as

$$\operatorname{Hom}_{\mathsf{P}_{\operatorname{I}_{\operatorname{u}},\operatorname{I}_{\operatorname{u}}}^{\wedge}}(\mathscr{T},\mathscr{T}')\xrightarrow{\sim}\varprojlim_{m}\operatorname{Hom}_{\mathsf{P}_{\operatorname{I}_{\operatorname{u}},\operatorname{I}_{\operatorname{u}}}^{\wedge}}(\mathscr{T},\mathsf{C}_{m}(\mathscr{T}')).$$

There is an obvious (bifunctorial) map from the left-hand side to the right-hand side, for any $\mathscr{T}, \mathscr{T}'$ in $\mathsf{P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$; we will prove that this map is an isomorphism when

 $\mathcal T$ has a standard filtration and $\mathcal T'$ has a costandard filtration, by induction on the sum of the lengths of these filtrations.

First, if $\mathscr{T} = \Delta_w^{\stackrel{\scriptstyle\wedge}{}}$ and $\mathscr{T}' = \nabla_y^{\wedge}$ for some $w, y \in W$, then we have

$$\operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\mathscr{T},\mathscr{T}') \cong \begin{cases} \mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge}) & \text{if } w = y; \\ 0 & \text{otherwise.} \end{cases}$$

On the other hand, from the description of $C_m(\nabla_y^{\wedge})$ in the proof of Lemma 9.1 we deduce that

$$\operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\mathscr{T},\mathsf{C}_{m}(\mathscr{T}'))\cong\begin{cases} \mathscr{O}(T^{\vee}_{\Bbbk})/\mathcal{J}^{m}\cdot\mathscr{O}(T^{\vee}_{\Bbbk}) & \text{if } w=y;\\ 0 & \text{otherwise.} \end{cases}$$

The claim is therefore clear in this case.

Now, assume that the object ${\mathscr T}$ admits a standard filtration, that we have an exact sequence

$$0 \to \nabla_{y}^{\wedge} \to \mathscr{T}' \to \mathscr{T}'' \to 0$$

where \mathscr{T}'' has a costandard filtration, and that the claim is known for the pairs $(\mathscr{T}, \nabla_y^{\wedge})$ and $(\mathscr{T}, \mathscr{T}'')$. By Proposition 9.4, for any $m \geq 1$ we then have an exact sequence

$$0 \to \mathsf{C}_m(\nabla_y^{\wedge}) \to \mathsf{C}_m(\mathscr{T}') \to \mathsf{C}_m(\mathscr{T}'') \to 0.$$

The description of $C_m(\nabla_u^{\wedge})$ in the proof of Lemma 9.1 shows that

$$\operatorname{Ext}^{1}_{\mathsf{P}^{\wedge}_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}}}(\mathscr{T},\mathsf{C}_{m}(\nabla^{\wedge}_{y}))=0;$$

we therefore obtain an exact sequence

$$\begin{split} 0 \to \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\mathscr{T},\mathsf{C}_{m}(\nabla_{y}^{\wedge})) \to \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\mathscr{T},\mathsf{C}_{m}(\mathscr{T}')) \\ & \to \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}}(\mathscr{T},\mathsf{C}_{m}(\mathscr{T}')) \to 0. \end{split}$$

The inverse system $(\operatorname{Hom}_{\mathsf{P}_{\mathbf{I}_u},\mathbf{I}_u}(\mathscr{T},\mathsf{C}_m(\nabla_y^{\wedge})): m \geq 1)$ is an inverse system of finitedimensional k-vector spaces; it therefore automatically satisfies the Mittag-Leffler condition, which implies that the sequence

$$0 \to \varprojlim_{m} \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{T},\mathsf{C}_{m}(\nabla_{y}^{\wedge})) \to \varprojlim_{m} \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{T},\mathsf{C}_{m}(\mathscr{T}')) \\ \to \varprojlim_{m} \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{T},\mathsf{C}_{m}(\mathscr{T}')) \to 0$$

is exact. Similarly we have an exact sequence

$$0 \to \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\wedge}}(\mathscr{T}, \nabla_{y}^{\wedge}) \to \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\wedge}}(\mathscr{T}, \mathscr{T}') \to \operatorname{Hom}_{\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\wedge}}(\mathscr{T}, \mathscr{T}'') \to 0.$$

Our maps for the pairs $(\mathscr{T}, \nabla_y^{\wedge})$, $(\mathscr{T}, \mathscr{T}')$ and $(\mathscr{T}, \mathscr{T}')$ define a morphism of exact sequences; since the first and third maps are isomorphisms by assumption, the second one is also an isomorphism by the five lemma.

Finally, very similar arguments show that if the object \mathscr{T}' admits a costandard filtration, if we have an exact sequence

$$0 \to \mathscr{T}'' \to \mathscr{T} \to \Delta_w^{\wedge} \to 0$$

such that \mathscr{T}'' has a standard filtration, and if the claim is known for the pairs $(\mathscr{T}'', \mathscr{T}')$ and $(\Delta_w^{\wedge}, \mathscr{T}')$, then it follows for the pair $(\mathscr{T}, \mathscr{T}')$, which finishes the proof.

(2) The claim will follow from the description of morphisms in a Serre quotient category provided we prove that $C_m(\mathscr{T})$ does not admit a nonzero morphism to an object of the form $\pi^{\dagger} \operatorname{For}_{I_u}^{I}(\mathscr{IC}_w)$ with $\ell(w) > 0$ and $C_m(\mathscr{T}')$ does not admit a nonzero morphism from such an object. Using Proposition 9.4 we obtain that if \mathscr{F} admits a standard, resp. costandard, filtration in $\mathsf{P}_{I_u,I_u}^{\wedge}$ then $C_m(\mathscr{F})$ admits a filtration whose subquotients have the form $C_m(\Delta_w^{\wedge})$, resp. $C_m(\nabla_w^{\wedge})$, with $w \in W$. The desired claim will therefore follow if we prove that for $y \in W$ the object $C_m(\nabla_y^{\wedge})$ does not admit a nonzero morphism to an object of the form $\pi^{\dagger}\operatorname{For}_{I_u}(\mathscr{IC}_w)$ with $\ell(w) > 0$, and $C_m(\Delta_y^{\wedge})$ does not admit a nonzero morphism to an object of the form $\pi^{\dagger}\operatorname{For}_{I_u}(\mathscr{IC}_w)$ with $\ell(w) > 0$, and $C_m(\Delta_y^{\wedge})$ does not admit a nonzero morphism from such an object. However, from the proof of Lemma 9.1 we know that $C_m(\nabla_y^{\wedge})$ is an extension of copies of $\pi^{\dagger}\operatorname{For}_{I_u}(\nabla_y^{I})$ and $C_m(\Delta_y^{\wedge})$ is an extension of copies of $\pi^{\dagger}\operatorname{For}_{I_u}(\Delta_y^{I})$; the claim therefore follows from the fact that the head of ∇_y^{I} and the socle of Δ_y^{I} are both of the form \mathscr{IC}_z with $\ell(z) = 0$, see [BRR, Lemma 4.5].

10. Perverse sheaves on $\widetilde{\mathrm{Fl}}_G$

We continue with the setting of Section 8, and make the following assumptions:

- (1) the quotient of $X^*(T)$ by the root lattice of (G, T) is free;
- (2) the quotient of $X_*(T)$ by the coroot lattice of (G,T) has no ℓ -torsion;
- (3) for any indecomposable factor in the root system of (G, T), ℓ is strictly bigger than the corresponding value in Figure 1.1.

(As in §4.8, we expect that the third assumption can be weakened. What will be used below is that the main result of [BRR] holds.) Our goal is to prove the first main result of the paper, Theorem 1.3.

10.1. **Statement.** We will use the constructions of Section 2–3, for the group $\mathbf{G} = G_{\Bbbk}^{\vee}$, its Borel subgroup B_{\Bbbk}^{\vee} , and its maximal torus T_{\Bbbk}^{\vee} . In particular, we fix a Steinberg section $\Sigma \subset G_{\Bbbk}^{\vee}$ as in §2.2. Then we have the universal centralizer $\mathbb{J}_{\Sigma} \subset G_{\Bbbk}^{\vee} \times \Sigma$, a smooth affine group scheme over Σ . We have a canonical morphism $\mathbf{D} \to \Sigma$ obtained by composing the obvious projection $\mathbf{D} \to T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$ with the inverse of the isomorphism $\Sigma \xrightarrow{\sim} T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$, and the smooth affine group scheme

$$\mathbb{J}_{\mathbf{D}} = \mathbf{D} \times_{\Sigma} \mathbb{J}_{\Sigma}$$

over the affine scheme **D**. Consider as in §3.2 the abelian category $\text{Rep}(\mathbb{J}_{\mathbf{D}})$ of representations of $\mathbb{J}_{\mathbf{D}}$ which are finitely generated over $\mathscr{O}(\mathbf{D})$, and its monoidal product \circledast . Recall that this product is right exact on both sides, and has as unit object $\mathscr{O}_{\Delta T_{\Bbbk}^{\vee}}$ where $\Delta T_{\Bbbk}^{\vee} \subset \mathbf{D}$ is the diagonal copy of T_{\Bbbk}^{\vee} , with the trivial structure as a representation. We will also denote by

$\mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$

the full subcategory of $\mathsf{Rep}(\mathbb{J}_{\mathbf{D}})$ whose objects are the representations which are set-theoretically supported on the base point $(e, e) \in \mathbf{D}$ (i.e., whose restriction to the open complement vanishes). Using the notation of §3.3, the objects in this subcategory can also be described as those on which the ideal \mathcal{I} acts nilpotently, or equivalently as those on which \mathcal{J} acts nilpotently.

The subcategory $\operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}}) \subset \operatorname{Rep}(\mathbb{J}_{\mathbf{D}})$ is a nonunital monoidal subcategory. If $\operatorname{Coh}_0(\mathbf{D})$ is as in §8.4, we have a fully faithful exact monoidal functor

(10.1)
$$\operatorname{Coh}_0(\mathbf{D}) \to \operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}})$$

sending a coherent sheaf to itself with the trivial structure as a representation. (This justifies our choice of notation for the monoidal products.)

Let now $\mathbf{u} \in \Sigma \subset G_{\Bbbk}^{\vee}$ be the point corresponding to the image of $e \in T_{\Bbbk}^{\vee}$ in $T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$. Then \mathbf{u} is a regular unipotent element, so that as explained in §4.8 the constructions of [BRR] provide an equivalence of abelian monoidal categories

$$\Phi_{\mathrm{I},\mathrm{I}}: \left(\mathsf{P}^{0}_{\mathrm{I},\mathrm{I}},\star^{0}_{\mathrm{I}}\right) \xrightarrow{\sim} \left(\mathsf{Rep}(Z_{G^{\vee}_{\Bbbk}}(\mathsf{u})),\otimes\right).$$

The fiber of $\mathbb{J}_{\mathbf{D}}$ over (e, e) is by definition the fiber of \mathbb{J}_{Σ} over \mathfrak{u} , which identifies with $Z_{G_{\Bbbk}^{\vee}}(\mathfrak{u})$. The functor of pushforward along the closed embedding $\{(e, e)\} \hookrightarrow \mathbf{D}$ therefore defines an exact fully faithful functor

(10.2)
$$\operatorname{Rep}(Z_{G_{\mathbb{L}}^{\vee}}(\mathsf{u})) \to \operatorname{Rep}_{0}(\mathbb{J}_{\mathbf{D}}),$$

which is easily seen to admit a canonical monoidal structure. (The essential image of this functor consists of representations on which the ideal \mathcal{I} acts trivially.)

On the other hand, recall the exact fully faithful monoidal functor

$$\pi_0^{\dagger} : \left(\mathsf{P}_{\mathrm{I},\mathrm{I}}^0, \star_{\mathrm{I}}^0\right) \to \left(\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^0, {}^{\mathrm{p}}\!\star_{\mathrm{I}_{\mathrm{u}}}^0\right)$$

considered in $\S5.3$, and the exact monoidal functor (8.2). In this section we will prove the following theorem, which is a more precise version of Theorem 1.3.

Theorem 10.1. There exists an equivalence of abelian monoidal categories

$$\Phi_{I_u,I_u}: \left(\mathsf{P}^0_{I_u,I_u}, {}^{\mathrm{p}}\!\!\star^0_{I_u}\right) \xrightarrow{\sim} (\mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}}), \circledast)$$

such that the following diagrams commute up to isomorphism:

$$(10.3) \qquad \begin{array}{c} \mathsf{P}_{\mathrm{I},\mathrm{I}}^{0} \xrightarrow{\Phi_{\mathrm{I},\mathrm{I}}} & \mathsf{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})) & \mathsf{P}_{U,U}^{0} \xrightarrow{\Phi_{U,U}} & \mathsf{Coh}_{0}(\mathbf{D}) \\ & & & \\ & & \\ & & \\ & & \\ \mathsf{P}_{0}^{\dagger} \bigvee & & \\ \mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\dagger} \xrightarrow{\Phi_{\mathrm{I}_{u},\mathrm{I}_{u}}} & \mathsf{Rep}_{0}(\mathbb{J}_{\mathbf{D}}). & \mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{0} \xrightarrow{\Phi_{\mathrm{I}_{u},\mathrm{I}_{u}}} & \mathsf{Rep}_{0}(\mathbb{J}_{\mathbf{D}}). \end{array}$$

The proof of this theorem will occupy the whole section. Our strategy will be to define an appropriate "deformation" of the functor $\Phi_{I,I}$ as described in §4.8, and check that this functor has the required properties by reducing most of them to the similar properties of $\Phi_{I,I}$ or $\Phi_{U,U}$.

10.2. Truncation and completion of representations. Recall from §3.3 the scheme \mathbf{D}^{\wedge} and the smooth affine group scheme

$$\mathbb{J}_{\mathbf{D}}^{\wedge} = \mathbf{D}^{\wedge} imes_{\mathbf{D}} \mathbb{J}_{\mathbf{D}} \cong \mathbf{D}^{\wedge} imes_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma}$$

over the affine scheme \mathbf{D}^{\wedge} . Recall that the category $\mathsf{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ of representations of this group scheme which are of finite type over $\mathscr{O}(\mathbf{D}^{\wedge})$ admits a natural monoidal product \circledast . We have a natural fully faithful exact monoidal functor

(10.4)
$$(\operatorname{\mathsf{Rep}}_0(\mathbb{J}_{\mathbf{D}}),\circledast) \to (\operatorname{\mathsf{Rep}}(\mathbb{J}_{\mathbf{D}}^\wedge),\circledast)$$

whose essential image consists of modules on which \mathcal{I} (equivalently, \mathcal{J}) acts nilpotently.

We also have "truncation" operations which produce objects in $\mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$ out of objects in $\mathsf{Rep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$. Namely, for $m \geq 1$ we can consider the functor

$$\mathsf{D}_m: \mathsf{Rep}(\mathbb{J}^\wedge_{\mathbf{D}}) \to \mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$$

given by restriction to the closed subscheme $\mathbf{D}^{(m)} \subset \mathbf{D}^{\wedge}$ from §8.5. The following claim is clear from definitions.

Lemma 10.2. Let $M, M' \in \mathsf{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$. For any $m \ge 1$ we have $\mathsf{D}_m(M \circledast M') \cong \mathsf{D}_m(M) \circledast \mathsf{D}_m(M').$

10.3. Extension of $\widehat{\mathscr{R}}$ to coherent sheaves and definition of \mathscr{R}^{\wedge} . Recall the category $\mathsf{Coh}_{\mathrm{fr}}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee})$ considered in §4.5. (Here the action of G_{\Bbbk}^{\vee} on itself is the adjoint action.) Applying Lemma 4.5 to the monoidal functor

$$\mathscr{Z}$$
: $\mathsf{Rep}(G^{\vee}_{\Bbbk}) \to \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$

and its automorphism $\widehat{\mathfrak{m}}_{(-)}$ (see §7.4) we obtain a canonical monoidal functor

$$\widehat{\mathscr{Z}}^{\mathsf{Coh}}:\mathsf{Coh}_{\mathrm{fr}}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk})\to\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

taking values in the subcategory $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. Recall that for any V in $\mathsf{Rep}(G^{\vee}_{\Bbbk})$ and \mathscr{F} in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ the isomorphism $\widehat{\sigma}_{\mathcal{S}^{-1}(V),\mathscr{F}}$ from Theorem 7.8(4) provides a canonical isomorphism

(10.5)
$$\widehat{\mathscr{Z}}(V) \mathbin{\widehat{\star}} \mathscr{F} \xrightarrow{\sim} \mathscr{F} \mathbin{\widehat{\star}} \widehat{\mathscr{Z}}(V),$$

or in other words a canonical isomorphism

(10.6)
$$\widehat{\mathscr{X}}^{\mathsf{Coh}}(V \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}) \mathbin{\widehat{\star}} \mathscr{F} \xrightarrow{\sim} \mathscr{F} \mathbin{\widehat{\star}} \widehat{\mathscr{X}}^{\mathsf{Coh}}(V \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}})$$

We note the following property for later use.

Lemma 10.3. The isomorphisms (10.6) define an isomorphism of bifunctors from $\operatorname{Coh}_{\mathrm{fr}}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee}) \times \mathsf{D}_{\mathrm{In},\mathrm{In}}^{\wedge}$ to $\mathsf{D}_{\mathrm{In},\mathrm{In}}^{\wedge}$.

Proof. Consider the monoidal k-linear additive category A of k-linear endofunctors of $\mathsf{D}^{\wedge}_{\mathsf{I}_{u},\mathsf{I}_{u}}$ (with monoidal structure given by composition). We have two k-linear monoidal functors from $\mathsf{Rep}(G_{\Bbbk}^{\vee})$ to A, sending respectively V to the endofunctors $\widehat{\mathscr{Z}}(V) \mathbin{\widehat{\star}}(-)$ and $(-) \mathbin{\widehat{\star}} \widehat{\mathscr{Z}}(V)$. Each of these functors admits an automorphism, given by $\widehat{\mathsf{m}}_{V} \mathbin{\widehat{\star}}(-)$ and $(-) \mathbin{\widehat{\star}} \widehat{\mathscr{R}}(V)$. Each of these functors admits an automorphism, given by $\widehat{\mathsf{m}}_{V} \mathbin{\widehat{\star}}(-)$ and $(-) \mathbin{\widehat{\star}} \widehat{\mathscr{R}}(V)$. Lemma 4.5 provides extensions of these functors determined by the corresponding automorphism, which are given by $V \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}} \mapsto \widehat{\mathscr{Z}}^{\mathsf{Coh}}(V) \mathbin{\widehat{\star}}(-)$ and $V \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}} \mapsto (-) \mathbin{\widehat{\star}} \widehat{\mathscr{Z}}^{\mathsf{Coh}}(V)$ respectively. Now the isomorphism (10.5) intertwines $\widehat{\mathsf{m}}_{V} \mathbin{\widehat{\star}} \mathrm{id}_{\mathscr{F}}$ and $\mathrm{id}_{\mathscr{F}} \mathbin{\widehat{\star}} \widehat{\mathsf{m}}_{V}$, see Theorem 7.8(4). By unicity in Lemma 4.5, this means that (10.6) is an isomorphism of bifunctors, as desired. \Box

Recall (see (2.2)) that the adjoint quotient $G_{\Bbbk}^{\vee}/G_{\Bbbk}^{\vee}$ identifies with $T_{\Bbbk}^{\vee}/W_{\rm f}$. The quotient morphism $G_{\Bbbk}^{\vee} \to G_{\Bbbk}^{\vee}/G_{\Bbbk}^{\vee}$ provides, for any \mathscr{F} in $\mathsf{Coh}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee})$, a canonical algebra morphism

$$\mathscr{O}(G^{\vee}_{\Bbbk}/G^{\vee}_{\Bbbk}) \to \operatorname{End}_{\operatorname{\mathsf{Coh}}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk})}(\mathscr{F}),$$

and therefore an algebra morphism $\mathscr{O}(T^{\vee}_{\Bbbk}/W_{\mathrm{f}}) \to \mathrm{End}(\mathscr{F})$. With these morphisms, the category $\mathsf{Coh}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk})$ becomes an $\mathscr{O}(T^{\vee}_{\Bbbk}/W_{\mathrm{f}})$ -linear category.

Lemma 10.4. For any V in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$, the composition

$$\mathscr{O}(T^{\vee}_{\Bbbk}/W_{\mathrm{f}}) \to \mathrm{End}_{\mathsf{Coh}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk})}(V \otimes_{\Bbbk} \mathscr{O}_{G^{\vee}_{\Bbbk}}) \to \mathrm{End}_{\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\widehat{\mathscr{Z}}(V))$$

(where the second map is induced by $\widehat{\mathscr{Z}}^{\mathsf{Coh}}$) coincides with the restriction of the morphism μ_V (see (7.7)) to $\mathscr{O}(T^{\vee}_{\Bbbk}/W_{\mathrm{f}})$.

Proof. First we consider the case when $V = \Bbbk$ is the trivial G_{\Bbbk}^{\vee} -module. In this case, $\widehat{\mathscr{X}}(\Bbbk)$ is the unit object δ^{\wedge} in $\mathsf{D}_{I_{\mathrm{u}},I_{\mathrm{u}}}^{\wedge}$. If M is a finite-dimensional G_{\Bbbk}^{\vee} -module, then we denote by $\mathrm{ch}_{M} \in \mathscr{O}(G_{\Bbbk}^{\vee}/G_{\Bbbk}^{\vee}) = \mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})$ the associated character. It is well known that these elements generate $\mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})$ as a vector space, so that to prove the desired claim it suffices to check that our maps coincide on such elements. Now ch_{M} can be interpreted as the composition

$$\mathscr{O}(G^\vee_\Bbbk) \to M \otimes \mathscr{O}(G^\vee_\Bbbk) \otimes M^* \to M \otimes \mathscr{O}(G^\vee_\Bbbk) \otimes M^* \to \mathscr{O}(G^\vee_\Bbbk)$$

where the first (resp. third) morphism is induced by the canonical map $\mathbb{k} \to M \otimes M^*$ (resp. $M \otimes M^* \to \mathbb{k}$), and the middle one is $\mathsf{m}_M^{\mathrm{taut}} \otimes \mathrm{id}_{M^*}$. (See §4.5 for the definition of $\mathsf{m}_M^{\mathrm{taut}}$.) Therefore, its image in $\mathrm{End}(\widehat{\delta})$ is the composition

$$\widehat{\delta} \to \widehat{\mathscr{Z}}(M) \mathbin{\widehat{\star}} \widehat{\mathscr{Z}}(M^*) \xrightarrow{\operatorname{\mathsf{in}}_M \mathbin{\widehat{\star}} \operatorname{id}_{\widehat{\mathscr{Z}}(M^*)}} \widehat{\mathscr{Z}}(M) \mathbin{\widehat{\star}} \widehat{\mathscr{Z}}(M^*) \to \widehat{\delta},$$

where the first and third morphisms are the images of the maps considered above. This map has been computed in Lemma 7.10, and is known to equal $\mu_{\hat{\delta}}(ch_M \otimes 1) = \mu_{\mathbb{k}}(ch_M)$; this proves the desired claim in this case.

Now we deduce the general case. It is clear that the canonical morphism

$$\mathscr{O}(T^{\vee}_{\Bbbk}/W_{\mathrm{f}}) \to \mathrm{End}_{\mathsf{Coh}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk})}(V \otimes_{\Bbbk} \mathscr{O}_{G^{\vee}_{\Bbbk}})$$

is the composition

$$\mathscr{O}(T^\vee_\Bbbk/W_{\mathrm{f}}) \to \operatorname{End}_{\mathsf{Coh}^{G^\vee_\Bbbk}(G^\vee_\Bbbk)}(\mathscr{O}_{G^\vee_\Bbbk}) \to \operatorname{End}_{\mathsf{Coh}^{G^\vee_\Bbbk}(G^\vee_\Bbbk)}(V \otimes_\Bbbk \mathscr{O}_{G^\vee_\Bbbk})$$

where the first map is the canonical morphism associated with the object $\mathscr{O}_{G_{\Bbbk}^{\vee}}$, and the second one is induced by the tensor product (on the left) with $V \otimes_{\Bbbk} \mathscr{O}_{G_{\Bbbk}^{\vee}}$. Since $\widehat{\mathscr{T}^{\mathsf{Coh}}}$ is monoidal, using the case already treated, it follows that its composition with the morphism induced by $\widehat{\mathscr{T}}$ is the composition

$$\mathscr{O}(T^\vee_\Bbbk/W_\mathrm{f}) \xrightarrow{\mu_\Bbbk} \mathrm{End}_{\mathsf{P}^\wedge_{\mathrm{I}_\mathrm{u}},\mathrm{I}_\mathrm{u}}(\widehat{\mathscr{X}}(\Bbbk)) \to \mathrm{End}_{\mathsf{P}^\wedge_{\mathrm{I}_\mathrm{u}},\mathrm{I}_\mathrm{u}}(\widehat{\mathscr{X}}(V))$$

where the second map is induced by convolution on the left with $\widehat{\mathscr{Z}}(V)$. Now, interpreting μ_V in terms of right monodromy, it is clear that $\mu_V = \operatorname{id}_{\widehat{\mathscr{Z}}(V)} \widehat{\ast} \mu_{\Bbbk}$, which completes the proof.

Remark 10.5. Theorem 7.8(1) and the unicity in Lemma 4.5 imply that we have $\pi_{\dagger} \circ \widehat{\mathscr{Z}^{\mathsf{Coh}}} \cong \mathscr{Z}^{\mathsf{Coh}}$. Hence Lemma 10.4 implies that the composition of (4.9) with the natural morphism

$$\mathscr{O}(T^{\vee}_{\Bbbk}/W_{\mathrm{f}}) \to \mathrm{End}_{\mathsf{Coh}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk})}(V \otimes_{\Bbbk} \mathscr{O}_{G^{\vee}_{\Bbbk}})$$

factors through the quotient $\mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}}) \to \mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})/\mathcal{J} = \Bbbk$.

Since the complement of the open subset $G_{\Bbbk,\mathrm{reg}}^{\vee} \subset G_{\Bbbk}^{\vee}$ is known to have codimension at least 3 (see [Hu, §4.13]), restriction induces an isomorphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \xrightarrow{\sim} \mathscr{O}(G_{\Bbbk,\mathrm{reg}}^{\vee})$, hence a fully faithful monoidal functor

$$\operatorname{Coh}_{\mathrm{fr}}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk}) \to \operatorname{Coh}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk,\mathrm{reg}})$$

Recall that restriction to Σ induces an equivalence of monoidal categories

$$\operatorname{Coh}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk,\operatorname{reg}}^{\vee}) \xrightarrow{\sim} \operatorname{Rep}(\mathbb{J}_{\Sigma}),$$

see Proposition 2.20. We use this equivalence and the functor above to see the category $\mathsf{Coh}_{\mathrm{fr}}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee})$ as a full subcategory in $\mathsf{Rep}(\mathbb{J}_{\Sigma})$. In these terms, the canonical functor $\mathsf{Rep}(G_{\Bbbk}^{\vee}) \to \mathsf{Rep}(\mathbb{J}_{\Sigma})$ is given by $V \mapsto V \otimes_{\Bbbk} \mathscr{O}_{\Sigma}$, and the $\mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})$ -linear structure on $\mathsf{Coh}_{\mathrm{fr}}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee})$ corresponds to the natural $\mathscr{O}(\Sigma)$ -linear structure on $\mathsf{Rep}(\mathbb{J}_{\Sigma})$ via the identification $\Sigma \xrightarrow{\sim} T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$.

Let us consider $\mathscr{O}(G_{\Bbbk}^{\vee})$ with the G_{\Bbbk}^{\vee} -module structure induced by multiplication on the left. In §4.5 we have considered a morphism

(10.7)
$$\mathscr{O}(G_{\Bbbk}^{\vee}) \to \operatorname{End}_{\operatorname{Ind-Coh}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee})}(\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}})$$

constructed using the morphism

$$\mathscr{O}(G^{\vee}_{\Bbbk}) \to \operatorname{Hom}_{\operatorname{Ind-Coh}^{G^{\vee}_{\Bbbk}}(G^{\vee}_{\Bbbk})}(\mathscr{O}_{G^{\vee}_{\Bbbk}}, \mathscr{O}(G^{\vee}_{\Bbbk}) \otimes \mathscr{O}_{G^{\vee}_{\Bbbk}}) = \mathscr{O}(G^{\vee}_{\Bbbk} \times G^{\vee}_{\Bbbk})^{G^{\vee}_{\Bbbk}}$$

induced by the map $G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee} \to G_{\Bbbk}^{\vee}$ given by $(g,h) \mapsto g^{-1}hg$. In the terms above the right-hand side identifies with the space $\mathscr{O}(G_{\Bbbk}^{\vee} \times \Sigma)^{\mathbb{J}_{\Sigma}}$ of \mathbb{J}_{Σ} -invariant functions on $G_{\Bbbk}^{\vee} \times \Sigma$, and our morphism

$$\mathscr{O}(G^{\vee}_{\Bbbk}) \to \mathscr{O}(G^{\vee}_{\Bbbk} \times \Sigma)^{\mathbb{J}_{\Sigma}}$$

is induced by the morphism $G_{\Bbbk}^{\vee} \times \Sigma \to G_{\Bbbk}^{\vee}$ given again by $(g, h) \mapsto g^{-1}hg$.

We now consider the ind-object $\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \in \operatorname{Ind}-\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. This object is a ring ind-object, and taking the images of the morphisms above we obtain a ring morphism

$$\mathscr{O}(G^{\vee}_{\Bbbk}) \to \operatorname{Hom}_{\operatorname{Ind}\text{-}\mathsf{D}^{\wedge}_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}}}(\delta^{\wedge}, \widehat{\mathscr{X}}(\mathscr{O}(G^{\vee}_{\Bbbk}))) \subset \operatorname{End}_{\operatorname{Ind}\text{-}\mathsf{D}^{\wedge}_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}}}(\widehat{\mathscr{X}}(\mathscr{O}(G^{\vee}_{\Bbbk}))),$$

where the embedding is as in §4.5. We can therefore consider the tensor product

$$\mathscr{R}^{\wedge} := \widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\Sigma)$$

in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ (based on the general construction recalled in §B.1), which is the quotient of $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ by a left ideal. The same considerations as for $\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ based on the fact that $\widehat{\mathscr{Z}}$ is a central functor (see [B1, p. 73] for details) imply that any left ideal in $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ is also a right ideal, so that \mathscr{R}^{\wedge} also has a canonical structure of ring object in $\mathrm{Ind}\text{-}\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, such that the surjection

(10.8)
$$\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \to \mathscr{R}'$$

is a ring morphism.

Remark 10.6. From the definition we see that the restriction of (10.7) to the subalgebra $\mathscr{O}(G_{\Bbbk}^{\vee}/G_{\Bbbk}^{\vee}) \cong \mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})$ coincides with the morphism considered in Lemma 10.4; by this lemma, it therefore coincides with the restriction of monodromy. As a consequence, the action of $\mathscr{O}(\Sigma)$ on \mathscr{R}^{\wedge} induced by the obvious action on $\mathscr{O}(\Sigma)$ coincides, via the identification $\mathscr{O}(\Sigma) \cong \mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})$, with the monodromy action of $\mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})$.

10.4. Some properties of \mathscr{R}^{\wedge} . In this subsection we prove a number of properties of the object \mathscr{R}^{\wedge} .

Lemma 10.7. For any $V \in \mathsf{Rep}(G^{\vee}_{\Bbbk})$ we have a canonical isomorphism

$$\mathscr{R}^{\wedge} \stackrel{\sim}{\star} \mathscr{Z}(V) \cong \mathscr{R}^{\wedge} \otimes_{\Bbbk} V$$

in Ind- $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

Proof. By exactness of the functor $(-) \stackrel{\sim}{\star} \widehat{\mathscr{Z}}(V)$ (see Theorem 7.8(5)) we have

(10.9)
$$\mathscr{R}^{\wedge} \widehat{\star} \widehat{\mathscr{Z}}(V) \cong \left(\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \widehat{\star} \widehat{\mathscr{Z}}(V)\right) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\Sigma),$$

where $\mathscr{O}(G_{\Bbbk}^{\vee})$ acts on $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \stackrel{\sim}{\star} \widehat{\mathscr{Z}}(V)$ via its action on $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$. Now by monoidality of $\widehat{\mathscr{Z}}$ (see Theorem 7.8(3)) we have

$$\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \mathbin{\widehat{\star}} \widehat{\mathscr{Z}}(V) \cong \widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk}) \otimes_{\Bbbk} V)$$

where G_{\Bbbk}^{\vee} acts diagonally on $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes_{\Bbbk} V$. By standard arguments this tensor product identifies canonically with the similar tensor product where G_{\Bbbk}^{\vee} acts trivially on V, which provides an isomorphism

(10.10)
$$\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \stackrel{\sim}{\star} \widehat{\mathscr{Z}}(V) \cong \widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\Bbbk} V.$$

This isomorphism is the image under $\widehat{\mathscr{X}}^{\mathsf{Coh}}$ of an isomorphism

$$\left(\mathscr{O}(G^{\vee}_{\Bbbk})\otimes_{\Bbbk}\mathscr{O}_{G^{\vee}_{\Bbbk}}\right)\otimes_{\mathscr{O}_{G^{\vee}_{\Bbbk}}}\left(V\otimes_{\Bbbk}\mathscr{O}_{G^{\vee}_{\Bbbk}}\right)\cong \left(\mathscr{O}(G^{\vee}_{\Bbbk})\otimes_{\Bbbk}\mathscr{O}_{G^{\vee}_{\Bbbk}}\right)\otimes_{\Bbbk}V$$

in $\operatorname{Coh}_{\operatorname{fr}}^{G_{\Bbbk}^{\vee}}(G_{\Bbbk}^{\vee})$, which is easily seen to be $\mathscr{O}(G_{\Bbbk}^{\vee})$ -equivariant where on each side $\mathscr{O}(G_{\Bbbk}^{\vee})$ acts via its action on $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes_{\Bbbk} \mathscr{O}_{G_{\Bbbk}^{\vee}}$. Hence (10.10) is $\mathscr{O}(G_{\Bbbk}^{\vee})$ -equivariant where on each side $\mathscr{O}(G_{\Bbbk}^{\vee})$ acts via its action on $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$. Combining this with (10.9) we deduce the desired isomorphism.

Since \mathscr{R}^{\wedge} is defined as a quotient of $\widehat{\mathscr{L}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$, the following claim follows from the similar property of $\widehat{\mathscr{L}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ proved at the end of §7.4.

Lemma 10.8. The monodromy morphism $\mu_{\mathscr{R}^{\wedge}}$ factors through the multiplication morphism $\mathscr{O}(T_{\Bbbk}^{\vee} \times T_{\Bbbk}^{\vee}) = \mathscr{O}(T_{\Bbbk}^{\vee}) \otimes \mathscr{O}(T_{\Bbbk}^{\vee}) \to \mathscr{O}(T_{\Bbbk}^{\vee}).$

Recall that since the category $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is defined as a quotient of $\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, any object in this category admits a canonical action of $\mathscr{O}(T^{\vee}_{\Bbbk} \times T^{\vee}_{\Bbbk})$ which factors through an action of $\mathscr{O}(T^{\vee}_{\Bbbk} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} T^{\vee}_{\Bbbk})$ (see Lemma 6.7), and these actions commute with any morphism in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. From Lemma 10.8 and (6.7)–(6.8) one obtains that for any \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ we have

(10.11)
$$\mu_{\mathscr{R}^{\wedge}\widehat{\star}^{0}\mathscr{F}} = \mathrm{id}_{\mathscr{R}^{\wedge}} \widehat{\star}^{0} \mu_{\mathscr{F}}$$

Let consider the object $\pi_{\dagger}(\mathscr{R}^{\wedge})$ in Ind- $D_{I_u,I}$. The category Ind- $D_{I_u,I}$ is not triangulated in any obvious way, nor can we consider any form of "perverse" t-structure on it. However, for any *n* the perverse cohomology functor ${}^{p}\mathscr{H}^{n}: D_{I_u,I} \to \mathsf{P}_{I_u,I}$ induces a functor on ind-objects; we can therefore consider the object ${}^{p}\mathscr{H}^{0}(\pi_{\dagger}(\mathscr{R}^{\wedge}))$ in Ind- $\mathsf{P}_{I_u,I}$. Recall also the object \mathscr{R} in Ind- $\mathsf{P}_{I,I}$ considered in (4.16).

Lemma 10.9. We have a canonical isomorphism

$${}^{\mathrm{p}}\mathscr{H}^{0}(\pi_{\dagger}(\mathscr{R}^{\wedge})) \cong \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{R})$$

in Ind- $P_{I_u,I}$.

Proof. Choose a presentation of $\mathscr{O}(G_{\Bbbk}^{\vee})$ -modules

$$\mathscr{O}(G^{\vee}_{\Bbbk})^{\oplus r} \to \mathscr{O}(G^{\vee}_{\Bbbk}) \to \mathscr{O}(\Sigma) \to 0,$$

and consider the induced exact sequence

$$\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk}))^{\oplus r} \to \widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \to \mathscr{R}^{\wedge} \to 0$$

in Ind- $P^{\wedge}_{I_{u},I_{u}}$. Since π_{\dagger} is right t-exact (see §6.2), the functor

$${}^{\mathbf{p}}\mathscr{H}^{0}(\pi_{\dagger}(-)):\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

is right exact. By [KS, Corollary 8.6.8] it follows that the induced functor on indobjects is also right exact, and using Theorem 7.8(1) we deduce an exact sequence

$$\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{Z}(\mathscr{O}(G^{\vee}_{\Bbbk})))^{\oplus r} \to \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{Z}(\mathscr{O}(G^{\vee}_{\Bbbk}))) \to {^{\mathrm{p}}}\!\mathscr{H}^{0}(\pi_{\dagger}\mathscr{R}^{\wedge}) \to 0,$$

which shows that

$${}^{\mathbf{p}}\mathscr{H}^{0}(\pi_{\dagger}\mathscr{R}^{\wedge}) \cong \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\Sigma))$$

where the action of $\mathscr{O}(G_{\Bbbk}^{\vee})$ on $\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ is as in §4.5, or equivalently is obtained from the action on $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ by application of π_{\dagger} . By Remarks 10.5 and 10.6, the restriction of this action to $\mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})$ factors through the quotient morphism $\mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}}) \to \mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})/\mathcal{J} = \Bbbk$, so that the action of $\mathscr{O}(G_{\Bbbk}^{\vee})$ on $\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ factors through an action of the subscheme

$$G_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathrm{Spec}(\Bbbk),$$

where the morphism $\operatorname{Spec}(\Bbbk) \to T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$ corresponds to the image of $e \in T_{\Bbbk}^{\vee}$. We deduce that

$$\mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\Sigma) = \mathscr{Z}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathrm{Spec}(\Bbbk))} \mathscr{O}(\Sigma \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathrm{Spec}(\Bbbk)).$$

Now since the morphism $\Sigma \to T^{\vee}_{\Bbbk}/W_{\rm f}$ is an isomorphism we have

$$\Sigma \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} \mathrm{Spec}(\Bbbk) = \{\mathsf{u}\},\$$

so that finally

$$\mathscr{Z}(\mathscr{O}(G^{\vee}_{\Bbbk})) \otimes_{\mathscr{O}(G^{\vee}_{\Bbbk})} \mathscr{O}(\Sigma) = \mathscr{Z}(\mathscr{O}(G^{\vee}_{\Bbbk})) \otimes_{\mathscr{O}(G^{\vee}_{\Bbbk})} \mathscr{O}(\{\mathsf{u}\}) = \mathscr{R},$$

which finishes the proof.

10.5. The coaction morphism. Consider the comultiplication morphism (4.13). As in §4.8 this morphism defines a morphism of ind-objects

(10.12)
$$\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \to \widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\Bbbk} \mathscr{O}(G_{\Bbbk}^{\vee}).$$

Here we can interpret the right-hand side as the tensor product

$$\mathscr{Z}(\mathscr{O}(G^{\vee}_{\Bbbk}))\otimes_{\mathscr{O}(G^{\vee}_{\Bbbk})}\mathscr{O}(G^{\vee}_{\Bbbk} imes G^{\vee}_{\Bbbk})$$

where $\mathscr{O}(G_{\Bbbk}^{\vee})$ acts on $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ as in the definition of \mathscr{R}^{\wedge} and the morphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathscr{O}(G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee})$ is induced by the first projection. Hence using the morphism $\mathscr{O}(G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee}) \to \mathscr{O}(\mathbb{J}_{\Sigma})$ induced by the composition

$$\mathbb{J}_{\Sigma} \hookrightarrow \Sigma \times G^{\vee}_{\Bbbk} \hookrightarrow G^{\vee}_{\Bbbk} \times G^{\vee}_{\Bbbk}$$

we obtain a canonical morphism

$$\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \to \widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \otimes_{\mathscr{O}(G^{\vee}_{\Bbbk})} \mathscr{O}(\mathbb{J}_{\Sigma})$$

where the morphism $\mathscr{O}(G^{\vee}_{\Bbbk}) \to \mathscr{O}(\mathbb{J}_{\Sigma})$ is induced by the composition

$$\mathbb{J}_{\Sigma} \hookrightarrow \Sigma \times G^{\vee}_{\Bbbk} \to \Sigma \hookrightarrow G^{\vee}_{\Bbbk}$$

where the second morphism is the obvious projection. Now, using (B.1) we obtain a canonical isomorphism

$$\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \otimes_{\mathscr{O}(G^{\vee}_{\Bbbk})} \mathscr{O}(\mathbb{J}_{\Sigma}) \cong \mathscr{R}^{\wedge} \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma}),$$

where the morphism $\mathscr{O}(\Sigma) \to \mathscr{O}(\mathbb{J}_{\Sigma})$ is induced by the natural projection $\mathbb{J}_{\Sigma} \to \Sigma$. We can therefore consider our morphism as a morphism

(10.13)
$$\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \to \mathscr{R}^{\wedge} \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma})$$

Lemma 10.10. The morphism (10.13) factors (uniquely) through a morphism

 $\operatorname{coact}_{\mathscr{R}^{\wedge}}:\mathscr{R}^{\wedge}\to\mathscr{R}^{\wedge}\otimes_{\mathscr{O}(\Sigma)}\mathscr{O}(\mathbb{J}_{\Sigma}).$

Proof. Consider the action of $\mathscr{O}(G^{\vee}_{\Bbbk}) \otimes \mathscr{O}(G^{\vee}_{\Bbbk})$ on

$$\widehat{\mathscr{F}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \otimes_{\Bbbk} \mathscr{O}(G^{\vee}_{\Bbbk})$$

where the left copy acts via the action on $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ used in the definition of \mathscr{R}^{\wedge} , and the right copy acts via multiplication on $\mathscr{O}(G_{\Bbbk}^{\vee})$. Of course, via the identification

$$\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \otimes_{\Bbbk} \mathscr{O}(G^{\vee}_{\Bbbk}) \cong \widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \otimes_{\mathscr{O}(G^{\vee}_{\Bbbk})} \mathscr{O}(G^{\vee}_{\Bbbk} \times G^{\vee}_{\Bbbk})$$

this action corresponds to the action on the right-hand side induced by the obvious action of $\mathscr{O}(G^{\vee}_{\Bbbk} \times G^{\vee}_{\Bbbk})$ on itself.

An explicit computation using the Hopf algebra operations in $\mathscr{O}(G_{\Bbbk}^{\vee})$ shows that the morphism (10.12) is $\mathscr{O}(G_{\Bbbk}^{\vee})$ -linear, where the action on the left-hand side is as in the definition of \mathscr{R}^{\wedge} and that on the right-hand side is obtained from the action of $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}(G_{\Bbbk}^{\vee})$ considered above via the morphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}(G_{\Bbbk}^{\vee})$ induced by the map

$$G^\vee_\Bbbk \times G^\vee_\Bbbk \to G^\vee_\Bbbk$$

given by $(g,h) \mapsto h^{-1}gh$. Hence our morphism

a a a at

$$\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \to \widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \otimes_{\mathscr{O}(G^{\vee}_{\Bbbk})} \mathscr{O}(G^{\vee}_{\Bbbk} \times G^{\vee}_{\Bbbk})$$

is $\mathscr{O}(G_{\Bbbk}^{\vee})$ -linear where the action on the right-hand side is induced by the same morphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}(G_{\Bbbk}^{\vee})$ and the obvious action of $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}(G_{\Bbbk}^{\vee})$. If follows that the morphism

$$\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \to \widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\mathbb{J}_{\Sigma})$$

used to define (10.13) is $\mathscr{O}(G_{\Bbbk}^{\vee})$ -linear where the action on the right-hand side is obtained from the obvious action of $\mathscr{O}(\mathbb{J}_{\Sigma})$ on itself via the morphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathscr{O}(\mathbb{J}_{\Sigma})$ induced by the map

$$\mathbb{J}_{\Sigma} \to G^{\vee}_{\mathbb{k}}$$

given by $(s,h) \mapsto h^{-1}sh$. By definition of \mathbb{J}_{Σ} this morphism coincides with the projection $\mathbb{J}_{\Sigma} \to \Sigma$; in particular, this action of $\mathscr{O}(G_{\Bbbk}^{\vee})$ on $\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\mathbb{J}_{\Sigma})$ factors through the restriction morphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathscr{O}(\Sigma)$, which implies the desired property for (10.13).

It is clear by construction that the morphism $\operatorname{coact}_{\mathscr{R}^{\wedge}}$ from Lemma 10.10 is "counital" in the sense that the composition

$$\mathscr{R}^{\wedge} \xrightarrow{\operatorname{coact}_{\mathscr{R}^{\wedge}}} \mathscr{R}^{\wedge} \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma}) \to \mathscr{R}'$$

(where the second map is induced by restriction to the identity section of \mathbb{J}_{Σ}) is $\mathrm{id}_{\mathscr{R}^{\wedge}}$, and "coassociative" in the sense that the composition

$$\mathscr{R}^{\wedge} \xrightarrow{\operatorname{coact}_{\mathscr{R}^{\wedge}}} \mathscr{R}^{\wedge} \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma}) \to \mathscr{R}^{\wedge} \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma}) \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma})$$
(where the second map is induced by the comultiplication map for the group scheme \mathbb{J}_{Σ}) coincides with the composition

$$\mathscr{R}^{\wedge} \xrightarrow{\operatorname{coact}_{\mathscr{R}^{\wedge}}} \mathscr{R}^{\wedge} \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma}) \xrightarrow{\operatorname{coact}_{\mathscr{R}^{\wedge}} \otimes \operatorname{id}} \mathscr{R}^{\wedge} \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma}) \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma}).$$

10.6. A monoidality morphism. In this subsection we explain the construction of a morphism which will be an ingredient in the construction of the monoidal structure on the functor Φ_{I_u,I_u} .

Recall that the product $\widehat{\mathscr{X}}(V) \stackrel{\sim}{\star} \mathscr{F}$ is perverse for any $V \in \mathsf{Rep}(G_{\Bbbk}^{\vee})$ and $\mathscr{F} \in \mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, see Theorem 7.8(5). In view of the construction of the product $\stackrel{\sim}{\star}^{0}$ (see §6.4), it follows that for any $V \in \mathsf{Rep}(G_{\Bbbk}^{\vee})$ and $\mathscr{F} \in \mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{0}$ the product $\widehat{\mathscr{X}}(V) \stackrel{\sim}{\star}^{0} \mathscr{F}$ belongs to $\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{0}$.

Lemma 10.11. Let $\mathscr{F} \in \mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. We have a canonical identification

$$\mathscr{R}^{\wedge p} \widehat{\star}^{0} \mathscr{F} = (\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \widehat{\star}^{0} \mathscr{F}) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\Sigma),$$

where the tensor product in the right-hand side is taken in the abelian category Ind- $\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, and the action of $\mathscr{O}(G^{\vee}_{\Bbbk})$ is induced by that on $\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk}))$.

Proof. By definition we have $\mathscr{R}^{\wedge} = \widehat{\mathscr{L}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\Sigma)$. Choosing a presentation as in the proof of Lemma 10.9 we obtain an exact sequence

$$\widehat{\mathscr{Z}}(\mathscr{O}(G^\vee_\Bbbk))^{\oplus r} \to \widehat{\mathscr{Z}}(\mathscr{O}(G^\vee_\Bbbk)) \to \mathscr{R}^\wedge \to 0.$$

By Lemma 6.9 the functor $(-) {}^{p} \widehat{\star}^{0} \mathscr{F} : \mathsf{P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}} \to \mathsf{P}^{0}_{\mathrm{I}_{u},\mathrm{I}_{u}}$ is right exact. By [KS, Corollary 8.6.8], it follows that the same is true for the extension of this functor to ind-objects, and we deduce an exact sequence

$$(\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk}))\,\widehat{\star}^{0}\,\mathscr{F})^{\oplus r}\to\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk}))\,\widehat{\star}^{0}\,\mathscr{F}\to\mathscr{R}^{\wedge}\,{^{\mathrm{p}}}\widehat{\star}^{0}\,\mathscr{F}\to 0$$

in Ind- $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. It follows that

$$\mathscr{R}^{\wedge p} \widehat{\star}^{0} \mathscr{F} = (\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \widehat{\star}^{0} \mathscr{F}) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(\Sigma),$$

as desired.

Using this lemma, we will now explain how to construct, for \mathscr{F}, \mathscr{G} in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$, a canonical morphism

(10.14)
$$(\mathscr{R}^{\wedge p} \widehat{\star}^{0} \mathscr{F}) \, {}^{p} \star^{0}_{I_{u}} \, (\mathscr{R}^{\wedge p} \widehat{\star}^{0} \, \mathscr{G}) \to \mathscr{R}^{\wedge p} \widehat{\star}^{0} \, (\mathscr{F} \, {}^{p} \star^{0}_{I_{u}} \, \mathscr{G})$$

in Ind- P_{I_u,I_u}^0 . First, from Lemma 10.11 and the right t-exactness of ${}^{p}\star_{I_u}^0$ (see §5.2) we deduce a canonical isomorphism

$$\begin{split} (\mathscr{R}^{\wedge \ \mathbf{p}} \widehat{\star}^{0} \ \mathscr{F}) \ ^{\mathbf{p}} & \star^{0}_{\mathbf{I}_{\mathbf{u}}} \left(\mathscr{R}^{\wedge \ \mathbf{p}} \widehat{\star}^{0} \ \mathscr{G} \right) = \\ & \left(\left(\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \ \widehat{\star}^{0} \ \mathscr{F} \right) \ ^{\mathbf{p}} \star^{0}_{\mathbf{I}_{\mathbf{u}}} \left(\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \ \widehat{\star}^{0} \ \mathscr{G} \right) \right) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee})} \ \mathscr{O}(\Sigma \times \Sigma), \end{split}$$

where $\mathscr{O}(G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee}) = \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}(G_{\Bbbk}^{\vee})$ acts via its action on the two factors $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$. Hence to construct (10.14) it suffices to construct a morphism

108

which is annihilated by the ideal of $\Sigma \times \Sigma \subset G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee}$. Now from the definition of ${}^{p}\star_{L_{\nu}}^{0}$, (7.8) and (6.13) we obtain isomorphisms

$$\begin{split} (\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \mathbin{\widehat{\star}}^{0} \mathscr{F}) \, {}^{\mathbf{p}} \star_{\mathrm{I}_{\mathrm{u}}}^{0} \left(\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \mathbin{\widehat{\star}}^{0} \mathscr{G} \right) = \\ & {}^{\mathrm{p}} \mathscr{H}^{0} \big((\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \mathbin{\widehat{\star}}^{0} \mathscr{F}) \star_{\mathrm{I}_{\mathrm{u}}}^{0} \left(\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \mathbin{\widehat{\star}}^{0} \mathscr{G}) \right) \cong \\ & {}^{\mathrm{p}} \mathscr{H}^{0} \big((\mathscr{F} \mathbin{\widehat{\star}}^{0} \ \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee}))) \star_{\mathrm{I}_{\mathrm{u}}}^{0} \left(\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \mathbin{\widehat{\star}}^{0} \mathscr{G}) \right) \cong \\ & {}^{\mathrm{p}} \mathscr{H}^{0} \big((\mathscr{F} \mathbin{\widehat{\star}}^{0} \ (\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \mathbin{\widehat{\star}} \ \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})))) \star_{\mathrm{I}_{\mathrm{u}}}^{0} \mathscr{G} \big). \end{split}$$

By Lemma 10.3 these isomorphisms commute with the actions of $\mathscr{O}(G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee})$ induced by the actions on the factors $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$. Now multiplication in $\mathscr{O}(G_{\Bbbk}^{\vee})$ induces a morphism

$$\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \mathbin{\widehat{\star}} \widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \to \widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk}))$$

which is $\mathscr{O}(G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee})$ -equivariant, where the action on the right-hand side is the composition of the product morphism $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}(G_{\Bbbk}^{\vee}) \to \mathscr{O}(G_{\Bbbk}^{\vee})$ with the given action of $\mathscr{O}(G_{\Bbbk}^{\vee})$. We deduce a canonical morphism

$$(\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk}))\,\widehat{\star}^{0}\,\mathscr{F})\,{}^{\mathrm{p}}\!\!\star^{0}_{\mathrm{I}_{\mathrm{u}}}\,(\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk}))\,\widehat{\star}^{0}\,\mathscr{G})\to{}^{\mathrm{p}}\!\mathscr{H}^{0}((\mathscr{F}\,\widehat{\star}^{0}\,\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})))\,\star^{0}_{\mathrm{I}_{\mathrm{u}}}\,\mathscr{G}).$$

Using (7.8) and (6.14), the right-hand side identifies with

$$\widehat{\mathscr{Z}}(\mathscr{O}(G^{\vee}_{\Bbbk})) \mathbin{\widehat{\star}}^{0} (\mathscr{F} {}^{\mathrm{p}} \star^{0}_{\mathrm{I}_{\mathrm{u}}} \mathscr{G});$$

it therefore admits a canonical morphism to $\mathscr{R}^{\wedge p_{\star}^{\circ}0}$ ($\mathscr{F}^{\mathsf{p}_{\star}^{\circ}0}_{\mathsf{I}_{\mathrm{u}}}\mathscr{G}$). Combining these morphisms we obtain a morphism (10.15), and from the construction and the comments above on equivariance one can check that this morphism is indeed annihilated by the ideal of $\Sigma \times \Sigma \subset G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee}$; it therefore induces the whished-for morphism (10.14).

10.7. **Exactness.** This (technical) subsection is devoted to the proof of the following claim, which will be crucial for our considerations below.

Lemma 10.12. The functor

(10.16)
$$\mathscr{R}^{\wedge p} \widehat{\star}^{0}(-) : \mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathrm{Ind} \mathsf{-} \mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

is exact. Moreover, for ${\mathscr G}$ in ${\mathsf P}^0_{I,I}$ we have a canonical isomorphism

(10.17)
$$\mathscr{R}^{\wedge p} \widehat{\star}^{0} (\pi_{0}^{\dagger} \mathscr{G}) \cong \pi_{0}^{\dagger} (\mathscr{R}^{0} \star_{\mathrm{I}}^{0} \mathscr{G})$$

To prove this lemma we will need some preliminary results. Let us choose a complex of $\mathscr{O}(G^\vee_\Bbbk)\text{-modules}$

(10.18)
$$\cdots \to 0 \to P^{-2} \xrightarrow{a} P^{-1} \xrightarrow{b} P^0 \to 0 \to \cdots$$

where each P^j is free of finite rank (and placed in degree j), the natural morphism $\operatorname{Im}(a) \to \ker(b)$ is an isomorphism (in other words, our complex is exact in degree -1) and $\operatorname{coker}(b) \cong \mathscr{O}(\Sigma)$. Tensoring with $\mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee}))$ we deduce a complex

of objects in $Ind-P_{I,I}^0$.

Lemma 10.13. The natural morphism $\operatorname{Im}(\widetilde{a}) \to \ker(\widetilde{b})$ is an isomorphism.

Proof. Recall from §4.8 the equivalence of categories $\Phi_{I,I} : P_{I,I}^0 \xrightarrow{\sim} \operatorname{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))$. Passing to ind-objects we deduce an equivalence $\operatorname{Ind} \operatorname{P}_{I,I}^0 \xrightarrow{\sim} \operatorname{Ind} \operatorname{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))$. Now the category $\operatorname{Ind} \operatorname{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))$ identifies with the category $\operatorname{Rep}^{\infty}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))$ of all algebraic $Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})$ -modules (see [KS, §6.3]), and under the equivalence

Ind-
$$\mathsf{P}^0_{\mathrm{I},\mathrm{I}} \xrightarrow{\sim} \mathsf{Rep}^\infty(Z_{G^\vee_\Bbbk}(\mathsf{u}))$$

the object $\mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee}))$ corresponds to $\mathscr{O}(G_{\Bbbk}^{\vee})$, with the structure of $Z_{G_{\Bbbk}^{\vee}}(\mathfrak{u})$ -module induced by multiplication on the left on G_{\Bbbk}^{\vee} . Through these identifications, the action of $\mathscr{O}(G_{\Bbbk}^{\vee})$ on $\mathscr{Z}^0(\mathscr{O}(G_{\Bbbk}^{\vee}))$ corresponds to the action on $\mathscr{O}(G_{\Bbbk}^{\vee})$ where $\varphi \in \mathscr{O}(G_{\Bbbk}^{\vee})$ acts by multiplication by the function $g \mapsto \varphi(g^{-1}\mathfrak{u}g)$ (see Remark 4.6). To prove our claim, it therefore suffices to prove that the complex of $Z_{G_{\Bbbk}^{\vee}}(\mathfrak{u})$ -modules

$$\dots \to 0 \to \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} P^{-2} \to \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} P^{-1} \\ \to \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} P^{0} \to 0 \to \dots$$

has no cohomology in degree -1. Now, the cohomology in degree -1 of this complex is

$$\operatorname{Tor}_{1}^{\mathscr{O}(G_{\Bbbk}^{\vee})}(\mathscr{O}(G_{\Bbbk}^{\vee}),\mathscr{O}(\Sigma)).$$

If we let G_{\Bbbk}^{\vee} act on $\mathscr{O}(G_{\Bbbk}^{\vee})$ via the right regular action, then for the action above $\mathscr{O}(G_{\Bbbk}^{\vee})$ becomes a G_{\Bbbk}^{\vee} -equivariant $\mathscr{O}(G_{\Bbbk}^{\vee})$ -module (where G_{\Bbbk}^{\vee} acts on the algebra $\mathscr{O}(G_{\Bbbk}^{\vee})$ via conjugation). The desired claim therefore follows from Corollary 2.10.

Now we can come to the main step towards Lemma 10.12. Here, as for Lemma 10.9 we will use the fact that it makes sense to apply a functor ${}^{\mathrm{p}}\mathscr{H}^{n}$ to an object in Ind-D⁰_{In I}, even though there is no "perverse t-structure" on this category.

Lemma 10.14. We have

$${}^{p}\hspace{-0.5mm}\mathscr{H}^{n}(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\pi_{\dagger}\mathscr{R}^{\wedge})) = \begin{cases} \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{R}^{0}) & \textit{if } n = 0; \\ 0 & \textit{if } n > 0 \textit{ or } n = -1. \end{cases}$$

Proof. By definition of the perverse t-structure on $\mathsf{D}^{0}_{\mathrm{I}_{u},\mathrm{I}}$ the functor $\Pi^{0}_{\mathrm{I}_{u},\mathrm{I}} : \mathsf{D}_{\mathrm{I}_{u},\mathrm{I}} \to \mathsf{D}^{0}_{\mathrm{I}_{u},\mathrm{I}}$ is t-exact with respect to the perverse t-structures; this functor therefore commutes with the functor ${}^{p}\mathscr{H}^{n}$, and then the same property holds for the extensions to ind-objects. Using Lemma 10.9, we deduce the case n = 0. Similarly, since \mathscr{R}^{\wedge} belongs to $\mathrm{Ind}\operatorname{-P}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}}$, and since π_{\dagger} is right t-exact (see §6.2), we have ${}^{p}\mathscr{H}^{n}(\pi_{\dagger}\mathscr{R}^{\wedge}) = 0$ for any n > 0, which implies the claim in this case.

It remains to treat the case n = -1. Consider again the complex (10.18), and the complex

$$(10.19) \quad \dots \to 0 \to \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} P^{-2} \xrightarrow{f} \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} P^{-1}$$
$$\xrightarrow{g} \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} P^{0} \to 0 \to \dots$$

in Ind- $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ obtained by tensoring with $\widehat{\mathscr{F}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$. Let us denote by A the full subcategory of $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ whose objects are the perverse sheaves \mathscr{F} such that $\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\pi_{\dagger}(\mathscr{F}))$ belongs to the heart of the perverse t-structure. Then A is an additive category, and the natural functor Ind-A \rightarrow Ind- $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is fully faithful by [KS, Proposition 6.1.10]. The object $\widehat{\mathscr{Z}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ belongs to the essential image of this functor, see Theorem 7.8(1); the same is therefore true for any term of our complex (10.19).

For an additive category B, let us denote by $C^{[-2,0]}(B)$ the category of complexes of objects of B whose components are zero in all degrees except possibly -2, -1and 0. By [KS, Lemma 15.4.1], the natural functor

Ind-
$$C^{[-2,0]}(A) \to C^{[-2,0]}(Ind-A)$$

is an equivalence of categories. This implies that there exist a filtrant category I, inductive systems $(\mathscr{M}_i^{-2}: i \in I), (\mathscr{M}_i^{-1}: i \in I)$ and $(\mathscr{M}_i^0: i \in I)$ of objects of A, and morphisms of inductive systems $(f_i: \mathscr{M}_i^{-2} \to \mathscr{M}_i^{-1})_{i \in I}, (g_i: \mathscr{M}_i^{-1} \to \mathscr{M}_i^0)_{i \in I}$ such that $g_i \circ f_i = 0$ for any i, and such that (10.19) is isomorphic to

$$\cdots \to 0 \to \underset{i \in I}{\overset{\underset{i \in I}{\boxplus}}{\longrightarrow}} \mathscr{M}_i^{-2} \xrightarrow{\overset{\underset{i \in I}{\amalg}}{\longrightarrow}} \underset{i \in I}{\overset{\underset{i \in I}{\amalg}}{\longrightarrow}} \mathscr{M}_i^{-1} \xrightarrow{\overset{\underset{i \in I}{\amalg}}{\longrightarrow}} \underset{i \in I}{\overset{\underset{i \in I}{\amalg}}{\longrightarrow}} \mathscr{M}_i^0 \to 0 \to \cdots$$

(as a complex of objects in $\operatorname{Ind}-\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$). For any $i \in I$ we set $\mathscr{Q}_{i} := \operatorname{coker}(g_{i})$; these objects define in a natural way an inductive system of objects in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. The object \mathscr{R}^{\wedge} is isomorphic to the cokernel of g; in view of the description of cokernels in ind-objects in an abelian category (see [KS, Lemma 8.6.4(ii)]), we therefore have

$$\mathscr{R}^{\wedge} \cong \underset{i \in I}{\overset{\circ}{\amalg}} \mathscr{Q}_i,$$

so that what we have to prove is that

(10.20)
$$\underset{i \in I}{\overset{\text{"Im}}{\longrightarrow}} \mathcal{H}^{-1}\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathcal{Q}_{i}) = 0.$$

Note that with these notations, Lemma 10.13 (combined with the t-exactness of $For_{I}^{I,0}$) says that the complex

$$(10.21) \quad \dots \to 0 \to \underset{i \in I}{\overset{\text{``Im}}{\longrightarrow}} \Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \pi_{\dagger}(\mathscr{M}_i^{-2}) \xrightarrow{\overset{\text{``Im}}{\longrightarrow}} \Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \pi_{\dagger}(f_i) \\ \xrightarrow{\overset{\text{``Im}}{\longrightarrow}} \Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \pi_{\dagger}(g_i) \\ \xrightarrow{\overset{\text{``Im}}{\longrightarrow}} \Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \pi_{\dagger}(\mathscr{M}_i^0) \to 0 \to \dots$$

of objects in $\operatorname{Ind}-\mathsf{P}^0_{\mathrm{I}_n,\mathrm{I}}$ has no cohomology in degree -1.

Recall from (6.11) the equivalence of triangulated categories

$$D^{\mathrm{b}}\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \xrightarrow{\sim} \mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

provided by the "realization functor." For any $i \in I$ we consider the complex

$$\widetilde{\mathscr{S}_i} = (\dots \to 0 \to \mathscr{M}_i^{-2} \xrightarrow{f_i} \mathscr{M}_i^{-1} \xrightarrow{g_i} \mathscr{M}_i^0 \to 0 \to \dots)$$

of objects in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ (seen as an object in $D^{\mathrm{b}}\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$), and denote by \mathscr{S}_{i} its image in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. If we set

$$\widetilde{\mathscr{S}'_i} = (\dots \to 0 \to \mathscr{M}_i^{-2} \xrightarrow{f_i} \ker(g_i) \to 0 \to \dots)$$

(seen as an object in $D^{\mathrm{b}}\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$) where \mathscr{M}_{i}^{-2} is in degree -2, and denote by \mathscr{S}'_{i} its image in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ then we have a distinguished triangle

$$\widetilde{\mathscr{S}'_i} \to \widetilde{\mathscr{S}_i} \to \mathscr{Q}_i \xrightarrow{[1]}$$

in $D^{\mathrm{b}}\mathsf{P}^{\wedge}_{\mathrm{Iu},\mathrm{Iu}}$, hence a distinguished triangle

$$\mathscr{S}'_i \to \mathscr{S}_i \to \mathscr{Q}_i \xrightarrow{[1]}$$

in $D^{\wedge}_{I_u,I_u}$. Applying the triangulated functor $\Pi^0_{I_u,I}\pi_{\dagger}$, then taking the long exact sequence in perverse cohomology, and finally formal direct limits, we deduce an exact sequence

of objects in Ind- $\mathsf{P}^{0}_{\mathrm{I}_{u},\mathrm{I}}$. By right t-exactness of the functor π_{\dagger} , and since each \mathscr{S}'_{i} is concentrated in negative perverse degrees, the third term in this sequence vanishes. As a consequence, to prove (10.20) it suffices to prove that

(10.22)
$$\underset{i \in I}{``\mathrm{lim}} \overset{\mathrm{vp}}{\mathcal{H}}^{-1} \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \pi_{\dagger}(\mathscr{S}_{i}) = 0.$$

Now we set

$$\widetilde{\mathscr{S}}_i'' = (\dots \to 0 \to \mathscr{M}_i^{-2} \xrightarrow{f_i} \mathscr{M}_i^{-1} \to 0 \to \dots)$$

(seen as an object in $D^{\mathrm{b}}\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$) where \mathscr{M}_{i}^{-2} is in degree -2, and denote by $\mathscr{S}_{i}^{\prime\prime}$ its image in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. We have distinguished triangles

$$\mathcal{M}_i^0 \to \mathcal{S}_i \to \mathcal{S}_i'' \xrightarrow{[1]}, \qquad \mathcal{M}_i^{-1}[1] \to \mathcal{S}_i'' \to \mathcal{M}_i^{-2}[2] \xrightarrow{[1]}$$

where in the second triangle the morphism $\mathcal{M}_i^{-2}[2] \to \mathcal{M}_i^{-1}[2]$ is $f_i[2]$, and in the first triangle the morphism $\mathcal{S}_i'' \to \mathcal{M}_i^0[1]$ is the unique morphism whose composition with the map $\mathcal{M}_i^{-1}[1] \to \mathcal{S}_i''$ appearing in the second triangle is $g_i[1]$. (The existence and unicity of this morphism is guaranteed by the long exact sequence obtained by applying $\operatorname{Hom}(-, \mathcal{M}_i^0[1])$ to the second triangle.) Applying the triangulated functor $\Pi_{0,u}^{1}\pi_{\dagger}$, we obtain distinguished triangles

$$\begin{split} \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{M}^{0}_{i}) \to \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}_{i}) \to \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}''_{i}) \xrightarrow{[1]},\\ \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{M}^{-1}_{i})[1] \to \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}''_{i}) \to \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{M}^{-2}_{i})[2] \xrightarrow{[1]} \end{split}$$

Since $\Pi^0_{I_u,I}\pi_{\dagger}(\mathscr{M}_i^{-2})$ is perverse by definition of A, taking the long exact sequence of perverse cohomology associated with the second triangle we obtain an exact sequence

(10.23)
$$\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{M}_{i}^{-2}) \to \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{M}_{i}^{-1}) \to {}^{\mathrm{p}}\mathscr{H}^{-1}\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}_{i}'') \to 0,$$

which identifies ${}^{p}\mathscr{H}^{-1}\Pi^{0}_{\mathrm{I}_{u},\mathrm{I}}\pi_{\dagger}(\mathscr{S}''_{i})$ with $\operatorname{coker}(\Pi^{0}_{\mathrm{I}_{u},\mathrm{I}}\pi_{\dagger}(f_{i}))$. On the other hand, the same procedure applied to the first distinguished triangle produces an exact sequence

$$0 \to {}^{\mathbf{p}}\mathscr{H}^{-1}\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}_{i}) \to {}^{\mathbf{p}}\mathscr{H}^{-1}\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}_{i}'') \to \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{M}_{i}^{0}).$$

Here, by construction the composition of the right morphism with the surjection $\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{M}_i^{-1}) \to {}^{\mathrm{p}}\mathscr{H}^{-1}\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}''_i)$ from (10.23) is $\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(g_i)$. Hence, through the identification ${}^{\mathrm{p}}\mathscr{H}^{-1}\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}''_i) \cong \operatorname{coker}(\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(f_i))$, this exact sequence identifies ${}^{\mathrm{p}}\mathscr{H}^{-1}\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}_i)$ with the kernel of the morphism

$$\operatorname{coker}(\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(f_i)) \to \Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{M}^0_i)$$

induced by $\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(g_i)$. Passing to formal direct limits and then using the description of kernels and cokernels in ind-objects in an abelian category (see [KS, Lemma 8.6.4(ii)]), we deduce that " \varinjlim_i " ${}^{\mathfrak{m}}\mathcal{H}^{-1}\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{S}_i)$ identifies with the kernel of the morphism

$$\operatorname{coker}(\underset{i}{\overset{\shortparallel}{\amalg}} \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(f_{i})) \to \underset{i}{\overset{\shortparallel}{\amalg}} \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\pi_{\dagger}(\mathscr{M}_{i}^{0})$$

induced by " \varinjlim_{i} " $\Pi^{0}_{I_{u},I}\pi_{\dagger}(g_{i})$. The exactness of the complex (10.21) in degree -1 exactly says that this morphism is injective, which shows (10.22) and finishes the proof.

Proof of Lemma 10.12. By construction, the bifunctor

$$\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \times \mathsf{D}^{0}_{\mathrm{I},\mathrm{I}} \to \mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

given by $(\mathscr{F},\mathscr{G}) \mapsto \mathscr{F}^0(\pi_0^{\dagger}\mathscr{G})$ is the unique bifunctor through which the bifunctor

$$\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \times \mathsf{D}_{\mathrm{I},\mathrm{I}} \to \mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

given by $(\mathscr{F}, \mathscr{G}) \mapsto \Pi^0_{I_u, I_u} (\mathscr{F} \widehat{\star} (\pi^{\dagger} \mathsf{For}^I_{I_u} (\mathscr{G})))$ factors. Now, by (6.5)–(6.6), for \mathscr{F} in $\mathsf{D}^{\wedge}_{I_u, I_u}$ and \mathscr{G} in $\mathsf{D}_{I, I}$ we have a canonical isomorphism

$$\mathscr{F} \mathbin{\widehat{\star}} (\pi^{\dagger} \mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}(\mathscr{G})) \cong \pi^{\dagger}((\pi_{\dagger}\mathscr{F}) \star_{\mathrm{I}} \mathscr{G}).$$

We deduce, for \mathscr{F} in $\mathsf{D}^\wedge_{\mathrm{I}_\mathrm{u},\mathrm{I}_\mathrm{u}}$ and \mathscr{G} in $\mathsf{D}^0_{\mathrm{I},\mathrm{I}}$, a bifunctorial isomorphism

$$\mathscr{F} \mathbin{\widehat{\star}}^0(\pi_0^{\dagger} \mathscr{G}) \cong \pi^{\dagger,0} \big((\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \pi_{\dagger} \mathscr{F}) \star^0_{\mathrm{I}} \mathscr{G} \big)$$

(where the bifunctor \star_{I}^{0} here is that defined in §4.7, and the functor $\pi^{\dagger,0}$ is defined in §5.3) and then, by t-exactness of $\pi^{\dagger,0}$ and \star_{I}^{0} , for any $n \in \mathbb{Z}$ we deduce an isomorphism

$${}^{\mathbf{p}}\mathscr{H}^{n}(\mathscr{F} \mathbin{\widehat{\star}}^{0}(\pi_{0}^{\dagger}\mathscr{G})) \cong \pi^{\dagger,0} \big({}^{\mathbf{p}}\mathscr{H}^{n}(\Pi^{0}_{\mathrm{I}_{u},\mathrm{I}}\pi_{\dagger}\mathscr{F}) \star^{0}_{\mathrm{I}} \mathscr{G} \big).$$

This isomorphism extends to ind-objects, and provides for \mathscr{G} in $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$ and $n \in \mathbb{Z}$ an isomorphism

(10.24)
$${}^{\mathbf{p}}\mathscr{H}^{n}(\mathscr{R}^{\wedge} \widehat{\star}^{0}(\pi_{0}^{\dagger}\mathscr{G})) \cong \pi^{\dagger,0}({}^{\mathbf{p}}\mathscr{H}^{n}(\Pi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}^{0}\pi_{\dagger}\mathscr{R}^{\wedge}) \star_{\mathrm{I}}^{0}\mathscr{G})$$

Applying (10.24) in case n = 0 and using Lemma 10.14, we obtain an isomorphism

$$\mathscr{R}^{\wedge \ \mathrm{p}} \widehat{\star}^0 \left(\pi_0^\dagger \mathscr{G} \right) \cong \pi^{\dagger,0} ((\mathrm{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0} \mathscr{R}^0) \star_{\mathrm{I}}^0 \mathscr{G}).$$

Using (4.11) and (5.2), we deduce the isomorphism (10.17).

The isomorphism (10.24) and Lemma 10.14 also imply that for any ${\mathscr G}$ in ${\sf P}^0_{{\rm I},{\rm I}}$ we have

$${}^{\mathbf{p}}\mathscr{H}^{n}(\mathscr{R}^{\wedge} \,\widehat{\star}^{0}(\pi_{0}^{\dagger}\mathscr{G})) = 0 \quad \text{if } n > 0 \text{ or } n = -1.$$

We claim that in fact, for any \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ we have

(10.25)
$${}^{\mathrm{p}}\mathscr{H}^{n}(\mathscr{R}^{\wedge} \widehat{\star}^{0} \mathscr{F}) = 0 \text{ if } n > 0 \text{ or } n = -1$$

Indeed, write $\mathscr{R}^{\wedge} = " \varinjlim_{i} " \mathscr{R}^{\wedge}_{i}$ with each \mathscr{R}^{\wedge}_{i} in $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. Given an exact sequence

$$\mathscr{F}_1 \hookrightarrow \mathscr{F}_2 \twoheadrightarrow \mathscr{F}_2$$

in $\mathsf{P}^{0}_{\mathrm{I}_{u},\mathrm{I}_{u}}$, for any *i* we have a distinguished triangle

$$\mathscr{R}_{i}^{\wedge} \mathbin{\widehat{\star}}^{0} \mathscr{F}_{1} \to \mathscr{R}_{i}^{\wedge} \mathbin{\widehat{\star}}^{0} \mathscr{F}_{2} \to \mathscr{R}_{i}^{\wedge} \mathbin{\widehat{\star}}^{0} \mathscr{F}_{3} \xrightarrow{[1]}$$

in $D^0_{I_{u},I_{u}}$, and then a long exact sequence

$$\cdots \to {}^{\mathbf{p}}\mathscr{H}^{n-1}(\mathscr{R}_{i}^{\wedge} \widehat{\star}^{0} \, \mathscr{F}_{3}) \to {}^{\mathbf{p}}\mathscr{H}^{n}(\mathscr{R}_{i}^{\wedge} \widehat{\star}^{0} \, \mathscr{F}_{1}) \to {}^{\mathbf{p}}\mathscr{H}^{n}(\mathscr{R}_{i}^{\wedge} \widehat{\star}^{0} \, \mathscr{F}_{2}) \to {}^{\mathbf{p}}\mathscr{H}^{n}(\mathscr{R}_{i}^{\wedge} \widehat{\star}^{0} \, \mathscr{F}_{3}) \to {}^{\mathbf{p}}\mathscr{H}^{n+1}(\mathscr{R}_{i}^{\wedge} \widehat{\star}^{0} \, \mathscr{F}_{1}) \to \cdots$$

in $\mathsf{P}^0_{\mathbf{I}_n,\mathbf{I}_n}$. Taking the formal inductive limit, we deduce a long exact sequence

$$\cdots \to {}^{\mathbf{p}}\mathscr{H}^{n-1}(\mathscr{R}^{\wedge} \widehat{\star}^{0} \mathscr{F}_{3}) \to {}^{\mathbf{p}}\mathscr{H}^{n}(\mathscr{R}^{\wedge} \widehat{\star}^{0} \mathscr{F}_{1}) \to {}^{\mathbf{p}}\mathscr{H}^{n}(\mathscr{R}^{\wedge} \widehat{\star}^{0} \mathscr{F}_{2}) \to {}^{\mathbf{p}}\mathscr{H}^{n}(\mathscr{R}^{\wedge} \widehat{\star}^{0} \mathscr{F}_{3}) \to {}^{\mathbf{p}}\mathscr{H}^{n+1}(\mathscr{R}^{\wedge} \widehat{\star}^{0} \mathscr{F}_{1}) \to \cdots$$

in Ind- $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$. This exact sequence shows that if (10.25) is true for two objects, then it follows for any extension between them. Since this statement is known for any object of the form $\pi_0^{\dagger}\mathscr{G}$ with \mathscr{G} in $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$, and since any object in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$ is a successive extension of objects of this form, this proves (10.25) for all \mathscr{F} .

Now that (10.25) is known, the same long exact sequence as above shows that the functor (10.16) transforms exact sequences in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ into exact sequences in $\mathrm{Ind}-\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, i.e. is exact.

10.8. **Definition of the functor.** We start with the following observation: consider a category A, a pro-object

$$X = \underset{i \in I}{``} \underbrace{\lim}_{i \in I} ``X_i,$$

in A, and an ind-object

$$Y = "\lim_{j \in J} "Y_j$$

in A. Then A embeds in the category Ind-A of ind-objects in A, and also in the category Pro-A of pro-objects in A. Using the induced functors on categories of pro-objects and ind-objects respectively, we can see X and Y either as objects in Pro-Ind-A, or as objects in Ind-Pro-A. The spaces of morphisms from X to Y in these two categories coincide: they both canonically identify with

$$\lim_{i \in I} \lim_{j \in J} \operatorname{Hom}_{\mathsf{A}}(X_i, Y_j) = \lim_{(i,j) \in I \times J} \operatorname{Hom}_{\mathsf{A}}(X_i, Y_j) = \lim_{j \in J} \lim_{i \in I} \operatorname{Hom}_{\mathsf{A}}(X_i, Y_j),$$

where the equalities follow from [KS, Proposition 2.1.7]. This space will simply be denoted $\text{Hom}_{A}(X, Y)$.

In the present setting we have the pro-object $\Xi_!^{\wedge}$ in D_{I_u,I_u} ; applying (the functor on pro-objects induced by) $\Pi^0_{I_u,I_u}$ we deduce a pro-object $\Pi^0_{I_u,I_u}(\Xi_!^{\wedge})$ in $\mathsf{D}^0_{I_u,I_u}$. (In other words, $\Pi^0_{I_u,I_u}(\Xi_!^{\wedge})$ is the image of the pro-object $\Pi^0_{U,U}(\Xi_!^{\wedge})$ considered in §8.2 under the functor on pro-objects induced by (8.1).) On the other hand, given \mathscr{F} in $\mathsf{P}^0_{I_u,I_u}$ we have the ind-object $\mathscr{R}^{\wedge} \, \mathscr{P} \, \mathscr{T} \, \mathscr{P} \, \mathscr{P}$ in $\mathsf{P}^0_{I_u,I_u}$. Now we have a fully faithful functor Ind- $\mathsf{P}^0_{I_u,I_u} \to \operatorname{Ind}-\mathsf{D}^0_{I_u,I_u}$, see [KS, Proposition 6.1.10]; $\mathscr{R}^{\wedge p} \, \mathscr{F} \, \mathscr{P} \, \mathscr{F}$ can therefore also be seen as an ind-object in $\mathsf{D}^0_{I_u,I_u}$. Using the notation above we can therefore consider the vector space

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}) := \mathrm{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}}}^{0},\mathrm{I}_{\mathrm{u}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}), \mathscr{R}^{\wedge p} \widehat{\star}^{0} \mathscr{F}\right).$$

For \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, monodromy endows $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F})$ with a canonical action of $\mathscr{O}(T_{\Bbbk}^{\vee} \times T_{\Bbbk}^{\vee})$, which by Lemma 6.7 factors through an action of $\mathscr{O}(\mathbf{D})$. (In view of Remark 10.6, the restriction of this action to $\mathscr{O}(T_{\Bbbk}^{\vee}/W_{\mathrm{f}})$ coincides with the action of

 $\mathscr{O}(\Sigma)$ induced by the action on \mathscr{R}^{\wedge} .) In this way, Φ_{I_u,I_u} can be seen as a functor

$$\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{Mod}(\mathscr{O}(\mathbf{D})).$$

Again for \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, using the right exactness of the functor $(-) \, {}^{\mathrm{p}} \widehat{\mathscr{F}}^0 \, \mathscr{F}$ (see Lemma 6.9), the morphism of Lemma 10.10 provides a canonical morphism

$$\mathscr{R}^{\wedge p} \widehat{\star}^{0} \mathscr{F} \to \left(\mathscr{R}^{\wedge p} \widehat{\star}^{0} \mathscr{F} \right) \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma})$$

in Ind- $\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. (Here the action of $\mathscr{O}(\Sigma)$ on $\mathscr{R}^{\wedge p} \widehat{\star}^{0} \mathscr{F}$ is induced by that on \mathscr{R}^{\wedge} , or equivalently by monodromy.)

Lemma 10.15. For any \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, there exists a canonical isomorphism

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F})\otimes_{\mathscr{O}(\Sigma)}\mathscr{O}(\mathbb{J}_{\Sigma})\xrightarrow{\sim}\mathrm{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\big(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),(\mathscr{R}^{\wedge}\stackrel{\mathrm{p}^{\ast}}{\star}^{0}\mathscr{F})\otimes_{\mathscr{O}(\Sigma)}\mathscr{O}(\mathbb{J}_{\Sigma})\big).$$

Proof. The object $\Xi_{!}^{\wedge}$ is perverse; by Proposition D.4 it can therefore be written as " \lim_{n} " \mathscr{A}_{n} for some projective system ($\mathscr{A}_{n} : n \geq 0$) of objects in ${}^{\mathrm{p}}\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\leq 0}$. On the other hand, $\mathscr{R}^{\wedge} {}^{\mathrm{p}} \widehat{\star}^{0} \mathscr{F}$ is an ind-object in $\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{0}$; it can therefore be written as " \lim_{i} " \mathscr{G}_{i} for some objects \mathscr{G}_{i} in $\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{0}$. We then have

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}) = \varinjlim_{n,i} \mathrm{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{A}_{n},\mathscr{G}_{i}) = \varinjlim_{n,i} \mathrm{Hom}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}({}^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{A}_{n}),\mathscr{G}_{i}).$$

If we write $\mathscr{O}(\mathbb{J}_{\Sigma}) = \varinjlim_{j} M_{j}$ for some finitely generated $\mathscr{O}(\Sigma)$ -modules M_{j} , then we similarly have

$$\operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}), (\mathscr{R}^{\wedge p}^{*} \mathscr{F}) \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma})\right) \\ = \lim_{n,i,j} \operatorname{Hom}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}({}^{\mathsf{p}} \mathscr{H}^{0}(\mathscr{A}_{n}), \mathscr{G}_{i} \otimes_{\mathscr{O}(\Sigma)} M_{j}),$$

where $\mathscr{O}(\Sigma)$ acts on \mathscr{G}_i via monodromy. By Lemma B.1, for any n, i, j we have a canonical morphism

(10.26)
$$\operatorname{Hom}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{A}_{n}),\mathscr{G}_{i})\otimes_{\mathscr{O}(\Sigma)}M_{j}\to \operatorname{Hom}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(^{\mathrm{p}}\mathscr{H}^{0}(\mathscr{A}_{n}),\mathscr{G}_{i}\otimes_{\mathscr{O}(\Sigma)}M_{j}),$$

which defines the desired morphism.

To prove that this morphism is an isomorphism, we observe that since $\mathscr{O}(\mathbb{J}_{\Sigma})$ is flat over $\mathscr{O}(\Sigma)$ (see Lemma 2.17), by Lazard's theorem (see e.g. [SP, Tag 058G]) the objects M_j can be chosen to be finite free $\mathscr{O}(\Sigma)$ -modules. Then (10.26) is an isomorphism for any n, i, j, which concludes the proof.

This lemma shows that the morphism of Lemma 10.10 induces, for any \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{D}},\mathrm{I}_{\mathrm{D}}}$, a canonical morphism

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}) \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma}),$$

which is easily seen to define a structure of $\mathscr{O}(\mathbb{J}_{\Sigma})$ -comodule on $\Phi_{I_u,I_u}(\mathscr{F})$. Combining these structures, we see that Φ_{I_u,I_u} defines a functor from $\mathsf{P}^0_{I_u,I_u}$ to the category of $\mathscr{O}(\mathbb{J}_{\mathbf{D}})$ -comodules.

We now explain how to construct, for \mathscr{F}, \mathscr{G} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, a bifunctorial morphism

(10.27)
$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{G}) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F} \stackrel{\mathrm{p}}{\star}^{0}_{\mathrm{I}_{\mathrm{u}}}\mathscr{G}).$$

First, the same construction as for (8.4) provides, for any \mathscr{F}, \mathscr{G} in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$, a canonical morphism

$$\begin{split} \operatorname{Hom}_{\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^\wedge_!),\mathscr{F})\otimes_{\mathscr{O}(\Sigma)}\operatorname{Hom}_{\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^\wedge_!),\mathscr{G})\\ \to \operatorname{Hom}_{\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^\wedge_!),\mathscr{F} \, {}^{\mathrm{p}}\!\star^0_{\mathrm{I}_{\mathrm{u}}}\,\mathscr{G}). \end{split}$$

We then deduce a similar morphism for ind-objects, which provides a canonical morphism

$$\Phi_{I_u,I_u}(\mathscr{F}) \circledast \Phi_{I_u,I_u}(\mathscr{G}) \to \operatorname{Hom}_{\mathsf{D}^0_{I_u,I_u}}(\Pi^0_{I_u,I_u}(\Xi^\wedge_!), (\mathscr{R}^\wedge \, {^{\mathrm{p}}}_{\star}^{\circ}^{\circ} \, \mathscr{F}) \, {^{\mathrm{p}}}_{\star}^0_{I_u} \, (\mathscr{R}^\wedge \, {^{\mathrm{p}}}_{\star}^{\circ}^{\circ} \, \mathscr{G})).$$

Composing this morphism with the morphism

$$\begin{split} \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),(\mathscr{R}^{\wedge}\stackrel{\mathrm{p}^{\wedge}}{\star}^{0}\mathscr{F})\stackrel{\mathrm{p}^{\bullet}}{\star}^{0}(\mathscr{R}^{\wedge}\stackrel{\mathrm{p}^{\wedge}}{\star}^{0}\mathscr{G})) \to \\ \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge}\stackrel{\mathrm{p}^{\wedge}}{\star}^{0}(\mathscr{F}\stackrel{\mathrm{p}^{\bullet}}{\star}^{0}\mathscr{G})) \end{split}$$

induced by (10.14), we deduce the whished-for morphism (10.27).

We will later see that (10.27) is an isomorphism for any \mathscr{F}, \mathscr{G} in $\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, but this will require proving first some other properties of $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

10.9. **Image of monodromy.** Recall the monodromy construction with respect to the loop rotation action, see Remark 5.3. This morphism provides, for any \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, a functorial automorphism $\mu^{\mathrm{rot}}_{\mathscr{F}}(x): \mathscr{F} \xrightarrow{\sim} \mathscr{F}$.

On the other hand, consider the category $\operatorname{Rep}^{\infty}(\mathbb{J}_{\mathbf{D}})$ of representations of the group scheme $\mathbb{J}_{\mathbf{D}}$, or in other words of $\mathscr{O}(\mathbb{J}_{\mathbf{D}})$ -comodules. As explained in Remark 2.16, the group scheme \mathbb{J}_{Σ} admits a canonical section, hence so does $\mathbb{J}_{\mathbf{D}}$. This implies that any M in $\operatorname{Rep}^{\infty}(\mathbb{J}_{\mathbf{D}})$ admits a "tautological" automorphism, defined as the composition

$$M \to M \otimes_{\mathscr{O}(\mathbf{D})} \mathscr{O}(\mathbb{J}_{\mathbf{D}}) \to M \otimes_{\mathscr{O}(\mathbf{D})} \mathscr{O}(\mathbf{D}) = M$$

where the first morphism is the coaction, and the second one is induced by restriction to the canonical section. Here again this automorphism is functorial (in the sense that it defines an automorphism of the identity functor).

Lemma 10.16. For any \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mu^{\mathrm{rot}}_{\mathscr{F}}(x)^{-1})$ is the tautological automorphism of $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F})$.

Proof. Since the loop rotation action is trivial on G/U, for any \mathscr{G} in $\mathsf{P}^0_{U,U}$, seen as an object in P^0_{U,U_*} , we have $\mu^{\mathrm{rot}}_{\mathscr{G}}(x) = \mathrm{id}$. On the other hand, by (6.9) we have

$$\mu^{\mathrm{rot}}_{\mathscr{R}^{\wedge} \mathsf{p}_{\widehat{\star}^{0}} \mathscr{F}}(x) = \mu^{\mathrm{rot}}_{\mathscr{R}^{\wedge}}(x) \, {}^{\mathrm{p}}_{\widehat{\star}^{0}} \, \mu^{\mathrm{rot}}_{\mathscr{F}}(x)$$

where $\mu_{\mathscr{R}^{\wedge}}^{\mathrm{rot}}$ is the automorphism induced by $\mu_{\mathscr{T}(\mathscr{O}(G_{\Bbbk}^{\vee}))}^{\mathrm{rot}}$. Since any morphism in $\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ commutes with monodromy, it follows that $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mu_{\mathscr{F}}^{\mathrm{rot}}(x)^{-1})$ coincides with the automorphism of $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F})$ induced by $\mu_{\mathscr{R}^{\wedge}}^{\mathrm{rot}}(x)$. By definition, this automorphism is obtained by passage to the quotient from $\mu_{\mathscr{T}(\mathscr{O}(G_{\Bbbk}^{\vee}))}^{\mathrm{rot}}(x)$, which by Theorem 7.8(6) coincides with $(\widehat{\mathfrak{m}}_{\mathscr{O}(G_{\Bbbk}^{\vee})})^{-1}$. By construction of this functor (see §10.3), the latter automorphism is the image under $\mathscr{\widehat{T}^{\mathsf{Coh}}$ of the inverse of the tautological automorphism of $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}$ as considered in §4.5.

On the other hand, by definition of the coaction on $\Phi_{I_u,I_u}(\mathscr{F})$, the tautological automorphism of this representation is induced by the automorphism of \mathscr{R}^{\wedge} given by the composition

$$\mathscr{R}^{\wedge} \xrightarrow{\operatorname{coact}_{\mathscr{R}^{\wedge}}} \mathscr{R}^{\wedge} \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\mathbb{J}_{\Sigma}) \to \mathscr{R}^{\wedge} \otimes_{\mathscr{O}(\Sigma)} \mathscr{O}(\Sigma) = \mathscr{R}^{\wedge}$$

where the second morphism is induced by restriction to the canonical section. This automorphism is obtained by passage to the quotient from the automorphism of $\widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee}))$ given by the composition

$$\begin{aligned} \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \to \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\Bbbk} \mathscr{O}(G_{\Bbbk}^{\vee}) &= \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee}) \\ &\to \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \otimes_{\mathscr{O}(G_{\Bbbk}^{\vee})} \mathscr{O}(G_{\Bbbk}^{\vee}) = \widehat{\mathscr{X}}(\mathscr{O}(G_{\Bbbk}^{\vee})) \end{aligned}$$

where the first morphism and the first identification are as in the construction of $\operatorname{coact}_{\mathscr{R}^{\wedge}}$, and the second morphism is induced by restriction to the diagonal. The latter morphism is the image under $\widehat{\mathscr{Z}^{\mathsf{Coh}}}$ of the automorphism of $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}$ given by the composition

$$\begin{split} \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}} & \to \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}} = \left(\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}\right) \otimes \mathscr{O}_{(G_{\Bbbk}^{\vee})} \mathscr{O}(G_{\Bbbk}^{\vee} \times G_{\Bbbk}^{\vee}) \\ & \to \left(\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}\right) \otimes \mathscr{O}_{(G_{\Bbbk}^{\vee})} \mathscr{O}(G_{\Bbbk}^{\vee}) = \mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}} \end{split}$$

where the first morphism is induced by the comultiplication in $\mathscr{O}(G_{\Bbbk}^{\vee})$ and the second one by restriction to the diagonal.

These remarks show that the claim will follow if we check that the two given automorphisms of $\mathscr{O}(G_{\Bbbk}^{\vee}) \otimes \mathscr{O}_{G_{\Bbbk}^{\vee}}$ coincide. This is an easy exercise of manipulation with the Hopf algebra $\mathscr{O}(G_{\Bbbk}^{\vee})$.

10.10. Exactness and compatibility with $\Phi_{I,I}$ and $\Phi_{U,U}$. Our goal in this subsection is to show that the functor Φ_{I_u,I_u} constructed in §10.8 factors through an exact functor

$$\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$$

which is moreover compatible with the functor $\Phi_{I,I}$ from §4.8 and the functor $\Phi_{U,U}$ from Theorem 8.3.

Lemma 10.17. For any \mathscr{F} in $\mathsf{P}^0_{I,I}$, we have a canonical isomorphism of $\mathscr{O}(\mathbb{J}_{\mathbf{D}})$ -comodules

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\pi_{0}^{\dagger}\mathscr{F}) \cong \Phi_{\mathrm{I},\mathrm{I}}(\mathscr{F})$$

(where the coaction on the right-hand side is provided by the functor (10.2)), and moreover

$$\operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge \mathrm{p}}\widehat{\star}^{0}(\pi^{\dagger}_{0}\mathscr{F})[n]\right)=0$$

if $n \neq 0$.

Proof. By Lemma 10.12 and (5.2), for \mathscr{F} in $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$ we have

(10.28)
$$\mathscr{R}^{\wedge p} \widehat{\star}^{0} (\pi_{0}^{\dagger} \mathscr{F}) \cong \pi_{0}^{\dagger} (\mathscr{R}^{0} \star_{\mathrm{I}}^{0} \mathscr{F}) \cong \pi^{\dagger, 0} \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}, 0} (\mathscr{R}^{0} \star_{\mathrm{I}}^{0} \mathscr{F}).$$

On the other hand, it is easily seen that the functor

$$\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \circ \pi_{\dagger} : \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

factors through a triangulated functor

$$\pi_{\dagger,0}:\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

which is left adjoint to $\pi^{\dagger,0}$. Moreover, since $\pi_{\dagger}(\Xi_{1}^{\wedge}) = \Xi_{1}$ we have

(10.29)
$$\pi_{\dagger,0} \circ \Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^\wedge_{!}) = \Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi_{!}).$$

(In particular, this pro-object in $\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ in fact belongs to $\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$.) We can now compute using these considerations: for $n \in \mathbb{Z}$ we have

$$\begin{split} \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}} \left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}), \mathscr{R}^{\wedge \ \mathrm{p}}^{\star 0} \pi^{\dagger}_{0} \mathscr{F}[n] \right) \\ & \cong \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}} \left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}), \pi^{\dagger,0} \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{R}^{0} \star^{0}_{1} \mathscr{F})[n] \right) \\ & \cong \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}} \left(\pi_{\dagger,0}^{0} \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}), \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{R}^{0} \star^{0}_{1} \mathscr{F})[n] \right) \\ & \cong \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}} \left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi_{!}), \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{R}^{0} \star^{0}_{1} \mathscr{F})[n] \right) \end{split}$$

(Here, the last space is simply a space of morphisms in the category $Ind-D^0_{I_{u},I}$.)

To proceed further, consider the "Iwahori–Whittaker" category of sheaves on Fl_G considered in [BRR, §7.1], and which will be denoted $D_{\mathcal{IW},I}$ here. As for the "finite" flag variety in §6.6, we have "averaging" functors

$$\mathsf{Av}_{\mathrm{I}_{\mathrm{u}},!}:\mathsf{D}_{\mathcal{I}\mathcal{W},\mathrm{I}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}},\qquad\mathsf{Av}_{\mathcal{I}\mathcal{W}}:\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\to\mathsf{D}_{\mathcal{I}\mathcal{W},\mathrm{I}}$$

such that $Av_{I_u,l}$ is left adjoint to Av_{IW} and both functors are t-exact; moreover, by construction we have

$$\Xi_{!} = \mathsf{Av}_{\mathrm{I}_{\mathrm{u}}, !} \circ \mathsf{Av}_{\mathcal{I}\mathcal{W}} \circ \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\delta)$$

If we denote by $\mathsf{P}_{\mathcal{IW},\mathrm{I}}$ the heart of the perverse t-structure on $\mathsf{D}_{\mathcal{IW},\mathrm{I}}$, then the simple objects in $\mathsf{P}_{\mathcal{IW},\mathrm{I}}$ are parametrized in a natural way by the subset of W consisting of elements w which have minimal length in their coset $W_{\mathrm{f}}w$. We can therefore consider the Serre and triangulated subcategories of $\mathsf{P}_{\mathcal{IW},\mathrm{I}}$ and $\mathsf{D}_{\mathcal{IW},\mathrm{I}}$ respectively generated by the simple objects labelled by elements of positive length, and then the corresponding Serre quotient $\mathsf{P}^0_{\mathcal{IW},\mathrm{I}}$ and Verdier quotient $\mathsf{D}^0_{\mathcal{IW},\mathrm{I}}$, and the quotient functor $\Pi^0_{\mathcal{IW},\mathrm{I}} : \mathsf{D}_{\mathcal{IW},\mathrm{I}} \to \mathsf{D}^0_{\mathcal{IW},\mathrm{I}}$. By Lemma A.2 there exists a unique t-structure on $\mathsf{D}^0_{\mathcal{IW},\mathrm{I}}$ such that $\Pi^0_{\mathcal{IW},\mathrm{I}}$ is t-exact; this t-structure is bounded, and its heart identifies with $\mathsf{P}^0_{\mathcal{IW},\mathrm{I}}$. The methods of [BGS, §§3.2–3.3] can be used to show that the realization functor

$$D^{\mathrm{b}}\mathsf{P}_{\mathcal{IW},\mathrm{I}} \to \mathsf{D}_{\mathcal{IW},\mathrm{I}}$$

is an equivalence of categories; combining this with Proposition A.3, we obtain an equivalence of triangulated categories

$$(10.30) D^{\mathrm{b}}\mathsf{P}^{0}_{\mathcal{IW},\mathrm{I}} \xrightarrow{\sim} \mathsf{D}^{0}_{\mathcal{IW},\mathrm{I}}$$

whose restriction to $\mathsf{P}^0_{\mathcal{IW},I}$ is the obvious embedding.

One can easily check that the functor $\Pi^0_{I_u,I} \circ Av_{I_u,!}$, resp. $\Pi^0_{\mathcal{IW},I} \circ Av_{\mathcal{IW}}$, factors uniquely through a t-exact triangulated functor

$$\mathsf{Av}^0_{\mathrm{I}_{\mathrm{u}},!}:\mathsf{D}^0_{\mathcal{I}\mathcal{W},\mathrm{I}}\to\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}},\quad\mathrm{resp.}\quad\mathsf{Av}^0_{\mathcal{I}\mathcal{W}}:\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\to\mathsf{D}^0_{\mathcal{I}\mathcal{W},\mathrm{I}};$$

and that $\mathsf{Av}^0_{\mathrm{I}_u,!}$ is left adjoint to $\mathsf{Av}^0_{\mathcal{IW}}.$ We then obtain that

$$\begin{split} \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}} & \left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi_{!}), \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{R}^{0} \star_{\mathrm{I}}^{0}\mathscr{F})[n] \right) \\ & \cong \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}} \left(\mathsf{Av}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{!}} \mathsf{Av}^{0}_{\mathscr{T}\mathcal{W}} \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0}), \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{R}^{0} \star_{\mathrm{I}}^{0}\mathscr{F})[n] \right) \\ & \cong \operatorname{Hom}_{\mathsf{D}^{0}_{\mathscr{T}\mathcal{W},\mathrm{I}}} \left(\mathsf{Av}^{0}_{\mathscr{T}\mathcal{W}} \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0}), \mathsf{Av}^{0}_{\mathscr{T}\mathcal{W}} \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{R}^{0} \star_{\mathrm{I}}^{0}\mathscr{F})[n] \right) \\ & \cong \operatorname{Hom}_{D^{\mathrm{b}}\mathsf{P}^{0}_{\mathscr{T}\mathcal{W},\mathrm{I}}} \left(\mathsf{Av}^{0}_{\mathscr{T}\mathcal{W}} \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0}), \mathsf{Av}^{0}_{\mathscr{T}\mathcal{W}} \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\mathscr{R}^{0} \star_{\mathrm{I}}^{0}\mathscr{F})[n] \right) \end{split}$$

where the last step uses (10.30). (Following our conventions, the last space means morphisms in the category $\text{Ind}-D^{b}\mathsf{P}_{\mathcal{IW},\mathbf{I}}^{0}$.)

It follows from [BRR, Corollary 9.2] and the "transitivity" of the Serre quotient that the functor

$$\mathsf{Av}^0_{\mathcal{I}\mathcal{W}} \circ \mathsf{For}^{\mathrm{I},0}_{\mathrm{I}_{\mathrm{u}}} : \mathsf{P}^0_{\mathrm{I},\mathrm{I}} \to \mathsf{P}^0_{\mathcal{I}\mathcal{W},\mathrm{I}}$$

is an equivalence of categories. We deduce an equivalence of categories

$$D^{\mathrm{b}}\mathsf{P}^{0}_{\mathrm{I},\mathrm{I}} \to D^{\mathrm{b}}\mathsf{P}^{0}_{\mathcal{IW},\mathrm{I}},$$

and finally an isomorphism

$$\operatorname{Hom}_{\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\big(\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^\wedge_!), \mathscr{R}^\wedge \stackrel{\mathrm{p}}{\star}^0(\pi_0^\dagger \mathscr{F})[n]\big) \cong \operatorname{Hom}_{D^\mathrm{b}\mathsf{P}^0_{\mathrm{I},\mathrm{I}}}(\delta^0, \mathscr{R}^0 \star^0_{\mathrm{I}} \mathscr{F}[n]),$$

where in the right-hand side we mean morphisms in $\text{Ind}-D^{b}\mathsf{P}_{I,I}^{0}$.

Now consider the case n = 0. Since the natural functor $\operatorname{Ind}-\mathsf{P}^0_{\mathrm{I},\mathrm{I}} \to \operatorname{Ind}-D^{\mathrm{b}}\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$ is fully faithful (see [KS, Proposition 6.1.10]), we can compute the morphism space above in $\operatorname{Ind}-\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$; by definition we recover $\Phi_{\mathrm{I},\mathrm{I}}(\mathscr{F})$, which proves the isomorphism of the lemma.

If $n \neq 0$, we use the monoidal equivalence

$$\Phi_{\mathrm{I},\mathrm{I}}:\mathsf{P}^{0}_{\mathrm{I},\mathrm{I}}\xrightarrow{\sim}\mathsf{Rep}(Z_{G^{\vee}}(\mathsf{u}))$$

that sends \mathscr{R}^0 to $\mathscr{O}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))$ (see §4.8) to obtain an isomorphism

 $\operatorname{Hom}_{\operatorname{Ind}\text{-}D^{\operatorname{b}}\mathsf{P}^{0}_{\operatorname{I},\operatorname{I}}}(\delta^{0},\mathscr{R}^{0}\star^{0}_{\operatorname{I}}\mathscr{F}[n])\cong$

 $\operatorname{Hom}_{\operatorname{Ind}\text{-}D^{\mathrm{b}}\mathsf{Rep}(Z_{G^{\vee}}(\mathsf{u}))}(\Bbbk, \mathscr{O}(Z_{G^{\vee}_{\Bbbk}}(\mathsf{u})) \otimes_{\Bbbk} \Phi_{\mathrm{I},\mathrm{I}}(\mathscr{F})[n]).$

If we write

$$\mathscr{O}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})) = "\varinjlim_{i \in I} "M_i$$

where I is filtrant and each M_i belongs to $\operatorname{\mathsf{Rep}}(Z_{G_k^{\vee}}(\mathsf{u}))$, then we have

$$\operatorname{Hom}_{\operatorname{Ind}\text{-}D^{\mathrm{b}}\mathsf{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))}(\Bbbk, \mathscr{O}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})) \otimes_{\Bbbk} \Phi_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}}(\mathscr{F})[n]) = \lim_{\substack{i \in I} \atop i \in I} \operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))}(\Bbbk, M_{i} \otimes_{\Bbbk} \Phi_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}}(\mathscr{F})[n]).$$

It is known that the natural functor $D^{\mathrm{b}}\mathsf{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})) \to D^{\mathrm{b}}\mathsf{Rep}^{\infty}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))$ is fully faithful. (This follows e.g. from the much more general results in [AB, Corollary 2.11 and its proof].) We deduce that for any $i \in I$ we have

$$\operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Rep}(Z_{G^{\vee}}(\mathsf{u}))}(\Bbbk, M_{i} \otimes_{\Bbbk} \Phi_{\mathrm{I},\mathrm{I}}(\mathscr{F})[n]) \cong \mathsf{H}^{n}(Z_{G^{\vee}_{\Bbbk}}(\mathsf{u}), M_{i} \otimes_{\Bbbk} \Phi_{\mathrm{I},\mathrm{I}}(\mathscr{F})),$$

and then using the fact that cohomology commutes with filtrant direct limits (see [Ja, Lemma I.4.17]) that

$$\begin{split} \operatorname{Hom}_{\operatorname{Ind}\text{-}D^{\mathrm{b}}\mathsf{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))}(\Bbbk, \mathscr{O}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})) \otimes_{\Bbbk} \Phi_{\mathrm{I},\mathrm{I}}(\mathscr{F})[n]) = \\ \mathsf{H}^{n}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}), \mathscr{O}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})) \otimes_{\Bbbk} \Phi_{\mathrm{I},\mathrm{I}}(\mathscr{F})). \end{split}$$

Finally, we use the fact that $\mathscr{O}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})) \otimes_{\Bbbk} \Phi_{\mathrm{I},\mathrm{I}}(\mathscr{F})$ is injective in $\mathsf{Rep}^{\infty}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))$ (see [Ja, §§I.3.9–I.3.10]) to conclude that this space vanishes. \Box

As a consequence of Lemma 10.17 we obtain the following properties.

Proposition 10.18. The functor Φ_{I_u,I_u} is exact, and takes values in $\mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$. Moreover, the left diagram in (10.3) commutes.

Proof. Recall that $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is a finite-length category, and that its simple objects are the objects $\pi^{\dagger}_{0}(\mathscr{F})$ with \mathscr{F} simple in $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$. Given a short exact sequence

$$\mathscr{F}_1 \hookrightarrow \mathscr{F}_2 \twoheadrightarrow \mathscr{F}_3$$

in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$, by Lemma 10.12 we have a short exact sequence

(10.31)
$$\mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}_{1} \hookrightarrow \mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}_{2} \twoheadrightarrow \mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}_{3}$$

in Ind- P_{I_u,I_u}^0 . By [KS, Proposition 8.6.6(1)], there exists a filtrant category I and an inductive system of short exact sequences

$$\mathscr{M}_i^1 \hookrightarrow \mathscr{M}_i^2 \twoheadrightarrow \mathscr{M}_i^3$$

in $\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ from which (10.31) is obtained by taking formal direct limits. Write also

$$\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^\wedge_!) = \underset{n}{``\varprojlim_n} "\mathscr{G}_n$$

for some objects \mathscr{G}_n in $\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. Then for any n and i we have an exact sequence

$$\begin{aligned} \operatorname{Hom}_{\mathsf{D}^{0}_{\mathsf{I}_{\mathrm{u}},\mathsf{I}_{\mathrm{u}}}}(\mathscr{G}_{n},\mathscr{M}^{1}_{i}[-1]) &\to \operatorname{Hom}_{\mathsf{D}^{0}_{\mathsf{I}_{\mathrm{u}},\mathsf{I}_{\mathrm{u}}}}(\mathscr{G}_{n},\mathscr{M}^{2}_{i}[-1]) \to \operatorname{Hom}_{\mathsf{D}^{0}_{\mathsf{I}_{\mathrm{u}},\mathsf{I}_{\mathrm{u}}}}(\mathscr{G}_{n},\mathscr{M}^{3}_{i}[-1]) \\ &\to \operatorname{Hom}_{\mathsf{D}^{0}_{\mathsf{I}_{\mathrm{u}},\mathsf{I}_{\mathrm{u}}}}(\mathscr{G}_{n},\mathscr{M}^{1}_{i}) \to \operatorname{Hom}_{\mathsf{D}^{0}_{\mathsf{I}_{\mathrm{u}},\mathsf{I}_{\mathrm{u}}}}(\mathscr{G}_{n},\mathscr{M}^{2}_{i}) \to \operatorname{Hom}_{\mathsf{D}^{0}_{\mathsf{I}_{\mathrm{u}},\mathsf{I}_{\mathrm{u}}}}(\mathscr{G}_{n},\mathscr{M}^{3}_{i}) \\ &\to \operatorname{Hom}_{\mathsf{D}^{0}_{\mathsf{I}_{\mathrm{u}},\mathsf{I}_{\mathrm{u}}}}(\mathscr{G}_{n},\mathscr{M}^{1}_{i}[1]) \to \operatorname{Hom}_{\mathsf{D}^{0}_{\mathsf{I}_{\mathrm{u}},\mathsf{I}_{\mathrm{u}}}}(\mathscr{G}_{n},\mathscr{M}^{2}_{i}[1]) \to \operatorname{Hom}_{\mathsf{D}^{0}_{\mathsf{I}_{\mathrm{u}},\mathsf{I}_{\mathrm{u}}}}(\mathscr{G}_{n},\mathscr{M}^{3}_{i}[1]). \end{aligned}$$

By exactness of filtrant direct limits we deduce an exact sequence

$$\begin{split} &\operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}_{1}[-1]\right) \to \operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}_{2}[-1]\right) \\ \to \operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}_{3}[-1]\right) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}_{1}) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}_{2}) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}_{3}) \\ \to \operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}_{2}[1]\right) \to \operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}_{2}[1]\right) \\ \to \operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}_{3}[1]\right). \end{split}$$

Using these exact sequences and Lemma 10.17 one proves by induction on the length that for any \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$ the module $\Phi_{\mathrm{I}_u,\mathrm{I}_u}(\mathscr{F})$ is finite-dimensional and annihilated by a power of \mathcal{J} , and that

$$\operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge} \stackrel{\mathrm{p}}{\star}^{0} \mathscr{F}[-1]\right) = 0 = \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}\left(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi^{\wedge}_{!}),\mathscr{R}^{\wedge} \stackrel{\mathrm{p}}{\star}^{0} \mathscr{F}[1]\right).$$

The first property shows that Φ_{I_u,I_u} takes values in $\mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$, and the second one implies (using again the exact sequence above) the exactness of Φ_{I_u,I_u} .

The commutativity of the left diagram in (10.3) has been established in Lemma 10.17.

We can now prove the compatibility of Φ_{I_u,I_u} with the equivalence $\Phi_{U,U}$ of Theorem 8.3.

Proposition 10.19. The right diagram in (10.3) commutes; in other words, for any \mathscr{F} in $\mathsf{P}^0_{U,U}$ there exists a canonical isomorphism of $\mathbb{J}_{\mathbf{D}}$ -modules

$$\Phi_{U,U}(\mathscr{F}) \xrightarrow{\sim} \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F})$$

where the left-hand side is endowed with the trivial structure as a representation.

Proof. From the definition of the functors we see that there exists a functorial morphism

$$\Phi_{U,U}(\mathscr{F}) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F})$$

induced by the unit morphism $\delta^{\wedge} \to \mathscr{R}^{\wedge}$ and the functor (8.1). Using Lemma 10.17 we see that this morphism is an isomorphism when \mathscr{F} is the unique simple object in $\mathsf{P}_{U,U}^0$, namely $\pi^{\dagger}(\mathsf{For}_{\mathrm{I}_u}^{\mathrm{I}}\mathscr{I}_e)$. Since both functors are exact, the five-lemma implies that this morphism is invertible for any \mathscr{F} in $\mathsf{P}_{U,U}^0$, which finishes the proof. \Box

10.11. Images of some truncated Wakimoto sheaves. Recall the representations of $\mathbb{J}_{\mathbf{D}}^{\wedge}$ introduced in §3.3, and the "truncation" functors introduced in §6.5 and §10.2. By construction, given $\mathscr{F} \in \mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\wedge}$, resp. $M \in \mathsf{Rep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$, these functors provide a projective system $(\mathsf{C}_{m}^{0}(\mathscr{W}_{w_{o}(\lambda)}^{\wedge}) : m \geq 1)$ of objects in $\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{0}$, resp. a projective system $(\mathsf{D}_{m}(M) : m \geq 1)$ of objects in $\mathsf{Rep}_{0}(\mathbb{J}_{\mathbf{D}})$. Our goal in this subsection is to prove the following claims.

Proposition 10.20. (1) For any $\lambda \in X^+_*(T)$, there exists an isomorphism of projective systems

$$\left(\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\mathscr{W}_{w_{\diamond}(\lambda)}^{\wedge})): m \geq 1\right) \cong \left(\mathsf{D}_{m}(\mathscr{M}_{\mathsf{t}(w_{\diamond}(\lambda))}^{\wedge}): m \geq 1\right).$$

(2) For any $\omega \in \Omega$, there exists an isomorphism of projective systems

$$\left(\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\Delta_{\omega}^{\wedge})):m\geq 1\right)\cong\left(\mathsf{D}_{m}(\mathscr{M}_{\omega}^{\wedge}):m\geq 1\right).$$

We will need some preliminaries.

Lemma 10.21. For any $V \in \mathsf{Rep}(G_{\Bbbk}^{\vee})$ and \mathscr{F} in $\mathsf{P}^{0}_{I_{\mathrm{u}},I_{\mathrm{u}}}$ we have a canonical isomorphism of $\mathbb{J}_{\mathbf{D}}$ -modules

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\left(\widehat{\mathscr{X}}(V) \stackrel{\mathrm{p}^{\ast}}{\star}^{0} \mathscr{F}\right) \cong V \otimes_{\Bbbk} \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}).$$

Proof. If we write $\mathscr{F} = \prod_{I_u,I_u}^0(\mathscr{G})$ with $\mathscr{G} \in \mathsf{P}_{I_u,I_u}$ then by exactness of the functor $\widehat{\mathscr{L}}(V) \mathbin{\widehat{\star}}(-)$ (see Theorem 7.8(5)) we have

$$\widehat{\mathscr{Z}}(V) \stackrel{\mathrm{p}}{\star}^{0} \mathscr{F} = \Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\widehat{\mathscr{Z}}(V) \stackrel{\sim}{\star} \mathscr{G}).$$

Using Lemma 10.7 we deduce an isomorphism

$$\mathscr{R}^{\wedge p}\widehat{\star}^{0}(\mathscr{\hat{Z}}(V)^{p}\widehat{\star}^{0}\mathscr{F}) \cong (\mathscr{R}^{\wedge p}\widehat{\star}^{0}\mathscr{F}) \otimes_{\Bbbk} V,$$

and then an isomorphism

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\widehat{\mathscr{Z}}(V) \stackrel{\mathrm{p}}{\star}^{0} \mathscr{F}) \cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}) \otimes_{\Bbbk} V,$$

as desired.

By exactness of the functor $\widehat{\mathscr{Z}}(V) \widehat{\star}(-)$ (see Theorem 7.8(5)), for any $m \ge 1$ we have

$$\mathsf{C}_m^0(\widehat{\mathscr{Z}}(V)) \cong \widehat{\mathscr{Z}}(V) \,\widehat{\star}^0 \, \mathsf{C}_m^0(\delta^{\wedge}).$$

In view of Lemma 10.21 and Proposition 10.19, we deduce a canonical isomorphism

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\widehat{\mathscr{Z}}(V))) \cong V \otimes_{\Bbbk} \Phi_{U,U}(\mathsf{C}_{m}^{0}(\delta^{\wedge})).$$

Now if we denote by $\Delta(T_{\Bbbk}^{\vee})^{(m)}$ the spectrum of $\mathscr{O}(T_{\Bbbk}^{\vee})/(\mathscr{J}^m \cdot \mathscr{O}(T_{\Bbbk}^{\vee}))$, embedded diagonally as a closed subscheme of **D**, it is clear that $\Phi_{U,U}(\mathsf{C}_m^0(\delta^{\wedge})) \cong \mathscr{O}_{\Delta(T_{\Bbbk}^{\vee})^{(m)}}$, so that we finally obtain an isomorphism

(10.32)
$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\mathscr{Z}(V))) \cong V \otimes_{\Bbbk} \mathscr{O}_{\Delta(T_{\Bbbk}^{\vee})^{(m)}}.$$

By Theorem 7.8(6) and Lemma 10.16, under this identification the automorphism of the left-hand side induced by \widehat{m}_V corresponds to the tautological automorphism of the $\mathbb{J}_{\mathbf{D}}$ -module $V \otimes_{\Bbbk} \mathscr{O}_{\Delta(T_k^{\vee})^{(m)}}$. By Remark 4.4, this automorphism can also be obtained from the tautological automorphism of $V \otimes \mathscr{O}_{G_k^{\vee}} \in \mathsf{Coh}^{G_k^{\vee}}(G_k^{\vee})$ by pullback under the composition

$$\Delta(T^{\vee}_{\Bbbk})^{(m)} \hookrightarrow \mathbf{D} \hookrightarrow \mathsf{St} \to G^{\vee}_{\Bbbk}.$$

(Here, the second embedding is provided by the identification of **D** with $\Sigma_{\mathbf{D}}$.)

Proof of Proposition 10.20. (1) We fix $\lambda \in X^+_*(T)$, and consider $V \in \operatorname{Rep}(G^{\vee}_{\Bbbk})$ which has highest weight λ (in the sense of §2.5). Recall from Theorem 7.8(7) that the perverse sheaf $\widehat{\mathscr{Z}}(V)$ admits a Wakimoto filtration whose subquotients have as labels the weights of V. In particular there exists an embedding $\mathscr{W}^{\wedge}_{w_{\circ}(\lambda)} \hookrightarrow \widehat{\mathscr{Z}}(V)$ whose cokernel admits a Wakimoto filtration. By Proposition 9.4, the induced morphism

(10.33)
$$\mathsf{C}^{0}_{m}(\mathscr{W}^{\wedge}_{w_{\circ}(\lambda)}) \to \mathsf{C}^{0}_{m}(\widehat{\mathscr{Z}}(V))$$

is injective for any $m \geq 1$. By Lemma 7.2(6) and Theorem 7.8(6), the endomorphism $f := \widehat{\mathfrak{m}}_V - \mu_{\widehat{\mathscr{F}}(V)}(e^{w_{\circ}(\lambda)} \otimes 1)$ of $\widehat{\mathscr{F}}(V)$ vanishes on the image of $\mathscr{W}^{\wedge}_{w_{\circ}(\lambda)}$. We deduce that, for any $m \geq 1$, $C^0_m(f)$ vanishes on the image of $C^0_m(\mathscr{W}^{\wedge}_{w_{\circ}(\lambda)})$.

For any $m \geq 1$ we can consider the finitely generated $\mathscr{O}(\mathbf{D}^{(m)})$ -module

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\mathscr{W}^{\wedge}_{w_{\circ}(\lambda)})),$$

and, by exactness of Φ_{I_u,I_u} , for any $m' \ge m$ we have

$$\mathscr{O}(\mathbf{D}^{(m)}) \otimes_{\mathscr{O}(\mathbf{D})} \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m'}(\mathscr{W}^{\wedge}_{w_{\diamond}(\lambda)})) \cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\mathscr{W}^{\wedge}_{w_{\diamond}(\lambda)})).$$

By [EGA1, Proposition 7.2.9], it follows that

$$K_{\lambda} := \varprojlim_{m} \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\mathscr{W}_{w_{\circ}(\lambda)}^{\wedge}))$$

is a finitely generated $\mathscr{O}(\mathbf{D}^{\wedge})$ -module such that

(10.34)
$$\mathscr{O}(\mathbf{D}^{(m)}) \otimes_{\mathscr{O}(\mathbf{D}^{\wedge})} K_{\lambda} \cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\mathscr{W}_{w_{\mathrm{o}}(\lambda)}^{\wedge}))$$

for any $m \ge 1$. From the embeddings (10.33), and by exactness of Φ_{I_u,I_u} , we obtain an embedding

$$K_{\lambda} \hookrightarrow \varprojlim_{m} \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\mathscr{Z}(V))),$$

i.e., using (10.32), an embedding

$$K_{\lambda} \hookrightarrow V \otimes \mathscr{O}(\Delta(T_{\Bbbk}^{\vee})^{\wedge})$$

where $\Delta(T_{\Bbbk}^{\vee})^{\wedge}$ is the spectrum of the completion of $\mathscr{O}(T_{\Bbbk}^{\vee})$ with respect to $\mathcal{J} \cdot \mathscr{O}(T_{\Bbbk}^{\vee})$, embedded diagonally in \mathbf{D}^{\wedge} .

Now, recall from Lemma 2.11 that $V \otimes \mathcal{O}_{\Delta \text{Groth}}$ has a canonical endomorphism whose kernel identifies with $\mathcal{O}_{\Delta \text{Groth}}(w_{\circ}(\lambda))$. Restricting to $\operatorname{St}_{\operatorname{reg}}$, using the equivalence of Proposition 2.20 and Lemma 2.21, and then completing, we deduce that $V \otimes \mathcal{O}(\Delta(T_{\Bbbk}^{\vee})^{\wedge})$ has a canonical endomorphism whose kernel is isomorphic to $\mathscr{M}_{w_{\circ}(\lambda)}^{\wedge}$. By construction and the comments just before the proof, this endomorphism vanishes on the image of K_{λ} , which provides an embedding

$$K_{\lambda} \hookrightarrow \mathscr{M}^{\wedge}_{w_{\alpha}(\lambda)}.$$

We will prove that this embedding is also surjective, hence an isomorphism, which will conclude the proof in view of (10.34).

By [EGA1, Corollaire 7.1.14], to prove this surjectivity it suffices to prove that the induced morphism

$$K_{\lambda}/(\mathcal{J}\cdot K_{\lambda}) \to \mathscr{M}^{\wedge}_{w_{\alpha}(\lambda)}/(\mathcal{J}\cdot \mathscr{M}^{\wedge}_{w_{\alpha}(\lambda)})$$

is surjective. Now, by construction this morphism is injective, so that to conclude it suffices to prove that

$$\dim(K_{\lambda}/(\mathcal{J}\cdot K_{\lambda})) = \dim(\mathscr{M}_{w_{\circ}(\lambda)}^{\wedge}/(\mathcal{J}\cdot \mathscr{M}_{w_{\circ}(\lambda)}^{\wedge})).$$

It is clear that the right-hand side equals $\dim(\mathscr{O}(T^{\vee}_{\Bbbk})/(\mathcal{J} \cdot \mathscr{O}(T^{\vee}_{\Bbbk})))$. For the left-hand side, we remark that

$$K_{\lambda}/(\mathcal{J}\cdot K_{\lambda}) = \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{1}^{0}(\mathscr{W}_{w_{\circ}(\lambda)}^{\wedge})).$$

From the proof of Lemma 9.1 one sees that the object $C_1^0(\mathscr{W}^{\wedge}_{w_{\diamond}(\lambda)})$ is an extension of $\dim(\mathscr{O}(T^{\vee}_{\Bbbk})/(\mathcal{J}\cdot\mathscr{O}(T^{\vee}_{\Bbbk})))$ many copies of the simple object $\Pi^0_{\mathrm{I},\mathrm{I}}(\Delta^{\mathrm{I}}_{\mathrm{t}(w_{\diamond}(\lambda))})$. (The fact that this object is simple follows from [BRR, Lemma 4.5].) We deduce that

$$\lim(K_{\lambda}/(\mathcal{J}\cdot K_{\lambda})) = \dim(\mathscr{O}(T^{\vee}_{\Bbbk})/(\mathcal{J}\cdot \mathscr{O}(T^{\vee}_{\Bbbk}))),$$

as desired.

(2) Since $\widetilde{\mathrm{Fl}}_{G,\omega}$ is closed the natural morphism $\Delta_{\omega}^{\wedge} \to \nabla_{\omega}^{\wedge}$ is an isomorphism; in the statement one can therefore replace Δ_{ω}^{\wedge} by ∇_{ω}^{\wedge} . Let us write $\omega = wt(\lambda)$ with $w \in W_{\mathrm{f}}$ and $\lambda \in X_*(T)$; then we have $\omega t(-\lambda) = w$ with $\ell(\omega t(\lambda)) = \ell(\omega) + \ell(t(-\lambda))$, hence by Lemma 6.3(1)–(2) we have

$$\nabla_{\omega}^{\wedge} \cong \nabla_{w}^{\wedge} \,\widehat{\star} \, \Delta_{\mathsf{t}(\lambda)}^{\wedge}.$$

After fixing such an isomorphism, using Proposition 9.6 we obtain for any $m \geq 1$ an isomorphism

$$\mathsf{C}_m^0(\nabla_{\omega}^{\wedge}) \cong \mathsf{C}_m^0(\nabla_w^{\wedge}) \, {}^{\mathrm{p}} \!\!\star_{\mathrm{I}_{\mathrm{u}}}^0 \, \mathsf{C}_m^0(\Delta_{\mathsf{t}(\lambda)}^{\wedge}).$$

Applying (10.27), one deduces a canonical morphism

$$\Phi_{\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\nabla_{w}^{\wedge})) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\Delta_{\mathsf{t}(\lambda)}^{\wedge})) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\nabla_{\omega}^{\wedge}).$$

From (3.1) one sees that λ is antidominant; using (1), we deduce isomorphisms

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\Delta^{\wedge}_{\mathsf{t}(\lambda)})) \cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\mathscr{W}^{\wedge}_{\lambda})) \cong \mathsf{D}_{m}(\mathscr{M}^{\wedge}_{\mathsf{t}(\lambda)}).$$

On the other hand, using Lemma 8.6 and Proposition 10.19 one obtains isomorphisms

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\Delta_{w}^{\wedge})) \cong \mathsf{D}_{m}(\mathscr{M}_{w}^{\wedge})$$

for any $m \ge 1$. Using Lemma 10.2 and (3.6), we deduce that our construction provides morphisms

$$\mathsf{D}_m(\mathscr{M}^\wedge_\omega) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^0_m(\Delta^\wedge_\omega))$$

defining a morphism of projective systems, and to conclude it suffices to prove that these morphisms are isomorphisms. Computing as in (1) one sees that the two modules involved have the same dimension, so that it suffices to prove that these morphisms are surjective. Now $\mathscr{O}(\mathbf{D}^{(m)})$ is a local ring, with maximal ideal the image of the ideal \mathcal{I} considered in §8.3. By Nakayama's lemma, to prove surjectivity it therefore suffices to prove that the induced morphism

$$\mathsf{D}_m(\mathscr{M}_\omega^\wedge)/\mathcal{I} \cdot \mathsf{D}_m(\mathscr{M}_\omega^\wedge) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_m^0(\Delta_\omega^\wedge))/\mathcal{I} \cdot (\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_m^0(\Delta_\omega^\wedge)))$$

is an isomorphism. Here by exactness of Φ_{I_u,I_u} the right-hand side identifies with $\Phi_{I,I}(\Pi^0_{I,I}(\pi^{\dagger}\Delta^I_{\omega})))$, and the morphism is an isomorphism by monoidality of $\Phi_{I,I}$. \Box

10.12. Fully faithfulness. We will now prove that the functor Φ_{I_u,I_u} is fully faithful. The proof will rely on the following easy lemma.

Lemma 10.22. Let A, A' be abelian categories, and let $F : A \to A'$ be an exact functor. Assume that every object in A has finite length, and that for any simple objects M, M' in A:

• the morphism

$$\operatorname{Hom}_{\mathsf{A}}(M, M') \to \operatorname{Hom}_{\mathsf{A}'}(F(M), F(M'))$$

induced by F is an isomorphism;

• the morphism

$$\operatorname{Ext}^{1}_{\mathsf{A}}(M, M') \to \operatorname{Ext}^{1}_{\mathsf{A}'}(F(M), F(M'))$$

induced by F is injective.

Then F is fully faithful.

Proof. One proves, by induction on the sum of the lengths of the objects involved and using the four- and five-lemma, that for any objects $M, M' \in A$ the morphism

$$\operatorname{Hom}_{\mathsf{A}}(M, M') \to \operatorname{Hom}_{\mathsf{A}'}(F(M), F(M')),$$

resp.

$$\operatorname{Ext}^{1}_{\mathsf{A}}(M, M') \to \operatorname{Ext}^{1}_{\mathsf{A}'}(F(M), F(M')),$$

is an isomorphism, resp. is injective, which implies the claim.

In order to use Lemma 10.22 we will need to describe some groups of extensions in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u},$ which we will relate to some groups of extensions in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}}$. First, the forgetful functor

$$\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{\mathrm{u}}}:\mathsf{D}_{\mathrm{I},\mathrm{I}}\to\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

induces a fully faithful functor $\mathsf{P}^0_{I,I}\to\mathsf{P}^0_{I_u,I};$ we deduce a canonical injective morphism

 $(10.35) \qquad \qquad \mathrm{Ext}^{1}_{\mathsf{P}^{0}_{\mathrm{I},\mathrm{I}}}(\delta^{0},\delta^{0}) \to \mathrm{Ext}^{1}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\mathsf{For}^{\mathrm{I},0}_{\mathrm{I}_{\mathrm{u}}}(\delta^{0}),\mathsf{For}^{\mathrm{I},0}_{\mathrm{I}_{\mathrm{u}}}(\delta^{0})).$

On the other hand, we also have the similar quotient category $\mathsf{P}_{U,B}^0$ constructed out of the category $\mathsf{P}_{U,B}$ considered in the proof of Lemma 6.10. The closed embedding $G/B \to \mathrm{Fl}_G$ induces a fully faithful functor $\mathsf{P}_{U,B}^0 \to \mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}^0$, and $\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^0)$ is the image under this functor of a canonical object of $\mathsf{P}_{U,B}^0$ denoted in the same way. We deduce another canonical injective morphism

$$(10.36) \qquad \operatorname{Ext}_{\mathsf{P}_{U,B}^{0}}^{1}(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0}),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0})) \to \operatorname{Ext}_{\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}^{1}(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0}),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0})).$$

Lemma 10.23. The morphisms (10.35) and (10.36) induce an isomorphism of \Bbbk -vector spaces

$$\operatorname{Ext}_{\mathsf{P}^{0}_{\mathrm{I},\mathrm{I}}}^{1}(\delta^{0},\delta^{0}) \oplus \operatorname{Ext}_{\mathsf{P}^{0}_{U,B}}^{1}(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0}),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0})) \xrightarrow{\sim} \operatorname{Ext}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}^{1}(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0}),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0})).$$

Proof. Recall from the proof of Lemma 10.17 the exact functor

$$\mathsf{Av}^0_{\mathcal{IW}}:\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} o\mathsf{P}^0_{\mathcal{IW},\mathrm{I}}$$

such that $Av_{\mathcal{IW}}^0 \circ For_{I_u}^{I,0}$ induces an equivalence $\mathsf{P}_{I,I}^0 \xrightarrow{\sim} \mathsf{P}_{\mathcal{IW},I}^0$. The functor $Av_{\mathcal{IW}}^0$ induces a morphism

$$\begin{array}{ll} (10.37) & \operatorname{Ext}_{\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}^1(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^0),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^0)) \to \\ & & \\ & \operatorname{Ext}_{\mathsf{P}^0_{\mathcal{I}_{\mathcal{W}},\mathrm{I}}}^1(\mathsf{Av}_{\mathcal{I}\mathcal{W}}^0(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^0)),\mathsf{Av}_{\mathcal{I}\mathcal{W}}^0(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^0))) \end{array}$$

whose composition with (10.35) is an isomorphism; to prove the lemma it therefore suffices to prove that its kernel is the image of (10.36).

Now, recall also (from the same proof) the functor

$$\mathsf{Av}^0_{\mathrm{I}_{\mathrm{u}},!}:\mathsf{D}^0_{\mathcal{IW},\mathrm{I}} o\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

which is left adjoint to $\mathsf{Av}^0_{\mathcal{IW}}:\mathsf{D}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\to\mathsf{D}^0_{\mathcal{IW},\mathrm{I}},$ and which satisfies

$$\mathsf{Av}^0_{I_u,!} \circ \mathsf{Av}^0_{\mathcal{IW}} \circ \mathsf{For}^{I,0}_{I_u}(\delta^0) = \Pi^0_{I_u,I}(\Xi_!).$$

By adjunction and standard properties of t-structures we have

$$\begin{split} \mathrm{Ext}^{1}_{\mathsf{P}^{0}_{\mathcal{I}\mathcal{W},\mathrm{I}}}(\mathsf{Av}^{0}_{\mathcal{I}\mathcal{W}}(\mathsf{For}^{\mathrm{I},0}_{\mathrm{I}_{\mathrm{u}}}(\delta^{0})), \mathsf{Av}^{0}_{\mathcal{I}\mathcal{W}}(\mathsf{For}^{\mathrm{I},0}_{\mathrm{I}_{\mathrm{u}}}(\delta^{0}))) = \\ \mathrm{Hom}_{\mathsf{D}^{0}_{\mathcal{I}\mathcal{W},\mathrm{I}}}(\mathsf{Av}^{0}_{\mathcal{I}\mathcal{W}}(\mathsf{For}^{\mathrm{I},0}_{\mathrm{I}_{\mathrm{u}}}(\delta^{0})), \mathsf{Av}^{0}_{\mathcal{I}\mathcal{W}}(\mathsf{For}^{\mathrm{I},0}_{\mathrm{I}_{\mathrm{u}}}(\delta^{0}))[1]) \cong \\ \mathrm{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi_{!}), \mathsf{For}^{\mathrm{I},0}_{\mathrm{I}_{\mathrm{u}}}(\delta^{0})[1]), \end{split}$$

and the morphism (10.37) identifies with the morphism

$$\operatorname{Hom}_{\mathsf{D}^0_{I_{\mathrm{u}},\mathrm{I}}}(\mathsf{For}^{\mathrm{I},0}_{I_{\mathrm{u}}}(\delta^0),\mathsf{For}^{\mathrm{I},0}_{I_{\mathrm{u}}}(\delta^0)[1]) \to \operatorname{Hom}_{\mathsf{D}^0_{I_{\mathrm{u}},\mathrm{I}}}(\Pi^0_{I_{\mathrm{u}},\mathrm{I}}(\Xi_!),\mathsf{For}^{\mathrm{I},0}_{I_{\mathrm{u}}}(\delta^0)[1])$$

induced by the natural surjection $\Pi^0_{I_u,I}(\Xi_!) \twoheadrightarrow \mathsf{For}^{I,0}_{I_u}(\delta^0)$. If we denote by \mathscr{K} the kernel of this morphism, we have a long exact sequence

$$\begin{array}{ll} (10.38) & 0 \to \operatorname{Hom}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0}),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0})) \to \operatorname{Hom}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi_{!}),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0})) \\ & \xrightarrow{f} \operatorname{Hom}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\mathscr{K},\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0})) \to \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0}),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^{0})) \\ & \to \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi_{!}),\delta^{0}) \to \cdots, \end{array}$$

which identifies the kernel of (10.37) with $\operatorname{coker}(f)$. But since all the objects involved belong to the full subcategory $\mathsf{P}_{U,B}^0$, we also have a similar long exact sequence

$$\begin{split} 0 &\to \operatorname{Hom}_{\mathsf{P}_{U,B}^{0}}(\mathsf{For}_{I_{u}}^{I,0}(\delta^{0}),\mathsf{For}_{I_{u}}^{I,0}(\delta^{0})) \to \operatorname{Hom}_{\mathsf{P}_{U,B}^{0}}(\Pi_{I_{u},I}^{0}(\Xi_{!}),\mathsf{For}_{I_{u}}^{I,0}(\delta^{0})) \\ \xrightarrow{g} \operatorname{Hom}_{\mathsf{P}_{U,B}^{0}}(\mathscr{K},\mathsf{For}_{I_{u}}^{I,0}(\delta^{0})) \to \operatorname{Ext}_{\mathsf{P}_{U,B}^{0}}^{1}(\mathsf{For}_{I_{u}}^{I,0}(\delta^{0}),\mathsf{For}_{I_{u}}^{I,0}(\delta^{0})) \\ &\to \operatorname{Ext}_{\mathsf{P}_{U,B}^{0}}^{1}(\Pi_{I_{u},I}^{0}(\Xi_{!}),\mathsf{For}_{I_{u}}^{I,0}(\delta^{0})) \to \cdots \end{split}$$

Since $\Xi_!$ is the projective cover of $\mathsf{For}_{\mathrm{I}_u}^{\mathrm{I}}(\delta)$ in $\mathsf{P}_{U,B}$ (see the proof of Lemma 6.10), $\Pi^0_{\mathrm{I}_u,\mathrm{I}}(\Xi_!)$ is the projective cover of $\mathsf{For}_{\mathrm{I}_u}^{\mathrm{I},0}(\delta^0)$ in $\mathsf{P}_{U,B}^0$, which implies that

$$\mathrm{Ext}^{1}_{\mathsf{P}^{0}_{U,B}}(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi_{!}),\mathsf{For}^{\mathrm{I},0}_{\mathrm{I}_{\mathrm{u}}}(\delta^{0}))=0,$$

and therefore allows to identify $\operatorname{Ext}_{\mathsf{P}_{U,B}^0}^1(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^0), \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^0))$ with $\operatorname{coker}(g)$. By fully faithfulness of the functor $\mathsf{P}_{U,B}^0 \to \mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}^0$ the domains and codomains of f and g identify, in a way compatible with these morphisms, so that their cokernels are canonically isomorphic.

Gathering these identifications we obtain an identification of the kernel of (10.37) with $\operatorname{Ext}_{\mathsf{P}_{U,B}^0}^1(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^0), \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}(\delta^0))$; it is clear by construction that this identification is induced by the morphism (10.36), which finishes the proof.

The other ingredient we will need is a way to "pass from δ^0 to δ^0_{ω} ," which will be provided by the following lemma. (See §4.6 for the definition of the object δ^0_{ω} .) Here, given a simple object \mathscr{F} in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$ we will denote by $\langle \mathscr{F} \rangle_{\mathrm{Serre}}$ the Serre subcategory generated by \mathscr{F} , i.e. the full subcategory whose objects are those all of whose composition factors are isomorphic to \mathscr{F} .

Lemma 10.24. For any $\omega \in \Omega$, the equivalence

$$\Delta_{\omega}^{\wedge} \widehat{\star}^{0} (-) : \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{0} \xrightarrow{\sim} \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{0}$$

restricts to an equivalence $\langle \pi_0^{\dagger} \delta^0 \rangle_{\text{Serre}} \xrightarrow{\sim} \langle \pi_0^{\dagger} \delta_{\omega}^0 \rangle_{\text{Serre}}$. Moreover the following diagram commutes:

$$\begin{array}{c} \langle \pi_0^{\dagger} \delta^0 \rangle_{\operatorname{Serre}} & \xrightarrow{\Delta_{\omega}^{\wedge} \widehat{\pi}^0(-)} \\ & & \\ \Phi_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}} \\ & & \\ \operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}}) & \xrightarrow{\mathscr{M}_{\omega}^{\wedge} \circledast(-)} \\ \end{array} \\ & & \\ \end{array} \\ \times \begin{array}{c} \langle \pi_0^{\dagger} \delta_{\omega}^0 \rangle_{\operatorname{Serre}} \\ & & \\ & & \\ \varphi_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}} \\ & & \\ & & \\ \end{array} \\ & & \\ \end{array}$$

Proof. Using (6.5)–(6.6) we see that

$$\Delta^{\wedge}_{\omega} \,\widehat{\star}^0 \,\pi^{\dagger}_0 \delta^0 \cong \pi^{\dagger}_0 \delta^0_{\omega},$$

which implies the first claim. In order to prove the second claim, we consider some object \mathscr{F} in $\langle \pi_0^{\dagger} \delta^0 \rangle_{\text{Serre}}$, and choose $m \geq 1$ such that \mathcal{J}^m acts trivially on \mathscr{F} . Then using Lemma 6.9 we obtain a canonical isomorphism

$$\Delta^{\wedge}_{\omega} \,\widehat{\star}^0 \,\mathscr{F} \cong \mathsf{C}^0_m(\Delta^{\wedge}_{\omega}) \,{}^{\mathrm{p}} \!\!\star^0_{\mathrm{I}_{\mathrm{u}}} \,\mathscr{F}$$

Now (10.27) provides a functorial morphism

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\Delta^{\wedge}_{\omega})) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}) \to \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\Delta^{\wedge}_{\omega}) \overset{\mathrm{p}}{\star}^{0}_{\mathrm{I}_{\mathrm{u}}}\mathscr{F}).$$

Using the identification above and Proposition 10.20(2), this morphism can be interpreted as a functorial morphism

$$\mathscr{M}^{\wedge}_{\omega} \circledast \Phi_{I_{u},I_{u}}(\mathscr{F}) \to \Phi_{I_{u},I_{u}}(\Delta^{\wedge}_{\omega} \,\widehat{\star}^{0} \, \mathscr{F}),$$

which does not depend on the choice of m. When $\mathscr{F} = \pi_0^{\dagger} \delta^0$, this morphism is an isomorphism by Proposition 10.18 and monoidality of $\Phi_{I,I}$. By the five lemma it is then an isomorphism for any \mathscr{F} , which concludes the proof.

We can finally prove the desired property.

Proposition 10.25. The functor Φ_{I_u,I_u} is fully faithful.

Proof. Recall that the simple objects in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$ are the images under π^\dagger_0 of the simple objects in $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$. In view of Lemma 10.22, to prove the proposition it therefore suffices to check that for any simple objects \mathscr{F}, \mathscr{G} in $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$ the functor $\Phi_{\mathrm{I}_u,\mathrm{I}_u}$ induces an isomorphism

$$\operatorname{Hom}_{\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\pi_0^{\dagger}\mathscr{F},\pi_0^{\dagger}\mathscr{G}) \xrightarrow{\sim} \operatorname{Hom}_{\mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})}(\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\pi_0^{\dagger}\mathscr{F}),\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\pi_0^{\dagger}\mathscr{G}))$$

and an injection

$$\operatorname{Ext}^{1}_{\mathsf{P}^{0}_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}}}(\pi_{0}^{\dagger}\mathscr{F},\pi_{0}^{\dagger}\mathscr{G}) \hookrightarrow \operatorname{Ext}^{1}_{\mathsf{Rep}_{0}(\mathbb{J}_{\mathbf{D}})}(\Phi_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}}(\pi_{0}^{\dagger}\mathscr{F}),\Phi_{\operatorname{I}_{\mathrm{u}},\operatorname{I}_{\mathrm{u}}}(\pi_{0}^{\dagger}\mathscr{G})).$$

For the Hom-spaces, this simply follows from the commutativity of the left diagram in (10.3) (proved in Proposition 10.18) since $\Phi_{I,I}$ is known to be an equivalence, hence in particular fully faithful.

To prove the claim about the Ext¹ spaces, recall that nonisomorphic simple objects in $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$ are supported on distinct connected components of Fl_G ; hence if $\mathscr{F} \ncong \mathscr{G}$ then $\mathrm{Ext}^1_{\mathsf{P}^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\pi_0^{\dagger}\mathscr{F},\pi_0^{\dagger}\mathscr{G}) = 0$, and there is nothing to prove. Otherwise we can assume that $\mathscr{F} = \mathscr{G} = \delta^0_{\omega}$ for some $\omega \in \Omega$. In fact, Lemma 10.24 reduces this case to the case when $\mathscr{F} = \mathscr{G} = \delta^0$. (Indeed the horizontal arrows in the diagram of Lemma 10.24 are equivalences, hence induce isomorphisms on Ext¹ spaces.) In this case we have $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\pi_0^{\dagger}\delta^0) = \Bbbk$, seen as the skyscraper sheaf at $\{(e, e)\}$, endowed with the trivial structure as a representation.

The same considerations as for (10.35) and (10.36) provide canonical embeddings

$$\begin{split} & \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{I,I}}(\delta^{0},\delta^{0}) \to \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{I_{\mathrm{u}},I_{\mathrm{u}}}}(\pi^{\dagger}_{0}\delta^{0},\pi^{\dagger}_{0}\delta^{0}), \\ & \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{U,U}}(\pi^{\dagger}_{0}\delta^{0},\pi^{\dagger}_{0}\delta^{0}) \to \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{I_{\mathrm{u}},I_{\mathrm{u}}}}(\pi^{\dagger}_{0}\delta^{0},\pi^{\dagger}_{0}\delta^{0}). \end{split}$$

We claim that these morphisms induce an isomorphism

$$\operatorname{Ext}^{1}_{\mathsf{P}^{0}_{\mathrm{I},\mathrm{I}}}(\delta^{0},\delta^{0}) \oplus \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{U,U}}(\pi^{\dagger}_{0}\delta^{0},\pi^{\dagger}_{0}\delta^{0}) \xrightarrow{\sim} \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\pi^{\dagger}_{0}\delta^{0},\pi^{\dagger}_{0}\delta^{0}).$$

Indeed, using the adjunction $(\pi_{\dagger,0}, \pi^{\dagger,0})$ (see the proof of Lemma 10.17) we obtain an isomorphism

$$\operatorname{Ext}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}^{1}(\pi_{0}^{\dagger}\delta^{0},\pi_{0}^{\dagger}\delta^{0}) \cong \operatorname{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\pi_{\dagger,0}\pi^{\dagger,0}\delta^{0},\delta^{0}[1]) \cong \operatorname{Ext}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}^{1}(\delta^{0},\delta^{0}) \oplus \mathsf{H}_{c}^{2r-1}(T;\Bbbk)^{*}$$

where $r = \dim(T)$, since $\pi_{\dagger,0}\pi^{\dagger,0}\delta^{0} \cong \bigoplus_{n} \delta^{0} \otimes_{\Bbbk} \mathsf{H}_{c}^{n}(T;\Bbbk)[2r-n]$. Similarly we have

$$\operatorname{Ext}^{1}_{\mathsf{P}^{0}_{U,U}}(\pi^{\dagger}_{0}\delta^{0},\pi^{\dagger}_{0}\delta^{0}) = \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{U,B}}(\delta^{0},\delta^{0}) \oplus \mathsf{H}^{2r-1}_{c}(T;\Bbbk)^{*},$$

so that our claim follows from Lemma 10.23.

In view of this isomorphism, to conclude the proof we need to check that the morphism

(10.39)
$$\operatorname{Ext}^{1}_{\mathsf{P}^{0}_{I,I}}(\delta^{0}, \delta^{0}) \oplus \operatorname{Ext}^{1}_{\mathsf{P}^{0}_{U,U}}(\pi^{\dagger}_{0}\delta^{0}, \pi^{\dagger}_{0}\delta^{0}) \to \operatorname{Ext}^{1}_{\mathsf{Rep}_{0}(\mathbb{J}_{\mathbf{D}})}(\mathbb{K}, \mathbb{K})$$

induced by Φ_{I_u,I_u} is injective. Note that on the first summand this morphism identifies with that induced by $\Phi_{I,I}$ (via the morphism induced by the fully faithful functor (10.2)), and on the second summand it identifies with the morphism induced by $\Phi_{U,U}$ (via the morphism induced by the fully faithful functor (10.1)). Consider elements $c_1 \in \operatorname{Ext}_{\mathsf{P}_{I,I}^0}^1(\delta^0, \delta^0)$ and $c_2 \in \operatorname{Ext}_{\mathsf{P}_{U,U}^0}^1(\pi_0^{\dagger}\delta^0, \pi_0^{\dagger}\delta^0)$ such that the image of c_1+c_2 vanishes. We have a forgetful functor $\operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}}) \to \operatorname{Coh}_0(\mathbf{D})$; the composition of the induced morphism

$$\operatorname{Ext}^{1}_{\operatorname{\mathsf{Rep}}_{0}(\mathbb{J}_{\mathbf{D}})}(\mathbb{k},\mathbb{k}) \to \operatorname{Ext}^{1}_{\operatorname{\mathsf{Coh}}_{0}(\mathbf{D})}(\mathbb{k},\mathbb{k})$$

with (10.39) vanishes on the first summand, and identifies with the isomorphism induced by $\Phi_{U,U}$ on the second summand. We deduce that $c_2 = 0$, and then that $c_1 = 0$ since $\Phi_{I,I}$ is an equivalence, which completes the proof.

10.13. Essential surjectivity. We will now prove that the functor Φ_{I_u,I_u} is essentially surjective. For this we need a preliminary lemma. Given $V \in \operatorname{\mathsf{Rep}}(G_{\Bbbk}^{\vee})$ and $M \in \operatorname{\mathsf{Coh}}(\mathbf{D})$, the coherent sheaf $V \otimes_{\Bbbk} M$ has a canonical structure of module for the group scheme $G_{\Bbbk}^{\vee} \times \mathbf{D}$ over \mathbf{D} . Now by construction $\mathbb{J}_{\mathbf{D}}$ is a subgroup scheme of $G_{\Bbbk}^{\vee} \times \mathbf{D}$, hence by restriction $V \otimes_{\Bbbk} M$ has a natural structure of $\mathbb{J}_{\mathbf{D}}$ -module. If M belongs to $\operatorname{\mathsf{Coh}}(\mathbf{D})$, then this module belongs to $\operatorname{\mathsf{Rep}}_0(\mathbb{J}_{\mathbf{D}})$. Recall also the closed subschemes $\mathbf{D}^{(m)} \subset \mathbf{D}$ considered in §8.5.

Lemma 10.26. Any object in $\operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}})$ is a quotient of a module of the form $V \otimes_{\mathbb{k}} \mathscr{O}_{\mathbf{D}^{(m)}}$ with $V \in \operatorname{Rep}(G_{\mathbb{k}}^{\vee})$ and $m \geq 1$.

Proof. We will in fact prove that any M in $\operatorname{Rep}(\mathbb{J}_{\mathbf{D}})$ is a quotient of a module $V \otimes_{\Bbbk} \mathscr{O}_{\mathbf{D}}$ with $V \in \operatorname{Rep}(G_{\Bbbk}^{\vee})$; if M is in $\operatorname{Rep}_{0}(\mathbb{J}_{\mathbf{D}})$ the corresponding surjection $V \otimes_{\Bbbk} \mathscr{O}_{\mathbf{D}} \twoheadrightarrow M$ will necessarily factor through a surjection $V \otimes_{\Bbbk} \mathscr{O}_{\mathbf{D}^{(m)}} \twoheadrightarrow M$ for some $m \geq 1$. Recall the equivalence of categories

$$\mathsf{Coh}^{G^{\vee}_{\Bbbk}}(T^{\vee}_{\Bbbk} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} (G^{\vee}_{\Bbbk})_{\mathrm{reg}} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} T^{\vee}_{\Bbbk}) \xrightarrow{\sim} \mathsf{Rep}(\mathbb{J}_{\mathbf{D}})$$

induced by restriction to $\Sigma_{\mathbf{D}}$, see Proposition 2.20. If \mathscr{F} is the equivariant coherent sheaf corresponding to M, then there exists a G_{\Bbbk}^{\vee} -equivariant coherent sheaf \mathscr{F}' on $T_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} G_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} T_{\Bbbk}^{\vee}$ whose restriction to the open subscheme $T_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} (G_{\Bbbk}^{\vee})_{\mathrm{reg}} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} T_{\Bbbk}^{\vee}$ is \mathscr{F} ; see e.g. [AriB, Lemma 2.12]. Now since $T_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} G_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} T_{\Bbbk}^{\vee}$ is affine, there exists $V \in \mathsf{Rep}(G_{\Bbbk}^{\vee})$ and a surjection of G_{\Bbbk}^{\vee} -equivariant coherent sheaves

$$V \otimes_{\Bbbk} \mathscr{O}_{T^{\vee}_{\Bbbk} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} G^{\vee}_{\Bbbk} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} T^{\vee}_{\Bbbk}} \twoheadrightarrow \mathscr{F}'.$$

Restricting to $T_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} (G_{\Bbbk}^{\vee})_{\mathrm{reg}} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} T_{\Bbbk}^{\vee}$ and then to $\Sigma_{\mathbf{D}}$, we obtain the desired surjection $V \otimes_{\Bbbk} \mathscr{O}_{\mathbf{D}} \twoheadrightarrow M$.

Proposition 10.27. The functor

$$\Phi_{I_u,I_u}:\mathsf{P}^0_{I_u,I_u}\to\mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$$

is essentially surjective.

Proof. We now know that Φ_{I_u,I_u} is exact (see Proposition 10.18) and fully faithful (see Proposition 10.25). By Lemma 10.26, any object in $\operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}})$ is isomorphic to the cokernel of a morphism between objects of the form $V \otimes_{\Bbbk} M$ with $V \in \operatorname{Rep}(G_{\Bbbk}^{\vee})$ and $M \in \operatorname{Coh}_0(\mathbf{D})$, to prove the proposition it therefore suffices to prove that any such object is isomorphic to the image of an object of $\mathsf{P}^0_{I_u,I_u}$. Let us fix $V \in$ $\operatorname{Rep}(G_{\Bbbk}^{\vee})$ and $M \in \operatorname{Coh}_0(\mathbf{D})$. Since $\Phi_{U,U}$ is essentially surjective (see Theorem 8.3), Proposition 10.19 implies that there exists \mathscr{F} in $\mathsf{P}^0_{I_u,I_u}$ such that $\Phi_{I_u,I_u}(\mathscr{F}) \cong M$. Then Lemma 10.21 implies that

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\widehat{\mathscr{Z}}(V) \stackrel{\mathrm{p}_{\widehat{\star}}^0}{\longrightarrow} \mathscr{F}) \cong V \otimes_{\Bbbk} M,$$

which finishes the proof.

10.14. Monoidality. Combining Propositions 10.18, 10.19, 10.25 and 10.27, we have now proved that Φ_{I_u,I_u} is an equivalence of categories, and that the diagrams in (10.3) are commutative. To conclude the proof of Theorem 10.1, it therefore only remains to prove that Φ_{I_u,I_u} is monoidal. Recall that in (10.27) we have defined a canonical "monoidality" morphism; what we will now prove is that this morphism is an isomorphism for any \mathscr{F}, \mathscr{G} in P_{I_u,I_u}^0 .

We start with a special case.

Lemma 10.28. The morphism (10.27) is an isomorphism in case

$$\mathscr{F} = \widehat{\mathscr{Z}}(V) \widehat{\star}^0 \mathscr{F}' \quad and \quad \mathscr{G} = \widehat{\mathscr{Z}}(V') \widehat{\star}^0 \mathscr{G}'$$

for some $V, V' \in \mathsf{Rep}(G^{\vee}_{\Bbbk})$ and $\mathscr{F}', \mathscr{G}' \in \mathsf{P}^0_{U,U}$.

Proof. Assume that \mathscr{F} and \mathscr{G} are as in the lemma. By Lemma 10.21 and Proposition 10.19 we have

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}) \cong V \otimes_{\Bbbk} \Phi_{U,U}(\mathscr{F}'), \quad \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{G}) \cong V' \otimes_{\Bbbk} \Phi_{U,U}(\mathscr{G}').$$

On the other hand, using (7.8) we obtain a canonical isomorphism

$$\mathscr{F} \, {}^{\mathrm{p}} \! \star^{0}_{\mathrm{I}_{\mathrm{u}}} \mathscr{G} \cong \widehat{\mathscr{Z}}(V \otimes_{\Bbbk} V') \, \widehat{\star}^{0} \, (\mathscr{F}' \, \star^{0}_{U} \, \mathscr{G}'),$$

so that similarly we have

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathscr{F}\star^{0}_{\mathrm{I}_{\mathrm{u}}}\mathscr{G})\cong (V\otimes_{\Bbbk}V')\otimes_{\Bbbk}\Phi_{U,U}(\mathscr{F}'\star^{0}_{U}\mathscr{G}')$$

Under these identifications the morphism (10.27) is induced by the isomorphism

$$\Phi_{U,U}(\mathscr{F}' \star^0_U \mathscr{G}') \cong \Phi_{U,U}(\mathscr{F}') \circledast \Phi_{U,U}(\mathscr{G}')$$

defining the monoidal structure of $\Phi_{U,U}$ (see Theorem 8.3); it is therefore an isomorphism.

Proposition 10.29. The morphism (10.27) is an isomorphism for all \mathscr{F}, \mathscr{G} in $\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$; in other words, the functor $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ admits a canonical monoidal structure.

Proof. By right exactness of the bifunctors $\star^0_{I_u}$ and \circledast and the five lemma, if given $\mathscr{F}, \mathscr{G}, \mathscr{G}'$ in $\mathsf{P}^0_{I_u, I_u}$ the claim is known for the pairs of objects $(\mathscr{F}, \mathscr{G})$ and $(\mathscr{F}, \mathscr{G}')$, then it will hold for the pair $(\mathscr{F}, \mathscr{G}'')$ for any cokernel \mathscr{G}'' of a morphism $\mathscr{G} \to \mathscr{G}'$. The same property holds when the order of the factors is switched. In view of Lemma 10.28, to conclude the proof it therefore suffices to prove that any object in $\mathsf{P}^0_{I_u, I_u}$ is isomorphic to the cokernel of a morphism between objects of the form $\widehat{\mathscr{T}}(V) \stackrel{*}{\star}^0 \mathscr{F}'$ with $V \in \mathsf{Rep}(G^{\vee}_\Bbbk)$ and $\mathscr{F}' \in \mathsf{P}^0_{U,U}$. In view of the computation in the

proof of Lemma 10.28, this follows from Lemma 10.26 and the fact that $\Phi_{U,U}$ and Φ_{I_u,I_u} are equivalences of categories.

11. A Soergel-type description of tilting perverse sheaves

We continue with the assumptions of Section 10. In this section we explain how to deduce Theorem 1.5 from Theorem 1.3.

11.1. Tilting completed perverse sheaves. Recall the category $\mathsf{P}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and its subcategory $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ from §6.3. Recall also that $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ is stable under the monoidal product $\widehat{\star}$. In §6.6 we have defined, for any $s \in S_{\mathrm{f}}$, an object $\Xi^{\wedge}_{s,!}$ in $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$. On the other hand, in §3.1 we have chosen, for any $s \in S \smallsetminus S_{\mathrm{f}}$, elements $w \in W$ and $s' \in S_{\mathrm{f}}$ such that $\ell(ws') = \ell(w) + 1$ and $s = ws'w^{-1}$. We then have

$$\Delta_w^{\wedge} \mathbin{\widehat{\star}} \Delta_{s'}^{\wedge} \cong \Delta_s^{\wedge} \mathbin{\widehat{\star}} \Delta_w^{\wedge} \quad \text{and} \quad \nabla_{w^{-1}}^{\wedge} \mathbin{\widehat{\star}} \nabla_s^{\wedge} \cong \nabla_{s'}^{\wedge} \mathbin{\widehat{\star}} \nabla_{w^{-1}}^{\wedge}$$

by Lemma 6.3(2), hence

$$\Delta_s^{\wedge} \cong \Delta_w^{\wedge} \mathbin{\widehat{\star}} \Delta_{s'}^{\wedge} \mathbin{\widehat{\star}} \nabla_{w^{-1}}^{\wedge} \quad \text{and} \quad \nabla_s^{\wedge} \cong \Delta_w^{\wedge} \mathbin{\widehat{\star}} \nabla_{s'}^{\wedge} \mathbin{\widehat{\star}} \nabla_{w^{-1}}^{\wedge}$$

by Lemma 6.3(1). From these isomorphisms we deduce that the object

$$\Xi_{s,!}^{\wedge} := \Delta_w^{\wedge} \mathbin{\widehat{\star}} \Xi_{s',!}^{\wedge} \mathbin{\widehat{\star}} \nabla_{w^{-1}}^{\wedge}$$

is a representative for the indecomposable tilting object \mathscr{T}_s^{\wedge} . On the other hand, for any $\omega \in \Omega$ the natural morphism $\Delta_{\omega}^{\wedge} \to \nabla_{\omega}^{\wedge}$ is an isomorphism since $\widetilde{\mathrm{Fl}}_{G,\omega}$ is closed, which shows that these perverse sheaves are tilting.

Recall that the category $D^{\wedge}_{I_u,I_u}$ is Krull–Schmidt, see §6.1. Standard arguments (see e.g. [BR1, Remark 7.9]) show that any object in $T^{\wedge}_{I_u,I_u}$ is a direct sum of direct summands of objects of the form

$$\Delta^{\wedge}_{\omega} \mathbin{\widehat{\star}} \Xi^{\wedge}_{s_1,!} \mathbin{\widehat{\star}} \cdots \mathbin{\widehat{\star}} \Xi^{\wedge}_{s_i,!}$$

with $\omega \in \Omega$ and $s_1, \cdots, s_i \in S$.

Let us recall also that for any V in $\operatorname{Rep}(G_{\Bbbk}^{\vee})$ tilting the object

$$\Xi^{\wedge}_{!} \,\widehat{\star} \,\widehat{\mathscr{Z}}(V) \cong \widehat{\mathscr{Z}}(V) \,\widehat{\star} \,\Xi^{\wedge}_{!}$$

belongs to $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$, see Proposition 7.9.

11.2. Soergel representations. Recall the category

 $\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$

defined in §3.3. Note that any object in this category is finite and flat, i.e. finite and projective, as an $\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})$ -module for the action on the right.

Let us note the following technical property for later use.

Lemma 11.1. The scheme $(T_{\Bbbk}^{\vee})^{\wedge} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma}$ is integral.

Proof. First, we note that since T_{\Bbbk}^{\vee} is isomorphic to a product of copies of the multiplicative group, the scheme $(T_{\Bbbk}^{\vee})^{\wedge}$ is integral. Recall the open subscheme $(T_{\Bbbk}^{\vee})_{\circ} \subset T_{\Bbbk}^{\vee}$ introduced in §2.6; in concrete terms, $\mathscr{O}((T_{\Bbbk}^{\vee})_{\circ})$ is the spectrum of the localization of $\mathscr{O}(T_{\Bbbk}^{\vee})$ with respect to the elements $\alpha - 1$ where α runs over the coroots of (G, T). We define similarly $(T_{\Bbbk}^{\vee})_{\circ}^{\wedge}$ as the spectrum of the localization of $\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})$ with respect to the elements $\alpha - 1$ where α runs over the coroots of (G, T).

(G,T). Then $(T_{\Bbbk}^{\vee})^{\wedge}_{\circ}$ is integral, and $\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge}_{\circ})$ is also the localization of $\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})$ viewed as an $\mathscr{O}(T_{\Bbbk}^{\vee})$ -module; in other words we have a canonical identification

(11.1)
$$(T^{\vee}_{\Bbbk})_{\circ} \times_{T^{\vee}_{\Bbbk}} (T^{\vee}_{\Bbbk})^{\wedge} \xrightarrow{\sim} (T^{\vee}_{\Bbbk})^{\wedge}_{\circ}.$$

Since \mathbb{J}_{Σ} is flat over $T_{\mathbb{k}}^{\vee}/W_{\mathrm{f}}$ (see Lemma 2.17), the natural morphism

$$\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma}) \to \mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge}_{\circ} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma})$$

is injective. To prove our claim it therefore suffices to prove that the right-hand side is a domain. However, in view of (11.1) we have

$$(T^{\vee}_{\Bbbk})^{\wedge}_{\circ} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma} = (T^{\vee}_{\Bbbk})^{\wedge} \times_{T^{\vee}_{\Bbbk}} \left((T^{\vee}_{\Bbbk})_{\circ} \times_{T^{\vee}_{\Bbbk}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma} \right).$$

By Lemma 2.19 the right-hand side identifies with

$$(T^{\vee}_{\Bbbk})^{\wedge} \times_{T^{\vee}_{\Bbbk}} ((T^{\vee}_{\Bbbk})_{\circ} \times T^{\vee}_{\Bbbk}) = (T^{\vee}_{\Bbbk})^{\wedge}_{\circ} \times T^{\vee}_{\Bbbk},$$

which is integral since it is a split torus over the integral scheme $(T_{\Bbbk}^{\vee})_{\circ}^{\wedge}$.

131

11.3. **Statement.** Our main application of Theorem 10.1 is the following statement.

Theorem 11.2. There exists a canonical equivalence of additive monoidal categories

$$\Phi_{\mathsf{T}}:(\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}},\widehat{\star})\xrightarrow{\sim}(\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}}),\circledast)$$

which satisfies the following properties:

- (1) $\Phi_{\mathsf{T}}(\Xi_{s,!}^{\wedge}) \cong \mathscr{B}_{s}^{\wedge}$ for any $s \in S$;
- (2) $\Phi_{\mathsf{T}}(\Delta_{\omega}^{\wedge}) \cong \mathscr{M}_{\omega}^{\wedge}$ for any $\omega \in \Omega$;
- (3) $\Phi_{\mathsf{T}}(\widehat{\mathscr{Z}}(V) \mathbin{\widehat{\star}} \Xi^{\wedge}_{\mathsf{L}}) \cong V \otimes_{\Bbbk} \mathscr{O}(\mathbf{D}^{\wedge}) \text{ for any } V \in \mathsf{Rep}(G^{\vee}_{\Bbbk}) \text{ tilting.}$

Note that property (3) shows in particular that $V \otimes_{\Bbbk} \mathscr{O}(\mathbf{D}^{\wedge})$ belongs to $\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ for any $V \in \mathsf{Rep}(G^{\vee}_{\Bbbk})$ tilting, which is not clear from definitions. The proof of Theorem 11.2 occupies the rest of the section. The idea of the proof is to give "sequential" descriptions of the categories $\mathsf{T}^{\wedge}_{\mathrm{I}_u,\mathrm{I}_u}$ and $\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$, in terms of the categories $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$ and $\mathsf{Rep}_0(\mathbb{J}_{\mathbf{D}})$ respectively, and then use Theorem 10.1 to relate these two descriptions.

More specifically, on the perverse side, we will denote by $\mathsf{P}_{I_u,I_u}^{\wedge,\mathrm{seq}}$ the monoidal additive k-linear category with

• objects the projective systems $(\mathscr{F}_m : m \ge 1)$ of objects in $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$ such that \mathcal{J}^m acts trivially on \mathscr{F}_m for any $m \ge 1$, and for $m' \ge m$ the morphism

$$\mathscr{O}((T^{\vee}_{\Bbbk})^{(m)}) \otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})} \mathscr{F}_{m'} \to \mathscr{F}_{m}$$

induced by the transition morphism $\mathscr{F}_{m'} \to \mathscr{F}_m$ is an isomorphism;

• morphisms from $(\mathscr{F}_m : m \ge 1)$ to $(\mathscr{G}_m : m \ge 1)$ the inverse limit

$$\varprojlim_{m} \operatorname{Hom}_{\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}}(\mathscr{F}_{m},\mathscr{G}_{m})$$

where for $m' \geq m$ the morphism $\operatorname{Hom}(\mathscr{F}_{m'}, \mathscr{G}_{m'}) \to \operatorname{Hom}(\mathscr{F}_m, \mathscr{G}_m)$ is induced by the functor $\mathscr{O}((T^{\vee}_{\Bbbk})^{(m)}) \otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})} (-);$

• monoidal product sending a pair of objects $((\mathscr{F}_m : m \ge 1), (\mathscr{G}_m : m \ge 1))$ to the object $(\mathscr{F}_m \, {}^{p}\star^0_{\Gamma_u} \, \mathscr{G}_m : m \ge 1)$. (This projective system is indeed an object of our category by right exactness of ${}^{p}\star^0_{\Gamma_u}$.)

With this definition, Proposition 9.6 and Lemma 9.7 imply that the assignment

$$\mathscr{T} \mapsto (\mathsf{C}^0_m(\mathscr{T}) : m \ge 1)$$

defines a fully faithful monoidal functor

(11.2)
$$\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \to \mathsf{P}^{\wedge,\mathrm{seq}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}.$$

11.4. Truncation functors for representations: fully faithfulness. We now consider the category $\text{Rep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$, and the functor D_m introduced in §10.2.

Lemma 11.3. Let $M, M' \in \mathsf{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$. If M' is projective for the right action of $\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge})$, then the functors D_m induce an isomorphism

$$\operatorname{Hom}_{\operatorname{\mathsf{Rep}}(\mathbb{J}_{\mathbf{D}}^{\wedge})}(M,M') \xrightarrow{\sim} \varprojlim_{m} \operatorname{Hom}_{\operatorname{\mathsf{Rep}}_{0}}(\mathbb{J}_{\mathbf{D}})(\mathsf{D}_{m}(M),\mathsf{D}_{m}(M')).$$

Proof. For any M, M' in $\operatorname{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ and any $m \geq 1$ the forgetful functor $\operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}}) \to \operatorname{Coh}_0(\mathbf{D})$ induces an embedding

$$\operatorname{Hom}_{\operatorname{\mathsf{Rep}}_0(\mathbb{J}_{\mathbf{D}})}(\mathsf{D}_m(M),\mathsf{D}_m(M')) \hookrightarrow \operatorname{Hom}_{\operatorname{\mathsf{Coh}}_0(\mathbf{D})}(\mathsf{D}_m(M),\mathsf{D}_m(M')).$$

These embeddings give rise to a morphism fitting as the right vertical arrow in the following commutative diagram (where the other arrows are the obvious maps):

The lower horizontal arrow in this diagram is an isomorphism by [EGA1, Chap. 0, Corollaire 7.2.10]. It follows that the upper horizontal arrow, which is the morphism we need to study, is injective.

To prove surjectivity, we consider a projective system

$$(f_m)_{m\geq 1}\in \varprojlim_m \operatorname{Hom}_{\operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}})}(\mathsf{D}_m(M),\mathsf{D}_m(M')).$$

Each f_m is a morphism of $\mathscr{O}(\mathbf{D})$ -modules; this collection therefore defines a morphism $f: M \to M'$ of $\mathscr{O}(\mathbf{D}^{\wedge})$ -modules, and what we have to show is that f is also a morphism of $\mathscr{O}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ -comodules. We consider the second projection $\mathbf{D}^{\wedge} \to (T^{\vee}_{\Bbbk})^{\wedge}$. Then we have

$$\mathbb{J}_{\mathbf{D}}^{\wedge} = \mathbf{D}^{\wedge} \times_{(T_{\Bbbk}^{\vee})^{\wedge}} \left((T_{\Bbbk}^{\vee})^{\wedge} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma} \right),$$

so that M and M' can be considered as representations of $(T_{\Bbbk}^{\vee})^{\wedge} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma}$. From this point of view, to prove the desired claim it suffices to show that f is a morphism of $\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma})$ -comodules. Now \mathbb{J}_{Σ} is smooth over the smooth \Bbbk -scheme $T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$, and $(T_{\Bbbk}^{\vee})^{\wedge}$ is flat over $T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$ as the composition of flat morphisms

$$(T^{\vee}_{\Bbbk})^{\wedge} \to (T^{\vee}_{\Bbbk}/W_{\mathrm{f}})^{\wedge} \to T^{\vee}_{\Bbbk}/W_{\mathrm{f}}$$

(see Lemma 3.3 for the first map). By Lemma C.1 and Corollary C.4, this implies that $(T_{\Bbbk}^{\vee})^{\wedge} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma}$ is infinitesimally flat. This group scheme is also integral by Lemma 11.1, and noetherian since it is of finite type over the noetherian scheme $(T_{\Bbbk}^{\vee})^{\wedge}$. We can therefore apply Lemma C.2 which reduces the proof to checking that f is a morphism of $\mathrm{Dist}((T_{\Bbbk}^{\vee})^{\wedge} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma})$ -modules. However each f_m is a morphism of $\mathrm{Dist}((T_{\Bbbk}^{\vee})^{\wedge} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \mathbb{J}_{\Sigma})$ -modules, so that this claim is clear.

Let us denote by $\mathsf{Rep}^{\mathrm{seq}}(\mathbb{J}^\wedge_{\mathbf{D}})$ the monoidal additive $\Bbbk\text{-linear category with:}$

• objects the projective systems $(M_m : m \ge 1)$ of objects of $\operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}})$ such that \mathcal{J}^m acts trivially on M_m for any $m \ge 1$, and for $m' \ge m$ the morphism

$$\mathsf{D}_m(M_{m'}) \to M_m$$

- induced by the transition morphism $M_{m'} \to M_m$ is an isomorphism;
- morphisms from $(M_m : m \ge 1)$ to $(N_m : m \ge 1)$ the inverse limit

$$\varprojlim_{m} \operatorname{Hom}_{\operatorname{\mathsf{Rep}}_{0}(\mathbb{J}_{\mathbf{D}})}(M_{m}, N_{m})$$

where for $m' \ge m$ the transition morphism

$$\operatorname{Hom}_{\operatorname{\mathsf{Rep}}_0(\mathbb{J}_{\mathbf{D}})}(M_{m'}, N_{m'}) \to \operatorname{Hom}_{\operatorname{\mathsf{Rep}}_0(\mathbb{J}_{\mathbf{D}})}(M_m, N_m)$$

is induced by the functor D_m ;

• monoidal product sending a pair of objects $((M_m : m \ge 1), (N_m : m \ge 1))$ to the object $(M_m \circledast N_m : m \ge 1)$.

With this definition, Lemma 10.2 and Lemma 11.3 imply that the assignment

$$M \mapsto (\mathsf{D}_m(M) : m \ge 1)$$

defines a monoidal functor

(11.3) $\operatorname{\mathsf{Rep}}(\mathbb{J}^{\wedge}_{\mathbf{D}}) \to \operatorname{\mathsf{Rep}}^{\operatorname{seq}}(\mathbb{J}^{\wedge}_{\mathbf{D}})$

which is fully faithful on the full subcategory of objects which are projective for the right action of $\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})$. In particular, objects of $\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ satisfy this condition (see §11.2), hence (11.3) is fully faithful on $\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$.

11.5. Construction of Φ_{T} . Comparing the definitions of the categories $\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge,\mathrm{seq}}$ (see §11.3) and $\mathsf{Rep}^{\mathrm{seq}}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ (see §11.4), we see that Theorem 10.1 provides a canonical equivalence of monoidal additive k-linear categories

$$\Phi^{\mathrm{seq}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}:\mathsf{P}^{\wedge,\mathrm{seq}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\xrightarrow{\sim}\mathsf{Rep}^{\mathrm{seq}}(\mathbb{J}^{\wedge}_{\mathbf{D}})$$

In the remaining subsections we will prove the following claim.

Lemma 11.4. (1) For any $s \in S$ there exists an isomorphism of projective systems

$$\left(\Phi_{\mathbf{I}_{u},\mathbf{I}_{u}}\left(\mathsf{C}_{m}^{0}(\Xi_{s,!}^{\wedge})\right):m\geq1\right)\cong\left(\mathsf{D}_{m}(\mathscr{B}_{s}^{\wedge}):m\geq1\right).$$

(2) There exists an isomorphism of projective systems

$$\left(\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\left(\mathsf{C}_{m}^{0}(\Xi_{!}^{\wedge})\right):m\geq1\right)\cong\left(\mathsf{D}_{m}(\mathscr{O}(\mathbf{D}^{\wedge})):m\geq1\right).$$

For now, let us explain why Lemma 11.4 allows to complete the proof of Theorem 11.2.

Proof of Theorem 11.2. Lemma 11.4(1) implies that the equivalence Φ_{I_u,I_u}^{seq} matches the image of each $\Xi_{s,!}^{\wedge}$ ($s \in S$) under (11.2) with the image of \mathscr{B}_s^{\wedge} under (11.3). Similarly, Proposition 10.20(2) implies that Φ_{I_u,I_u}^{seq} matches the image of each Δ_{ω}^{\wedge} ($\omega \in \Omega$) under (11.2) with the image of $\mathscr{M}_{\omega}^{\wedge}$ under (11.3). In view of the discussion in §11.1 and the definition of $\mathsf{SRep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$, these properties and monoidality imply that Φ_{I_u,I_u}^{seq} identifies the essential images of the functor (11.2) with the image of $\mathsf{SRep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$ under (11.3). We deduce the equivalence Φ_{T} , and also that this equivalence satisfies properties (1)–(2). Finally, consider some $V \in \mathsf{Rep}(G_{\Bbbk}^{\vee})$ which is tilting. As explained in §11.1, we have the object $\widehat{\mathscr{Z}}(V) \stackrel{\sim}{\star} \Xi_{!}^{\wedge}$ in $\mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$. Lemma 11.4(2) and Lemma 10.21 provide an isomorphism of projective systems

$$\left(\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\left(\mathsf{C}_{m}^{0}(\widehat{\mathscr{Z}}(V)\,\widehat{\star}\,\Xi_{!}^{\wedge})\right):m\geq1\right)\cong\left(\mathsf{D}_{m}(V\otimes_{\Bbbk}\mathscr{O}(\mathbf{D}^{\wedge})):m\geq1\right),$$

which allows to identify the images of $\Phi_{\mathsf{T}}(\widehat{\mathscr{X}}(V)\widehat{\star}\Xi_{!}^{\wedge})$ and $V \otimes_{\Bbbk} \mathscr{O}(\mathbf{D}^{\wedge})$ under (11.3). Now $\Phi_{\mathsf{T}}(\widehat{\mathscr{X}}(V)\widehat{\star}\Xi_{!}^{\wedge})$ satisfies the assumption in Lemma 11.3 because it belongs to $\mathsf{SRep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$, and $V \otimes_{\Bbbk} \mathscr{O}(\mathbf{D}^{\wedge})$ also does by Lemma 3.3(1). We deduce that these objects are isomorphic, proving that Φ_{T} satisfies property (3).

11.6. The case of G/U. Recall the category $\mathsf{D}_{U,U}^{\wedge}$ from §6.6, the heart $\mathsf{P}_{U,U}^{\wedge}$ of the perverse t-structure on this category, and the full subcategory $\mathsf{T}_{U,U}^{\wedge}$ of tilting perverse sheaves. Recall also that we denote by $\mathsf{P}_{U,U}$ the heart of the perverse t-structure on $\mathsf{D}_{U,U}$. The pushforward along the closed embedding $G/U \hookrightarrow \widetilde{\mathsf{Fl}}_{G}$ defines a fully faithful monoidal functor $(\mathsf{D}_{U,U}^{\wedge}, \widehat{\star}_U) \to (\mathsf{D}_{I_u,I_u}^{\wedge}, \widehat{\star})$, which restricts to a fully faithful monoidal functor $(\mathsf{T}_{U,U}^{\wedge}, \widehat{\star}_U) \to (\mathsf{T}_{I_u,I_u}^{\wedge}, \widehat{\star})$. As noted in §8.1, for any $m \geq 1$ the restriction of the functor C_m to $\mathsf{P}_{U,U}^{\wedge}$ factors through a functor $\mathsf{P}_{U,U}^{\wedge} \to \mathsf{P}_{U,U}$, which will be denoted similarly.

Consider also the category $\mathsf{Coh}(\mathbf{D}^{\wedge})$ of coherent sheaves on the noetherian scheme \mathbf{D}^{\wedge} . We have a natural fully faithful functor

$$(11.4) \qquad \qquad \mathsf{Coh}(\mathbf{D}^{\wedge}) \to \mathsf{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$$

sending a coherent sheaf to itself with the trivial structure as a representation of $\mathbb{J}_{\mathbf{D}}^{\wedge}$, whose essential image contains the objects \mathscr{B}_{s}^{\wedge} for $s \in S_{\mathrm{f}}$. It is clear from definition that the monoidal product \circledast on $\mathsf{Rep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$ restricts to a monoidal product on $\mathsf{Coh}(\mathbf{D}^{\wedge})$ (which will be denoted similarly). It is clear also that for any $m \geq 1$ the restriction of the functor D_{m} to $\mathsf{Coh}(\mathbf{D}^{\wedge})$ factors through a functor $\mathsf{Coh}(\mathbf{D}^{\wedge}) \to \mathsf{Coh}_{0}(\mathbf{D})$, which will be denoted similarly. We will denote by

$$SCoh(\mathbf{D}^{\wedge})$$

the full monoidal additive subcategory of $\mathsf{Coh}(\mathbf{D}^{\wedge})$ generated (under the monoidal product, direct sums and direct summands) by the unit object \mathscr{M}_{e}^{\wedge} and the objects \mathscr{B}_{s}^{\wedge} for $s \in S_{\mathrm{f}}$. With this definition, (11.4) identifies $\mathsf{SCoh}(\mathbf{D}^{\wedge})$ with a full subcategory of $\mathsf{SRep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$.

In [BR1, Theorem 11.8] we have shown that the functor $\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi_{!}^{\wedge}, -)$ defines an equivalence of monoidal categories

$$\Phi_{\mathsf{T},U}: (\mathsf{T}^{\wedge}_{U,U}, \widehat{\star}_U) \xrightarrow{\sim} (\mathsf{SCoh}(\mathbf{D}^{\wedge}), \circledast)$$

which satisfies

(11.5) $\Phi_{\mathsf{T},U}(\Xi_{s,!}^{\wedge}) \cong \mathscr{B}_{s}^{\wedge} \quad \text{for } s \in S_{\mathrm{f}}$

and

(11.6)
$$\Phi_{\mathsf{T},U}(\Xi^{\wedge}_{!}) \cong \mathscr{O}(\mathbf{D}^{\wedge}).$$

Lemma 11.5. For any $m \geq 1$ and any \mathscr{T} in $\mathsf{T}^{\wedge}_{U,U}$ we have a canonical isomorphism

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\mathscr{T})) \cong \mathsf{D}_{m}(\Phi_{\mathsf{T},U}(\mathscr{T})),$$

where the right-hand side is seen as an object of $\mathsf{Coh}_0(\mathbb{J}_{\mathbf{D}})$ via (10.1).

Proof. The object $\mathsf{C}_m^0(\mathscr{T})$ belongs to the subcategory $\mathsf{P}_{U,U}^0 \subset \mathsf{P}_{\mathrm{I}_u,\mathrm{I}_u}^0$. By Proposition 10.19, this implies that

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\mathscr{T})) \cong \Phi_{U,U}(\mathsf{C}^{0}_{m}(\mathscr{T})) \cong \mathrm{Hom}_{\mathsf{P}^{\wedge}_{U,U}}(\Xi^{\wedge}_{!},\mathsf{C}_{m}(\mathscr{T})),$$

where the last step uses Lemma 8.2. Now by projectivity of $\Xi_{!}^{\wedge}$ (see Lemma 6.10) we have

$$\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi^{\wedge}_{!},\mathsf{C}_{m}(\mathscr{T}))\cong\operatorname{Hom}_{\mathsf{P}_{U,U}^{\wedge}}(\Xi^{\wedge}_{!},\mathscr{T})\otimes_{\mathscr{O}(T^{\vee}_{\Bbbk})}\mathscr{O}((T^{\vee}_{\Bbbk})^{(m)}),$$

which finishes the proof in view of the definitions of $\Phi_{\mathsf{T},U}$ and D_m .

11.7. Images of truncated standard and costandard objects. We now need to describe the images under Φ_{I_u,I_u} of (images in $\mathsf{P}^0_{I_u,I_u}$ of) truncated standard and costandard free-monodromic perverse sheaves. Recall that the images of truncated costandard perverse sheaves have been described in Lemma 8.6. We next prove the analogous statement for standard objects.

Lemma 11.6. For any $w \in W_f$, there exists an isomorphism of projective systems of $\mathscr{O}(\mathbf{D})$ -modules

$$(\Phi_{U,U}(\mathsf{C}^0_m(\Delta_w^{\wedge})): m \ge 1) \cong (\mathscr{M}_w/\mathcal{J}^m \cdot \mathscr{M}_w: m \ge 1).$$

Proof. By Lemma 6.3(1) we have an isomorphism

$$\Delta_w^{\wedge} \mathbin{\widehat{\star}} \nabla_{w^{-1}}^{\wedge} \cong \delta^{\wedge}.$$

After fixing such an isomorphism, using Proposition 9.6 we deduce for any $m \ge 1$ an isomorphism

$$\mathsf{C}^0_m(\Delta^\wedge_w) \,{}^{\mathrm{p}}\!\!\star^0_{\mathrm{I}_{\mathrm{u}}} \,\mathsf{C}^0_m(\nabla^\wedge_{w^{-1}}) \cong \mathsf{C}^0_m(\delta^\wedge),$$

which by monoidality of $\Phi_{U,U}$ provides an isomorphism

(11.7)
$$\Phi_{U,U}(\mathsf{C}^0_m(\Delta^\wedge_w)) \circledast \Phi_{U,U}(\mathsf{C}^0_m(\nabla^\wedge_{w^{-1}})) \cong \Phi_{U,U}(\mathsf{C}^0_m(\delta^\wedge))$$

Now, consider

$$M_w := \lim_{m \ge 1} \Phi_{U,U}(\mathsf{C}^0_m(\Delta_w^\wedge)).$$

For any $m \geq 1$, the $\mathscr{O}(\mathbf{D})$ -module $\Phi_{U,U}(\mathsf{C}^0_m(\Delta^\wedge_w))$ is finitely generated, and, by exactness of $\Phi_{U,U}$, for any $m' \geq m$ the natural morphism

$$\left(\mathscr{O}(\mathbf{D})/\mathcal{J}^m\cdot\mathscr{O}(\mathbf{D})\right)\otimes_{\mathscr{O}(\mathbf{D})}\Phi_{U,U}(\mathsf{C}_{m'}^0(\Delta_w^\wedge))\to\Phi_{U,U}(\mathsf{C}_m^0(\Delta_w^\wedge))$$

is an isomorphism. By [EGA1, Chap. 0, Proposition 7.2.9], it follows that M_w is a finitely generated $\mathscr{O}(\mathbf{D}^{\wedge})$ -module, and that for any $m \geq 1$ the natural morphism

$$M_w/\mathcal{J}^m \cdot M_w \to \Phi_{U,U}(\mathsf{C}^0_m(\Delta_w^\wedge))$$

is an isomorphism. To conclude the proof, it therefore suffices to construct an isomorphism of $\mathscr{O}(\mathbf{D}^{\wedge})$ -modules $M_w \cong \mathscr{M}_w^{\wedge}$. Now the isomorphisms (11.7) and the description of images of costandard objects in Lemma 8.6 provide an isomorphism

$$M_w \otimes_{\mathscr{O}((T_{\Bbbk}^{\vee})^{\wedge})} \mathscr{M}_{w^{-1}}^{\wedge} \cong \mathscr{M}_e^{\wedge}$$

Since $\mathscr{M}_{w^{-1}}^{\wedge}$ is invertible for the product \circledast , with inverse \mathscr{M}_{w}^{\wedge} , we deduce the desired isomorphism $M_{w} \cong \mathscr{M}_{w}^{\wedge}$.

We can finally prove the desired general statement.

Lemma 11.7. For any $w \in W$, there exist isomorphisms of projective systems

$$\left(\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\Delta_{w}^{\wedge})): m \geq 1\right) \cong \left(\mathsf{D}_{m}(\mathscr{M}_{w}^{\wedge}): m \geq 1\right)$$

and

$$\left(\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\nabla_{w}^{\wedge})): m \geq 1\right) \cong \left(\mathsf{D}_{m}(\mathscr{M}_{w}^{\wedge}): m \geq 1\right).$$

Proof. First we prove the isomorphisms when $w = s \in S$. In case $s \in S_{\rm f}$, they follow from Lemma 8.6 and Lemma 11.6. Now assume that $s \in S_{\rm f} \setminus \{s\}$. Then there exist $w \in W_{\rm f}$ and $\lambda \in X_*(T)$ antidominant such that $s = wt(\lambda)$ with $\ell(t(\lambda)) = \ell(w) + 1$. By Lemma 6.3(1)–(2) we then have

$$\Delta_s^{\wedge} \cong \nabla_w^{\wedge} \,\widehat{\star} \, \Delta_{\mathsf{t}(\lambda)}^{\wedge}$$

Using Proposition 9.6 and the monoidality of $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ we deduce for any $m\geq 1$ an isomorphism

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\Delta^{\wedge}_{s})) \cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\nabla^{\wedge}_{w})) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\Delta^{\wedge}_{\mathsf{t}(\lambda)})).$$

Now $\Phi_{I_u,I_u}(C_m^0(\nabla_w^\wedge))$ is described in Lemma 8.6, and $\Phi_{I_u,I_u}(C_m^0(\Delta_{t(\lambda)}^\wedge))$ is described in Proposition 10.20(1). Using these descriptions, Lemma 10.2 and (3.6), we deduce the desired isomorphism

$$\left(\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\Delta_{s}^{\wedge})):m\geq1\right)\cong\left(\mathsf{D}_{m}(\mathscr{M}_{s}^{\wedge}):m\geq1\right).$$

The isomorphism

$$\left(\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\nabla_{s}^{\wedge})):m\geq1\right)\cong\left(\mathsf{D}_{m}(\mathscr{M}_{s}^{\wedge}):m\geq1\right)$$

follows, using the same arguments as in the proof of Lemma 11.6.

Note that in case $w \in \Omega$, we have $\Delta_w^{\wedge} \cong \nabla_w^{\wedge}$, so that both isomorphisms follow from Proposition 10.20(2).

Finally we treat the general case. Given a reduced expression $w = \omega s_1 \cdots s_r$ (with $\omega \in \Omega$, $s_1, \cdots, s_r \in S$ and $r = \ell(w)$), by Lemma 6.3(2) we have

$$\Delta_w^{\wedge} \cong \Delta_\omega^{\wedge} \mathbin{\widehat{\star}} \Delta_{s_1}^{\wedge} \mathbin{\widehat{\star}} \cdots \mathbin{\widehat{\star}} \Delta_{s_r}^{\wedge}, \quad \nabla_w^{\wedge} \cong \nabla_\omega^{\wedge} \mathbin{\widehat{\star}} \nabla_{s_1}^{\wedge} \mathbin{\widehat{\star}} \cdots \mathbin{\widehat{\star}} \nabla_{s_r}^{\wedge}$$

Using Proposition 9.6 and monoidality of Φ_{I_u,I_u} we deduce isomorphisms

$$\begin{split} \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\Delta_{w}^{\wedge})) &\cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\Delta_{\omega}^{\wedge})) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\Delta_{s_{1}}^{\wedge})) \circledast \cdots \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\Delta_{s_{r}}^{\wedge})), \\ \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\nabla_{w}^{\wedge})) &\cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\nabla_{\omega}^{\wedge})) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\nabla_{s_{1}}^{\wedge})) \circledast \cdots \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}_{m}^{0}(\nabla_{s_{r}}^{\wedge})). \end{split}$$

We deduce the desired isomorphisms using the case of the elements ω and s_i , together with Lemma 10.2 and (3.6).

Remark 11.8. Since Φ_{I_u,I_u} is an equivalence, Lemma 11.7 implies in particular that the projective systems ($C_m^0(\Delta_w^{\wedge}): m \geq 1$) and ($C_m^0(\nabla_w^{\wedge}): m \geq 1$) are isomorphic. In other words, in the quotient $\mathsf{P}^0_{I_u,I_u}$ one cannot see the difference between standard and costandard objects.

11.8. Completion of the proof. We can finally complete the proof of Lemma 11.4.

Proof of Lemma 11.4. From Lemma 11.5 and (11.5) we deduce (1) in case $s \in S_{\rm f}$. Similarly, (2) follows from Lemma 11.5 and (11.6).

Finally, consider some $s \in S \setminus S_f$, and recall the elements w, s' chosen in §3.1. Then by definition we have

$$\Xi_{s,!}^{\wedge} = \Delta_w^{\wedge} \,\widehat{\star} \, \Xi_{s',!}^{\wedge} \,\widehat{\star} \, \nabla_{w^{-1}}^{\wedge}.$$

Using Proposition 9.6 and monoidality of Φ_{I_u,I_u} we deduce for any $m \ge 1$ an isomorphism

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\Xi^{\wedge}_{s,!})) \cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\Delta^{\wedge}_{w})) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\Xi^{\wedge}_{s',!})) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\nabla^{\wedge}_{w^{-1}})).$$

Here we know that $\Phi_{I_u,I_u}(C_m^0(\Xi_{s',!}^\wedge)) \cong \mathsf{D}_m(\mathscr{B}_{s'}^\wedge)$. On the other hand, by Lemma 11.7 we have isomorphisms

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\Delta^{\wedge}_{w})) \cong \mathsf{D}_{m}(\mathscr{M}^{\wedge}_{w}), \quad \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathsf{C}^{0}_{m}(\nabla^{\wedge}_{w^{-1}})) \cong \mathsf{D}_{m}(\mathscr{M}^{\wedge}_{w^{-1}}).$$

We deduce isomorphisms

$$\Phi_{\mathrm{I}_{u},\mathrm{I}_{u}}(\mathsf{C}_{m}^{0}(\Xi_{s,!}^{\wedge})) \cong \mathsf{D}_{m}(\mathscr{M}_{w}^{\wedge}) \circledast \mathsf{D}_{m}(\mathscr{B}_{s'}^{\wedge}) \circledast \mathsf{D}_{m}(\mathscr{M}_{w^{-1}}^{\wedge}).$$

Using Lemma 10.2 and the definition (see (3.4)), one sees that the right-hand side identifies with $\mathsf{D}_m(\mathscr{B}^\wedge_s)$, which concludes the proof.

Remark 11.9. The objects $\Xi_{s,!}^{\wedge}$ considered above have been defined in a canonical way for $s \in S_{\rm f}$, and in this case the isomorphism in Theorem 11.2(1) is canonical. But we do not know any canonical construction in case $s \in S \setminus S_{\rm f}$, and the objects Δ_{ω}^{\wedge} ($\omega \in \Omega$) are defined only up to isomorphism. However, one can always fix some objects $\Xi_s^{\wedge,{\rm can}}$ ($s \in S$) and $\Delta_{\omega}^{\wedge,{\rm can}}$ ($\omega \in \Omega$) in $\mathsf{T}_{{\rm I}_{\rm u},{\rm I}_{\rm u}}^{\wedge}$ and identifications

$$\Phi_{\mathsf{T}}(\Xi^{\wedge,\mathrm{can}}_s) = \mathscr{B}^{\wedge}_s, \qquad \Phi_{\mathsf{T}}(\Delta^{\wedge,\mathrm{can}}_{\omega}) = \mathscr{M}^{\wedge}_{\omega}$$

for $s \in S \setminus S_{\rm f}$ and $\omega \in \Omega$. Using Lemma 3.5 and monoidality of $\Phi_{\rm T}$, we deduce canonical isomorphisms

(11.8)
$$\Delta_{\omega}^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \Xi_s^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \Delta_{\omega^{-1}}^{\wedge,\operatorname{can}} \cong \Xi_{\omega s \omega^{-1}}^{\wedge,\operatorname{can}}$$

for any $s \in S$ and $\omega \in \Omega$ One can then define the category $\mathsf{T}_{\mathrm{I}_{n},\mathrm{I}_{n}}^{\wedge,\mathrm{BS}}$ with

- objects the collections $(\omega, s_1, \cdots, s_i)$ with $\omega \in \Omega$ and $s_1, \cdots, s_i \in S$;
- morphisms from $(\omega, s_1, \cdots, s_i)$ to $(\omega', s'_1, \cdots, s'_i)$ given by

$$\operatorname{Hom}_{\mathsf{T}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\wedge}}(\Delta_{\omega}^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \Xi_{s_{1}}^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \cdots \mathbin{\widehat{\star}} \Xi_{s_{i}}^{\wedge,\operatorname{can}}, \Delta_{\omega'}^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \Xi_{s_{1}'}^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \cdots \mathbin{\widehat{\star}} \Xi_{s_{j}'}^{\wedge,\operatorname{can}}).$$

Using the isomorphisms (11.8) one can define on $\mathsf{T}_{\mathrm{I}_{u},\mathrm{I}_{u}}^{\wedge,\mathrm{BS}}$ a canonical monoidal structure, such that we have a canonical equivalence of monoidal categories

(11.9)
$$T^{\wedge,\mathrm{BS}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}} \xrightarrow{\sim} \mathsf{BSRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$$

which is the identity on objects, where $\mathsf{BSRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$ is as in §3.3 (for $\mathbf{G} = G^{\vee}_{\Bbbk}$). We also have a canonical fully faithful monoidal functor

$$\mathsf{T}^{\wedge,\mathrm{BS}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$$

sending $(\omega, s_1, \dots, s_i)$ to $\Delta_{\omega}^{\wedge, \operatorname{can}} \widehat{\star} \Xi_{s_1}^{\wedge, \operatorname{can}} \widehat{\star} \dots \widehat{\star} \Xi_{s_i}^{\wedge, \operatorname{can}}$, and $\mathsf{T}_{\mathrm{I}_{\mathrm{u}}, \mathrm{I}_{\mathrm{u}}}^{\wedge}$ identifies with the Karoubian closure of the additive hull of $\mathsf{T}_{\mathrm{I}_{\mathrm{u}}, \mathrm{I}_{\mathrm{u}}}^{\wedge, \mathrm{BS}}$.

12. The case of Fl_G

We continue with the assumptions of Sections 10–11. In this section we briefly indicate how the constructions of Sections 8, 10 and 11 can be adapted to give a description of the more familiar category of tilting I_u -equivariant perverse sheaves on Fl_G .

12.1. The case of G/B. We start with the analogue of Theorem 8.3. Recall the categories $\mathsf{P}_{U,B}$ and $\mathsf{P}_{U,B}^0$ considered in §10.12, and denote by $\Pi_{U,B}^0 : \mathsf{P}_{U,B} \to \mathsf{P}_{U,B}^0$ the quotient functor. We can also consider the Verdier quotient of the category $\mathsf{D}_{U,B}$ from the proof of Lemma 6.10 by the Sere subcategory generated by simple perverse sheaves $\mathsf{For}_{I_u}^{\mathrm{I}}(\mathscr{IC}_w)$ with $w \in W_{\mathrm{f}} \smallsetminus \{e\}$. The functor $\pi^{\dagger,0}$ from §5.3 is t-exact, and it restricts to a functor $\mathsf{D}_{U,B}^0 \to \mathsf{D}_{U,U}^0$; it therefore restricts to an exact functor

(12.1)
$$\mathsf{P}^0_{U,B} \to \mathsf{P}^0_{U,U},$$

which is moreover fully faithful (because so is the restriction of π^{\dagger} to perverse sheaves). The same construction as for the convolution product ${}^{p}\star^{0}_{U}$ (see §8.1) provides a left action of the category $\mathsf{P}^{0}_{U,U}$ on $\mathsf{P}^{0}_{U,B}$, via a bifunctor also denoted ${}^{p}\star^{0}_{U}$, and which is right exact on both sides.

On the other hand, consider the finite k-scheme

$$\mathbf{E} := \mathbf{D} \times_{T_{\Bbbk}^{\vee}} \{e\} = T_{\Bbbk}^{\vee} \times_{T_{\Bbbk}^{\vee}/W_{\mathrm{f}}} \{e\}$$

where the morphism $\mathbf{D} \to T_{\Bbbk}^{\vee}$ is the projection on the second factor and $e \in T_{\Bbbk}^{\vee}$ is the unit element. The closed embedding $\mathbf{E} \hookrightarrow \mathbf{D}$ induces an exact fully faithful functor

(12.2)
$$\operatorname{Coh}(\mathbf{E}) \to \operatorname{Coh}_0(\mathbf{D})$$

We also have a natural left action of $\mathsf{Coh}_0(\mathbf{D})$ on $\mathsf{Coh}(\mathbf{E})$, via a bifunctor denoted \circledast .

Theorem 12.1. There exists a canonical equivalence of abelian categories

$$\Phi_{U,B}: \mathsf{P}^0_{U,B} \xrightarrow{\sim} \mathsf{Coh}(\mathbf{E})$$

such that the following diagram commutes up to isomorphism:

$$\begin{array}{c|c}
\mathsf{P}_{U,B}^{0} & \xrightarrow{(12.1)} & \mathsf{P}_{U,U}^{0} \\
 & \Phi_{U,B} & \downarrow & \downarrow & \downarrow \\
 & \Phi_{U,B} & \downarrow & \downarrow & \downarrow \\
 & & \downarrow & \downarrow & \downarrow \\
 & & \mathsf{Coh}(\mathbf{E}) & \xrightarrow{(12.2)} & \mathsf{Coh}_{0}(\mathbf{D}).
\end{array}$$

Moreover, this equivalence intertwines the actions of $\mathsf{P}^0_{U,U}$ on $\mathsf{P}^0_{U,B}$ and of $\mathsf{Coh}_0(\mathbf{D})$ on $\mathsf{Coh}(\mathbf{E})$ via the equivalence $\Phi_{U,U}$.

Proof. Recall the object $\Xi_!$ from §6.6, which is a projective cover of the simple object δ in $\mathsf{P}_{U,B}$. The object $\Pi^0_{U,B}(\Xi_!)$ is then the projective cover of the unique simple object in $\mathsf{P}^0_{U,B}$, so that the functor $\operatorname{Hom}_{\mathsf{P}^0_{U,B}}(\Pi^0_{U,B}(\Xi_!), -)$ induces an equivalence of abelian categories

$$\mathsf{P}^0_{U,B} \xrightarrow{\sim} \mathsf{Mod}^{\mathrm{fg}}_{\mathrm{r}}(\mathrm{End}_{\mathsf{P}^0_{U,B}}(\Pi^0_{U,B}(\Xi_!))).$$

Now, a much simplified version of the proof of Lemma 8.2 shows that for any \mathscr{F} in $\mathsf{P}_{U,B}$ the morphism

$$\operatorname{Hom}_{\mathsf{P}_{U,B}}(\Xi_{!},\mathscr{F}) \to \operatorname{Hom}_{\mathsf{P}_{U,B}^{0}}(\Pi_{U,B}^{0}(\Xi_{!}), \Pi_{U,B}^{0}(\mathscr{F}))$$

induced by $\Pi^0_{U,B}$ is an isomorphism. In particular, we have an algebra isomorphism

$$\operatorname{End}_{\mathsf{P}_{U,B}}(\Xi_!) \xrightarrow{\sim} \operatorname{End}_{\mathsf{P}_{U,B}^0}(\Pi_{U,B}^0(\Xi_!)).$$

By [BR1, Corollary 9.2], the left-hand side identifies (via monodromy) with $\mathscr{O}(\mathbf{E})$, which provides the desired equivalence $\Phi_{U,B}$.

The compatibility with $\Phi_{U,U}$ can be checked using adjunction and (10.29). The compatibility with the action of $\mathsf{P}^0_{U,U}$ follows from (a simplified version of) the same arguments as for the monoidality of $\Phi_{U,U}$.

12.2. Description of the regular quotient. We now set

$$\mathbb{J}_{\mathbf{E}} := \mathbf{E} \times_{\mathbf{D}} \mathbb{J}_{\mathbf{D}}.$$

Since the fiber of \mathbb{J}_{Σ} over the image of e in $T_{\Bbbk}^{\vee}/W_{\mathrm{f}}$ is $Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})$, we in fact have $\mathbb{J}_{\mathbf{E}} \cong \mathbf{E} \times Z_{G_{\Bbbk}^{\vee}}(\mathsf{u})$ as group schemes over \mathbf{E} . We will consider the abelian category $\mathsf{Rep}(\mathbb{J}_{\mathbf{E}})$ of representations of $\mathbb{J}_{\mathbf{E}}$ on coherent $\mathscr{O}_{\mathbf{E}}$ -modules. Pushforward along the embeddings

$$\{(e,e)\} \hookrightarrow \mathbf{E} \hookrightarrow \mathbf{D}$$

provides exact and fully faithful functors

(12.3)
$$\operatorname{\mathsf{Rep}}(Z_{G_{\mathbb{L}}^{\vee}}(\mathsf{u})) \to \operatorname{\mathsf{Rep}}(\mathbb{J}_{\mathbf{E}}) \to \operatorname{\mathsf{Rep}}_{0}(\mathbb{J}_{\mathbf{D}})$$

whose composition is (10.2). We also have a canonical left action of $\operatorname{Rep}_0(\mathbb{J}_{\mathbf{D}})$ on $\operatorname{Rep}(\mathbb{J}_{\mathbf{E}})$, and a canonical right action of $\operatorname{Rep}(Z_{G_{\Bbbk}^{\vee}}(\mathsf{u}))$ on $\operatorname{Rep}(\mathbb{J}_{\mathbf{E}})$; the corresponding bifunctors will be denoted \circledast and \otimes respectively. (The latter bifunctor is exact on both sides, but the former one is only right exact on both sides.) Finally, we have a natural exact fully faithful functor

$$(12.4) \qquad \qquad \mathsf{Coh}(\mathbf{E}) \to \mathsf{Rep}(\mathbb{J}_{\mathbf{E}})$$

sending a coherent sheaf to itself with the trivial structure as a representation.

On the other hand, recall from §4.7 the category $D^0_{I_u,I}$, and the heart $P^0_{I_u,I}$ of its perverse t-structure. We have canonical exact and fully faithful functors

$$\mathsf{P}_{\mathrm{I},\mathrm{I}}^{0} \xrightarrow{\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I},0}} \mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}^{0} \xrightarrow{\pi^{\dagger,0}} \mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{0}.$$

As explained in §4.7, we also have a natural right action of $\mathsf{P}^0_{\mathrm{I},\mathrm{I}}$ on $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}}$ (via a bifunctor denoted \star^0_{I} , which is exact on both sides), and as for the construction of the convolution product ${}^{\mathrm{p}}\star^0_{\mathrm{I}_u}$ (see §5.2) we have a natural left action of $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}_u}$ on $\mathsf{P}^0_{\mathrm{I}_u,\mathrm{I}}$, via a bifunctor also denoted ${}^{\mathrm{p}}\star^0_{\mathrm{I}_u}$ (and which is also right exact on both sides). We have a natural embedding $G/B = \overline{\mathrm{Fl}_{G,w_o}} \subset \mathrm{Fl}_G$, which provides via pushforward an exact and fully faithful functor

$$(12.5) \qquad \qquad \mathsf{P}^{\mathsf{0}}_{U,B} \to \mathsf{P}^{\mathsf{0}}_{\mathrm{I}_{u},\mathrm{I}}$$

Theorem 12.2. There exists a canonical equivalence of abelian categories

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}:\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\xrightarrow{\sim}\mathsf{Rep}(\mathbb{J}_{\mathbf{E}})$$

such that the following diagrams commute (up to isomorphisms), where on the lefthand side the lower horizontal row is as in (12.3):

Moreover, this equivalence intertwines the actions of $\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and $\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ on $\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ and of $\mathsf{Rep}(Z_{G_{\mathrm{v}}^{\vee}}(\mathsf{u}))$ and $\mathsf{Rep}_{0}(\mathbb{J}_{\mathbf{D}})$ on $\mathsf{Rep}(\mathbb{J}_{\mathbf{E}})$ via the equivalences $\Phi_{\mathrm{I},\mathrm{I}}$ and $\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

12.3. (Sketch of) proof of Theorem 12.2. The same considerations as in §6.4 provide a left action of the category $D^{\wedge}_{I_u,I_u}$ on $D^0_{I_u,I}$, via a bifunctor denoted $\hat{\star}^0$, and which satisfies

(12.6)
$$\pi^{\dagger,0}(\mathscr{F} \widehat{\star}^0 \mathscr{G}) \cong \mathscr{F} \widehat{\star}^0 (\pi^{\dagger,0} \mathscr{G})$$

for any \mathscr{F} in $\mathsf{D}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and \mathscr{G} in $\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$. Taking 0-th perverse cohomology we then define ${}^{p}\widehat{\star}^{0}$ as in the setting of $\mathsf{P}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$.

One can deduce from Lemma 6.7 that the monodromy action of $\mathscr{O}(T_{\Bbbk}^{\vee})$ on any object in $\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}}$ (induced by the *T*-action on Fl_{G} by multiplication on the left) factors through an action of $\mathscr{O}(\mathbf{E})$. We can then define the functor $\Phi_{\mathrm{I}_{u},\mathrm{I}}$, initially with values in $\mathscr{O}(\mathbf{E})$ -modules, by setting

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\mathscr{F}) = \mathrm{Hom}_{\mathsf{D}^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\Pi^{0}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi_{!}), \mathscr{R}^{\wedge p}\widehat{\star}^{0} \mathscr{F}).$$

(Here $\Pi^0_{I_u,I}(\Xi_!)$ is a honest object in $\mathsf{D}^0_{I_u,I}$, and $\mathscr{R}^{\wedge p} \mathscr{F}^0 \mathscr{F}$ is an ind-object in $\mathsf{D}^0_{I_u,I}$; morphisms are taken in the category $\operatorname{Ind}-\mathsf{D}^0_{I_u,I}$.) Using (12.6), exactness of $\pi^{\dagger,0}$, adjunction and (10.29) one sees that for \mathscr{F} in $\mathsf{P}^0_{I_u,I}$ we have a canonical isomorphism

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\pi^{\dagger,0}(\mathscr{F})) \cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\mathscr{F}).$$

This implies that $\Phi_{I_u,I}$ is exact and takes values in finite-dimensional $\mathscr{O}(\mathbf{E})$ -modules. The same considerations as in §10.8 can be used to endow $\Phi_{I_u,I}(\mathscr{F})$ with the structure of a representation of $\mathbb{J}_{\mathbf{E}}$, so that we have in fact constructed an exact functor

$$\Phi_{I_u,I}: \mathsf{P}^0_{I_u,I} \to \mathsf{Rep}(\mathbb{J}_{\mathbf{E}}).$$

The considerations above show that the left-hand diagram in Theorem 12.2 commutes. One can next check commutativity of the right-hand diagram as in Lemma 10.19.

Using the comparison with Φ_{I_u,I_u} we see also that $\Phi_{I_u,I}$ is fully faithful. Essential surjectivity can be checked as in §10.13: namely, using pushforward to **D**, Lemma 10.26, and then pullback to **E**, one sees that any object in $\text{Rep}(\mathbb{J}_{\mathbf{E}})$ is a quotient of an object of the form $V \otimes \mathscr{O}(\mathbf{E})$ with V in $\text{Rep}(G_{\mathbb{k}}^{\vee})$. This allows to conclude since

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\widehat{\mathscr{Z}}(V)\,\widehat{\star}^0\,\Pi^0_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\Xi_!))\cong V\otimes\mathscr{O}(\mathbf{E}).$$

Finally, compatibility with the actions can be checked by considerations similar to those used to prove monoidality of Φ_{I_u,I_u} (see §10.6 and §10.14).

12.4. Description of tilting perverse sheaves. We finally explain how to adapt Theorem 11.2 to the present setting. We will denote by $T_{I_u,I}$ the subcategory of tilting objects in the highest weight category $P_{I_u,I}$. We have a natural functor

$$\pi_{\dagger}:\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}\to\mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}},$$

and an action of the category $\mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}^{\wedge}$ on $\mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ (via $\widehat{\star}$), from which the functor π_{\dagger} can be recoved as the action on the object $\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\delta)$. Standard considerations show that $\mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ is the smallest additive full subcategory of $\mathsf{P}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ that contains $\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\delta)$ and is stable under direct summands and the action of the objects $\Xi_{s,!}^{\wedge}$ ($s \in S$) and Δ_{ω}^{\wedge} ($\omega \in \Omega$).

The following claim follows from [BR1, Proposition 5.9(2)].

Lemma 12.3. For any \mathscr{F}, \mathscr{G} in $\mathsf{T}^{\wedge}_{\mathsf{I}_{\mathsf{u}},\mathsf{I}_{\mathsf{u}}}$, the functor π_{\dagger} induces an isomorphism

$$\operatorname{Hom}_{\mathsf{T}_{\operatorname{I}_{u}}^{\wedge}}(\mathscr{F},\mathscr{G})\otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \Bbbk \xrightarrow{\sim} \operatorname{Hom}_{\mathsf{T}_{\operatorname{I}_{u},\operatorname{I}}}(\pi_{\dagger}(\mathscr{F}),\pi_{\dagger}(\mathscr{G})),$$

where the action of $\mathscr{O}(T^{\vee}_{\Bbbk})$ is via monodromy associated with the right action of T on $\widetilde{\mathrm{Fl}}_G$, and \Bbbk is viewed as an $\mathscr{O}(T^{\vee}_{\Bbbk})$ -module via evaluation at $e \in T^{\vee}_{\Bbbk}$.

On the other hand, we have a natural functor

$$\mathsf{Rep}(\mathbb{J}^{\wedge}_{\mathbf{D}}) \to \mathsf{Rep}(\mathbb{J}_{\mathbf{E}})$$

given by restriction to $\mathbf{E} \subset \mathbf{D}^{\wedge}$. We also have an action of $\mathsf{Rep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$ on $\mathsf{Rep}(\mathbb{J}_{\mathbf{E}})$ by convolution (the corresponding bifunctor will once again be denoted \circledast), and the functor above is given by the action on the skyscraper sheaf at the base point. We will denote by $\mathsf{SRep}(\mathbb{J}_{\mathbf{E}})$ the full additive subcategory of $\mathsf{Rep}(\mathbb{J}_{\mathbf{E}})$ generated under direct sums and direct summands by the image of $\mathsf{SRep}(\mathbb{J}_{\mathbf{D}})$ under this functor. In other words, $\mathsf{SRep}(\mathbb{J}_{\mathbf{E}})$ is the smallest Karoubian additive subcategory of $\mathsf{Rep}(\mathbb{J}_{\mathbf{E}})$ containing the skyscraper sheaf at the base point and stable under action by the objects \mathscr{B}_{s}^{\wedge} ($s \in S$) and $\mathscr{M}_{\omega}^{\wedge}$ ($\omega \in \Omega$).

Theorem 12.4. There exists a canonical equivalence of additive categories

$$\Psi_{\mathsf{T}}:\mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\xrightarrow{}\mathsf{SRep}(\mathbb{J}_{\mathbf{E}})$$

which sends $\operatorname{For}_{I_{u},I_{u}}^{I}(\delta)$ to the skyscraper sheaf at the base point and intertwines the actions of $\mathsf{T}_{I_{u},I_{u}}^{\wedge}$ on $\mathsf{T}_{I_{u},I}$ and $\operatorname{SRep}(\mathbb{J}_{\mathbf{D}}^{\wedge})$ on $\operatorname{SRep}(\mathbb{J}_{\mathbf{E}})$ via the equivalence Φ_{T} .

Proof. It is a standard fact that the quotient functor $\Pi^0_{I_u,I}$ is fully faithful on the subcategory $\mathsf{T}_{I_u,I}$ (see [BBM, §2.1] for the similar claim on G/B). Since the equivalence $\Phi^0_{I_u,I}$ of Theorem 12.2 sends $\mathsf{For}^{I}_{I_u}(\delta)$ to the skyscraper sheaf at the base point, to prove the theorem it therefore suffices to check that this functor satisfies

$$\Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi^{\wedge}_{s,!} \stackrel{\mathrm{p}_{\widehat{\star}}^{0}}{\mathscr{F}}) \cong \mathscr{B}^{\wedge}_{s} \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\mathscr{F}), \quad \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Delta^{\wedge}_{\omega} \stackrel{\mathrm{p}_{\widehat{\star}}^{0}}{\mathscr{F}}) \cong \mathscr{M}^{\wedge}_{\omega} \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\mathscr{F})$$

for $s \in S$, $\omega \in \Omega$ and \mathscr{F} in $\mathsf{P}^{0}_{\mathrm{I}_{u},\mathrm{I}}$. However, \mathcal{J} acts trivially on \mathscr{F} . As in the proof of Lemma 9.5, one can check that we have a canonical isomorphism

$$\Xi_{s,!}^{\wedge} \stackrel{\mathrm{p}\widehat{\star}^0}{\mathscr{F}} \cong \mathsf{C}_1^0(\Xi_{s,!}^{\wedge}) \stackrel{\mathrm{p}}{\star}^0_{\mathrm{I}_{\mathrm{u}}} \mathscr{F}.$$

It follows that

$$\begin{split} \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\Xi_{s,!}^{\wedge} \stackrel{\mathrm{p}_{\widehat{\star}}^{0}}{\mathscr{F}}) &\cong \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}(\mathrm{C}_{1}^{0}(\Xi_{s,!}^{\wedge})) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\mathscr{F}) \\ &\cong \mathrm{D}_{1}(\mathscr{B}_{s}^{\wedge}) \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\mathscr{F}) \cong \mathscr{B}_{s}^{\wedge} \circledast \Phi_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}(\mathscr{F}), \end{split}$$

which proves the first isomorphism. The second one can be checked similarly. \Box

Remark 12.5. Comparing Theorem 11.2 and Theorem 12.4, and using Lemma 12.3, we see that for any M, N in $\mathsf{SRep}(\mathbb{J}^{\wedge}_{\mathbf{D}})$, there exists a canonical isomorphism

$$\operatorname{Hom}_{\mathsf{SRep}(\mathbb{J}_{\mathbf{D}}^{\wedge})}(M,N) \otimes_{\mathscr{O}(T_{\Bbbk}^{\vee})} \mathbb{k} \xrightarrow{\sim} \operatorname{Hom}_{\mathsf{SRep}(\mathbb{J}_{\mathbf{E}})}(M_{|\mathbf{E}},N_{|\mathbf{E}}).$$

Such a property is standard in the usual theory of Soergel bimodules, see e.g. [R3, Proposition 1.13].

13. Applications

We continue with the assumptions of Sections 10–11–12, and assume furthermore either that $G = GL_n$ or that ℓ is very good for G.

13.1. Parity complexes and ℓ -Kazhdan–Lusztig polynomials. We consider the category $D_{I,I}$ from §4.2, with its convolution product \star_I , and the notion of *parity complexes* in this category from [JMW]. The full subcategory of $D_{I,I}$ whose objects are the parity complexes will be denoted $Par_{I,I}$. This subcategory has a more "concrete" description as follows. For any $s \in S$, the simple perverse sheaf \mathscr{I}_s^I is just the constant sheaf on the smooth variety $\overline{Fl}_{G,s}$, shifted by 1; in particular it is a parity complex. On the other hand, if $\omega \in \Omega$ then the orbit $Fl_{G,\omega}$ is just a point; in particular, \mathscr{I}_{ω}^{I} is the skyscraper sheaf at that point, and is also a parity complex. We will denote by Par_{LI}^{BS} the category with

- objects the collections $(s_1, \dots, s_i, \omega, n)$ with $s_1, \dots, s_i \in S$, $\omega \in \Omega$ and $n \in \mathbb{Z}$;
- morphisms from $(s_1, \dots, s_i, \omega, n)$ to $(s'_1, \dots, s'_j, \omega', n')$ given by

$$\operatorname{Hom}_{\mathsf{Par}_{\mathrm{I},\mathrm{I}}}(\mathscr{I}\!\mathscr{C}^{\mathrm{I}}_{s_{1}}\star_{\mathrm{I}}\cdots\star_{\mathrm{I}}\mathscr{I}\!\mathscr{I}\!\mathscr{C}^{\mathrm{I}}_{s_{i}}\star_{\mathrm{I}}\mathscr{I}\mathscr{I}_{\omega}^{\mathrm{I}}[n],\mathscr{I}\!\mathscr{C}^{\mathrm{I}}_{s_{1}'}\star_{\mathrm{I}}\cdots\star_{\mathrm{I}}\mathscr{I}\!\mathscr{I}\!\mathscr{C}^{\mathrm{I}}_{s_{j}'}\star_{\mathrm{I}}\mathscr{I}\mathscr{I}\!\mathscr{I}_{\omega'}^{\mathrm{I}}[n']).$$

By definition there exists a canonical fully faithful functor

(13.1)
$$\operatorname{Par}_{\mathrm{LI}}^{\mathrm{BS}} \to \mathsf{D}_{\mathrm{I},\mathrm{I}}.$$

It is easily seen that for any $\omega \in \Omega$ and $s \in S$ there is a canonical isomorphism

$$\mathscr{I}\!\!\mathscr{C}^{\mathrm{I}}_{\omega}\star_{\mathrm{I}}\mathscr{I}\!\!\mathscr{C}^{\mathrm{I}}_{s}\star_{\mathrm{I}}\mathscr{I}\!\!\mathscr{C}^{\mathrm{I}}_{\omega^{-1}}\cong \mathscr{I}\!\!\mathscr{C}^{\mathrm{I}}_{\omega s \omega^{-1}}.$$

Using this property one obtains that there exists a natural convolution product (still denoted \star_I) on $\mathsf{Par}_{I,I}^{\mathrm{BS}}$ which is defined on objects by

$$(s_1, \cdots, s_i, \omega, n) \star_{\mathbf{I}} (s'_1, \cdots, s'_j, \omega', n') = (s_1, \cdots, s_i, \omega s'_1 \omega^{-1}, \cdots, \omega s'_j \omega^{-1}, \omega \omega', n+n')$$

and such that (13.1) is monoidal. For any $n \in \mathbb{Z}$ the cohomological shift functor [n] induces an autoequivalence of $\mathsf{Par}_{I,I}^{\mathrm{BS}}$, which will again be denoted [n].

It is well known that the category $D_{I,I}$ is Krull–Schmidt, and that an object in $D_{I,I}$ is a parity complex if and only if it is a direct sum of direct summands of objects of $\mathsf{Par}_{I,I}^{\mathrm{BS}}$. In other words, the functor (13.1) identifies $\mathsf{Par}_{I,I}$ with the Karoubian envelope of the additive hull of the category $\mathsf{Par}_{I,I}^{\mathrm{BS}}$.

The theory developed in [JMW] provides a classification of the indecomposable objects in $\mathsf{Par}_{\mathrm{I},\mathrm{I}}$. More specifically, for any $w \in W$ there exists a unique (up to isomorphism) indecomposable object \mathscr{E}_w in $\mathsf{Par}_{\mathrm{I},\mathrm{I}}$ which is supported on $\overline{\mathrm{Fl}}_{G,w}$ and whose restriction to $\mathrm{Fl}_{G,w}$ is $\underline{\Bbbk}_{\mathrm{Fl}_{G,w}}[\ell(w)]$. Then the assignment $(w,n) \mapsto \mathscr{E}_w[n]$ induces a bijection between $W \times \mathbb{Z}$ and the set of isomorphism classes of indecomposable objects in $\mathsf{Par}_{\mathrm{I},\mathrm{I}}$.

Recall the category $\mathsf{BSK}_{\mathrm{add}}$ constructed in §3.6, which we consider here in the case when $\mathbf{G} = G_{\Bbbk}^{\vee}$ (with the usual choice of Borel subgroup and maximal torus). In fact, we will rather consider the "right" variant of this category constructed as in Remark 3.9, which we will denote $\mathsf{BSK}_{\mathrm{r,add}}$. It is a standard fact that we have identifications

$$\mathsf{H}^{\bullet}_{\mathsf{I}}(\mathsf{pt}; \Bbbk) = \mathsf{H}^{\bullet}_{\mathcal{T}}(\mathsf{pt}; \Bbbk) = \mathsf{S}(\Bbbk \otimes_{\mathbb{Z}} X^{*}(T))$$

where S denotes the symmetric algebra (over \Bbbk) and the right-hand side is seen as a graded ring with $\Bbbk \otimes_{\mathbb{Z}} X^*(T)$ in degree 2. Moreover $\Bbbk \otimes_{\mathbb{Z}} X^*(T)$ identifies canonically with the Lie algebra **t** of T_{\Bbbk}^{\vee} ; in this way, $\mathsf{H}^{\bullet}_{\mathsf{I}}(\mathsf{pt}; \Bbbk)$ identifies with the graded algebra $\mathscr{O}(\mathbf{t}^*)$ considered in §3.6. The category $\mathsf{BSK}_{r,\mathrm{add}}$ is related to $\mathsf{Par}_{I,I}^{\mathrm{BS}}$ as follows.

Theorem 13.1. There exists a canonical equivalence of monoidal categories

$$\mathsf{BSK}_{\mathrm{r,add}} \cong \mathsf{Par}_{\mathrm{I,I}}^{\mathrm{BS}}$$

which intertwines the shift functors (1) and [1], and is the identity on objects.

Proof. This theorem is essentially obtained as the combination of [RW1, Theorem 10.7.1] and the main result of [Ab1]. More precisely, these references provide a canonical equivalence of monoidal categories with the expected properties between the full subcategories in $\mathsf{BSK}_{r,\mathrm{add}}$ and $\mathsf{Par}_{I,I}^{\mathrm{BS}}$ whose objects are of the form (s_1, \cdots, s_i, e) with $s_1, \cdots, s_i \in S$. However, it is easily seen that this equivalence intertwines, for any $\omega \in \Omega$, the equivalences given by

$$M \mapsto F_{\omega} \star M \star F_{\omega^{-1}}$$
 and $\mathscr{F} \mapsto \mathscr{I}\mathscr{C}^{\mathrm{I}}_{\omega} \star_{\mathrm{I}} \mathscr{F} \star_{\mathrm{I}} \mathscr{I}\mathscr{C}^{\mathrm{I}}_{\omega^{-1}}.$

Using this property one sees that the equivalence above can be "extended" to the equivalence of the theorem. $\hfill \Box$

Remark 13.2. A different (and more direct) proof of Theorem 13.1 can be obtained following the constructions in [Ab2, §3]. We will not pursue this here.

We also have similar notions in the category $D_{I_u,I}$; by definition, an object \mathscr{F} in $D_{I,I}$ is a parity complex if and only if $\mathsf{For}_{I_u}^{I}(\mathscr{F})$ is a parity complex. If we denote by $\mathsf{Par}_{I_u,I}^{\mathrm{BS}}$ the category with

- objects the collections $(s_1, \dots, s_i, \omega, n)$ with $s_1, \dots, s_i \in S$, $\omega \in \Omega$ and $n \in \mathbb{Z}$;
- morphisms from $(s_1, \dots, s_i, \omega, n)$ to $(s'_1, \dots, s'_i, \omega', n')$ given by

and by $\mathsf{Par}_{I_u,I}$ the full subcategory of $\mathsf{D}_{I_u,I}$ whose objects are the parity complexes, then $\mathsf{Par}_{I_u,I}$ identifies with the Karoubian envelope of the additive hull of the category $\mathsf{Par}_{I_u,I}^{\mathrm{BS}}$.

The right action of the category $\mathsf{D}_{\mathrm{I},\mathrm{I}}$ on $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ (by convolution) induces a right action of $\mathsf{Par}_{\mathrm{I},\mathrm{I}}^{\mathrm{BS}}$ on $\mathsf{Par}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}^{\mathrm{BS}}$, and of $\mathsf{Par}_{\mathrm{I},\mathrm{I}}$ on $\mathsf{Par}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}^{\mathrm{L}}$. The corresponding bifunctors will again be denoted \star_{I} . For any $n \in \mathbb{Z}$ the cohomological shift functor [n] induces an autoequivalence of $\mathsf{Par}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}^{\mathrm{BS}}$, which will again be denoted [n].

If D is one of the categories $D_{I,I}$, $D_{I_u,I}$, $\mathsf{Par}_{I,I}$, $\mathsf{Par}_{I_u,I}$, $\mathsf{Par}_{I,I}^{\mathrm{BS}}$ or $\mathsf{Par}_{I_u,I}^{\mathrm{BS}}$ and \mathscr{F}, \mathscr{G} are objects in D, then we will set

$$\operatorname{Hom}_{\mathsf{D}}^{\bullet}(\mathscr{F},\mathscr{G}) = \bigoplus_{n \in \mathbb{Z}} \operatorname{Hom}_{\mathsf{D}}(\mathscr{F},\mathscr{G}[n]).$$

(Depending on the context, this space will be considered either as a graded vector space, or a plain vector space.) We will see \Bbbk as a graded $H_{I}^{\bullet}(\text{pt}; \Bbbk)$ -module concentrated in degree 0, in the standard way. The following lemma is a particular case of [MR, Lemma 2.2].
Lemma 13.3. For any \mathscr{F}, \mathscr{G} in $\mathsf{Par}_{I,I}$, the functor $\mathsf{For}_{I_u}^l$ induces an isomorphism of graded vector spaces

$$\Bbbk \otimes_{\mathsf{H}^{\bullet}_{I}(\mathrm{pt};\Bbbk)} \mathrm{Hom}^{\bullet}_{D_{I,I}}(\mathscr{F},\mathscr{G}) \xrightarrow{\sim} \mathrm{Hom}^{\bullet}_{D_{I_{u},I}}\big(\mathsf{For}^{I}_{I_{u}}(\mathscr{F}),\mathsf{For}^{I}_{I_{u}}(\mathscr{G})\big).$$

Below we will use the following consequences of this lemma:

(1) the category $\mathsf{Par}_{I_u,I}^{\mathrm{BS}}$ identifies with the category whose objects are those of $\mathsf{Par}_{I,I}^{\mathrm{BS}}$, and whose morphism space from \mathscr{F} to \mathscr{G} is given by the degree-0 part in

$$\mathbb{K} \otimes_{\mathsf{H}^{\bullet}_{\mathsf{I}}(\mathrm{pt};\mathbb{k})} \mathrm{Hom}^{\bullet}_{\mathsf{Par}^{\mathrm{BS}}_{\mathsf{I}}}(\mathscr{F},\mathscr{G});$$

(2) for any $w \in W$, the object $\mathsf{For}_{\mathrm{I}_{u}}^{\mathrm{I}}(\mathscr{E}_{w})$ is indecomposable; as a consequence, the assignment $(w, n) \mapsto \mathsf{For}_{\mathrm{I}_{u}}^{\mathrm{I}}(\mathscr{E}_{w})[n]$ induces a bijection between $W \times \mathbb{Z}$ and the set of isomorphism classes of indecomposable objects in $\mathsf{Par}_{\mathrm{I}_{u},\mathrm{I}}$ (see [MR, Lemma 2.4]).

One possible definition of the ℓ -Kazhdan-Lusztig polynomials attached to W is as follows: for $y, w \in W$ we set

$${}^{\ell}\!h_{y,w}(v) = \sum_{n \in \mathbb{Z}} \dim \mathsf{H}^{-\ell(w)-n} \big(\mathrm{Fl}_{G,y}, \mathscr{E}_{w \,|\mathrm{Fl}_{G,y}} \big) \cdot v^n.$$

(The fact that this definition coincides with that considered e.g. in [JW] follows from the results of [RW1, Part III]. In general, these are *Laurent polynomials* rather than polynomials in the usual sense.)

Below we will use the following standard properties of these polynomials. (For Item (1), see e.g. the proof of [JW, Proposition 4.2(4)]. For (2), see e.g. [JMW, Proposition 2.6].)

Lemma 13.4. (1) For any $w, y \in W$ we have ${}^{\ell}h_{y,w}(v) = {}^{\ell}h_{y^{-1},w^{-1}}(v)$. (2) For any $w, y \in W$ we have

$$\dim \left(\operatorname{Hom}_{\mathsf{Par}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}^{\bullet}(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{E}_{w}),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{E}_{y})) \right) = \sum_{z \in W} {}^{\ell} h_{z,w}(1) \cdot {}^{\ell} h_{z,y}(1).$$

13.2. A degrading functor. Recall the constructions of Remark 11.9. We will denote by $T_{I_{u},I}^{BS}$ the category with

- objects the collections $(\omega, s_1, \cdots, s_i)$ with $\omega \in \Omega$ and $s_1, \cdots, s_i \in S$;
- morphisms from $(\omega, s_1, \dots, s_i)$ to $(\omega', s'_1, \dots, s'_i)$ given by

$$\operatorname{Hom}_{\mathsf{T}_{\mathbf{I}_{u},\mathbf{I}}}\left(\pi_{\dagger}(\Delta_{\omega}^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \Xi_{s_{1}}^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \cdots \mathbin{\widehat{\star}} \Xi_{s_{i}}^{\wedge,\operatorname{can}}), \pi_{\dagger}(\Delta_{\omega'}^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \Xi_{s_{1}'}^{\wedge,\operatorname{can}} \mathbin{\widehat{\star}} \cdots \mathbin{\widehat{\star}} \Xi_{s_{j}'}^{\wedge,\operatorname{can}})\right)$$

In view of the comments in §12.4, we have a canonical fully faithful functor

$$(13.2) \mathsf{T}^{\mathrm{BS}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \to \mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

which identifies $T_{I_u,I}$ with the Karoubian closure of the additive hull of $T_{I_u,I}^{BS}$.

Theorem 13.5. There exist a functors

$$\mathsf{v}:\mathsf{Par}^{\mathrm{BS}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}
ightarrow\mathsf{T}^{\mathrm{BS}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}},$$

and an isomorphism $\varepsilon : \mathbf{v} \circ [1] \xrightarrow{\sim} \mathbf{v}$ such that:

(1) for any \mathscr{F}, \mathscr{G} in $\mathsf{Par}^{\mathrm{BS}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$, the functor v and the isomorphism ε induce an isomorphism

$$\operatorname{Hom}_{\mathsf{Par}_{\operatorname{Iu},\operatorname{I}}}^{\bullet}(\mathscr{F},\mathscr{G}) \xrightarrow{\sim} \operatorname{Hom}_{\mathsf{T}_{\operatorname{Iu},\operatorname{I}}}^{\operatorname{BS}}(\mathsf{v}(\mathscr{F}),\mathsf{v}(\mathscr{G}));$$

(2) for any
$$s_1, \dots, s_i \in S$$
, $\omega \in \Omega$ and $n \in \mathbb{Z}$ we have

$$\mathbf{v}(s_1,\cdots,s_i,\omega,n)=(\omega^{-1},s_i,\cdots,s_1).$$

Proof. Using Theorem 13.1 and the comments after Lemma 13.3 one obtains a canonical equivalence between the category $\mathsf{Par}_{\mathrm{I}_u,\mathrm{I}}^{\mathrm{BS}}$ and the category $\overline{\mathsf{BSK}}_{\mathrm{r,add}}$ defined as follows: its objects are those of $\mathsf{BSK}_{\mathrm{r,add}}$, and the morphisms from M to M' are given by the degree-0 part in

$$\Bbbk \otimes_{\mathscr{O}(\mathbf{t}^*)} \operatorname{Hom}_{\mathsf{BSK}_{\mathrm{r.add}}}^{\bullet}(M, M').$$

This equivalence is the identity on objects.

On the other hand, consider the category BSK^{\wedge} . Using the equivalences (3.15) and (11.9) together with Lemma 12.3, we obtain a canonical equivalence between $\mathsf{T}^{\mathrm{BS}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ and the category $\underline{\mathsf{BSK}}^{\wedge}$ defined as follows: its objects are those of BSK^{\wedge} , and the morphisms from M to M' are given by

$$\operatorname{Hom}_{\mathsf{BSK}^{\wedge}}(M, M') \otimes_{\mathscr{O}((T^{\vee}_{\Bbbk})^{\wedge})} \Bbbk.$$

Once again, this equivalence is the identity on objects.

As explained in Remark 3.9, we have a canonical equivalence of monoidal categories $\mathsf{BSK}_{r,\mathrm{add}} \xrightarrow{\sim} \mathsf{BSK}_{\mathrm{add}}$ sending $(s_1, \cdots, s_i, \omega)$ to $(\omega^{-1}, s_i, \cdots, s_1)$. This equivalence induces an equivalence between $\overline{\mathsf{BSK}}_{r,\mathrm{add}}$ and the category $\underline{\mathsf{BSK}}_{\mathrm{add}}$ which has the same objects as $\mathsf{BSK}_{\mathrm{add}}$, and morphisms from M to M' defined as

 $\operatorname{Hom}_{\mathsf{BSK}_{\mathrm{add}}}(M, M') \otimes_{\mathscr{O}(\mathbf{t}^*)} \Bbbk.$

Therefore, to conclude the proof of Theorem 13.5 it suffices to construct a functor

Λ

(13.3)
$$v_{\mathsf{BSK}} : \underline{\mathsf{BSK}}_{\mathrm{add}} \to \underline{\mathsf{BSK}}$$

sending each collection $(\omega, s_1, \dots, s_i, n)$ to $(\omega, s_1, \dots, s_i)$ and an isomorphism $v_{\mathsf{BSK}} \circ (1) \cong v_{\mathsf{BSK}}$ such that for any M, N in $\underline{\mathsf{BSK}}_{\mathrm{add}}$ these data induce an isomorphism

$$\bigoplus_{n \in \mathbb{Z}} \operatorname{Hom}_{\underline{\mathsf{BSK}}_{\mathrm{add}}}(M, N(n)) \xrightarrow{\sim} \operatorname{Hom}_{\underline{\mathsf{BSK}}^{\wedge}}(\mathsf{v}_{\mathsf{BSK}}(M), \mathsf{v}_{\mathsf{BSK}}(N)).$$

This functor is obtained from Proposition 3.16 and Lemma 3.17.

Remark 13.6. Theorem 13.5 has a variant relating the categories $\mathsf{Par}_{I,I}^{\mathrm{BS}}$ and $\mathsf{T}_{I_{\mathrm{u}},I_{\mathrm{u}}}^{\wedge,\mathrm{BS}}$, and involving the isomorphism appearing in Lemma 3.17. We leave it to the interested reader to formulate this statement, and modify the proof above accordingly.

13.3. Numerical consequence. In this subsection we discuss a "numerical" application of Theorem 13.5. Recall that if \mathscr{F} belongs to $\mathsf{T}_{\mathrm{I}_u,\mathrm{I}}$ and $w \in W$, we denote by $(\mathscr{F} : \Delta_w^{\mathrm{I}})$, resp. $(\mathscr{F} : \nabla_w^{\mathrm{I}})$, the number of occurrences of $\mathsf{For}_{\mathrm{I}_u}^{\mathrm{I}}(\Delta_w^{\mathrm{I}})$, resp. $\mathsf{For}_{\mathrm{I}_u}^{\mathrm{I}}(\nabla_w^{\mathrm{I}})$, in a standard, resp. costandard, filtration of \mathscr{F} . It is well known that these numbers are well defined; in fact we have

$$(\mathscr{F}:\Delta_w^{\mathrm{I}}) = \dim \operatorname{Hom}(\mathscr{F}, \mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\nabla_w^{\mathrm{I}})), \quad (\mathscr{F}:\nabla_w^{\mathrm{I}}) = \dim \operatorname{Hom}(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\Delta_w^{\mathrm{I}}), \mathscr{F}).$$

Using Verdier duality one also sees that for any \mathscr{F} in $\mathsf{T}_{\mathrm{I}_{u},\mathrm{I}}$ and $w \in W$ we have

(13.4)
$$(\mathscr{F}:\Delta^{\mathrm{I}}_{w}) = (\mathscr{F}:\nabla^{\mathrm{I}}_{w}).$$

It is a standard fact that if \mathscr{F}, \mathscr{G} belong to $\mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ we have

$$\dim \operatorname{Hom}(\mathscr{F},\mathscr{G}) = \sum_{w \in W} (\mathscr{F} : \Delta_w^{\mathrm{I}}) \cdot (\mathscr{G} : \nabla_w^{\mathrm{I}}),$$

which in view of (13.4) implies that

(13.5)
$$\dim \operatorname{Hom}(\mathscr{F},\mathscr{G}) = \sum_{w \in W} (\mathscr{F} : \Delta^{\mathrm{I}}_w) \cdot (\mathscr{G} : \Delta^{\mathrm{I}}_w).$$

If \mathscr{F} belongs to $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ and $w \in W$, we will similarly denote by $(\mathscr{F} : \Delta^{\wedge}_{w})$, resp. $(\mathscr{F} : \nabla^{\wedge}_{w})$, the number of occurrences of Δ^{\wedge}_{w} , resp. ∇^{\wedge}_{w} , in a standard, resp. costandard, filtration of \mathscr{F} . To see that these numbers are well defined, one simply notices that if \mathscr{F} is in $\mathsf{T}^{\wedge}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}_{\mathrm{u}}}$ the object $\pi_{\dagger}(\mathscr{F})$ is a tilting perverse sheaf, and that for $w \in W$ we have

(13.6)
$$(\mathscr{F}:\Delta_w^{\wedge}) = (\pi_{\dagger}\mathscr{F}:\Delta_w^{\mathrm{I}}), \quad (\mathscr{F}:\nabla_w^{\wedge}) = (\pi_{\dagger}\mathscr{F}:\nabla_w^{\mathrm{I}}).$$

These equalities and (13.4) also show that

$$(\mathscr{F}:\Delta_w^\wedge) = (\mathscr{F}:\nabla_w^\wedge)$$

for any $w \in W$.

Recall the object \mathscr{T}_w and \mathscr{T}_w^{\wedge} defined in §4.2 and §6.3 respectively.

Corollary 13.7. For any $w, y \in W$ we have

$$(\mathscr{T}_w^\wedge : \Delta_y^\wedge) = {}^\ell h_{y,w}(1).$$

Proof. Passing to Karoubian closures of additive hulls, the functor v of Theorem 13.5 induces a functor

$$\mathsf{Par}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} \to \mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

(still denoted v) which is a "degrading functor" in the sense that it satisfies property (1) of Theorem 13.5. By construction we have

(13.7)
$$\mathsf{v}\big(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{I}\mathscr{C}_{s_{1}}^{\mathrm{I}}\star_{\mathrm{I}}\cdots\star_{\mathrm{I}}\mathscr{I}\mathscr{C}_{s_{i}}^{\mathrm{I}}\star_{\mathrm{I}}\mathscr{I}\mathscr{C}_{\omega}^{\mathrm{I}})\big)\cong\pi_{\dagger}\big(\Delta_{\omega^{-1}}^{\wedge}\widehat{\star}\,\Xi_{s_{i},!}^{\wedge}\widehat{\star}\cdots\widehat{\star}\,\Xi_{s_{1},!}^{\wedge}\big)$$

for any $\omega \in \Omega$ and $s_1, \dots, s_i \in S$. For any $w \in W$, the finite-dimensional graded ring

$$\operatorname{Hom}_{\mathsf{Par}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}^{\bullet}(\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{E}_{w}),\mathsf{For}_{\mathrm{I}_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{E}_{w}))$$

has a local degree-0 part; it is therefore local as an ungraded ring, see [GG]. This observation and the "degrading" property of v show that $v(\operatorname{For}_{I_u}^{I}(\mathscr{E}_w))$ is indecomposable. Once this fact it known, it is not difficult to deduce from (13.7) that for any $w \in W$ we have

$$\mathsf{v}(\mathsf{For}^{\mathrm{I}}_{\mathrm{I}_{u}}(\mathscr{E}_{w})) \cong \mathscr{T}_{w^{-1}}.$$

We deduce that for any $w, y \in W$ we have

$$\dim \left(\operatorname{Hom}_{\mathsf{Par}_{I_{\mathrm{u}},\mathrm{I}}}^{\bullet}(\mathsf{For}_{I_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{E}_{w}),\mathsf{For}_{I_{\mathrm{u}}}^{\mathrm{I}}(\mathscr{E}_{y})) \right) = \dim \left(\operatorname{Hom}_{\mathsf{T}_{I_{\mathrm{u}},\mathrm{I}}}(\mathscr{T}_{w^{-1}},\mathscr{T}_{y^{-1}}) \right).$$

Comparing Lemma 13.4(2) and (13.5), one then deduces (by induction on w, and then by induction on y for fixed w) that for any $w, y \in W$ we have

$$(\mathscr{T}_w : \Delta_y^{\mathrm{I}}) = {}^{\ell} h_{y^{-1}, w^{-1}}(1).$$

Finally, the formula of the corollary follows, using Lemma 13.4(1) and (13.6).

13.4. Koszul duality. Another application of Theorem 13.5 is to an alternative construction of the "modular Koszul duality" of [AMRW], in the special case of affine flag varieties. This construction, based on the ideas of an earlier construction in the setting of ordinary flag varieties of reductive groups [AR2], gives more than the methods of [AMRW]: it also allows to construct a "forgetful functor" relating the "mixed perverse sheaves" of [AR2, AMRW] to ordinary perverse sheaves.

Recall that following [AR2] we define the "mixed derived category" of k-sheaves on Fl_G by setting

$$\mathsf{D}^{\min}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}} := K^{\mathrm{b}}\mathsf{Par}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}.$$

This category admits a "Tate twist" autoequivalence $\langle 1 \rangle$ defined as $\{-1\}[1]$ where $\{-1\}$ is the autoequivalence induced by the negative cohomological shift in the category $\mathsf{Par}_{\mathrm{I}_{u},\mathrm{I}}$, while [1] is the cohomological shift in the homotopy category. The constructions of [AR2, Section 2] endow $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}}^{\mathrm{mix}}$ with a "perverse t-structure" whose heart is a finite-length abelian category, stable under $\langle 1 \rangle$, and which will be denoted $\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}}^{\mathrm{mix}}$. By [AR2, §3.2 and Section 4] the category $\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}}^{\mathrm{mix}}$ admits a natural structure of graded highest weight category, defined by some families of "standard objects" ($\Delta_w^{\mathrm{mix}} : w \in W$) and "costandard objects" ($\nabla_w^{\mathrm{mix}} : w \in W$). In particular there is a notion of tilting objects in $\mathsf{P}_{\mathrm{I}_{u},\mathrm{I}}^{\mathrm{mix}}$, and the indecomposable tilting objects are parametrized in a natural way by $W \times \mathbb{Z}$; see [AR2, Proposition A.4]. For $w \in W$ we will denote by $\mathscr{T}_w^{\mathrm{mix}}$ the indecomposable tilting object associated with (w, 0); then for any $n \in \mathbb{Z}$ the object corresponding to (w, n) is $\mathscr{T}_w^{\mathrm{mix}}\langle n \rangle$. Any object in $\mathsf{Par}_{\mathrm{I}_u,\mathrm{I}}$ can also be seen as an object in $\mathsf{D}_{\mathrm{I}_u,\mathrm{I}}^{\mathrm{mix}}$, by identifying it with a complex concentrated in degree 0. In particular, the image of $\mathsf{For}_{\mathrm{I}_u}^{\mathrm{I}}(\mathscr{E}_w)$ will be denoted $\mathscr{E}_w^{\mathrm{mix}}$.

As terminology and notation suggest, one wants to think of $\mathsf{D}_{I_u,I}^{\min}$ as a "mixed version" of the category $\mathsf{D}_{I_u,I}$, and in fact the results of [AMRW, AR3] show that this category has properties similar to those of the category of mixed $\overline{\mathbb{Q}}_{\ell}$ -sheaves in the sense of Deligne (or, more precisely, a modification considered in [BGS]). However, from its construction we do not have a priori any formal relation between $\mathsf{D}_{I_u,I}^{\min}$ and $\mathsf{D}_{I_u,I}$. Point (2) of the following theorem exactly compensates this discrepancy.

Theorem 13.8. (1) There exists an equivalence of triangulated categories

 $\kappa:\mathsf{D}^{\mathrm{mix}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}\xrightarrow{\sim}\mathsf{D}^{\mathrm{mix}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$

which satisfies $\kappa \circ \langle 1 \rangle \cong \langle -1 \rangle [1] \circ \kappa$ and

$$\begin{aligned} \kappa(\Delta_w^{\min}) &\cong \Delta_{w^{-1}}^{\min}, \qquad \kappa(\nabla_w^{\min}) \cong \nabla_{w^{-1}}^{\min}, \\ \kappa(\mathscr{T}_w^{\min}) &\cong \mathscr{C}_{w^{-1}}^{\min}, \qquad \kappa(\mathscr{C}_w^{\min}) \cong \mathscr{T}_{w^{-1}}^{\min} \end{aligned}$$

for any $w \in W$.

(2) There exists a functor

$$\nu: \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}^{\mathrm{mix}} \to \mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$$

and an isomorphism of functors $\nu \circ \langle 1 \rangle \cong \nu$ such that for any \mathscr{F}, \mathscr{G} in $\mathsf{D}^{\mathrm{mix}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}$ the induced morphism

$$\bigoplus_{n \in \mathbb{Z}} \operatorname{Hom}_{\mathsf{D}^{\operatorname{mix}}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\mathscr{F}, \mathscr{G}\langle n \rangle) \to \operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},\mathrm{I}}}(\nu(\mathscr{F}), \nu(\mathscr{G}))$$

is an isomorphism. Moreover ν is t-exact for the perverse t-structures, and satisfies

$$\nu(\Delta_w^{\min}) \cong \Delta_w, \qquad \nu(\nabla_w^{\min}) \cong \nabla_w, \\ \nu(\mathscr{T}_w^{\min}) \cong \mathscr{T}_w, \qquad \nu(\mathscr{E}_w^{\min}) \cong \mathscr{E}_w$$

for any $w \in W$

Proof. The proofs are identical to those of [AR2, Theorem 5.4 and Proposition 5.5], starting from Theorem 13.5 instead of the main result of [AR1]. \Box

13.5. Application to the Finkelberg–Mirković conjecture. We now explain that the results of this paper allow to give a proof of a weak form of the conjecture formulated by Finkelberg–Mirković in [FM], which has had an important influence on recent work regarding the representation theory of reductive groups. Here for simplicity we assume that G is semisimple of adjoint type, i.e. that $X^*(T)$ is the root lattice of (G,T). We assume in addition that $\ell > h$ where h is the Coxeter number of G, and that $\ell \neq 19$, resp. $\ell \neq 31$, in case \Re has a component of type \mathbf{E}_7 , resp. \mathbf{E}_8 . These assumptions ensure in particular that those of Sections 10–11–12 are satisfied.

We let \mathbb{G} be a (simply-connected) semisimple algebraic group over \Bbbk whose Frobenius twist is G_{\Bbbk}^{\vee} . We also denote by \mathbb{B} , resp. \mathbb{T} , the Borel subgroup, resp. maximal torus, whose Frobenius twist is B_{\Bbbk}^{\vee} , resp. T_{\Bbbk}^{\vee} . We identify the lattice of characters of \mathbb{T} with $X^*(T_{\Bbbk}^{\vee}) = X_*(T)$ in such a way that the morphism $X^*(T_{\Bbbk}^{\vee}) \to X^*(\mathbb{T})$ induced by the Frobenius morphism is given by $\lambda \mapsto \ell \lambda$. Then the simple objects in the category $\operatorname{Rep}(\mathbb{G})$ of finite-dimensional \mathbb{G} -modules are classified, via their highest weight, by the dominant weights of $(\mathbb{G}, \mathbb{B}, \mathbb{T})$, i.e. by $X^+_*(T)$. The simple object attached to λ will be denoted $\mathbb{L}(\lambda)$.

The affine Weyl group of \mathbb{G} in the sense of [Ja] is the subgroup W' of W generated by S, i.e. the semidirect product of $W_{\rm f}$ with the coroot lattice of (G, T) (see §4.3). We consider the action \cdot_{ℓ} of W' on $X^*(\mathbb{T})$ defined by

$$(t_{\lambda}x) \cdot_{\ell} \mu = x(\mu + \rho) - \rho + \ell\lambda$$

for λ in the coroot lattice and $x \in W_{\mathrm{f}}$, where ρ is the halfsum of the positive roots of \mathbb{G} . We will denote by $\operatorname{Rep}_0(\mathbb{G})$ the Serre subcategory of $\operatorname{Rep}(\mathbb{G})$ generated by the simple objects $\mathbb{L}(\lambda)$ with $\lambda \in (W' \cdot_{\ell} 0) \cap X^+_*(T)$. It is a basic observation that the map $w \mapsto w \cdot_{\ell} 0$ defines a bijection between the subset of W' consisting of elements w which have minimal length in $W_{\mathrm{f}}w$ and $(W' \cdot_{\ell} 0) \cap X^+_*(T)$.

The category $\operatorname{\mathsf{Rep}}(\mathbb{G})$ admits a natural structure of highest weight category with underlying poset $X^+_*(T)$ (with the order \preceq), with standard, resp. costandard, object attached to λ the Weyl module, resp. induced module, of highest weight λ . These objects will be denoted $\mathbb{M}(\lambda)$ and $\mathbb{N}(\lambda)$ respectively. In particular we have a notion of *tilting objects* in this category, and the indecomposable tilting objects are classified by $X^+_*(T)$ (via their highest weight). The indecomposable tilting module with highest weight λ will be denoted $\mathbb{T}(\lambda)$. The *linkage principle* shows that $\operatorname{\mathsf{Rep}}_0(\mathbb{G})$ is a direct summand in $\operatorname{\mathsf{Rep}}(\mathbb{G})$. In particular, the standard, resp. costandard, resp. indecomposable tilting, modules whose highest weight is in $(W' \cdot_\ell 0) \cap X^+_*(T)$ belong to $\operatorname{\mathsf{Rep}}_0(\mathbb{G})$. We will denote by $\operatorname{Tilt}_0(\mathbb{G})$ the subcategory of $\operatorname{\mathsf{Rep}}_0(\mathbb{G})$ whose objects are the tilting modules.

On the other hand, consider the I_u-equivariant derived category $D_{I_u,L+G}$ of constructible k-sheaves on Gr_G . This category admits a natural perverse t-structure, whose heart will be denoted $\mathsf{P}_{\mathrm{I}_{\mathrm{u}},L^+G}$. Let also Gr_G° be the connected component of Gr_G containing the base point, and denote by $\mathsf{D}_{\operatorname{I}_{\mathrm{u}},L^+G,\circ} \subset \mathsf{D}_{\operatorname{I}_{\mathrm{u}},L^+G}$ and $\mathsf{P}_{\mathrm{I}_{\mathrm{u}},L^+G,\circ} \subset \mathsf{P}_{\mathrm{I}_{\mathrm{u}},L^+G}$ the subcategories of objects supported on Gr_G° . The I_u-orbits on $\operatorname{Gr}_G^{\circ}$ are in a natural bijection with the subset $W'^{\mathsf{f}} \subset W'$ consisting of elements $w \in W'$ which have minimal length in $wW_{\rm f}$. (This bijection sends w to the image of $\operatorname{Fl}_{G,w}$ in Gr_{G} .) We will denote by $\overline{\Delta}_w$, resp. $\overline{\nabla}_w$, resp. $\overline{\mathscr{IC}}_w$, the !-pushforward, resp. *-pushforward, resp. intermediate extension, of the constant local system on the orbit attached to w, shifted by its dimension. In fact, the orbit attached to $w \in W'^{\mathrm{f}}$ is isomorphic to an affine space (of dimension $\ell(w)$), which implies that $\overline{\Delta}_w$ and $\overline{\nabla}_w$ are perverse sheaves. Moreover $\overline{\mathscr{IC}}_w$ is the image of the unique (up to scalar) nonzero morphism $\overline{\Delta}_w \to \overline{\nabla}_w$, and the map sending w to the isomorphism class of $\overline{\mathscr{IC}}_w$ provides a bijection between W'^{f} and the set of isomorphism classes of simple objects in $\mathsf{P}_{\mathrm{I}_{\mathrm{n}},L^{+}G,\circ}$. The classical results of [BGS, §§3.2–3.3] ensure that $\mathsf{P}_{\mathrm{I}_{u},L^{+}G,\circ}$ admits a canonical structure of highest weight category with underlying poset W'^{f} (endowed with the restriction of the Bruhat order on W'), with standard, resp. costandard, object attached to w the perverse sheaf $\overline{\Delta}_w$, resp. $\overline{\nabla}_w$. In particular we can consider tilting objects in this category, and the indecomposable tilting objects are classified by $W^{\prime,f}$. The indecomposable tilting object attached to w will be denoted $\overline{\mathscr{T}}_w$.

There exists a canonical (left) action of the monoidal category $(\mathsf{D}^{\wedge}_{\mathrm{I}_{u},\mathrm{I}_{u}},\widehat{\star})$ on $\mathsf{D}_{\mathrm{I}_{u},L^{+}G}$, defined by a procedure similar to that used for the action on $\mathsf{D}_{\mathrm{I}_{u},\mathrm{I}}$. The corresponding bifunctor will also be denoted $\widehat{\star}$.

The following statement is the promised weak form of a conjecture formulated in [FM].

Theorem 13.9. There exists an equivalence of categories

$$\mathsf{P}_{\mathrm{I}_{\mathrm{u}},L^{+}G,\circ} \cong \mathsf{Rep}_{0}(\mathbb{G})$$

matching, for $w \in W'^{\text{f}}$, the object $\overline{\Delta}_w$, resp. $\overline{\nabla}_w$, resp. $\overline{\mathscr{IC}}_w$, resp. $\overline{\mathscr{I}}_w$, with $\mathbb{M}(w^{-1} \cdot_{\ell} 0)$, resp. $\mathbb{N}(w^{-1} \cdot_{\ell} 0)$, resp. $\mathbb{L}(w^{-1} \cdot_{\ell} 0)$, resp. $\mathbb{T}(w^{-1} \cdot_{\ell} 0)$.

Proof. Consider the full subcategory $\mathsf{Par}_{I,I,\circ}^{\mathrm{BS}}$ of $\mathsf{Par}_{I,I}^{\mathrm{BS}}$ whose objects are the sequences of the form (s_1, \cdots, s_i, e, n) . Let also $\mathsf{Par}_{I,I,\circ}$ be the Karoubian envelope of the additive hull of $\mathsf{Par}_{I,I,\circ}^{\mathrm{BS}}$; then $\mathsf{Par}_{I,I,\circ}$ identifies with the category of parity objects in the subcategory of $\mathsf{D}_{I,I}$ consisting of complexes supported on the connected component of the base point in Fl_G . We will denote by $\overline{\mathsf{Par}}_{I,I,\circ}$ the category which has the same objects as $\mathsf{Par}_{I,I,\circ}$, and whose morphism space from \mathscr{F} to \mathscr{G} is the quotient of $\mathrm{Hom}_{\mathsf{Par}_{I,I}}^{\bullet}(\mathscr{F},\mathscr{G})$ by the subspace consisting of sums of homogeneous morphisms factoring through objects of the form $\mathscr{IC}_{s_1}^{\mathrm{I}} \star_{\mathrm{I}} \cdots \star_{\mathrm{I}} \mathscr{IC}_{s_i}^{\mathrm{I}}[n]$ with $s_1 \in S_{\mathrm{f}}$.

For any $s \in S$ we can choose a weight μ_s on the wall of the fundamental alcove attached to s which does not belong to any other wall (see [Ja, §II.6.3]). Then we have translation functors $T_0^{\mu_s}$ and $T_{\mu_s}^0$, and we set $\Theta_s := T_{\mu_s}^0 \circ T_0^{\mu_s}$. (The definition of this functor requires some choices, but its isomorphism class is well defined; see e.g. [RW1, §3.2] for details.) It has been conjectured in [RW1], and proved independently in [BR2] and [Ci], that there exists a right action of the monoidal category $\mathsf{Par}_{\mathrm{I},\mathrm{I},\circ}^{\mathrm{BS}}$ on $\mathsf{Rep}_0(\mathbb{G})$ such that the object $\mathscr{IC}_s^{\mathrm{I}}$ acts via a functor isomorphic to Θ_s for any $s \in S$. As explained in [RW1, Theorem 5.2.1], once this statement is known one obtains an equivalence of additive categories

(13.8)
$$\overline{\mathsf{Par}}_{\mathrm{I},\mathrm{I},\circ} \xrightarrow{\sim} \mathrm{Tilt}_0(\mathbb{G})$$

sending an object to its action on the trivial module $\mathbb{k} = \mathbb{L}(0)$.

On the other hand, consider the subcategory $\mathsf{T}_{\mathrm{I}_u,\mathrm{I},\circ}$ of $\mathsf{T}_{\mathrm{I}_u,\mathrm{I}}$ consisting of tilting objects supported on the connected component of the base point in Fl_G , and denote by $\mathrm{Tilt}_{\mathrm{I}_u,L^+G,\circ}$ the full subcategory of $\mathsf{P}_{\mathrm{I}_u,L^+G,\circ}$ consisting of tilting objects. It is a standard fact that pushforward under the natural morphism $\mathrm{Fl}_G \to \mathrm{Gr}_G$ sends tilting perverse sheaves to tilting perverse sheaves. (In fact the proof of the corresponding fact for \mathbb{Q}_ℓ -perverse sheaves given in [Yu, Proposition 2.4.1] applies in the present setting also.) We deduce a canonical functor

$$\mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I},\circ} \to \mathsf{T}_{\mathrm{I}_{\mathrm{u}},L^{+}G,\circ}$$

It is easily seen that this functor sends each $\Xi_{s,!}$ with $s \in S_{\rm f}$ to 0. As a consequence, the composition

$$\mathsf{Par}_{\mathrm{I},\mathrm{I},\circ} \to \mathsf{T}_{\mathrm{I}_{\mathrm{u}},\mathrm{I},\circ} \to \mathsf{T}_{\mathrm{I}_{\mathrm{u}},L^{+}G,\circ}$$

(where the first functor is obtained from the composition of $\mathsf{For}^1_{\mathrm{I}_u}$ with the functor of Theorem 13.5 by passing to Karoubian envelopes of additive hulls) factors through a functor

$$(13.9) \qquad \qquad \mathsf{Par}_{\mathrm{I},\mathrm{I},\circ} \to \mathsf{T}_{\mathrm{I}_{\mathrm{u}},L^{+}G,\circ}$$

which for any $s \in S$ intertwines right convolution (via \star_{I}) with $\mathscr{I}\!\mathscr{C}^{\mathrm{I}}_{s}$ on the lefthand side and left convolution (via $\widehat{\star}$) with \mathscr{T}^{\wedge}_{s} on the right-hand side. In view of (13.8), (13.9) provides a functor

$$\operatorname{Tilt}_0(\mathbb{G}) \to \mathsf{T}_{\mathrm{I}_{\mathrm{u}},L^+G,\circ},$$

which is easily seen to send $\mathbb{T}(w^{-1} \cdot_{\ell} 0)$ to $\overline{\mathscr{T}}_w$ for any $w \in W'^{\text{f}}$.

The general theory of highest weight categories implies that the triangulated category $D^{\mathrm{b}}\mathsf{Rep}_0(\mathbb{G})$, resp. $\mathsf{D}_{\mathrm{I}_{\mathrm{u}},L^+G,\circ}$, identifies with the bounded homotopy category of $\mathrm{Tilt}_0(\mathbb{G})$, resp. $\mathsf{T}_{\mathrm{I}_{\mathrm{u}},L^+G,\circ}$. From the functor above we therefore deduce a triangulated functor

(13.10)
$$D^{\mathsf{D}}\mathsf{Rep}_{0}(\mathbb{G}) \to \mathsf{D}_{\mathrm{I}_{n},L^{+}G,\circ}$$

We claim that for any $w \in W'^{\text{f}}$ the functor (13.10) sends the module $\mathbb{M}(w^{-1} \cdot_{\ell} 0)$, resp. $\mathbb{N}(w^{-1} \cdot_{\ell} 0)$, to $\overline{\Delta}_w$, resp. $\overline{\nabla}_w$. We will prove this claim in the case of standard objects; the other case is similar. The proof proceeds by induction on $\ell(w)$. The case $\ell(w) = 0$, i.e. w = e, is clear since $\mathbb{M}(0) = \mathbb{T}(0)$ and $\overline{\Delta}_e = \overline{\mathscr{T}}_e$. Then choose $w \in W'^{\text{f}}$ and $s \in S$ such that $\ell(sw) = \ell(w) - 1$ (which implies that $ws \in W'^{\text{f}}$) and assume the claim is known for sw. Using adjunction and [Ja, Proposition II.7.11] one sees that we have

(13.11)
$$\dim \operatorname{Hom}(\mathbb{M}(w^{-1}s \cdot_{\ell} 0), \Theta_s \mathbb{M}(w^{-1}s \cdot_{\ell} 0)) = 1,$$

and by [Ja, Proposition II.7.12] there exists an exact sequence

$$\mathbb{M}(w^{-1}s \cdot_{\ell} 0) \hookrightarrow \Theta_s \mathbb{M}(w^{-1}s \cdot_{\ell} 0) \twoheadrightarrow \mathbb{M}(w^{-1} \cdot_{\ell} 0)$$

where, by (13.11), the first morphism can be chosen to be any nonzero morphism. Now, consider a nonzero morphism $\delta \to \mathscr{IC}_s[1]$ in $\mathsf{Par}_{\mathrm{I},\mathrm{I}}$. (Such a morphism exists, and is unique up to scalar.) Consider the image of this morphism under the composition of the functor of Theorem 13.5 and $\mathsf{For}_{\mathrm{I}_u}^{\mathrm{I}}$, and convolve on the right with $\overline{\Delta}_{sw}$. We deduce a nonzero morphism $\overline{\Delta}_{sw} \to \mathscr{T}_s \star_{\mathrm{I}} \overline{\Delta}_{sw} \cong \mathscr{T}_s^{\wedge} \widehat{\star} \overline{\Delta}_{sw}$, which fits in an exact sequence

$$\overline{\Delta}_{sw} \hookrightarrow \mathscr{T}_s^{\wedge} \mathbin{\widehat{\star}} \overline{\Delta}_{sw} \twoheadrightarrow \overline{\Delta}_w.$$

These considerations show that the morphism $\delta \to \mathscr{IC}_s[1]$, acting on $\mathbb{M}(w^{-1}s \cdot_{\ell} 0)$, is sent by (13.10) to a nonzero morphism $\overline{\Delta}_{sw} \to \mathscr{T}_s^{\wedge} \widehat{\star} \overline{\Delta}_{sw}$; the latter morphism is therefore also nonzero, and its cone (isomorphic to $\mathbb{M}(w^{-1} \cdot_{\ell} 0)$) is sent to $\overline{\Delta}_w$, which finishes the proof of our claim.

Once this claim is proved, we deduce by standard arguments that (13.10) is an equivalence of categories. Namely, for $w, y \in W'$, we have

$$\dim \operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Rep}_{0}(\mathbb{G})}(\mathbb{M}(w^{-1}\cdot_{\ell} 0), \mathbb{N}(y^{-1}\cdot_{\ell} 0)[n]) = \begin{cases} 1 & \text{if } w = y \text{ and } n = 0; \\ 0 & \text{otherwise} \end{cases}$$

and

$$\dim \operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u},L}+_{G,\circ}}}(\overline{\Delta}_{w}, \overline{\nabla}_{y}[n]) = \begin{cases} 1 & \text{if } w = y \text{ and } n = 0; \\ 0 & \text{otherwise.} \end{cases}$$

Fix $w \in W'^{,\mathrm{f}}$, and consider a nonzero morphism $\mathbb{M}(w^{-1}\cdot_{\ell}0) \to \mathbb{N}(w^{-1}\cdot_{\ell}0)$. The cone of this morphism belongs to the triangulated subcategory generated by the objects $\mathbb{M}(y^{-1}\cdot_{\ell}0)$ with $\ell(y) < \ell(w)$; the image of this morphism under (13.10) is therefore a morphism $\overline{\Delta}_w \to \overline{\nabla}_w$ whose cone belongs to the triangulated subcategory generated by the objects $\overline{\Delta}_y$ with $\ell(y) < \ell(w)$, hence is nonzero. These arguments show that our functor induces, for any $w, y \in W'^{,\mathrm{f}}$ and $n \in \mathbb{Z}$, an isomorphism

$$\operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Rep}_{0}(\mathbb{G})}(\mathbb{M}(w^{-1}\cdot_{\ell}0),\mathbb{N}(y^{-1}\cdot_{\ell}0)[n])\xrightarrow{\sim}\operatorname{Hom}_{\mathsf{D}_{\mathrm{I}_{\mathrm{u}},L^{+}G,\circ}}(\overline{\Delta}_{w},\overline{\nabla}_{y}[n]);$$

by Beilinson's lemma it is therefore fully faithful. This property implies that the essential image of this functor is a triangulated subcategory of $D_{I_u,L^+G,\circ}$; since it contains all the standard perverse sheaves it must coincide with $D_{I_u,L^+G,\circ}$.

The equivalence (13.10) is t-exact since it sends standard, resp. costandard, representations to standard, resp. costandard, perverse sheaves. It therefore restricts to an (exact) equivalence $\operatorname{Rep}_0(\mathbb{G}) \xrightarrow{\sim} \operatorname{P}_{\operatorname{I}_u,L+G,\circ}$. By exactness this equivalence sends the image of a nonzero morphism $\mathbb{M}(w^{-1} \cdot_{\ell} 0) \to \mathbb{N}(w^{-1} \cdot_{\ell} 0)$ to the image of a nonzero morphism $\overline{\Delta}_w \to \overline{\nabla}_w$, i.e. it sends $\mathbb{L}(w^{-1} \cdot_{\ell} 0)$ to $\overline{\mathscr{IC}}_w$. It must also send indecomposable tilting modules to indecomposable tilting perverse sheaves, and by consideration of highest weights we see that it more precisely sends $\mathbb{T}(w^{-1} \cdot_{\ell} 0)$ to $\overline{\mathscr{IC}}_w$, which finishes the proof.

- Remark 13.10. (1) In addition to the properties stated in Theorem 13.9, the conjecture in [FM] requires a compatibility of the equivalence with the Satake equivalence S; see e.g. [AR5, §1.2] for details. This compatibility is important for some applications, see e.g. [AR6]. We expect to obtain a proof of this stronger version in the third part of this project.
 - (2) In the course of the proof of Theorem 13.9 we have seen that the functor of Theorem 13.5 induces an equivalence of additive categories

$$\overline{\mathsf{Par}}_{\mathrm{I},\mathrm{I},\circ} \xrightarrow{\sim} \mathsf{T}_{\mathrm{I}_{\mathrm{u}},L^{+}G,\circ}.$$

On the other hand, consider the category $\operatorname{Par}_{I,I,o}$ whose objects are those of $\operatorname{Par}_{I,I}$ and whose morphisms are obtained from those in $D_{I,I}$ by quotient by morphisms factoring through an object of the form $\mathscr{IC}_{s_1}^I \star_I \cdots \star_I \mathscr{IC}_{s_i}^I[n]$ with $s_1 \in S_f$. Recall the category $D_{\mathcal{IW},I}$ considered in the course of the proof of Lemma 10.17. We have a notion of parity complexes in this category (see e.g. [RW1, §11.2]), giving rise to a full subcategory $\operatorname{Par}_{\mathcal{IW},I} \subset$

 $D_{\mathcal{IW},I}$. Considering objects supported on the connected component containing the base point, we obtain subcategories $\mathsf{Par}_{\mathcal{IW},I,\circ} \subset D_{\mathcal{IW},I,\circ}$. As explained in [RW1, Theorem 11.7.1], the averaging functor $\mathsf{D}_{I,I} \to \mathsf{D}_{\mathcal{IW},I}$ induces an equivalence of categories

$$\widetilde{\mathsf{Par}}_{\mathrm{I},\mathrm{I},\circ} \xrightarrow{\sim} \mathsf{Par}_{\mathcal{IW},\mathrm{I},\circ}.$$

The obvious "degrading" functor $\widetilde{\mathsf{Par}}_{I,I,\circ}\to\overline{\mathsf{Par}}_{I,I,\circ}$ therefore provides a functor

$$\mathsf{Par}_{\mathcal{IW},\mathrm{I},\circ} \to \mathsf{T}_{\mathrm{I}_{\mathrm{u}},L^{+}G,\circ}$$

with properties similar to those stated in Theorem 13.5. This provides a version of the construction of [AMRW, §7.2] involving "true" perverse sheaves rather than their mixed versions, and also explains the relation between the approach to $\text{Tilt}_0(\mathbb{G})$ proposed in [RW1] and the one that is suggested by the Finkelberg–Mirković conjecture.

(3) From Theorem 13.9 and standard arguments (involving in particular the considerations in [Wi, \$1.13]⁴) one can obtain yet another proof of Lusztig's conjecture on characters of simple modules in $\text{Rep}_0(\mathbb{G})$ [Lu] in large characteristic, which bypasses the comparison of characters of simple and indecomposable tilting modules (see [RW1, \$1.8]).

Appendix A. Verdier quotients, Serre quotients, and t-structures

A.1. Quotient categories. If A is an abelian category, recall that a *Serre subcate*gory of A is a nonempty strictly full subcategory which is stable under subquotients and extensions. Given a Serre subcategory $B \subset A$, one can form the quotient category A/B and the exact functor $Q : A \to A/B$ which satisfy the universal property that given an abelian category A' and an exact functor $F : A \to A'$ such that F(M) = 0 for any M in B, there exists a unique functor $G : A/B \to A'$ such that $F = G \circ Q$. (Moreover, in this situation G is exact.)

Remark A.1. There exist at least two different constructions of A/B: the initial construction of Gabriel given in [Gab, §III.1], and the construction as a localization explained in [SP, Tag 02MN]. Since these constructions provide categories satisfying the same universal property, the corresponding categories must be canonically equivalent.

Now, let D be a triangulated category, and $E \subset D$ a full triangulated subcategory. Then one can form the quotient category D/E and the quotient functor $\Pi : D \rightarrow D/E$ using a localization procedure as in [SP, Tag 05RA]. This category satisfies the following universal properties (see [SP, Tag 05RJ]):

- (1) for any triangulated category D' and any triangulated functor $F : D \to D'$ such that F(M) = 0 for any M in E, there exists a unique functor G : $D/E \to D'$ such that $F = G \circ \Pi$; moreover G is triangulated;
- (2) for any abelian category A and any cohomological functor $H : \mathsf{D} \to \mathsf{A}$ such that H(M) = 0 for any M in E , there exists a unique functor $H' : \mathsf{D}/\mathsf{E} \to \mathsf{A}$ such that $H = H' \circ \Pi$; moreover H' is a cohomological functor.

 $^{^{4}}$ Note that the combinatorial statement due to Kato discussed in this reference can also be proved geometrically by translating in the Grothendieck group the main result of [AR5] in the case of characteristic-0 coefficients. (This case was first proved by the first author with Arkhipov, Braverman, Gaitsgory and Mirković).

Another property which is checked similarly is that given a triangulated bifunctor $F : \mathsf{D} \times \mathsf{D} \to \mathsf{D}'$ such that F(X, Y) = 0 if either X or Y is in E, there exists a unique bifunctor $G : (\mathsf{D}/\mathsf{E}) \times (\mathsf{D}/\mathsf{E}) \to \mathsf{D}'$ such that $F = G(\Pi(-), \Pi(-))$, and moreover G is triangulated.

A.2. Verdier quotient and t-structures. Let D be a triangulated category equipped with a bounded t-structure $(D_{\leq 0}, D_{\geq 0})$. We will denote by A the heart of this t-structure, and by $(H^n : n \in \mathbb{Z})$ the associated cohomology functors. Recall that a bounded t-structure is automatically non-degenerate; in particular we have

$$D_{\leq 0} = \{ X \in \mathsf{D} \mid \forall n \in \mathbb{Z}_{>0}, H^n(X) = 0 \}; D_{>0} = \{ X \in \mathsf{D} \mid \forall n \in \mathbb{Z}_{<0}, H^n(X) = 0 \},\$$

see [BBDG, Proposition 1.3.7].

Let also $B \subset A$ be a Serre subcategory, and denote by D_B the full triangulated subcategory of D generated by B; it is easily seen that

$$\mathsf{D}_{\mathsf{B}} = \{ X \in \mathsf{D} \mid \forall n \in \mathbb{Z}, \, H^n(X) \in \mathsf{B} \}.$$

We set

$$D' := D/D_B$$
,

and denote the quotient functor by $\Pi : \mathsf{D} \to \mathsf{D}'$.

- **Lemma A.2.** (1) There exists a unique t-structure on D' such that Π is texact.
 - (2) This t-structure is bounded, and given explicitly by

$$\begin{split} \mathsf{D}'_{\leq 0} &= \{X \in \mathsf{D}' \mid \forall m \in \mathbb{Z}_{>0}, \, H^m(X) \in \mathsf{B}\};\\ \mathsf{D}'_{\geq 0} &= \{X \in \mathsf{D}' \mid \forall m \in \mathbb{Z}_{<0}, \, H^m(X) \in \mathsf{B}\}, \end{split}$$

where we identify the objects of D and D'.

(3) If A' is the heart of this t-structure, then the restriction of Π_D to A, seen as a functor $A \to A'$, factors through an equivalence of categories $A/B \xrightarrow{\sim} A'$.

Proof. We define the full subcategories $\mathsf{D}'_{\leq 0}$ and $\mathsf{D}'_{\geq 0}$ as in the statement of the lemma, and then set $\mathsf{D}'_{\leq n} := \mathsf{D}'_{\leq 0}[-n]$ and $\mathsf{D}'_{\geq n} := \mathsf{D}'_{\geq 0}[-n]$ for $n \in \mathbb{Z}$; we also have

$$\mathsf{D}'_{\leq n} = \{ X \in \mathsf{D}' \mid \forall m \in \mathbb{Z}_{>n}, \, H^m(X) \in \mathsf{B} \}; \\ \mathsf{D}'_{>n} = \{ X \in \mathsf{D}' \mid \forall m \in \mathbb{Z}_{$$

First, let us show that $(\mathsf{D}_{\leq 0}',\mathsf{D}_{\geq 0}')$ is a t-structure on $\mathsf{D}'.$

The first axiom we have to check is that if $X \in \mathsf{D}'_{\leq 0}$ and $Y \in \mathsf{D}'_{\geq 1}$ we have $\operatorname{Hom}_{\mathsf{D}'}(X,Y) = 0$. Recall that a morphism $f: X \to \overline{Y}$ in D' is the equivalence class of a diagram

$$X \xleftarrow{g} Z \xrightarrow{h} Y$$

where $Z \in D$, g, h are morphisms in D, and the cone C of g belongs to D_B . Since $X \in D'_{\leq 0}$, the long exact sequence in cohomology associated with the distinguished triangle

$$Z \xrightarrow{g} X \to C \xrightarrow{[1]}$$

shows that $H^n(Z) \in \mathsf{B}$ for any n > 0. It follows that the canonical morphism $\varphi : \tau_{\leq 0}Z \to Z$ is such that $\Pi(\varphi)$ is an isomorphism. Similarly the canonical

morphism $\psi: Y \to \tau_{\geq 1} Y$ is such that $\Pi(\psi)$ is an isomorphism. We deduce that f can be written as the composition

$$X \xrightarrow{\Pi(g)^{-1}} Z \xrightarrow{\Pi(\varphi)^{-1}} \tau_{\leq 0} Z \xrightarrow{\Pi(\varphi)} Z \xrightarrow{\Pi(h)} Y \xrightarrow{\Pi(\psi)} \tau_{\geq 1} Y \xrightarrow{\Pi(\psi)^{-1}} Y.$$

Now $\psi \circ h \circ \varphi = 0$ since $(\mathsf{D}_{\leq 0}, \mathsf{D}_{\geq 0})$ is a t-structure on D , hence f = 0.

It is clear that we have $D'_{\leq 0} \subset D'_{\leq 1}$ and $D'_{\geq 0} \supset D'_{\geq 1}$, so that our data satisfy the second axiom of a t-structure. Finally, for any $X \in D'$, considered as an object in D, the canonical triangle $\tau_{\leq 0}X \to X \to \tau_{\geq 1}X \xrightarrow{[1]}$ in D induces a distinguished triangle

$$\Pi(\tau_{<0}X) \to \Pi(X) \to \Pi(\tau_{>1}X) \xrightarrow{[1]}$$

in D', proving that the third axiom is also satisfied.

It is clear that the t-structure constructed above is bounded, and that Π is t-exact. The unicity of a t-structure on D' such that Π is t-exact follows from the standard claim that two t-structures on a given triangulated category such that the nonpositive part of the first one is contained in the nonpositive part of the second one and the nonnegative part of the first one is contained in the nonnegative part of the second one must coincide.

Let us now denote by A' the heart of the t-structure on D'. Since Π is t-exact it restricts to an exact functor $A \rightarrow A'$. It is clear that this functor sends all objects of B to 0; by the universal property of the Serre quotient it therefore factors through an exact functor

$$(A.1) A/B \to A'$$

On the other hand, consider the quotient functor $Q : A \to A/B$, and the cohomological functor $Q \circ H^0 : D \to A/B$. It is clear that this functor sends all objects of D_B to 0; it therefore factors through a cohomological functor $D' \to A/B$, which by restriction to A' provides an exact functor

$$(A.2) \qquad \qquad A' \to A/B$$

It is clear that (A.1) and (A.2) are quasi-inverse to each other, so that (A.1) is an equivalence of categories.

A.3. Quotients of derived categories. Let now A be an abelian category, and $B \subset A$ be a Serre subcategory. We consider as in §A.2 the full triangulated subcategory D_B of $D^b(A)$ generated by B. Consider the Serre quotient A/B, and the quotient functor $Q : A \to A/B$. The following statement is [Mi, Theorem 3.2].

Proposition A.3. The functor

$$D^{\mathrm{b}}(Q): D^{\mathrm{b}}(\mathsf{A}) \to D^{\mathrm{b}}(\mathsf{A}/\mathsf{B})$$

factors through an equivalence of triangulated categories

$$D^{\mathrm{b}}(\mathsf{A})/\mathsf{D}_{\mathsf{B}} \xrightarrow{\sim} D^{\mathrm{b}}(\mathsf{A}/\mathsf{B}).$$

By uniqueness, in this particular setting the t-structure obtained using Lemma A.2 from the standard t-structure on $D^{\rm b}(\mathsf{A})$ is just the standard t-structure on $D^{\rm b}(\mathsf{A}/\mathsf{B})$.

APPENDIX B. FLAT MODULES IN CATEGORIES

B.1. Modules in categories. Let k be a commutative ring, R be a k-algebra, and A be a k-linear abelian category. Recall (see [KS, §8.5]) that an R-module in A is a pair (X, ξ_X) where X is an object in A and $\xi_X : R \to \text{End}_A(X)$ is a k-algebra morphism. The R-modules in A are the objects in a k-linear abelian category Mod(R, A), where morphisms from (X, ξ_X) to (Y, ξ_Y) are defined as morphisms $f : X \to Y$ in A which satisfy $f \circ \xi_X(r) = \xi_Y(r) \circ f$ for any $r \in R$. Usually the morphism ξ_X will be omitted from notation. Note that if $X \in \text{Mod}(R, A)$ and $Y \in A$ the k-module $\text{Hom}_A(X, Y)$ admits a natural structure of right R-module, where $r \in R$ acts on a morphism $f : X \to Y$ by sending it to $f \circ \xi_X(r)$.

Given a right *R*-module *M* and an object $X \in Mod(R, A)$, if the functor

$$Y \mapsto \operatorname{Hom}_{R^{\operatorname{op}}}(M, \operatorname{Hom}_{\mathsf{A}}(X, Y))$$

is representable we denote by $M \otimes_R X$ the object that represents it. This condition is automatic in the following cases:

- A admits inductive limits, see [KS, Proposition 8.5.5(a)]. (In particular, this condition is satisfied in case A = Ind-C with C a k-linear abelian category, see [KS, Theorem 8.6.5(iii)].)
- (2) M is of finite presentation, see [KS, Remark 8.5.7].

If we denote by $\mathsf{Mod}_{\mathbf{r}}(R)$ the category of right *R*-modules, then in case (1) we therefore obtain a canonical bifunctor

$$(-)\otimes_R (-): \mathsf{Mod}_{\mathbf{r}}(R) \times \mathsf{Mod}(R, \mathsf{A}) \to \mathsf{A},$$

which is additive and right exact on both sides. It is clear that if S is another kalgebra, M is an (S, R)-bimodule and N is a right S-module, then if $X \in Mod(R, A)$ the object $M \otimes_R X$ has a canonical structure of S-module in A, and moreover we have a canonical isomorphism

(B.1)
$$N \otimes_S (M \otimes_R X) \cong (N \otimes_S M) \otimes_R X.$$

If we assume that R is right noetherian and denote by $\mathsf{Mod}_{r}^{\mathrm{fg}}(R)$ the category of finitely generated right R-modules, then in view of (2) above we similarly have a canonical bifunctor

$$(B.2) \qquad (-) \otimes_R (-) : \mathsf{Mod}_r^{\mathrm{tg}}(R) \times \mathsf{Mod}(R, \mathsf{A}) \to \mathsf{A}$$

which again is additive and right exact on both sides and satisfies (B.1) when M is finitely generated as a right R-module and N is finitely generated as a right S-module. Concretely, given a free presentation

$$R^{\oplus n} \xrightarrow{f} R^{\oplus m} \to M \to 0,$$

the morphism f, seen as a matrix with coefficients in R, defines via ξ_X a morphism $X^{\oplus n} \to X^{\oplus m}$, and $M \otimes_R X$ is canonically isomorphic to the cokernel of this map.

Lemma B.1. Let $X \in A$, $Y \in Mod(R, A)$ and $M \in Mod_r(R)$, and assume that the tensor product $M \otimes_R Y$ is defined. Then $Hom_A(X, Y)$ has a canonical structure of left *R*-module, and there exists a canonical morphism of *k*-modules

$$M \otimes_R \operatorname{Hom}_{\mathsf{A}}(X, Y) \to \operatorname{Hom}_{\mathsf{A}}(X, M \otimes_R Y)$$

Proof. The *R*-module structure on $\text{Hom}_A(X, Y)$ is defined so that an element $r \in R$ acts on a morphism f by sending it to $\xi_Y(r) \circ f$. By the standard adjunction for tensor products of *R*-modules, to construct a morphism as in the lemma we need to define a morphism of right *R*-modules

 $M \to \operatorname{Hom}_k(\operatorname{Hom}_A(X, Y), \operatorname{Hom}_A(X, M \otimes_R Y)).$

In other words, given $m \in M$ and a morphism $f : X \to Y$, we need to construct a morphism $X \to M \otimes_R Y$. By the Yoneda lemma, to construct such a morphism one needs to define, for any $Z \in A$, a morphism

$$\operatorname{Hom}_{\mathsf{A}}(M \otimes_R Y, Z) \to \operatorname{Hom}_{\mathsf{A}}(X, Z)$$

functorial in Z. Now by definition we have

$$\operatorname{Hom}_{\mathsf{A}}(M \otimes_{R} Y, Z) = \operatorname{Hom}_{R^{\operatorname{op}}}(M, \operatorname{Hom}_{\mathsf{A}}(Y, Z)).$$

To a morphism $\varphi: M \to \operatorname{Hom}_{\mathsf{A}}(Y, Z)$ one can associate the morphism

$$\varphi(m) \circ f : X \to Z$$

which provides the desired construction.

- Remark B.2. (1) One can define in a similar way the notion of right *R*-module in A, and the tensor product $X \otimes_R M$ if X is a right *R*-module in A and M is a left *R*-module. In practice, we will only consider this construction in case *R* is commutative, so that left and right *R*-modules are the same. We will choose the most convenient notation among $X \otimes_R M$ and $M \otimes_R X$, depending on the context.
 - (2) In case A is the category of k-modules, then an R-module in A is nothing but a left R-module in the usual sense. Moreover, the above definition of the tensor product coincides with the usual definition.

B.2. Flatness. Let again k be a commutative ring, A a k-linear abelian category, and R a k-algebra. We will assume in addition that R is right noetherian. We will say that an object $X \in Mod(R, A)$ is R-flat (or simply flat if R is clear from the context) if the functor

$$(-)\otimes_R X: \mathsf{Mod}^{\mathrm{fg}}_{\mathrm{r}}(R) \to \mathsf{A}$$

is exact, i.e. if for any injection $M_1 \hookrightarrow M_2$ of finitely generated *R*-modules the induced morphism $M_1 \otimes_R X \to M_2 \otimes_R X$ is injective.

Remark B.3. In view of [SP, Tag 00HD], this definition is equivalent to the usual definition of flatness for *R*-modules in case *R* is commutative⁵ and A is the category of *k*-modules (see Remark B.2(2)).

The next lemma states that this notion satisfies the usual properties of flat modules.

Lemma B.4. (1) Consider a short exact sequence

$$0 \to X_1 \to X_2 \to X_3 \to 0$$

in Mod(R, A). If X_3 is R-flat, then for any M in $Mod_r^{fg}(R)$ the induced sequence

$$0 \to M \otimes_R X_1 \to M \otimes_R X_2 \to M \otimes_R X_3 \to 0$$

 $^{{}^{5}}$ This assumption is probably not necessary, but we were not able to find an appropriate reference where this condition is omitted.

is a short exact sequence.

(2) Consider a short exact sequence

$$0 \to X_1 \to X_2 \to X_3 \to 0$$

in Mod(R, A). If X_1 and X_3 are R-flat, then so is X_2 .

Proof. (1) Assume that X_3 is flat, fix M in $\mathsf{Mod}_{\mathrm{r}}^{\mathrm{fg}}(R)$, and consider an exact sequence

$$0 \to M_1 \to M_2 \to M \to 0$$

in $\mathsf{Mod}_{r}^{\mathrm{fg}}(R)$, where M_{2} is free. Then we obtain a commutative diagram

in which all rows and columns are exact. Applying the snake lemma to the first two lines we obtain that the first map on the third line is injective, which proves the desired claim.

(2) Assume that X_1 and X_3 are flat, and consider an injective morphism $M_1 \hookrightarrow M_2$ in $\mathsf{Mod}_r^{\mathrm{fg}}(R)$. Then using (1) and our assumptions we obtain a commutative diagram

with exact rows and columns. The four-lemma implies that the morphism on the middle column is injective, proving that X_2 is flat.

Appendix C. Infinitesimal flatness

Let k be a commutative ring, and let H be an affine group scheme over $\operatorname{Spec}(k)$. Let $I \subset \mathcal{O}(H)$ be the ideal defining the unit in H, i.e. the kernel of the augmentation morphism $\mathcal{O}(H) \to k$ in the k-Hopf algebra $\mathcal{O}(H)$. Following [Ja, §I.7.9], we will say that H is *infinitesimally flat* if the quotient $\mathcal{O}(H)/I^n$ is a finite projective module (equivalently, is finitely presented and flat, see [SP, Tag 00NX]) over k for any $n \geq 1$. This notion behaves well under flat base change, as explained in the following lemma.

Lemma C.1. If H is infinitesimally flat, then for any flat morphism $k \to k'$ the group scheme $\operatorname{Spec}(k') \times_{\operatorname{Spec}(k)} H$ over k' is infinitesimally flat.

Proof. The claim is obvious from the fact that the ideal of the unit in the k'-group scheme $\operatorname{Spec}(k') \times_{\operatorname{Spec}(k)} H$ is $k' \otimes_k I \subset k' \otimes_k \mathcal{O}(H)$. \Box

The main interest of this notion comes from the following statement, copied from [Ja, Lemma I.7.16], where we denote by Dist(H) the distribution algebra of H (see [Ja, §I.7.7]).

Lemma C.2. Assume that H is infinitesimally flat, noetherian, and integral. Then for any H-modules M, M' such that M' is projective over k, the natural morphism

$$\operatorname{Hom}_H(M, M') \to \operatorname{Hom}_{\operatorname{Dist}(H)}(M, M')$$

is an isomorphism.

This notion is related to that of *regular immersion* (see [SP, Tag 063J]) as follows.

Lemma C.3. If the embedding of the unit $\text{Spec}(k) \to H$ is a regular immersion, then H is infinitesimally flat.

Proof. If the embedding Spec $(k) \to H$ is a regular immersion, then by [SP, Tag 063K] and [SP, Tag 063M] the quotient I/I^2 is a finite projective k-module, and for any $n \ge 1$ we have an isomorphism $\operatorname{Sym}_k^n(I/I^2) \xrightarrow{\sim} I^n/I^{n+1}$. Now the left-hand side is finite and projective by [SP, Tag 01CK]. Hence each $\mathcal{O}(H)/I^n$ is an extension of finite projective modules, and is therefore finite and projective.

We deduce the following property, in case k is a k-algebra for some field k.

Corollary C.4. Assume that Spec(k) is smooth over \Bbbk , and that H is smooth over k. Then H is infinitesimally flat.

Proof. The schemes Spec(k) and Spec(H) are smooth over \Bbbk , hence regular by [SP, Tag 056S]. Using [SP, Tag 0E9J] this implies that the immersion $\text{Spec}(k) \to H$ is regular, so that H is infinitesimally flat by Lemma C.3.

Appendix D. Complements on the perverse t-structure on the completed category

D.1. Statement. In this section we consider the setting of [BR1, Part I], in its étale variant. Namely, we consider an algebraically closed field \mathbb{F} , an \mathbb{F} -torus A, and a (Zariski locally trivial) A-torsor $\pi : X \to Y$ where X, Y are algebraic varieties over \mathbb{F} . We assume we are given a finite stratification

$$Y = \bigsqcup_{s \in \mathcal{S}} Y_s$$

J

where each Y_s is isomorphic to an affine space, and the restriction of π to $X_s := \pi^{-1}(Y_s)$ is a trivial torsor. For any $s \in S$ we denote by

$$j'_s: Y_s \to Y, \quad j_s: X_s \to X$$

the embeddings.

We fix an algebraic closure \Bbbk of a finite field of characteristic different from char(\mathbb{F}). We will assume that for any $s, t \in \mathcal{S}$ and any $n \in \mathbb{Z}$ the sheaf

$$\mathscr{H}^n((j'_t)^*(j'_s)_*\underline{\Bbbk}_{Y_s})$$

is constant. By base change this implies that each $\mathscr{H}^n((j_t)^*(j_s)_*\underline{\Bbbk}_{X_s})$ is constant too, and these conditions guarantee that the formalism of perverse sheaves applies in the category $D^{\mathrm{b}}_{\mathcal{S}}(Y,\underline{\Bbbk})$ of bounded complexes of $\underline{\Bbbk}$ -sheaves \mathscr{F} on Y such that

 $\mathscr{H}^n((j'_s)^*\mathscr{F})$ is constant of finite rank for any n, s, and in the category $D^{\mathrm{b}}_{\mathscr{S}}(X, \Bbbk)$ of bounded complexes of \Bbbk -sheaves \mathscr{F} on X such that $\mathscr{H}^n((j_s)^*\mathscr{F})$ is constant of finite rank for any n, s.

We consider the "completed category" $\widehat{D}_{\mathcal{S}}(X/\!\!/ A, \mathbb{k})$ as defined in [BR1, Definition 3.1]; this category is a certain full subcategory in the category of pro-objects in $D^{k}_{\mathcal{S}}(X,\mathbb{k})$. (This construction, as well as those considered below, and the proofs of their properties, are initially due to Yun, see [BY, Appendix A].) This category is triangulated, and as explained in [BR1, §5.2] it admits a (bounded) "perverse" t-structure $({}^{p}\widehat{D}_{\mathcal{S}}(X/\!/ A,\mathbb{k})^{\leq 0}, {}^{p}\widehat{D}_{\mathcal{S}}(X/\!/ A,\mathbb{k})^{\geq 0})$. We can of course consider the similar constructions for each variety X_{s} endowed with the trivial stratification; the corresponding completed category will be denoted $\widehat{D}_{\mathcal{S}}(X_{s}/\!\!/ A,\mathbb{k})$. For each s, the (derived) functors $(j_{s})_{*}, (j_{s})_{!}, j_{s}^{*}, j_{s}^{!}$ induce triangulated functors

(D.1) $(j_s)_*, (j_s)_! : \widehat{D}_{\mathcal{S}}(X_s/\!\!/ A, \Bbbk) \to \widehat{D}_{\mathcal{S}}(X/\!\!/ A, \Bbbk),$

(D.2)
$$j_s^*, j_s^! : \widehat{D}_{\mathcal{S}}(X/\!\!/ A, \Bbbk) \to \widehat{D}_{\mathcal{S}}(X_s/\!\!/ A, \Bbbk).$$

Our goal in this section is to prove the following claim.

Proposition D.1. The functors $(j_s)_*$, $(j_s)_!$ in (D.1) are t-exact with respect to the perverse t-structures.

This proposition will be deduced from the fact that the similar functors

(D.3)
$$(j_s)_*, (j_s)_! : D^{\mathrm{b}}_{\mathcal{S}}(X_s /\!\!/ A, \Bbbk) \to D^{\mathrm{b}}_{\mathcal{S}}(X /\!\!/ A, \Bbbk)$$

are t-exact with respect to the perverse t-structures, since j_s is affine (see [BBDG, Corollaire 4.1.3]). The other ingredient is a result from [BY, Appendix A]; since the proof of this claim is somewhat sketchy, and since the construction of the perverse t-structure in [BR1] is slightly different from the original construction in [BY], we provide an explicit proof of this result in our setting.

D.2. **Preliminaries on** R_A^{\wedge} -modules. As in [BR1] we denote by R_A the group algebra of the cocharacter lattice $X_*(A)$ over \Bbbk , and by R_A^{\wedge} the completion of R_A with respect to the natural augmention ideal $\mathfrak{m}_A \subset R_A$. We will denote by $\mathsf{Mod}^{\mathsf{nil}}(R_A^{\wedge})$ the full subcategory of $\mathsf{Mod}(R_A^{\wedge})$ whose objects are the nilpotent R_A^{\wedge} modules, i.e. the modules such that any element is annihilated by a power of \mathfrak{m}_A . We will also denote by $\mathsf{Mod}^{\mathsf{fg},\mathsf{nil}}(R_A^{\wedge})$ the full subcategory of $\mathsf{Mod}^{\mathsf{fg}}(R_A^{\wedge})$ whose objects are the modules which are both finitely generated and nilpotent. (These modules are necessarily finite-dimensional.)

Lemma D.2. For any M in $\mathsf{Mod}^{\mathrm{nil}}(R_A^{\wedge})$, there exists an injective R_A^{\wedge} -module N which belongs to $\mathsf{Mod}^{\mathrm{nil}}(R_A^{\wedge})$ and an embedding $M \hookrightarrow N$.

Proof. Choose an injective R_A^{\wedge} -module N' and an embedding $M \hookrightarrow N'$. Since M is nilpotent, this embedding necessarily factors through the submodule

$$N = \{ n \in N' \mid (\mathfrak{m}_A)^k \cdot n = 0 \text{ for } k \gg 0 \}.$$

Now N is an injective R_A^{\wedge} -module by [SP, Tag 08XW].

As in [BMR, §3.1.7], this lemma implies that the canonical functor

$$D^{\mathrm{b}}\mathsf{Mod}^{\mathrm{nul}}(R^{\wedge}_{A}) \to D^{\mathrm{b}}\mathsf{Mod}(R^{\wedge}_{A})$$

is fully faithful, and that its essential image consists of complexes all of whose cohomology objects belong to $\mathsf{Mod}^{\mathrm{nil}}(R_A^{\wedge})$.

Now, we consider a bounded complex M of finitely generated R_A^{\wedge} -modules. Then we can consider for any $n \ge 0$ the complex

$$\left(R_A^\wedge/(\mathfrak{m}_A)^n\cdot R_A^\wedge\right)\otimes_{R_A^\wedge}M \quad \in D^\mathrm{b}\mathsf{Mod}^{\mathrm{nil}}(R_A^\wedge),$$

and also the derived tensor product

$$\left(R_A^{\wedge}/(\mathfrak{m}_A)^n\cdot R_A^{\wedge}\right)\overset{L}{\otimes}_{R_A^{\wedge}} M \quad \in D^{\mathrm{b}}\mathsf{Mod}^{\mathrm{nil}}(R_A^{\wedge}).$$

(Note that R_A^{\wedge} has finite global dimension, so that this complex is indeed bounded.)

Lemma D.3. The pro-objects in $D^{\mathrm{b}}\mathsf{Mod}^{\mathrm{nil}}(R^{\wedge}_A)$

are canonically isomorphic.

Proof. There exists for any n a canonical morphism

$$\left(R_A^{\wedge}/(\mathfrak{m}_A)^n \cdot R_A^{\wedge}\right) \overset{L}{\otimes}_{R_A^{\wedge}} M \to \left(R_A^{\wedge}/(\mathfrak{m}_A)^n \cdot R_A^{\wedge}\right) \otimes_{R_A^{\wedge}} M;$$

we will prove that these morphisms define an isomorphism between the pro-objects under consideration. For that, by definition we need to show that for any $N \in D^{\mathrm{b}}\mathsf{Mod}^{\mathrm{nil}}(R^{\wedge}_A)$ the induced morphism

$$\varinjlim_{n} \operatorname{Hom}\left(\left(R_{A}^{\wedge}/(\mathfrak{m}_{A})^{n} \cdot R_{A}^{\wedge}\right) \otimes_{R_{A}^{\wedge}} M, N\right) \to \varinjlim_{n} \operatorname{Hom}\left(\left(R_{A}^{\wedge}/(\mathfrak{m}_{A})^{n} \cdot R_{A}^{\wedge}\right) \otimes_{R_{A}^{\wedge}} M, N\right)$$

is an isomorphism.

Fix a bounded below complex N^{\bullet} of injective and nilpotent R_A^{\wedge} -modules whose image in $D^+ \mathsf{Mod}^{\operatorname{nil}}(R_A^{\wedge})$ is N. (Such a complex exists by Lemma D.2.) Then for any bounded complex M' of finitely generated R_A^{\wedge} -modules, we observe that we have a canonical isomorphism

$$\varinjlim_{n} \operatorname{Hom}_{R_{A}^{\wedge}}^{\bullet} \left(\left(R_{A}^{\wedge} / (\mathfrak{m}_{A})^{n} \cdot R_{A}^{\wedge} \right) \otimes_{R_{A}^{\wedge}} M', N^{\bullet} \right) \xrightarrow{\sim} \operatorname{Hom}_{R_{A}^{\wedge}}^{\bullet} \left(M', N^{\bullet} \right),$$

where $\operatorname{Hom}_{R_A^{\wedge}}^{\bullet}$ denotes the complex of morphisms of R_A^{\wedge} -modules between two complexes of modules. Taking 0-th cohomology, and by exactness of filtrant direct limits, we deduce a canonical isomorphism

$$\varinjlim_{n} \operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Mod}(R^{\wedge}_{A})} \left(\left(R^{\wedge}_{A} / (\mathfrak{m}_{A})^{n} \cdot R^{\wedge}_{A} \right) \otimes_{R^{\wedge}_{A}} M', N \right) \xrightarrow{\sim} \operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Mod}(R^{\wedge}_{A})} \left(M', N \right)$$

Applying this with M' = M we obtain an isomorphism

$$\varinjlim_{n} \operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Mod}(R_{A}^{\wedge})} \left(\left(R_{A}^{\wedge}/(\mathfrak{m}_{A})^{n} \cdot R_{A}^{\wedge} \right) \otimes_{R_{A}^{\wedge}} M, N \right) \xrightarrow{\sim} \operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Mod}(R_{A}^{\wedge})} (M, N).$$

On the other hand, applying this isomorphism with M' a bounded projective resolution of M we obtain an isomorphism

$$\varinjlim_{n} \operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Mod}(R^{\wedge}_{A})} \left(\left(R^{\wedge}_{A} / (\mathfrak{m}_{A})^{n} \cdot R^{\wedge}_{A} \right) \overset{L}{\otimes}_{R^{\wedge}_{A}} M, N \right) \xrightarrow{\sim} \operatorname{Hom}_{D^{\mathrm{b}}\mathsf{Mod}(R^{\wedge}_{A})} (M, N).$$

This provides the desired identification.

In [BR1, §4.1] we consider a certain subcategory $\widehat{D}(R_A^{\wedge})$ of the category of proobjects in $D^{\mathrm{b}}\mathsf{Mod}^{\mathrm{nil}}(R_A^{\wedge})$. We show in [BR1, Proposition 4.5] that the functor

$$M \mapsto \underset{n}{\overset{\text{``}}{\xleftarrow{}}} (R_A^{\wedge} / (\mathfrak{m}_A)^n \cdot R_A^{\wedge}) \overset{L}{\otimes}_{R_A^{\wedge}} M$$

induces an equivalence of categories

$$D^{\mathrm{b}}\mathsf{Mod}^{\mathrm{fg}}(R^{\wedge}_A) \xrightarrow{\sim} \widehat{D}(R^{\wedge}_A).$$

Lemma D.3 shows that the image of a complex M of finitely generated R_A^{\wedge} -modules can also be described as

$$\underset{n}{``\varprojlim_n}"\left(R_A^\wedge/(\mathfrak{m}_A)^n\cdot R_A^\wedge\right)\otimes_{R_A^\wedge}M.$$

In particular, if for some m the complex M satisfies $\mathsf{H}^{i}(M) = 0$ for i > m, resp. for i < m, then its image in $\widehat{D}(R_{A}^{\wedge})$ can be written as " \varprojlim_{n} " M_{n} where each M_{n} satisfies $\mathsf{H}^{i}(M) = 0$ for i > m, resp. for i < m.

D.3. Proof of Proposition D.1. The following statement is [BY, Lemma A.6.2].

Proposition D.4. Let $\mathscr{F} \in \widehat{D}_{\mathcal{S}}(X/\!\![A, \Bbbk)$. Then \mathscr{F} belongs to ${}^{p}\widehat{D}_{\mathcal{S}}(X/\!\![A, \Bbbk)^{\leq 0}$, resp. to ${}^{p}\widehat{D}_{\mathcal{S}}(X/\!\![A, \Bbbk)^{\geq 0}$, if and only if there exists a projective system $(\mathscr{F}_{n} : n \geq 0)$ of objects in ${}^{p}D^{\mathrm{b}}_{\mathcal{S}}(X, \Bbbk)^{\leq 0}$, resp. in ${}^{p}D^{\mathrm{b}}_{\mathcal{S}}(X, \Bbbk)^{\geq 0}$, and an isomorphism $\mathscr{F} \cong \underset{m=n}{\overset{\circ}{\coprod}} \mathscr{F}_{n}$.

Proof. The proof of the "only if" direction is given in [BY, Lemma A.6.2]. It proceeds by induction on the number of strata in the support of \mathscr{F} , the base case (one stratum) being given by the comments at the end of §D.2. The "if" direction can be deduced as follows. Assume given a projective system $(\mathscr{F}_n : n \ge 0)$ of objects in ${}^pD^{\mathrm{b}}_{\mathcal{S}}(X,\Bbbk)^{\leq 0}$ such that " \varprojlim_n " \mathscr{F}_n belongs to $\widehat{D}_{\mathcal{S}}(X/\!\!/ A, \Bbbk)$. To prove that \mathscr{F} belongs to ${}^p\widehat{D}_{\mathcal{S}}(X/\!\!/ A, \Bbbk)^{\leq 0}$ it suffices to prove that for any \mathscr{G} in ${}^p\widehat{D}_{\mathcal{S}}(X/\!\!/ A, \Bbbk)^{\geq 1}$ we have $\operatorname{Hom}(\mathscr{F}, \mathscr{G}) = 0$. Now, by the "only if" direction, there exists a projective system $(\mathscr{G}_n : n \ge 0)$ of objects in ${}^pD^{\mathrm{b}}_{\mathcal{S}}(X, \Bbbk)^{\geq 1}$ such that $\mathscr{G} \cong "\varprojlim_n "\mathscr{G}_n$. We then have

$$\operatorname{Hom}(\mathscr{F},\mathscr{G}) = \varprojlim_n \varinjlim_m \operatorname{Hom}(\mathscr{F}_m,\mathscr{G}_n) = 0$$

since $\operatorname{Hom}(\mathscr{F}_m, \mathscr{G}_n) = 0$ for any n, m. The case of ${}^p\widehat{D}_{\mathcal{S}}(X/\!\!/ A, \Bbbk)^{\geq 0}$ is similar. \Box

Using this proposition we can give the proof of Proposition D.1.

Proof of Proposition D.1. Formal properties of the "recollement" formalism show that $(j_s)_!$ is right t-exact and $(j_s)_*$ is left t-exact. The fact that $(j_s)_!$ is also left t-exact and that $(j_s)_*$ is also right t-exact follows from the t-exactness of the functors (D.3), together with Proposition D.4 (applied in the "only if" direction on X_s , and in the "if" direction on X).

References

- [Ab1] N. Abe, A bimodule description of the Hecke category, Compos. Math. 157 (2021), 2133–2159.
- [Ab2] N. Abe, On singular Soergel bimodules, preprint arXiv:2004.09014.
- [AMRW] P. Achar, S. Makisumi, S. Riche, and G. Williamson, Koszul duality for Kac-Moody groups and characters of tilting modules, J. Amer. Math. Soc. 32 (2019), 261–310.

- [AR1] P. Achar and S. Riche, Modular perverse sheaves on flag varieties I: tilting and parity sheaves (with a joint appendix with G. Williamson), Ann. Sci. Éc. Norm. Supér. 49 (2016), 325–370.
- [AR2] P. Achar and S. Riche, Modular perverse sheaves on flag varieties II: Koszul duality and formality, Duke Math. J. 165 (2016), 161–215.
- [AR3] P. Achar and S. Riche, Reductive groups, the loop Grassmannian, and the Springer resolution, Invent. Math. 214 (2018), 289–436.
- [AR4] P. Achar and S. Riche, *Central sheaves on affine flag varieties*, book in preparation.
- [AR5] P. Achar and S. Riche, A geometric Steinberg formula, preprint arXiv:2109.11980, to appear in Transform. Groups (special issue in memory of J. Humphreys).
- [AR6] P. Achar and S. Riche, A geometric model for blocks of Frobenius kernels, preprint arXiv:2203.03530.
- [AriB] D. Arinkin and R. Bezrukavnikov, Perverse coherent sheaves, Mosc. Math. J. 10 (2010), 3–29, 271.
- [AB] S. Arkhipov and R. Bezrukavnikov, Perverse sheaves on affine flags and Langlands dual group (with an appendix by R. Bezrukavnikov and I. Mirković), Israel J. Math. 170 (2009), 135–183.
- [BBDG] A. Beĭlinson, J. Bernstein, P. Deligne, and O. Gabber, Faisceaux pervers, in Analyse et topologie sur les espaces singuliers, I (Luminy, 1981), 5–172, Astérisque 100 (1982).
- [BBM] A. Beĭlinson, R. Bezrukavnikov, and I. Mirković, *Tilting exercises*, Mosc. Math. J. 4 (2004), 547–557, 782.
- [BGS] A. Beïlinson, V. Ginzburg, and W. Soergel, Koszul duality patterns in representation theory, J. Amer. Math. Soc. 9 (1996), 473–527.
- [B1] R. Bezrukavnikov, On tensor categories attached to cells in affine Weyl groups, with an appendix by D. Gaitsgory, in Representation theory of algebraic groups and quantum groups, 69–100, Adv. Stud. Pure Math. 40, Math. Soc. Japan, 2004.
- [B2] R. Bezrukavnikov, On two geometric realizations of an affine Hecke algebra, Publ. Math. Inst. Hautes Études Sci. 123 (2016), 1–67.
- [BGMRR] R. Bezrukavnikov, D. Gaitsgory, I. Mirković, S. Riche, and L. Rider, An Iwahori-Whittaker model for the Satake category, J. Éc. polytech. Math. 6 (2019), 707–735.
- [BM] R. Bezrukavnikov and I. Mirković, Representations of semisimple Lie algebras in prime characteristic and the noncommutative Springer resolution, with an appendix by E. Sommers, Ann. of Math. (2) 178 (2013), 835–919.
- [BMR] R. Bezrukavnikov, I. Mirković, and D. Rumynin, Localization of modules for a semisimple Lie algebra in prime characteristic, with an appendix by R. Bezrukavnikov and S. Riche, Ann. of Math. 167 (2008), 945–991.
- [BR1] R. Bezrukavnikov and S. Riche, A topological approach to Soergel theory, preprint arXiv:1807.07614, to appear in Representation Theory and Algebraic Geometry: A Conference Celebrating the Birthdays of Sasha Beilinson and Victor Ginzburg (V. Baranovsky, N. Guay, T. Schedler, eds.), Trends in Mathematics, Birkhäuser.
- [BR2] R. Bezrukavnikov and S. Riche, *Hecke action on the principal block*, preprint arXiv:2009.10587, to appear in Compos. Math.
- [BRR] R. Bezrukavnikov, S. Riche, and L. Rider, Modular affine Hecke category and regular unipotent centralizer, I, preprint 2005.05583.
- [BY] R. Bezrukavnikov and Z. Yun, On Koszul duality for Kac-Moody groups, Represent. Theory 17 (2013), 1–98.
- [CYZ] X. W. Chen, Y. Ye, and P. Zhang, Algebras of derived dimension zero, Comm. Algebra 36 (2008), 1–10.
- [Ci] J. Ciappara, Hecke category actions via Smith-Treumann theory, preprint arXiv:2103.07091.
- [Co] S. Cotner, Centralizers of sections of a reductive group scheme, arXiv:2203.15133.
- [DCM] C. De Concini and A. Maffei, A generalized Steinberg section and branching rules for quantum groups at roots of 1, Mosc. Math. J. 12 (2012), 469–495, 668.
- [Del] P. Deligne, Exp. XIII : Le formalisme des cycles évanescents, in Groupes de monodromie en géométrie algébrique. II. Séminaire de Géométrie Algébrique du Bois-Marie 1967–1969 (SGA 7 II). Dirigé par P. Deligne et N. Katz, 82–115, Lecture Notes in Mathematics 340, Springer-Verlag, 1973.

- [Dem] M. Demazure, Invariants symétriques entiers des groupes de Weyl et torsion, Invent. Math. 21 (1973), 287–301.
- [Dod] C. Dodd, Equivariant coherent sheaves, Soergel bimodules, and categorification of affine Hecke algebras, preprint arXiv:1108.4028.
- [EGNO] P. Etingof, S. Gelaki, D. Nikshych, and V. Ostrik, *Tensor categories*, Mathematical Surveys and Monographs 205, American Mathematical Society, 2015.
- [EW] B. Elias and G. Williamson, Soergel calculus, Represent. Theory 20 (2016), 295–374.
- [Fi] P. Fiebig, The combinatorics of Coxeter categories, Trans. Amer. Math. Soc. 360 (2008), 4211–4233.
- [FM] M. Finkelberg and I. Mirković, Semi-infinite flags I. Case of global curve P¹, in Differential topology, infinite-dimensional Lie algebras, and applications, 81–112, Amer. Math. Soc. Transl. Ser. 2, vol. 194, Amer. Math. Soc., 1999.
- [Gab] P. Gabriel, Des catégories abéliennes, Bull. Soc. Math. France 90 (1962), 323-448.
- [Ga1] D. Gaitsgory, Construction of central elements in the affine Hecke algebra via nearby cycles, Invent. Math. 144 (2001), 253–280.
- [Ga2] D. Gaitsgory, *Braiding compatibilities*, appendix to [B1].
- [GG] R. Gordon and E. Green, Graded Artin algebras, J. Algebra 76 (1982), 111–137.
- [GW] U. Görtz and T. Wedhorn, Algebraic geometry I. Schemes with examples and exercises, Advanced Lectures in Mathematics, Vieweg + Teubner, 2010.
- [Go1] V. Gouttard, Perverse monodromic sheaves, preprint arXiv:2005.02655, to appear in J. Lond. Math. Soc.
- [Go2] V. Gouttard, Faisceaux pervers monodromiques, PhD thesis, available at https:// tel.archives-ouvertes.fr/tel-03355535/document.
- [EGA1] A. Grothendieck, Éléments de géométrie algébrique (rédigés avec la collaboration de J. Dieudonné) : I. Le langage des schémas, Inst. Hautes Études Sci. Publ. Math. 4 (1960).
- [Ha] R. Hartshorne, Algebraic geometry, Springer, 1977.
- [He] S. Herpel, On the smoothness of centralizers in reductive groups, Trans. Amer. Math. Soc. 365 (2013), 3753–3774.
- [Hu] J. E. Humphreys, Conjugacy classes in semisimple algebraic groups, American Mathematical Society, 1995.
- [Ja] J. C. Jantzen, Representations of Algebraic Groups, second edition, Mathematical Surveys and Monographs 107, Amer. Math. Soc., 2003.
- [JW] L. T. Jensen and G. Williamson, The p-canonical basis for Hecke algebras, in Categorification and higher representation theory, 333–361, Contemp. Math. 683, Amer. Math. Soc., 2017.
- [JMW] D. Juteau, C. Mautner, and G. Williamson, Parity sheaves, J. Amer. Math. Soc. 27 (2014), 1169–1212.
- [KS] M. Kashiwara and P. Schapira, *Categories and sheaves*, Grundlehren der Mathematischen Wissenschaften 332, Springer-Verlag, 2006.
- [Ku] J. F. Kurtzke, Centralizers of irregular elements in reductive algebraic groups, Pacific J. Math. 104 (1983), 133–154.
- [La] T. Y. Lam, A first course in noncommutative rings, Graduate Texts in Mathematics, vol. 131, Springer-Verlag, New York, 1991.
- [Lee] T. Y. Lee, Adjoint quotients of reductive groups, in Autour des schémas en groupes. Vol. III, 131–145, Panor. Synthèses 47, Soc. Math. France, 2015.
- [Let] E. Letellier, Fourier transforms of invariant functions on finite reductive Lie algebras, Lecture Notes in Math., vol. 1859, Springer-Verlag, 2005.
- [Lu] G. Lusztig, Some problems in the representation theory of finite Chevalley groups, in The Santa Cruz Conference on Finite Groups (Univ. California, Santa Cruz, Calif., 1979), pp. 313–317, Proc. Sympos. Pure Math. 37, Amer. Math. Soc., 1980.
- [MR] C. Mautner and S. Riche, Exotic tilting sheaves, parity sheaves on affine Grassmannians, and the Mirković-Vilonen conjecture, J. Eur. Math. Soc. (JEMS) 20 (2018), 2259–2332.
- [MRR] A. Mayeux, T. Richarz, and M. Romagny, Néron blowups and low-degree cohomological applications, preprint arXiv:2001.03597.
- [MV] I. Mirković and K. Vilonen, Geometric Langlands duality and representations of algebraic groups over commutative rings, Ann. of Math. 166 (2007), 95–143.

- [Mi] J.i. Miyachi, Localization of triangulated categories and derived categories, J. Algebra 141 (1991), 463–483.
- [R1] S. Riche, Koszul duality and modular representations of semi-simple Lie algebras, Duke Math. J. 154 (2010), 31–134.
- [R2] S. Riche, Kostant section, universal centralizer, and a modular derived Satake equivalence, Math. Z. 286 (2017), 223–261.
- [R3] S. Riche, La théorie de Hodge des bimodules de Soergel (d'après Soergel et Elias-Williamson), Astérisque 414, Séminaire Bourbaki. Vol. 2017/2018. Exposés 1136-1150 (2019), Exp. No. 1139.
- [RW1] S. Riche and G. Williamson, Tilting modules and the p-canonical basis, Astérisque 397 (2018).
- [RW2] S. Riche and G. Williamson, Smith-Treumann theory and the linkage principle, preprint arXiv:2003.08522, to appear in Publ. Math. Inst. Hautes Études Sci.
- [SGA1] Revêtements étales et groupe fondamental (SGA 1). Séminaire de géométrie algébrique du Bois Marie 1960–61. Directed by A. Grothendieck. With two papers by M. Raynaud. Updated and annotated reprint of the 1971 original. Documents Mathématiques 3, Société Mathématique de France, 2003.
- [SGA32] Schémas en groupes II: Groupes de type multiplicatif, et structure des schémas en groupes généraux, Séminaire de Géométrie Algébrique du Bois Marie 1962/64 (SGA 3). Directed by M. Demazure and A. Grothendieck. Lecture Notes in Mathematics 152, Springer-Verlag, 1970.
- [So] W. Soergel, Kazhdan-Lusztig-Polynome und unzerlegbare Bimoduln über Polynomringen, J. Inst. Math. Jussieu 6 (2007), 501–525.
- [SP] Stacks project, available at http://stacks.math.columbia.edu.
- R. Steinberg, Regular elements of semisimple algebraic groups, Inst. Hautes Études Sci. Publ. Math. 25 (1965), 49–80.
- [S2] R. Steinberg, On a theorem of Pittie, Topology 14 (1975), 173–177.
- [Ve] J.-L. Verdier, Spécialisation de faisceaux et monodromie modérée, in Analyse et topologie sur les espaces singuliers, II, III (Luminy, 1981), 332–364, Astérisque 101–102 (1983).
- [Wi] G. Williamson, Algebraic representations and constructible sheaves, Jpn. J. Math. 12 (2017), 211–259.
- [Yu] Z. Yun, Weights of mixed tilting sheaves and geometric Ringel duality, Selecta Math. (N.S.) 14 (2009), 299–320.

Department of Mathematics, Massachusetts Institute of Technology, Cambridge, MA, 02139, USA.

Email address: bezrukav@math.mit.edu

UNIVERSITÉ CLERMONT AUVERGNE, CNRS, LMBP, F-63000 CLERMONT-FERRAND, FRANCE. *Email address:* simon.riche@uca.fr